Copyright

by

Hak Joong Kim

May 2016

The Dissertation Committee for Hak Joong Kim Certifies that this is the approved version of the following dissertation:

Characterization of Two Radical S-Adenosyl-L-methionine Enzymes in

the Biosynthesis of Aminoglycosides

Committee:

Hung-wen Liu, Supervisor

Eric V. Anslyn

Guangbin Dong

Adrian T. Keatinge-Clay

Christian P. Whitman

Characterization of Two Radical S-Adenosyl-L-methionine Enzymes in the Biosynthesis of Aminoglycosides

by

Hak Joong Kim, B. S.; M. S.

Dissertation

Presented to the Faculty of the Graduate School of The University of Texas at Austin in Partial Fulfillment of the Requirements for the Degree of

Doctor of Philosophy

The University of Texas at Austin May, 2016

Dedication

To my parents and parents-in-law for their incessant support and love

and

To my beloved wife, Ji Eun, and my lovely son, Art, for completing me

Acknowledgements

It has greatly been pleased to study at the University of Texas at Austin for last seven years. Specially, it has been very special time for me to work at the Liu lab with respectable people. Here, I would like to remember and give many thanks for the people who have helped me in several aspects.

I really want to thank my advisor, Dr. Hung-wen (Ben) Liu, for his continuous confidence, patience, support, and encouragement to me. Without you, I might not finish this Ph.D course. Also I would like to thank my committee members, Drs. Eric Anslyn, Guangbin Dong, Adrian Keatinge-Clay, and Christian Whitman for kindly spending their precious time and for valuable comments for this dissertation. I want to thank Dr. Whitman once again for giving his sincere encouragement to me.

I would like to thank former and current Liu lab members to work together with me and share their priceless experiences. Dr. Yasushi Ogasawara worked with azinomycin project. He patiently taught me how to handle biological works, especially basic molecular techniques and enzyme purification methods, because I had worked for only organic synthesis before I joined to this lab. He was a good teacher. He had not given any solution directly to me and always made me think by myself to get solution even though it took more time. This training helped me to get knowledge and skill about enzymology slowly but surely. Dr. Yung-nan Li helped to prepare many enzymes in gentamicin project. She is not only an outstanding scientist, but also a good mother of Liu lab. I specially thank Dr. Reid McCarthy who is in heaven now. We worked together in GenK project. (Reid, I have felt regretful that I did not say "good-bye" to you at the last day before you left from Austin.) Jake LeVieux and Yu-Cheng Yeh have worked in AprD4/D3 project. They helped me prepare enzymes and substrates. Second I am very thankful to other lab members. First of all, I would like to thank two members, Mark and Jon, to help prepare this dissertation. Dr. Mark Ruszczycky reviewed all chapters and gave critical comments. Whenever I visited him and ask some questions, he always gave thoughtful advices to me. He is a person who does not hesitate to give straight and honest words. I have learned from him in many aspects. Jon Gengler spent his valuable time to review one chapter of my dissertation.

Previous lab member, Dr. Hak Joong Kim, picked me up in the airport and gave a ride in the first day when I arrived at Austin. He gave lots of help for me to settle down at this city. Also he guided me to join this group with Dr. Sei-hyun Choi. Sei was one of the hardest working people in the lab. Dr. Namho Kim was a good friend because he is the same age as me and took many same classes together. We had discussed about our own projects many times. Dr. Steve Mansoorabadi was a kind and smart person in the lab. Dr. Chia-I Lin gave many advices to me about my projects. I would also like to thank former lab members, Dr. Eita Sasaki, Dr. He (Grace) Sun, Dr. Wei-Chen, Chang, Dr. Meilan Wu, Dr. Anthony Romo, Dr. Hui Huang, Dr. Jordi Calveras, Dr. Ke-Yi Lin, Dr. Mel Cheng-hao Liu, to give cheers and helps to enjoy lab life.

I would like to thank the current lab members. Byungsun Jeon and Yeonjin Ko helped me in many aspects, even in personal life. I hope they will soon find best postdoc positions which they really want to go to. Gengmin Lin provided me references about some steps of organic synthesis. Jung-Kuei Chen and Aoshu Zhong helped me to do biological assays. The desk of Shao-An Wang was close to my one. We have talked about several interesting topics. I hope many other current lab members, Richiro Ushimaru, Jiawei Zhang, Zhang Chen, and Ronald Besandre, will have successful graduate student life and good lucks for their works. I would like to thank many other people who helped me to study and work at the University of Texas at Austin. I really want to thank Penny Kile, a former graduate programs coordinator in the department of chemistry. She gave generous supports for me to survive in this program. I am grateful to Betsy Hablen, a current graduate program coordinator in the department of chemistry, and Debra Foulds, a former program coordinator in the medicinal chemistry department, for their administrative supports. I would like to thank people in NMR facilities, Dr. Ben Shoulders, Angela Spangenberg, and Steve Sorey, and people in mass facilities, Dr. Ian Riddington, Jordan Dinser, and Kandyss Najjar. They took my many samples and also gave a lot of advices about interpretations of my data. I thank Michael Ronalter and Adam Kennedy in the glass shop to fix my glasswares and make novel ones.

Besides the school life, I want to give many thanks to church members. Specially, I would like to thank Baker family and Yi family. Also I really thank you Janet Sabb to teach me English and fix my English pronunciation personally.

Finally I would like to thank my family member for their endless supports and loves. I feel thankful and at the same time feel sorry for them with all my heart. My father-in-law and mother-in-law are taking care of my son, Art, during my wife, Ji Eun, works at her company in Korea. My parents have always supported me whatever I have done. I really thank Ji Eun for getting married with me and becoming my mate. I feel very sorry to my wife and my son about that I am not spending time with them. I hope we get together soon and will not separate with any reason. I love you.

Characterization of Two Radical S-Adenosyl-L-methionine Enzymes in the Biosynthesis of Aminoglycosides

Hak Joong Kim, Ph. D. The University of Texas at Austin, 2016

Supervisor: Hung-wen Liu

Biosynthetic studies of natural products are essential to the discovery and development of new drugs, because by understanding biosynthetic pathways and the enzymes that characterize them, new pathways can be engineered for the production of new compounds with improved clinical properties. Aminoglycosides have traditionally been used as important antibiotics, but resistance against aminoglycosides is well known. This has resulted in a need to better understand the biosynthesis of aminoglycosides. Besides the practical value, the discovery of new enzymes with unprecedented functions in the biosynthesis of aminoglycosides can expand our knowledge and advance our understanding of enzyme catalysis.

The work described in this dissertation focuses on the *in vitro* characterization of two radical *S*-adenosyl-L-methionine (SAM) enzymes in the biosynthesis of aminoglycosides. First, GenK is a cobalamin (Cbl)-dependent radical SAM enzyme that is responsible for catalyzing the methylation of gentamicin X_2 to produce G418. *In vitro* assays of purified and reconstituted GenK from *Micromonospora echinospora* showed that this enzyme is a radical SAM enzyme with one [4Fe-4S] cluster. Assays of GenK

with SAM, gentamic X_2 and Cbl confirmed that the methylation reaction occurs at an unactivated carbon during gentamicin biosynthesis. Isotope labeling experiments strongly suggested that SAM is the preliminary methyl donor to cobalamin, followed by secondary transfer from Me-Cbl to gentamicin X₂. It was demonstrated that GenK also accepts alternative aminoglycoside substrates besides gentamicin X₂. Second, three possible mechanisms for the GenK reaction were suggested and tested. Experiments investigating the stoichiometry of the GenK reaction revealed 5'-deoxyadenosine (5'dAdo), S-adenoxylhomocysteine (SAH), and G418 were produced in equal proportion and one-to-one with each methylation reaction catalyzed by GenK. The experiment with labeled substrates indicated that the 6'-pro-R-hydrogen atom of gentamicin X₂ is abstracted by 5'-dAdo• and that methylation occurs with retention of configuration at C6'. Several substrate derivatives were synthesized to investigate the manner of methyl transfer from cobalamin to the substrate. Finally, the in vitro characterization of AprD4 and AprD3 in the biosynthesis of C3'-deoxyaminoglycoside was also conducted. Purified and reconstituted AprD4 from Streptomyces tenebrarius is a radical SAM enzyme, catalyzing homolysis of SAM to 5'-dAdo in the presence of paromamine to produce 4oxolividamine. AprD3 from the same strain is shown to be a dehydrogenase acting as the reductase counterpart to AprD4 to catalyze the reduction of 4-oxolividamine to generate lividamine. The experiments with labeled compounds confirmed the regiochemistry of hydrogen atom abstraction by 5'-dAdo• and the stereochemical course of C3'deoxygenation of paromamine.

Table of Contents

List of Tables xiv
List of Figuresxv
Chapter 1. Radical <i>S</i> -adenosyl-L-methionine (SAM) Enzymes in Aminoglycoside Biosythesis
1.1. Aminoglycoside Biosynthesis1
1.1.1. Introduction1
1.1.2. Aminoglycoside Structure2
1.1.3. Biosynthesis fo Aminoglycosides
1.1.3.1. Biosynthesis of 2-Deoxystreptamine
1.1.3.2. Biosynthesis of Paromamine10
1.1.3.3. Biosynthesis of Ribostamycin, Neomycin C, and
Neomycin B11
1.1.3.4. Biosynthesis of Kanamycins12
1.1.3.5. Biosynthesis of Gentamicins14
1.1.3.6. Biosynthesis of Other Aminoglycosides
1.2. Radical S-Adenosyl-L-Methionine (SAM) Enzyme
1.3. Radical SAM Enzymes in Aminoglycosie Biosynthesis27
1.3.1. BtrN
1.3.2. NeoN
1.4. Dissertation Statement
Chapter 2. Characterization of GenK, a Cobalamin-Dependent Radical SAM Methyltransferase Enzyme
2.1. Introduction
2.2. Experimental Procedures
2.2.1. Materials and Equipment
2.2.2. Cloning and Expression of <i>Micromonospora echinospora</i> GenK
2.2.3. Refolding of GenK from Inclusion Bodies and Reconstitution

with Iron and Sulfide	49
2.2.4. Enzymatic Synthesis and Purification of S-Adenosyl-L-	
Methionine (SAM)	49
2.2.5. GenK Activity Assays	50
2.2.6. HPLC Detection of SAH and 5'-dAdo	51
2.2.7. HPLC Detection of Aminoglycoside	51
2.2.8. Mass Spectroscopic Analysis of GenK Activity	52
2.2.9. Synthesis of 2-Streptamine Derivative	52
2.2.10. Synthesis of Xylose Derivative	52
2.2.11. Synthesis of Gentamicin A ₂	53
2.2.12. Synthesis of 6'-Amino-6'-dehydroxy-gentamicin A2	61
2.2.13. Synthesis of 2'-Deamino-2'-hydroxy-gentamicin A2	65
2.2.14. Synthesis of 2'-Deamino-2'-hydroxy-6'-amino-6'-dehydro gentamicin A ₂	оху- 73
2.2.15. Synthesis of Garosamine Derivative	77
2.2.16. Synthesis of Gentamicin B	78
2.2.17. Synthesis of JI-20A	82
2.2.18. Synthesis of 2'-Deamino-2'-hydroxy-gentamicin X ₂	89
2.3. Results and Discussion	93
2.3.1. Gene Analysis	93
2.3.2. Purification of GenK	96
2.3.3. In vitro Activity Assays of GenK	97
2.3.4. Products of GenK Catalysis	98
2.3.5. Reducing Systems for GenK	100
2.3.6. Stoichiometry of Products Catalyzed by GenK	101
2.3.7. Methyl Transfer Manner of GenK Using Labeled SAM	103
2.3.8. Cobalamin Dependence of GenK Activity	108
2.3.9. Syntheses of Putative GenK Substrates	109
2.3.10. Substrates Catalyzed by GenK	113
2.4. Conclusion	117

Chapter 3. Mechanistic Studies of GenK	119
3.1 Introduction	119
3.2. Experimental Procedures	126
3.2.1. Materials and Equipment	126
3.2.2. Cloning, Expression, Refolding, and Reconstitution of	
GenK	127
3.2.3. GenK Activity Assays, HPLC detection of SAH, 5'-dAdo, Aminoglycosides, and Mass Spectroscopic Analysis of GenK Activity	C 128
3.2.4. Synthesis of [6'S- ² H]-Gentamicin X ₂	128
3.2.5. Synthesis of [6' <i>R</i> - ² H]-Gentamicin X ₂	137
3.2.6. Synthesis of [6'-F]-Gentamicin X ₂	146
3.2.7. Synthesis of $[6',6',6'-{}^{2}H,{}^{2}H,F]$ -Gentamicin X ₂	151
3.2.8. Synthesis of [6'-OMe]-Gentamicin X ₂	160
3.3. Results and Discussion	167
3.3.1. Proposed Mechanism	167
3.3.2. Synthesis of $[6'S^{-2}H]$ -GenX ₂ and $[6'R^{-2}H]$ -GenX ₂	169
3.3.3. Stereospecific Hydrogen Abstraction	172
3.3.4. UV-Vis Spectroscopic Experiments	178
3.3.5. Synthesis of $[6'-F]$ -GenX ₂ , $[6',6',6'-{}^{2}H,{}^{2}H,F]$ -GenX ₂ , and	
[6'-OMe]-GenX ₂	180
3.3.6. Assays with GenX ₂ Derivatives	181
3.4. Conclusion	183
Chapter 4. Characterization of Radical SAM Enzyme AprD4 and Its Reductas Partner AprD3	se 186
4.1 Introduction	186
4.2. Experimental Procedures	192
4.2.1. Materials and Equipment	192
4.2.2. Cloning and Expression of <i>Streptomyces tenebrarius</i>	
AprD3	193
 Chapter 4. Characterization of Radical SAM Enzyme AprD4 and Its Reductas Partner AprD3	se 186 186 192 192 193

4.2.3. Cloning, Expression, and Reconstitution of <i>Streptomyces tenebrarius</i> AprD4	194
4.2.4. AprD4 Activity Assays and AprD3 Activity Assays	196
4.2.5. HPLC Detction of 5'-Deoxyagenosine (5'-dAdo)	197
4.2.6. HPLC Detection of AprD4/AprD3 Reaction Product	197
4.2.7. Mass Spectroscopic Analysis of AprD3/D4 Activity and Apr Activity	prD4 197
4.2.8. Synthesis of Lividamine	198
4.2.9. Synthesis of [4'- ² H]-Paromamine	205
4.2.10. Synthesis of [3'- ² H]-Paromamine	214
4.3. Results and Discussion	226
4.3.1. Gene Analysis	226
4.3.2. Purification of AprD3 and AprD4	227
4.3.3. Synthesis of Lividamine	228
4.3.4. In vitro AprD3 and AprD4 Assay	229
4.3.5. Proposed Mechanism of AprD4 Catalysis	235
4.3.6. Synthesis of [4'- ² H]-Paromamine	236
4.3.7. Synthesis of [3'- ² H]-Paromamine	237
4.3.8. Mechanistic Studies of AprD4 Catalysis	238
4.4. Conclusion	242
Appendix	243
A.1 Spectral Data for Chapter 2	243
A.2 Spectral Data for Chapter 3	311
A.3 Spectral Data for Chapter 4	387
Bibliography	439

List of Tables

Table 1-1:	Gene analysis of gentamicin	19
Table 2-1:	The four classes of racial SAM methylases	39
Table 2-2:	Class B radical SAM methyltransferases characterized in vitro	45

List of Figures

Figure 1-1:	Structures of selected aminoglycoside antibiotics	5
Figure 1-2:	Gene cluster of gentamicin	17
Figure 1-3:	Mechanism for radical SAM cleavage	25
Figure 1-4:	Roles of the secondary [4Fe-4S] clusters in selected radical SAM	
	enzymes	26
Figure 1-5:	Proposed mechanism of the BtrN reaction	29
Figure 1-6:	Substrate specificity of BtrN	30
Figure 1-7:	NeoN catalysis	33
Figure 2-1:	Partial amino acid sequence alignment of demonstrated and putativ	e
	Cbl-dependent radical SAM <i>sp</i> ³ -carbon methyltransferases	95
Figure 2-2:	UV-visible absorbance spectra and SDS PAGE of isolated	
	GenK	97
Figure 2-3:	HPLC traces showing conversion of GenX ₂ to G418 in the presence	e
	of GenK	99
Figure 2-4:	HPLC traces showing the production of both 5'-dAdo and SAH	
	during the GenK reaction	00
Figure 2-5:	Effectiveness of various reducing systems for the activation of	
	GenK	01
Figure 2-6:	5'-dAdo and SAH production vs GenK concentration at different	
	times10	02
Figure 2-7:	Stoichiometry of 5'-dAdo and SAH production during the GenK	
	reaction10	03
Figure 2-8:	Mass spectra of GenX ₂ and G418	05

Figure 2-9:	Mass spectra of the Cbl $[M + H]^{2+}$ region	106
Figure 2-10:	Mass spectra of the Cbl $[M + H]^+$ region	107
Figure 2-11:	HPLC traces of GenK reaction in the presence of cobalamin	108
Figure 2-12:	GenK activity vs concentrations of [MeCbl]	109
Figure 2-13:	Eary biosynthetic pathway for gentamicins	112
Figure 2-14:	HPLC traces of GenK reaction with the putative substrates	115
Figure 2-15:	Mass spectroscopic data of GenK reaction	117
Figure 3-1:	Intial reaction of radical SAM enzymes	120
Figure 3-2:	Proposed mechanism for PhpK catalysis	121
Figure 3-3:	Proposed mechanism for TsrM catalysis	122
Figure 3-4:	Hypothesized mechanism for Fom3	123
Figure 3-5:	Proposed mechanism of GenD1	124
Figure 3-6:	Proposed mechanism of ThnK	125
Figure 3-7:	Proposed mechanism of Class B radical SAM	
	Methyltransferases	125
Figure 3-8:	NMR spectra of two labeled compounds, 20 and 29	172
Figure 3-9:	HPLC traces showing the production of 5'-dAdo (4) and SAH (5)	
	during GenK assay with 6'-labeled compounds	173
Figure 3-10:	Electrospray ionization (ESI) mass data of resulting 5'-dAdo (4)	174
Figure 3-11:	ESI MS data of resulting SAH (5)	175
Figure 3-12:	ESI MS data of resulting SAM (1)	175
Figure 3-13:	ESI MS data of resulting G418	176
Figure 3-14:	Chemical ionization (CI) mass data of 1-fluoro-2,4-dinitrobenzene	Э
	(FDNB)-derivatized G418	177
Figure 3-15:	UV-visible analysis of GenK assay xvi	179

Figure 3-16:	HPLC traces with GenK reaction with the GenX ₂ derivatives	181
Figure 3-17:	Formations of 5'-dAdo and SAH from the co-incubation of	
	[6'-F]-GenX ₂ and GenX ₂ in the GenK reaction for 20 h	182
Figure 3-18:	Formations of 5'-dAdo and SAH from the co-incubation of	
	[6'-F]-GenX ₂ and GenX ₂ in the GenK reaction for 20 h	183
Figure 4-1:	4'-O-adenylation of kanamycin A by ANT(4')	187
Figure 4-2:	3'-O-phosphorylation of kanamycin A by ANT(3')	187
Figure 4-3:	4'- <i>N</i> -acetylation of kanamycin A by AAC(6')	188
Figure 4-4:	Selected examples of aminoglycosides	189
Figure 4-5:	Mechanism of the diol dehydratases	190
Figure 4-6:	SDS PAGE of AprD3	228
Figure 4-7:	UV-visible absorption spectra and SDS PAGE of AprD4	228
Figure 4-8:	HPLC traces of AprD4 reaction mixtures for production of 5'-dAd	lo
	observed by UV detection at 260 nm	230
Figure 4-9:	Competence of various reducing systems for supporting AprD4	
	activity	231
Figure 4-10:	HPLC traces of AprD4 reaction mixtures showing the consumption	n
	of paromamine, minitored using a Corona CAD and UV detction	
	at 340 nm	232
Figure 4-11:	Low resolution and high resolution mass spectra of the collected	
	peak at 7.2 min shown in Figure 4-9A, trace g	232
Figure 4-12:	Electrospray ionization (ESI) mass spectra of the AprD4 reaction	
	with paromamine (11)	234

Figure 4-13:	Mass spectra of [5'- ² H]-5'-dAdo isolated from the incubation of SAM
	and AprD4/D3 with [4'- ² H]-paromamine (47) and assay mixture
	showing formation of product (13) in the incubation of $[4'-{}^{2}H]$ -
	paromamine (47) with AprD4/D3
Figure 4-14:	Mass spectra of $[5'-{}^{2}H_{2}]-5'-dAdo$ produced during the AprD4/AprD3
	reaction with $[5'-{}^{2}H_{2}]$ -SAM and unlabeled paromamine (11)239

Chapter 1: Radical S-adenosyl-L-methionine (SAM) Enzymes in Aminoglycoside Biosynthesis

1.1. AMINOGLYCOSIDE BIOSYNTHESIS

1.1.1. Introduction

Aminoglycosides belong to a long-standing and important class of antibiotics.¹⁻² Waksman reported the first aminoglycoside streptomycin from the soil bacteria, *Streptomyces griseus*, in 1943 and showed that it possesses antituberculosis activity.³ However, antibiotic resistance against aminoglycosides has been more common, and this has led investigators to search for new aminoglycosides. Consequently, several aminoglycosides like gentamicin,⁴ tobramycin,⁵ apramycin,⁶ lividomycin⁷ and others⁸ were subsequently discovered and shown to be clinically useful in the treatment of infections by Gram-negative and selective Gram-positive bacterial, fungi, and *Pseudomonas* spp. Also semisynthetic derivatives like dibekacin,⁹ amikacin,¹⁰ arbekacin,¹¹ isepamicin,¹² and netilmicin¹³ were introduced during the 1970s when modifications of aminoglycosides such as *N*-acetylation, *O*-phosphorylation, and *O*-adenylation were recognized as resistance mechanisms that reduce the affinity of the drugs for rRNAs.³

The mode of action of aminoglycoside antibiotics has been thoroughly studied.³ Aminoglycosides contain several hydroxyl and amino groups that are important for the binding interactions with bacterial ribosomal RNAs. Such binary complex formation prevents translocation or proofreading during translation resulting in the death of the bacteria.¹⁴ In general, the amino groups of aminoglycosides are positively charged at neutral pH and thus interact favorably with the negatively charged phosphate backbone of RNAs. Also crystal structure analysis of rRNA complexes with aminoglycoside antibiotics showed the specific interactions that promote binding of aminoglycosides to RNAs.¹⁵⁻¹⁶ Although most aminoglycosides have different binding interactions, they all adversely affect translation. The details as to how their effects on translation lead to the bacterial death are not clear. One possibility is that aminoglycosides combine with ribosomal RNA leading to misread proteins.¹⁷⁻¹⁸ These abnormal proteins combine with the inner membrane and lead to destabilization. Another possibility is that aminoglycosides perturbs metabolism and respiration of bacteria which results in oxidative stress due to the increased formation of highly toxic hydroxyl radicals and superoxide production.²⁰

1.1.2. Aminoglycoside Structure

The general structure of aminoglycosides consists of an aminocyclitol linked to at least one amino sugar. Aminoglycosides thus contain a number of free hydroxyl groups and at least two amino groups. Based on the structure of the aminocyclitol, the aminoglycosides can be divided into four different groups: a) *myo*-inositol-derived, b) 2-deoxy-*scyllo*-inose-derived, c) 2-*epi*-5-*epi*-valiolone-derived, and d) cyclopentitol-derived aminoglycosides (Figure 1-1). Figure 1-1 shows examples of theses aminoglycosides.²¹

myo-Inositol-derived aminoglycosides include streptomycin, bluensomycin, spectinomycin, the fortimicins, and kasugamycin. These molecules play critical roles in many cellular processes and are involved in cell wall biogenesis, stress response, and signal transduction pathways.²²⁻²⁴ These aminoglycosides are derived from *myo*-inositol

1-phosphate (MIP) which is biosynthesized from D-glucose 6-phosphate (G6P) by *myo*inositol-1-phosphate synthase found in both prokaryote and eukaryotes.²⁵ MIP is converted into aminoinositols via a series of reactions catalyzed by phosphatases, dehydrogenases, and aminotransferases.

2-Deoxy-scyllo-inose-derived aminoglycosides commonly contain 2deoxystreptamine (2-DOS) as a core scaffold and are decorated with several monosaccharides at the C4, C5 and C6 positions. These aminoglycosides include the neomycins, gentamicins, apramycin kanamycins, and butirosins. They are biosynthetically derived from paromamine. Clinically important aminoglycosides in this group are further subclassified according to the substitution pattern of DOS with aminosugars. Thus, neomycin and paromomycin are 4,5-disubstituted aminoglycosides, whereas gentamicin, kanamycin, and tobramycin are 4,6-disubstituted aminoglycosides. The details regarding the biosynthesis of 2-DOS containing aminoglycosides are described below.

The third class of aminoglycosides is characterized by a C₇N aminocyclitol core structure. These aminoglycosides include acarbose, the oligostatins, the trestatins, validamycin A, pyralomycin 1a, and cetoniacytone.²¹ Their chemical structures commonly contain an unsaturated aminocyclitol unit, valienamine, although some compounds contain modified forms of valienamine. The biosynthetic origin of valienamine in these aminoglycosides was established to be 2-*epi*-5-*epi*-valiolone by isotope-labeling experiments.²⁶⁻²⁸

The final class of aminoglycosides is relatively rare in nature. These aminoglycosides include pactamycin,²⁹ an antibiotic antitumor agent from *Streptomyces pactum*, and allosamidin, a chitinase inhibitor isolated from *Streptomyces* sp. No 1713.³⁰ These aminoglycosides contain a unique cyclopentitol moiety. Pactamicin consists of

three cyclic units, the cyclopentitol, a 6-methylsalicyclic acid functionality, and an *m*-aminoacetophenone moiety which are derived from different biosynthetic pathways. The cyclopentitol moiety is derived from glucose, while the other two are generated via polyketide synthase (PKS) chemistry and a pathway reminiscent of shikimate biosynthesis.³¹ Allosamidin has a distinctive structure consisting of a cyclopentitol moiety called allosamizoline and two monosaccharide units of *N*-acetyl-D-allosamine. Cyclopentitol moieties are also found in carbocyclic nucleotides like neplanocin A, aristeromycin, adecypenol, queuosine, and epoxyqueuosine, and bacterial derived hopanoids.²¹



Figure 1-1. Structures of selected aminoglycoside antibiotics

1.1.3. Biosynthesis of Aminoglycosides

This section focuses on the biosynthesis of aminoglycosides and 2-DOS. Biosynthetic studies of other aminoglycosides are covered by several reviews.^{8,21,32} Many laboratories are involved to characterize the biosynthetic genes and enzymes for aminoglycoside biosynthesis and their biotechnological applications. As aminoglycoside biosynthetic genes are often reported with different names, the set of gene annotations use by Piepersberg⁸ and Eguchi³³ will also be used consistently here.

Early feeding experiments with radioisotope- and stable-isotope primary metabolites such as glucose and glucosamine showed that the majority of aminoglycoside components are derived from sugars.³⁴ Mutant studies were also used to elucidate the structures of biosynthetic intermediates³⁵ and to determine the order of the biosynthetic steps.³⁶ However, the mechanisms of the enzymatic reactions and the timing of modifications in the biosynthesis of aminoglycosides were not fully elucidated until now.

The first genetic analysis of aminoglycoside biosynthesis was performed in 1987 and only a portion of the complete gene cluster for streptomycin was identified.³⁷ Several biosynthetic genes located both up- and downstream of the identified portion were discovered later using hybridization studies, and determining the nucleotide sequences.³⁸ Since then, analyses of more than 30 aminoglycoside biosynthetic gene clusters have been accomplished and deposited in the public gene data bank. The discovery of 2deoxy-*scyllo*-inose (2-DOS) synthases from the butirosin producing strain *Bacillus circulans* in 1999³⁹ boosted the identification of additional 2-DOS-containing aminoglycoside biosynthetic gene clusters such as butirosin, neomycin, ribostamycin, paromamine, lividomycin, kanamycin, tobramycin, gentamicin, sisomycin, istamycin, apramycin and hygromycin B. Comparison of the gene clusters has led to speculation that some conserved biosynthetic enzymes encoded in different aminoglycoside gene clusters are responsible for the biosynthesis of the common intermediates. There are also many cases of gene clusters encoding aminoglycosides that have mixed structures of other aminoglycosides, such as gentamicin gene cluster that shares modification patterns and subunits with kanamycin and fortimicin gene cluster. Under such situation, the information from biosynthetic gene cluster from one aminoglycoside predicts the modification of the intermediates step by step in the biosynthetic pathway of another aminoglycoside with similar gene cluster.

1.1.3.1. Biosynthesis of 2-Deoxystreptamine (DOS)

2-DOS, also known as 1,3-diamino-4,5,6-trihydroxycyclohexane, is found in the majority of aminoglycosides as a central moiety and plays a critical role in their biological function by interacting directly with the 16S ribosomal RNA subunit.^{1,40-41} The biosynthesis of 2-DOS has been investigated in detail.⁴²⁻⁴⁴ Early studies using ¹³C-labeled substrate in neomycin isolated from the culture of *Streptomyces fradiae* demonstrated that 2-DOS originates from D-glucose, with C-1 and C-2 carbons of 2-DOS corresponding respectively to the C-1 and C-6 carbons of glucose (Scheme 1-1).⁴⁵ The nitrogen from glucosamine is not incorporated into 2-DOS.³⁴ Experiments with 6,6-dideuterated glucose in ribostamycin from *Streptomyces ribosidificus* demonstrated that both deuteriums at C-6 are retained in 2-DOS and rearrangement proceeded with retention of stereochemistry.⁴⁶⁻⁴⁷ Labeling experiments showed that the C-4 proton of glucose is lost during the biosynthesis of 2-DOS.⁴⁸⁻⁴⁹ The *in vitro* assay with cell-free extracts of *S. fradiae* demonstrated that the conversion of glucose-6-phsophate to 2-DOS is NAD⁺-dependent.⁵⁰⁻⁵¹



Scheme 1-1. Biosynthesis of 2-DOS

BtrC, 2-deoxy-scyllo-inosose (2-DOI) synthase, was purified from the butirosinproducing bacterium Bacillus circulans, and shown to be dependent on both cobalt and NAD^{+, 52-54} Enzymes homologous to BtrC have been identified in all 2-DOS-containing aminoglycoside biosynthetic pathways (Scheme 1-1).55 The mechanism of BtrC was proposed based on its X-ray crystal structure and compared to that of 5-dehydroquinate (DHQ) synthase, because they have moderately similar amino acid sequences (26–34%) (Scheme 1-2).⁵⁶ The two enzymes have different conserved amino acids at positions 243 and 239: asparagine and arginine in DHQ vs glutamate and glycine in 2-DOI synthase. Also 2-DOI synthase has an additional conserved glutamate residue at 235. In the DHQ synthase reaction (Scheme 1-2), oxidation at C4 occurs in the first step. Asn268 is not directly involved in the abstraction of a proton at C5, but the substrate phosphate group can abstract a proton to provide an α,β -unsaturated ketone intermediate.⁵⁷ After reduction of the carbonyl group at C-4 and ring opening of the hemi acetal, the stereoselective intramolecular aldol condensation gives DHQ. The proposed mechanism of BtrC catalysis is similar to that of DHQ synthase. The difference between them is the acid-base chemistry during proton abstraction at C5 (the second step).⁵⁸

A. 3-dehydroquinate (DHQ) synthase



Scheme 1-2. Proposed mechanisms of DHQ synthase and 2-DOI synthases. The amino acid position numbers derived from *Aspergillus nidulans* DHQ and BtrC from 2-DOI synthase are indicated.

The conversion of 2-DOI to 2-deoxy-*scyllo*-inosamine (2-DOIA) is catalyzed by a PLP-dependent aminotransferase BtrS.³⁰ This enzyme was also found to catalyze the conversion of 2-deoxy-3-keto-*scyllo*-inosamine (keto-2-DOIA) to 2-deoxystreptamine (2-DOS), suggesting that this enzyme play a dual role. This phenomenon is also found in the biosynthesis of other aminoglycosides including neomycin, gentamicin, and kanamycin.²⁶

The next step from 2-DOIA to keto-2-DOIA is catalyzed by the radical SAM enzyme BtrN (this enzyme is described in detail in Section 1.3.1).²⁹ During 2-DOS biosynthesis, only the butirosin biosynthetic pathway involves a radical SAM enzyme for oxidation of hydroxyl group at C1. Other biosynthetic pathways involve the NADP⁺-dependent enzymes for this step.

1.1.3.2. Biosynthesis of Paromamine

Paromamine is the most common intermediate during biosynthesis of the 2-DOSbased aminoglycosides besides neamine. The biosynthesis of paromamine consists of two steps: glycosylation of 2-DOS with *N*-acetylglucosamine (GlcNAc) and deacetylation (Scheme 1-3). BtrM identified in the butirosin biosynthetic gene cluster and its homologues, NeoD in neomycin biosynthesis, GenM1 in gentamicin biosynthesis, KanM1 in kanamycin biosynthesis, RibM in ribostamycin biosynthesis, and TobM1 in tobramycin biosynthesis, were suggested to catalyze transfer of NDP-D-glucosamine to 2-DOS.⁵⁵ It was demonstrated with cell-free extracts of *E. coli* expressing *neoM* that NeoM catalyzes the glycosylation reaction.⁵⁹ The function of BtrD in the butirosin gene cluster is to catalyze 2'-*N*-acetylparomamine deacetylation, and was shown by studying the recombinant BtrD and 2'-*N*-acetylparomamine.⁶⁰ Also GenD in gentamicin, KanN in kanamycin, and RibD in ribostamycin were demonstrated to be 2'-*N*-acetylparomamine deacetylases by heterologous expression of the respective genes to produce paromamine in *S. venezuelae*.⁶⁰



Scheme 1-3. Biosynthesis of paromamine from 2-DOS

1.1.3.3. Biosynthesis of Ribostamycin, Neomycin C, and Neomycin B

Ribostamycin is an intermediate in the production of 4,5-disubstituted aminoglycosides, including butirosin and neomycin. The transformation of paromamine to neamine is catalyzed by the FAD-dependent dehydrogenase NeoQ and the PLP-dependent aminotransferase NeoB (Scheme 1-4).⁶¹ The formation of ribostamycin, a common intermediate fin the biosynthesis of neomycins and butirosin, from neamine occurs in two steps: ribosylation by BtrL using 5-phosphoribosyl-1-pyrophosphate (PRPP) as a ribosyl donor to form 5"-phosphoribostamycin and subsequent hydrolysis by BtrP.⁶² BtrL and BtrP also catalyze the ribosylation of paromamine to provide 6'-deamino-6'-hydroxyribostamycin.⁶² RibL and RibP in the ribostamycin producer *S. ribosidificus* have the same functions as BtrL and BtrP.⁶³

The last glycosylation in neomycin biosynthesis to attach the fourth ring is catalyzed by NeoF which transfers UDP-*N*-acetylglucosamine onto ribostamicin to generate 2"-*N*-acetyl-neomycin Y_2 .⁵⁹ The function of NeoF was also confirmed by an *in vivo* in-frame gene deletion of *neoF* in *S. fradiae*, resulting in the accumulation of ribostamycin.⁶⁴ The deacetylation of 2"-*N*-acetyl-neomycin Y_2 to provide neomycin Y_2 is catalyzed by NeoD, which also recognizes the pseudodisaccharide 2-acetylparomamine (Scheme 1-3 and Scheme 1-4).⁵⁹ In the gene cluster of neomycin, only one BtrD-homologous deacetylase, NeoD, is present. From the result that the *neoD S. fradiae*

mutant still produces neomycin from neamine, the second deacetylase in *S. fradiae* should exist for the deacetylation of tetrapseudosaccharide.⁶⁴ The aminotransfer at C6" is catalyzed by two enzymes, the dehydrogenase NeoQ and aminotransferase NeoB.⁶⁵⁻⁶⁶ NeoQ and NeoB can also recognize the pseudodisaccharide paromamine to provide neamine (Scheme 1-5).⁶⁷ The final epimerization is catalyzed by a radical SAM enzyme NeoN. NeoN is described in detail in section 1.3.2.



Scheme 1-4. Biosynthetic pathway for neomycin B

1.1.3.4. Biosynthesis of Kanamycins

Compared to 4,5-disubstituted aminoglycosides such as ribostamycin and neomycin, kanamycins isolated from S. kanamyceticus are 4,6-disubstituted pseudotrisaccharides with a 3-aminosugar moiety at the C6-hydroxyl group of 2-DOS. The Yoon group proposed parallel biosynthetic pathways based on heterologous gene expression of kanamycin genes in S. venezuelae (Scheme 1-5).⁶⁸ KanM1 is able to accept both UDP-glucosamine and UDP-glucose as the glycosyl donor to provide respectively paromamine and 2'-deamino-2'-hydroxyparomamine.⁶⁸ The glycosyltransferase KanM2 can use four different substrates such as paromamine, neamine, 2'-deamino-2'hydroxyparomamine, and 2'-deamino-2'-hydroxyneamine as glycosyl acceptors. Also KanM2 can accept two glycosyl donors such as UDP-glucose and UDP-kanosamine (the biosynthetic step using UDP-glucose is not shown in Scheme 1-5. After UDP-glucose is accepted by KanM2 to give 3'-deamino-3'-hydroxykanamycin C,⁶⁹ subsequent amino group introduction at C3" to produce kanamycin C is unclear even though KanD2 and KanS2 are assigned to this step from the gene analysis of kanamycin biosynthesis). The dehydrogenase KanQ and aminotransferase KanB also show substrate flexibility. They both pseudodisaccharides such as paromamine 2'-deamino-2'accept and hydroxyparomamine as well as pseudotrisaccharides such as kanamycin C and kanamycin X.⁶⁸ An α -ketoglutarate-dependent non-heme iron dioxygenase, KanJ, and a oxidoreductase, KanK, catalyze the deamination of kanamycin B to kanamycin A.⁷⁰ KanJ catalyzes the hydroxylation at C2' to provide a hemiaminal intermediate which is converted into 2'-oxokanamycin B by the release of ammonia. Then a NADH-dependent reductase KanK reduces the carbonyl group of 2'-oxokanamycin B to produce kanamycin A.



Scheme 1-5. Biosynthetic pathway for kanamycin A

1.1.3.5. Biosynthesis of Gentamicins

Gentamicins are aminoglycoside antibiotics produced by *Micromonospora echinispora*.⁴ Gentamicins are composed of a 2-deoxystraptamine (2-DOS) core substituted at C4 with a purpurosamine moiety and at C-6 with a garosamine moiety. Biosynthetic studies of gentamicin had been carried out involving isolation of gentamicin-related compounds from the wild type producers as well as mutants to identify the biosynthetic intermediates. The major components which compose approximately 80% of gentamicin include gentamicin C_1 , C_{1a} and C_2 , and the minor ones include gentamicin A, A₁, A₂, A₃, B, X₂,⁷¹⁻⁷³ and antibiotics JI-20A and JI-20B.⁷⁴ Using these compounds, Tilley *et al* carried out biosynthetic studies of gentamicin and showed interconversion of gentamicin-related components using biological catalysts and extracts.⁷⁵ They used *M. purpurea* Paro 346, a mutant that produces paromamine but none of the gentamicin C's. Since gentamicin C_{1a} was not converted into gentamicin C_2 or gentamicin C_1 , the *C*-methyl transferase responsible for methylation at the C6' position apparently could not recognize gentamicin C_{1a} as a substrate. Using the mutant, they also showed the transformation of JI-20A to gentamicin C_{1a} and gentamicin C_{2b} whereas JI-20B and G418 were converted to gentamicin C_2 and gentamicin C_1 . These results supported the hypothesis that two different pathways are operant for the production of different gentamicin C's. Furthermore, gentamicin A and gentamicin X_2 were transformed by the mutant into gentamicin C_{1a} , gentamicin C_2 and gentamicin C_1 , thus it appears that these species serve as precursors to the gentamicins C's before the branch point. From these data, a biosynthetic pathway was proposed (Scheme 1-6).

The gene cluster of gentamicin was sequenced by the Sohng group⁷⁶ and Wellington group⁷⁷ in 2004 and the Piepersberg group⁸ in 2006, all from *Micromonospora echinospora* ATCC 15835 (Figure 1-2). Homology search results for each genes in the cluster are summarized in Table 1-1. The majority of the genes in the gentamicin gene cluster show high similarity with those in the fortimicin,⁸ sisomicin,⁷⁸ Frankia Ccl3,³³ and istamicin³³ clusters (structures are shown in Figure 1-1). GenC, GenS1, GenE, GenM1, and GenD were assigned to be responsible for paromamine formation, similar to the butirosin biosynthetic pathway (Scheme 1-2 and Scheme 1-3). GenD1, GenD2, GenX, GenU, GenV, GenN, GenS2, and GenM2 were expected to be involved in the biosynthesis of the garosamine moiety. GenM2, GenD2, and GenS2 were

expected to be responsible for xylose transfer. GenD1 and GenN were estimated to be involved in the modification of the xylose moiety. GenK, GenB2, GenB4 and GenG are conserved in the gentamicin and fortimicin biosynthetic gene clusters. GenK may catalyze *C*-methylation of the purpurosamine moiety. GenH, GenI, GenQ, and GenB1 (or GenB3) are conserved in the gentamicin, sisomycin, fortimicin, and istamycin biosynthetic gene clusters. They seem to be responsible for the biosynthesis of the common characters of the purpurosamine moiety. GenX, GenU and GenV remain uncharacterized.

Another interesting finding is that the gentamicin, fortimicin, and Frankia clusters have queuosine biosynthetic genes such as GenO, GenW, GenA, and GenF, and a radical SAM enzyme like GenF. Blast search against known *Streptomyces* genomes (*S. griseus*, *S. avermitilis*, *S. coelicolor*, and *S. acabiei*) shows *Streptomyces* does not have queuosine biosynthetic genes. Furthermore, the Flankia CCl3 aminoglycoside (fortimicin) biosynthetic gene cluster has queuosine biosynthetic genes (Table 1-1). Considering queuosine biosynthetic genes are missing in known genomes of hi-GC Gram-positive bacteria, all gentamicin/fortimicin clusters have a set of genes for queuosine biosynthesis (QueD, QueE, QueC, QueF, and TGT) in the cluster. Moreover, other aminoglycoside biosynthetic gene clusters do not have those homologues. Since these genes are not present in the istamycin cluster, it is less likely that those genes are important for 3,4-dideoxygenation steps. One possible scenario is that homologues of QueD, QueE, QueC, and QueF produce preA1 and the TGT homologue catalyzes the transfer of proQ1 to rRNA for gentamicin/fortimicin self-resistance mechanism.



Figure 1-2. Gene cluster of gentamicin

Proposed Function	Piepers- berg	Welling- ton	Sohng	Siso- micin cluster	Forti- micin cluster	Frankia Ccl3 cluster	Ista- mycin cluster
hypothetical protein		GntT	GacN				
putative Trp-tRNA ligase		GntU	GacM				
queuosine/archaeosine tRNA-ribosyltransferase	GenO	GntV	GacL	Sis3	FosA	3378	
putative ribosomal methyltransferase	GmrB	GrmO	GacL	Sis4			
	GenB1	GntW	GacK				
gentamicin (hexosaminyl-	GenB2	GntL	GacE	Sis5 Sis18 F	ForB	3369	IstB
6-) aminotransferase	GenB3	GntJ	GacB				
	GenB4	GntH	GacC				
hexose-6'-dehydrogenase	GenQ	GntX	GacJ	Sis6	ForQ	3360	IstQ
SDR/putative gentamicin oxidoreductase	GenD3	GntY	GtmH	Sis7			
(<i>N</i> -acetyl)hexos- aminyltransferase	GenM1	GntZ	GtmG	Sis8	ForM	3362	IstM
putative ribosomal methyltransferase	GmrA	GrmA	GtmF	Sis9			

putative <i>L</i> - glutamine:ketocyclitol aminotransferase	GenS1	GntA	GtmB	Sis10 Sis15	ForS	3363	IstS
	GenS2	GntF	GtmD				
2-deoxy-scyllo-inose synthase	GenC	GntB	GtmA	Sis11			IstC
NAD(P) dependent oxidoreductase	GenD2	GntC	GtmC	Sis12			
gentamicin glycosyltransferase II	GenM2	GntD	GtmE	Sis13			
B12 radical SAM dependent oxidoreductase/ methyltransferase	GenD1	GntE	GtmI	Sis14			
GTP cyclohydrolase /NADPH dependent 7- cyclo-7-deazaguanine reductase	GenW	GntG	GacA	Sis16	FosC	3383	
biosynthetic aminoglycoside 3'-	GenP	GntI	GtmJ	Sis17	ForP	3368	
B12 radical SAM dependent C- methyltransferase	GenK	GntK	GtmJ		ForK	3370	
hypothetical protein	GenX	GntM	GacF	Sis19			
hypothetical protein	GenU	GntN	GacG	Sis20			
deoxyinosamine dehydrogenase	GenE	GntP	GacH	Sis22	ForE	3364	IstE
putative cation antiporter	GenY	GntQ	GacI	Sis23	ForY		
ATP utilize enzyme (QueC)	GenA	GntR		Sis24	FosE	3379	
radical SAM enzyme (QueE)	GenF	GntS		Sis25	FosF	3380	
Que homologue	GenG				FosG	3381	
export protein	GenV GenH GenI	GntO	GtmK	Sis21 Sis26	ForI ForH ForJ ForV	3359 3358	IstI IstH IstJ
SAM-dependent methyltransferase	GenT			Sis28	ForT	3373	
putative <i>N</i> - acetylhexosaminyl deacetylase	GenD	GtmM	Sis29	ForD	3367	IstD	
---	------	------	-------	-------	------	-------	
putative gentamicin methyltransferase	GenN		Sis30				
hypothetical				ForW	3361	IstW	
putative fortimicin <i>O</i> -methyltransferase				ForO	3365	IstO	
radical SAM family protein				ForL2	3366	IstL2	
fortimicin A <i>N</i> - fortimidoyltransferase				ForZ	3371	IstZ	
putative fortimicin production protein				ForX	3376	IstX	
putative myo-inositol 3- dehydrogenase				ForG	3374		
fortimicin <i>N</i> - methyltransferase				ForN	3373	IstN	
inositol-1- monophosphatase				ForA			
radical SAM protein				ForL	3357	IstL	
					3375	IstL3	

Table 1-1. Gene analysis of gentamicin







GenN or GenT

H₃C



GenD2 GenS2 GenN





















GenP ???







H_eC

HO JI-20Ba

HO-





7_NH2

6 юн

CH3

-0H gentamicin C₂ gentamicin A

GenD1

он

\ и́нсн₃ сн₃

GenK

gentamicin X₂







Scheme 1-6. Biosynthetic pathway for gentamicins. The activity of the enzymes with red colors has already been demonstrated.

Many biotransformations in the biosynthetic pathway of gentamicin have been demonstrated. The Yoon group confirmed that GenM2 is required for the glycosylation that forms the pseudotrisaccharide from paromamine and UDP-xylose.⁷⁹ They used an engineered paromamine-producing strain of S. venezuelae to generate gentamicin A₂. However, it was noted that the amount of gentamicin A_2 produced with S. venezuelae/pYJ505 was 4.5 µg/L, and could only be detected by LC-MS analysis. This was mainly due to limited amount of NDP-xylose present in S. venezuelae host cells. When cell free extracts of S. venezuelae/pYJ505 were supplied with UDP-xylose, gentamicin A₂ production was enhanced up to 8-fold, supporting that the glycosyl donor is UDP-xylose. Two oxidation and amination reactions occur in the biosynthetic pathway of gentamicins. One is the conversion of gentamicin A_2 to gentamicin A, and the other is the conversion of gentamicin X₂ to JI-20A or G-418 to JI-20B. Leadlay and Sun et al demonstrated through gene mutation experiments that the first oxidation and amination are catalyzed respectively by GenD2 and GenS2,⁸⁰ and the second transformations are catalyzed by GenB1 and GenQ.⁸¹ For the two *N*-methyltransfer reactions from gentamicin A2 to gentamicin A and from gentamicin $C_{1a}\xspace$ to gentamicin $C_{2b}\xspace$ or from gentamicin C2 to gentamicin C1, only the methylation reaction at N3" has been established to be catalyzed by GenN.^{80,82} Two C-methyltransfer reactions have now been shown to be catalyzed by cobalamin-dependent radical SAM enzymes (GenD1 and GenK). GenD1 catalyzes the methylation at C4" of garosamine,⁸⁰ and is described in detail in Sections 2.1 and 2.2. GenK catalyzes the methylation at C6' and is discussed in Chapters 2 and 3. Two epimerization reactions are also required for gentamicin biosynthesis: the first is from JI-20B to JI-20Ba,⁸¹ and the second is from gentamicin C2a to gentamicin C2.⁸³ These two epimerizations are both catalyzed only by GenB2 and are reversible.

The remaining uncharacterized steps are the deoxygenation at C3' and C4' and *N*-methylation at the end to complete the biosynthesis of gentamicins. While GenP is known to catalyze phosphorylation of the hydroxyl group at C3',⁸⁴ the remaining enzymatic reactions are enigmatic. From comparison of the genes in the gentamicin and fortimicin clusters, several genes could be involved in the dideoxygenation reaction, such as GenV, GenH, and GenI. These genes have been assigned as membrane associated exporters. However, these proteins do not resemble membrane proteins, and homologues of these enzymes are not present in any other organisms based on BLAST analysis.

1.1.3.6. Biosynthesis of Other Aminoglycosides

Biosynthetic studies of other aminoglycosides with 2-DOS like tobramycin, apramycin, and lividomycin have not been successful so far. Conserved biosynthetic enzymes are expected to be responsible for the construction of the common biosynthetic intermediate, 2-DOS and paromamine based on the gene analysis.^{8,33}

Tobramycin (3'-deoxykanamycin B) has been identified from two strains, *Streptoalloteichus hindustanus* and *Streptomyces tenebrarius* (Figure 1-1).⁸ Both microorganisms produce apramycin as well, indicating that the biosynthesis of tobramycin and apramycin share some biosynthetic genes. The biosynthesis of tobramycin includes deoxygenation of paromamine or kanamycin C. Comparison of the tobramycin/apramycin, kanamycin and lividomycin biosynthetic gene clusters has suggested that AprD3 and AprD4 catalyze deoxygenation reaction during the biosynthesis of tobramycin and apramycin.^{8,33} AprD3 and AprD4 are described in detail in Chapter 4.

1.2. RADICAL S-ADENOSYL-L-METHIONINE (SAM) ENZYME

S-Adenosyl-L-methionine (SAM) is a versatile compound in nature. This compound is primarily known for its electrophilic methyl that is used to methylate nucleotides, amino acids, or other molecules containing sulfur, amino, and hydroxyl nucleophiles. However, in the early 1970s, lysine 2,3-aminomutase is shown to use SAM in a novel way.⁸⁵⁻⁸⁶ This enzyme is sensitive to air and its activity depends on the presence of dithionite and SAM. Further studies showed that the 5'-deoxyadenosyl moiety of SAM is homolytically cleaved and the resulting 5'-deoxyadenosyl radical facilitates hydrogen abstraction from the substrate of lysine 2,3-aminomutase,⁸⁷ which is similar to the reactions catalyzed by adenosylcobalamin-dependent mutases using the 5'deoxyadenosyl radical. EPR analysis revealed the presence of a [4Fe-4S] cluster.⁸⁸⁻⁸⁹ Studies of several other enzymes such as pyruvate formate-lyase activase,⁹⁰⁻⁹⁴ biotin synthase,⁹¹ and ribonucleotide triphosphate reductase III,⁹⁵⁻⁹⁶ led to the characterization of a new class of enzymes that utilize a reduced iron-sulfur cluster and SAM to initiate their catalyses. This class enzymes, which has grown to approximately 64000 based on bioinformatics analysis,⁹⁷ are across all phylogenetic kingdoms and are now referred as the radical SAM superfamily.⁹⁸

Radical SAM enzymes have a characteristic Cx_3Cx_2C sequence motif conserved in most of superfamily members, although enzymes exhibiting variations of this motif are also known (see in Section 4.1). In the active site of a typical radical SAM enzyme, the three cysteine residues coordinate three of the four irons of a [4Fe-4S] cluster, and one of the four iron atoms is coordinated by the amino and carboxylate moieties of SAM (Figure 1-3).⁹⁹⁻¹⁰⁰ Reducing agents such as flavodoxin or dithionite can provide one electron to reduce the [4Fe-4S]²⁺ cluster to generate the catalytically active [4Fe-4S]⁺. The reduced [4Fe-4S]⁺ cluster can reduce SAM leading to homolysis of the S-C5' bond to give the a 5'-deoxyadenosyl radical (5'-dAdo•) and methionine. The 5'-dAdo• s then used to abstract a hydrogen atom from the substrate in a stereo- and regiospecific manner to generate a substrate radical which can undergo further reaction such as methylation, sulfur insertion, oxidation, dehydrogenation, methylthiolation, and complex carbon skeleton rearrangements.¹⁰¹ In most radical SAM enzymes, SAM is consumed during catalysis, and methionine and 5'-dAdo along with the product are produced in a 1:1:1 ratio (pathway A in Figure 1-3). However, when SAM is used catalytically, the product radical abstracts a hydrogen atom from 5'-dAdo to afford the product and 5'-dAdo•. The latter recombines with methionine to regenerate SAM (pathway B in Figure 1-3).

A second auxiliary iron-sulfur cluster has been found in many radical SAM enzymes. In some cases, the second cluster is involved in the reaction (Figure 1-4). For example, in the biosynthesis of biotin, one sulfur atom in the second cluster is inserted between C6 and C9 of dethiobiotin to generate biotin (Figure 1-4A).¹⁰¹ In the cases of MiaB and RimO, the nucleophilic persulfide-bound moiety of the second cluster accepts the methyl group from SAM (Figure 1-4B).¹⁰² The resulting methylated sulfur atom of the polysulfide chain reacts with the substrate radical to yield the methylthiolated product and a [4Fe-4S] cluster with a terminal persulfide. HydG has another auxiliary [4Fe-4S] cluster coordinated with a Cx₂Cx₂₂C motif (Figure 1-4C).¹⁰³ A tyrosine bound to an auxiliary *C*-terminal [4Fe-4S] cluster reacts with 5'-dAdo• generated by SAM and an *N*-terminal [4Fe-4S] cluster to provide a tyrosine radical. Heterolytic cleavage of this tyrosine radical at the C α -C β bond forms a transient 4-oxidobenzyl (4OB) radical and the dehydroglycine bond to the C-terminal [4Fe-4S] cluster, which has been detected by EPR spectroscopy.¹⁰⁴ Other secondary [4Fe-4S] clusters are expected to act as electron donors.¹⁰⁵



Figure 1-3. Mechanism for radical SAM cleavage



Figure 1-4. Roles of the secondary [4Fe-4S] clusters in selected radical SAM enzymes. A) BioB in biotin biosynthesis, B) RimO in the methylthiolation of aspartate89 of the ribosomal S12 protein, C) HydG in the biosynthesis of the H-cluster of the [FeFe]-hydrogenase

1.3. RADICAL SAM ENZYMES IN AMINOGLYCOSIDE BIOSYNTHESIS

In the biosynthetic pathways of aminoglycosides, several enzymes have been assigned as radical SAM enzymes such as BtrN in the butirosin pathway, NeoN in the neomycin pathway, RibN in the ribostamycin pathway, ParN in the paromomycin pathway, LivN and LivW in the lividomycin pathway, GenD1, GenK and GenF in the gentamicin pathway, IstL2, IstL, and IstL3 in the istamycin pathway, ForK, FosF, ForL2, and FosL in the fortimicin pathway, AprD4 in the apramycin/tobramycin pathway, HygY in the hygromycin pathway, and SpcY in the spectinomycin pathway,^{8,33} but only a few radical SAM enzymes (i.e. BtrN, NeoN, and GenD1) in aminoglycoside biosynthesis have been characterized *in vitro*. GenD1 is discussed in detail in Section 2.1 and Section 3.1. The *in vitro* characterizations of GenK and AprD4 are presented in this dissertation in detail.

1.3.1. BtrN

During the biosynthesis of 2-deoxystreptamine (2-DOS) in the butirosin producer *B. circulans*, BtrN catalyzes the oxidation of the alcohol at C1 of 2-deoxy-*scyllo*-inosamine (2-DOIA) to form keto-2-deoxy-*scyllo*-inosamine (keto-2-DOIA) (another named 3-amino-2,3-dideoxy-*scyllo*-inosose (amino-DOI)) under anaerobic conditions (Scheme 1-1).¹⁰⁶ From analysis of the butirosin biosynthetic gene cluster, BtrN has been shown to possess canonical the CxxxCxxC motif in its primary sequence, indicating that this enzyme is a radical SAM enzyme.¹⁰⁶ Electron paramagnetic resonance (EPR) studies performed at 50 °K with reconstituted BtrN but without SAM and 2-DOIA showed characteristic *g* values of 1.92 and 2.04 for the [4Fe-4S] cluster of radical SAM enzymes.

These *g* values can be compared with reconstituted BtrN in the presence of dithionite and SAM (g = 1.83, 1.99) or with additional SAM and 2-DOIA (g = 1.87, 1.96, 2.05).¹⁰⁷

Gene analysis of BtrN showed the existence of eight cysteine residues, suggesting that a second [4Fe-4S] cluster may exist.¹⁰⁸ Mutagenesis studies indicated that reconstituted BtrN contains two [4Fe-4S] clusters. When the three cysteine residues of the CxxxCxxC motif were changed to alanine, the enzyme was found to still contain a single [4Fe-4S] cluster based on iron/sulfur titration and Mössbauer spectroscopy.¹⁰⁸ Generation of variants in which the remaining cysteine residues were changed to alanines showed that the C69A mutant provided similar activity as the wild type, but the C235A mutant displayed reduced solubility and low activity. Mutation of Cys169, C187 and C232 produced insoluble proteins. These results led to the proposal that the three latter residues serve as ligands to bind the second [4Fe-4S] cluster.

Kinetics analysis revealed that BtrN proceeds with an ordered mechanism because uncompetitive substrate inhibition was observed with amino-DOI, but not with SAM.¹⁰⁶ In BtrN catalysis, one equivalent of SAM versus 2-DOIA is consumed and an equal number of moles of 5'-dAdo, methionine, and amino-DOI are generated. Assays of BtrN with [3-²H]-2-DOIA generated a mixture of unlabeled, monodeuterated, and dideuterated 5'-dAdo, while incubation of nonlabeled 2-DOIA in the deuterium buffer did not provide incorporation of deuterium at all. These results indicate that 5'-dAdo radical abstracts a hydrogen atom from the C3 position of 2-DOIA.

Determination of the X-ray structure of this enzyme with and without substrate further clarified the mechanism and role of the auxiliary cluster.¹⁰⁹ The auxiliary cluster is fully ligated both in the presence and in the absence of the substrate, which excludes the possibility that this cluster coordinates the substrate. In addition, 2-DOIA is bound in a hydrophilic pocket between SAM and the auxiliary clusters. Based on the observation

that 2-DOIA binds in an equatorial chair conformation and hydrogen bonding interactions with the functional group of the substrate, the auxiliary cluster in BtrN is suggested to function as an electron acceptor during the dehydrogenation reaction. Also the putative base, Arg152, involved in the BtrN catalysis was identified. Formation of the α -hydroxyalkyl radical after H-atom abstraction by 5'-dAdo radical promotes deprotonation of the C3-hydroxy functional group by decreasing its p K_a . The proposed mechanism of the BtrN reaction is shown in Figure 1-5.



Figure 1-5. Proposed mechanism of the BtrN reaction

The substrate flexibility of BtrN was examined using several cyclitol and sugar compounds (Figure 1-6).¹⁰⁶ Only three substrate analogues such as DOS, *scyllo*-inositol, and *myo*-inositol showed activity with BtrN in terms of the formation of 5'-dAdo. These results indicate that BtrN exhibits strict specificity for cyclitols and not for sugars.



Figure 1-6. Substrate specificity of BtrN

1.3.2. NeoN

In the neomycin biosynthetic pathway, NeoN was assigned to catalyze the epimerization at C5^{III} of neomycin C to generate neomycin B.^{8,33} ParN in paromomycin, and LivN in lividomycin are also assigned to have analogous functions because they have similar structures (Figure 1-1). These enzymes contain the canonical CxxxCxxC motif that is present in most radical SAM enzymes. The reconstituted and reduced NeoN with SAM and neomycin C under anaerobic conditions provided neomycin B.¹¹⁰ NeoN also catalyzes the conversion of neomycin Y_2 to neomycin Y_1 , indicating that the amino group at C6^{III} is not necessarily involved in substrate recognition (Scheme 1-7). In the NeoN catalytic cycle, SAM is not utilized catalytically based on the observation that one equivalent each of SAM and the substrate are consumed, while equal amounts of 5'-dAdo and neomycin B are generated.¹¹⁰



Scheme 1-7. Epimerization of NeoN

The NeoN reaction carried out in a buffer prepared with deuterium oxide was analyzed by LC-ESI-MS and NMR, demonstrating the incorporation of a single deuterium atom at the C-5" of neomycin B. Mutation studies of NeoN with a triple cysteine mutant for the radical SAM motif (C26/30/33A) did not show epimerization activity, indicating that these cysteine residues are critical for catalysis. The assay of NeoN with C249A provided a new product, a five-membered carbocyclic structure with an aminomethyl group confirmed by 2D NMR analysis (Figure 1-7B). This result suggests that the C249 thiol provides the hydrogen atom at C5" position in which 5'dAdo• abstracts to complete the epimerization of neomycin C. EPR spectroscopic studies performed at 10K with reduced and reconstituted NeoN showed characteristic g values of 2.03 and 1.92. However, in the presence of SAM, the g values changed to 2.00, 1.90, and 1.86, indicating the coordination of SAM to the $[4Fe-4S]^+$ cluster. Addition of neomycin C did not affect the shape of the signal and this result suggests that neomycin C does not coordinate directly with the [4Fe-4S]⁺ cluster even though the substrate may be located close to the cluster. EPR analysis with the reduced C249A variant, neomycin C and SAM at 10 K revealed almost the same signals compared to the wild type, but at 50 K a

hyperfine-coupled EPR signal was observed. This observation indicated trapping of an organic radical generated at the C-5" position of neomycin.

The iron and sulfur atom contents in NeoN were estimated respectively at 6.6 \pm 0.9 and 5.8 \pm 0.9 equivalents per protein monomer, which indicates that NeoN has a second [4Fe-4S] cluster.¹⁰⁷ UV spectroscopic analysis of the C26/30/33A mutants showed an absorption around 420 nm, which is characteristic for [4Fe-4S] clusters. Other mutation studies showed that the C226/246/271/274A mutants led to self-degradation of NeoN during protein purification. Sequence alignment of NeoN with BtrN¹⁰⁹ and anaerobic sulfatase maturating enzyme (anSME)¹¹¹ indicated that a putative second [4Fe-4S] cluster might coordinate a set of C226, C247, C271, and C274. Also it is also expected that C249 is located close to the second [4Fe-4S] cluster and might be able to interact with C247. From these analyses, catalytic mechanism was proposed for NeoN as shown in Figure 1-7A.



Figure 1-7. NeoN catalysis. A) Proposed mechanism of NeoN reaction. B) Reaction of C249A mutant of NeoN with neomycin C

1.4. DISSERTATION STATEMENT

Biosynthetic studies of natural products have become essential to the discovery and development of new drugs, because understanding the biosynthetic pathways and the chemical mechanisms of each enzyme in the pathways can facilitate efforts in pathway engineering to produce new compounds with improved clinical properties. Aminoglycosides are well known and important antibiotics, but antibiotic resistance to aminoglycosides is increasing. This has led to renewed interest in the biosynthesis of aminoglycosides. Radical SAM enzymes catalyze novel and miscellaneous reactions in the biosynthesis of natural products. Even though several enzymes in the biosynthesis of aminoglycosides are assigned to be radical SAM enzymes based on the gene analysis, only a few radical SAM enzymes have been characterized *in vitro*. This dissertation focuses on the *in vitro* characterization of two radical SAM enzymes in the biosynthesis of aminoglycosides.

Chapter 2 describes the *in vitro* characterization of a Cbl-dependent radical SAM methyltransferase, GenK. This enzyme catalyzes the methylation of an unactivated sp^3 carbon of gentamicin X₂ to produce G418 during the biosynthesis of gentamicin. Experiments to investigate the stoichiometry of GenK reaction showed that an equal number of equivalents of 5'-dAdo, SAH, and G418 are produced. Isotope-labeling experiments demonstrated that the *S*-methyl group from SAM is transferred via Cbl to the aminoglycoside product during the course of the reaction. Experiments with other aminoglycosides in the biosynthetic pathway of gentamicin demonstrated the promiscuity of GenK substrate recognition.

Chapter 3 describes efforts to elucidate the mechanism of the GenK reaction. From the results of Chapter 2, three mechanisms are proposed and examined using chemically synthesized substrate analogs. The GenK assays with stereospecifically labeled (*R*)- and (*S*)-[6'-²H]-gentamicin X₂ demonstrated the stereochemical course of the hydrogen atom abstraction by 5'-Ado• and the overall stereochemistry of the GenK catalyzed reaction. Various substrate derivatives of gentamicin X₂ modified at C6' were tested to investigate the manner of methyl transfer from methylcobalamin to the substrate to produce the methylated product.

Chapter 4 describes the *in vitro* characterization of a radical SAM enzyme AprD4 and its reductase partner AprD3. AprD4/D3 from *Streptomyces tenebrarius* were purified following heterologous expression in *E. coli* and examined *in vitro*. The C3'deoxygenation reaction of paromamine catalyzed by AprD4/D3 was confirmed by comparison of chemically synthesized lividamine. Mass spectroscopy experiments confirmed the function of AprD4 by the detection of 4'-oxolividamine. Experiments with isotope-labeled substrates showed the position of hydrogen atom abstraction by 5'-dAdo• and stereochemical course of proton transfer at C3'-deoxygenation of paromamine.

Chapter 2: Characterization of GenK, a Cobalamin-Dependent Radical SAM Methyltransferase Enzyme

2.1. INTRODUCTION

Methylation is a recurrent and significant reaction in biological processes.¹¹² Various biological molecules including natural products, metabolites. and macromolecules such as nucleic acids, proteins, and polysaccharides can all serve as substrates in methylation reaction whereupon their biological properties may be modified. For example, DNA methylation is a critical feature in the control of gene expression as methylated DNA sequences are subjected to transcriptional repression.¹¹³ One of the key molecules that serves as methyl donor in these reactions is S-adenosyl-Lmethionine (SAM).¹¹⁴ In such reactions, attack at the electrophilic methyl group of SAM by a nucleophile such as amino, alkoxide and thiolate functionality of the substrate results in methyl transfer to the substrate and concomitant production of S-adenosyl-Lhomocysteine (SAH). However, a number of methyltransferases that are mechanistically unique compared the standard paradigm involving Lewis acid-base chemistry have recently been described.¹¹⁵ The catalytic cycles of these enzymes instead involve radical chemistry and result in the methylation of non-nucleophilic carbon or phosphrous centers.¹¹⁶ These enzymes are members of radical SAM superfamily and thus catalyze the reductive homolysis of SAM via single electron transfer form an active site [4Fe-4S] cluster to produce methionine and a 5'-deoxyadenosyl radical (5'-dAdo•) that serves as a radical initiator in the subsequent transformation.¹¹⁷

In a recent review, Booker *et al*¹¹⁶ divided the radical SAM methyltransferase (RSMT) into four classes (i.e. classes A, B, C, and D) organized by protein architecture, expected enzymatic mechanism, and cofactor dependence (Table 2-1). Catalysis by class A RSMTs involves the generation of a methylene radical on two strictly conserved cysteines and results in the methylation of sp^2 -hybridized carbon centers. Representative enzymes in this class are RlmN¹¹⁸⁻¹²⁰ and Cfr¹²¹⁻¹²⁶, which contribute to the modification of rRNA in various organisms. Class B RSMTs are characterized by distinctive protein architecture that includes a cobalamin-binding domain at the N-terminus and a radical SAM domain at the C-terminus. These enzymes methylate unactivated sp^2 - or sp^3 hybrized carbons or phosphinates. While class B is the largest and most diverse to date, only five enzymes (TsrM¹²⁷⁻¹²⁹, Fom3¹³⁰⁻¹³¹, GenD1⁸¹, ThnK¹³² and PhpK¹³³⁻¹³⁵) have been studied in vitro. These enzymes are described in detail below. Class C RSMTs demonstrate significant sequence similarity with coproporpyphyrinogen III oxidase (HemN) and methylate sp^2 -hybridized carbons. These enzymes do not have the two conserved cysteines that are characteristic of the class A enzymes and are frequently part of the biosynthetic pathways responsible for complex secondary metabolites with antitumor and antibiotic behavior such as nosiheptide (NosN¹³⁶ and NocN¹³⁷), yatakemycin (YtkY¹³⁸⁻¹³⁹) and jawsamycin (Jaw5¹⁴⁰⁻¹⁴¹). Finally, in the more recently discovered class D radical SAM methyltransferases, which includes MJ0619.142 Methylenetetrahydrofolate is proposed to serve as the methyl donor rather than SAM.

Class	Enzyme	Function	Substrate
А	RlmN Cfr	rRNA and tRNA modification rRNA modification	
В	Fom3 Fms7 GenD1 and GenK ThnK, L, and P PacJ, N, and O Cndl Swb9 HpnP BchQ, BchR PoyB and C	Forsfomycin biosynthesis Fortimicin A biosynthesis Gentamicin biosynthesis Thienamycin biosynthesis Pactamycin biosynthesis Chondrochlorens biosynthesis Quinomycin biosynthesis 2-Methylhopanoid biosynthesis Chlorophyll biosynthesis Polytheonamide biosynthesis	H X Vy ys X=H, OH, NHR
	CloN6Clorobiocin biosynthesisCouN6Coumermycin biosynthesisTsrMThiostrepton biosynthesisSioTSiomycin biosynthesis		NH O
	BcpD/PhpK	Bialaphos biosynthesis	
	NosN NocN	Nosiheptide biosynthesis Nocathiacin biosynthesis	N N
С	TpdI TpdL and U	Thiomuracin biosynthesis GE2270 biosynthesis	H S N O
	Blm-Orf8 Tlm-Orf11 Zpm-Orf26	Bleomycin biosynthesis Tallysomycin biosynthesis Zorbamycin biosynthesis	H ₂ N N H ₂ N O
	YtkT	Yatakemycin biosynthesis	
	Jaw5	Jawsamycin biosynthesis	S-Enz
D	MJ0619	Methanopterin modification	

Table 2-1. The four classes of radical SAM methylases⁵

Among the four classes of radical SAM methylases, class B is of particular interest, because it includes GenK, which is the topic of this chapter. Other members of this class include PhpK, TsrM, Fom3, GenD1, and ThnK.

PhpK is isolated from the phosalacin producer Kitasatospora phosalacinea, and was the first class B RSMT to be purified and characterized in vitro.¹³³ PhpK catalyzes the methylation of the phosphinate group of 2-acetylamino-4-hydroxyphosphinyl butanoate (N-acetyldemethylphosphinothricin, NAcDMPT) to produce 2-acetylamino-4hydroxymethylphosphinylbutanoate (*N*-acetylphosphinothricin, NAcPT) which possesses the only carbon-phosphorus-carbon linkage currently known in nature (Scheme 2-1). PhpK overexpressed from E. coli was not soluble and found only in inclusion bodies. After refolding and purification of the enzyme by anionic exchange chromatography, its Fe/S cluster was anaerobically reconstituted. Reconstituted PhpK coordinates a single [4Fe-4S] cluster, which was confirmed by EPR spectroscopy at 10 K. Assays containing reconstituted PhpK, SAM, methylcobalamin, NAcDMPT and dithionite enable observation of formation of NAcPT. Two dimensional ¹H-³¹P gradient heteronuclear single-quantum correlation (gHSQC) spectroscopy displayed a strong H-P cross-peak indicative a methylated phosphorus center. However, the NAcPT-associated gHSQC cross peak was not found in the absence of dithionite, which is required for the reduction of the [4Fe-4S] cluster. When ¹³CH₃-methyl-Cbl(III) was used in PhpK reactions, H-P cross peaks from the passive couplings of ¹H and ³¹P to the attached ¹³C nuclei were found. This experiment demonstrated that MeCbl(III) is a methyl source for NAcPT.



R = COOH, L-Ala-L-Ala, or L-Ala-L-Leu

Scheme 2-1. P-Methyl transfer reactions catalyzed by PhpK

TsrM from *Streptomyces laurentii* catalyzes the methylation of sp^2 hybridized C2 carbon in tryptophan to produce 2-methyltryptophan during the biosynthesis of thiostrepton A (Scheme 2-2).¹²⁷⁻¹²⁹ This reaction is the first step in the conversion of tryptophan to the quinaldic acid moiety of thiostrepton A. Early feeding experiments using chirally labeled methionine showed that the methyl group is transferred with net retention of configuration. This observation suggested a double displacement mechanism that involves two S_N2 type reactions, which differs from the typical methionine-dependent methylations.¹⁴³ Berteau *et al* purified TsrM as a strep-tag fusion protein in *E. coli*. The purified protein has one mol of iron and one mol of sulfur per polypeptide.¹²⁷ When the reconstituted enzyme was incubated in the presence of tryptophan, SAM and sodium dithionite, there was no detectable formation of 5'-dAdo, which is a typical harbinger of radical-SAM chemistry. However, after methylcobalamin was included in the previous assay mixture, methyltryptophan and SAH were produced in a constant ratio. Furthermore, isotope-tracer experiments demonstrated that the methyl group ultimately comes from SAM while cobalamin acts as an intermediate methyl carrier. When 11 tryptophan derivatives with different substitution at C5 (5-Me-, 5-OH-, 5-F-Trp) or without the α -carboxy group of tryptophan or with other changes were evaluated as potential substrates for TsrM, all compounds that were substituted on the indole ring were found to be susceptible to the methyl transfer reaction.¹²⁸



Scheme 2-2. Methylation of tryptophan catalyzed by TsrM

Fom3 isolated from Streptomyces wedmorensis catalyzes methylation of 2hydroxyethylphosphonate (2-HEP) to produce 2-hydroxypropylphosphonate (2-HPP) in the biosynthetic pathway of fosfomycin (Scheme 2-3).¹³¹ Early mutation studies showed that a mutant strain with a block in the vitamin B_{12} biosynthetic pathway could not produce fosfomycin, but converted HPP to fosfomycin.¹⁴⁴ Feeding experiments with ¹⁴Clabled MeCbl and this mutant showed the formation of ¹⁴C-labeld fosfomycin from HPP.¹⁴⁵ Isotope labeling experiments indicated that methyl transfer occurs at pro-(R)hydrogen position.¹⁴⁶ An *in vitro* study of wild type (WT) Fom3 was conducted by Wang et al.¹³⁰ WT Fom3 overexpressed from E. coli was not soluble and found exclusively in the inclusion bodies. After refolding, however, the enzyme could be isolated and reconstituted with additional ion and sulfide. N-terminal hexahistidine tagged WT His-Fom3 was also prepared. WT Fom3 has a single [4Fe-4S] cluster and His₆-Fom3 has four [4Fe-4S] clusters. In low-temperature EPR experiments, WT Fom3 and His₆-Fom3 showed signals of the [4Fe-4S] cluster. Cysteines 282, 286, and 289 in the conserved radical SAM CX₃CX₂C motif were shown to be essential for cluster binding by mutation studies. ³¹P NMR spectroscopic data with His-Fom3, dithionite, 2-HEP, and MeCbl exhibited signals corresponding to HPP. However, no product formation has been detected due to the instability of the enzyme.



Scheme 2-3. Fosfomycin biosynthetic pathway

GenD1 is another putative radical SAM enzyme in the gentamicin biosynthetic pathway.⁸¹ GenD1 catalyzes the *C*-methylation at C-4" of gentamicin A to produce gentamicin X_2 (Scheme 2-4). GenD1 has a noncanonical Cx_4Cx_2C motif instead of the highly conserved Cx_3Cx_2C binding motif for the [4Fe-4S] cluster found in most radical SAM enzymes. *In vitro* studies of GenD1 were conducted by Leadlay and Sun *et al.*⁸¹ In contrast to other radical SAM methyltransferase, GenD1 could be expressed and isolated as the *N*-His₆-tagged protein in *E. coli* in soluble form and in excellent yield. After reconstitution of the iron-sulfur cluster, assays with GenA, SAM, dithiothreitol (DTT), methylcobalamin, methyl viologen and NADPH showed production of GenX₂ and 5'-deoxyadenosine (5'-dAdo) which were confirmed by HPLC and mass spectroscopy. As expected, no activity was observed with non-reconstituted enzyme or under aerobic conditions.



Scheme 2-4. GenD1 catalyzed methylation reaction

Townsend *et al* found that ThnK from *Streptomyces cattleya* in the biosynthetic pathway of thienamycin catalyzes sequential methylations to build out the C6-ethyl side chain in a stereocontrolled manner (Scheme 2-5).¹³² ThnK was expressed with a *C*-terminal His₆-tag in *E. coli* Rosetta 2(DE3) and enzyme production was conducted in ethanolamine-M9 medium, which facilitates uptake of externally supplied HOCbl into *E. coli*. Without further reconstitution of iron and sulfur, the as-isolated ThnK has 7.4 \pm 1.4 equivalents of iron and 3.7 \pm 0.8 equivalents of sulfide per polypeptide. Assays of ThnK with SAM, methyl viologen, NADPH and (*2R*,*3R*,*5R*)-carbapenam showed the formation of 5'-deoxyadenosine (5'-dAdo), *S*-adenosylhomocysteine (SAH) and an ethylated product. Labeling experiments and mass spectrometry using *S*-adenosyl-[CD₃]-*methyl*-methionine (*d*₃-SAM) demonstrated an M+3 *m/z* shift for the methylated product and an M+5 *m/z* shift for the twice-methylated product, consistent with the formation of an ethyl side chain at C6 of the substrate by successive methyl transfers.



Scheme 2-5. Double methylation catalyzed by ThnK

The class B radical SAM methyltransferases characterized *in vitro* so far are summarized in Table 2-2. TsrM and GenD1 gave different products from SAM. TsrM gave only SAH, not 5'-dAdo, but GenD1 gave opposite results. Most enzymes used MeCbl or HOCbl, indicating that Cbl plays a role as an a methyl carrier for the methyl transfer process. Reducing systems required to reduce the [4Fe-4S] cluster to the active +1 redox state were different. PhpK and Fom3 were reduced by dithionite. The GenD1 and ThnK reactions used methyl viologen (MV) or benzyl viologen (BV) and NADPH. However, TsrM used dithiothreitol (DTT) as the sole reductant.

Enzymes	PhpK	TsrM	Fom3	GenD1	ThnK
Fe S	5.9 4.4	5.0 ± 0.3 3.0 ± 0.2	5.7 ± 0.7 7.5 ± 1.0	No information	7.4 ± 1.4 3.7 ± 0.8
Substrate	Р	$sp^2 C$	sp^{3} C	sp^{3} C	sp^{3} C
SAM amount used	2-5	1	0.5-1	4	1
5'-dAdo formation	No information	No	No information	Yes	Yes
SAH formation	No information	Yes	No information	No	Yes
Cbl used	MeCbl	MeCbl HOCbl CNCbl AdoCbl	MeCbl	MeCbl HOCbl	MeCbl HOCbl
Reducing systems	dithionite	DTT	dithionite	NADPH/MV NADPH/BV	NADPH/MV

Table 2-2. Class B radical SAM methyltransferses characterized in vitro.

The biosynthesis of gentamicins is already described in the Chapter 1. Sequence analysis of the gentamicin C_1 biosynthetic gene cluster suggested that the methylation step at C-6' of gentamicin X_2 (Gen X_2) to yield the clinically useful geneticin (G418)^{75, 147} is likely catalyzed by a Cbl-dependent radical SAM enzyme, GenK (Scheme 2-6).⁷⁷ GenK belongs to class B of the radical SAM dependent methyltransferases. Recently gene mutation experiments showed that knock-out of the genK gene blocked the production of gentamicin C_1 or C_2 , while production of gentamicin C_{1a} was increased significantly.¹⁴⁸ The *vitro* activity of GenK and results of product stoichiometry, mechanism of methyl transfer, and alternative substrates are reported herein.



Scheme 2-6. GenK catalysis

2.2. EXPERIMENTAL PROCEDURES

2.2.1. Materials and Equipment

All chemicals were purchased from Fisher Science (Pittsburgh, PA) or Sigma-Aldrich (St. Louise, MO) and used without further purification unless otherwise specified. DNA gel extraction and spin minipreps were obtained from Qiagen (Valencia, CA). All reagents for sodium dodecyl sulfate polyacrylamide gel electrophoresis (SDS-PAGE) and Amicon and Microcon YM-10 filtration products were purchased from Bio-Rad (Hercules, CA) and Millipore (Billerica, MA), respectively. *Escherichia coli* DH5*a* cells were bought from Bethesda Research Laboratories (Muskegon, MI). The vector pET24b (+) and enzyme KOD DNA polymerase were purchased from Novagen (Madison, WI). DNA modifying enzymes (for restriction digestion and ligation), PCR primers, and the overexpression host *E. coli* BL21 star (DE3) were acquired from Invitrogen (Carlsbad, CA) and New England Biolabs (NEB, Beverly, MA). Luria Bertani

(LB) media are products of Difco (Detroit, MI) or Fisher Scientific (Pittsburgh, PA). Prestained protein markers were purchased from NEB. Protein concentrations were determined by Bradford Assay¹⁴⁹ using bovine serum albumin as the standard. The relative molecular mass and purity of enzyme samples were determined using SDS-PAGE. The general methods and protocols for recombinant DNA manipulations were as described by Sambrook et al.¹⁵⁰ DNA sequencing was performed at the Core Facilities of the Institute of Cellular and Molecular Biology, University of Texas at Austin. All reactions involving GenK were performed in a Coy Anaerobic Chamber (glovebox) under an atmosphere of approximately 97.5% N_2 and 2.5% H_2 with less than 1 ppm O_2 . All solutions used for GenK assays were deaerated by bubbling with nitrogen gas before they were transferred into the glovebox. The solvents in the glove box were stirred open to the anaerobic atmosphere overnight to allow equilibration before use. Gentamicin X_{2} , G-418, and GenA were purchased from TOKU-E (Bellingham, WA). Tetrahydrofuran (THF) was distilled from sodium benzophenone and dichloromethane (CH_2Cl_2) was distilled over calcium hydride under a nitrogen atmosphere. Other anhydrous solvents were purchased from Acros Organics (Pittsburgh, PA). Analytic thin layer chromatography (TLC) was performed on pre-coated TLC glass plates (silica gel, grade 60, F254, 0.25 mm layer thickness) purchased from EMD Chemicals (Madison, WI). Flash column chromatography was carried out on silica gel (230-400 mesh, grade 60) from Sorbent Technologies (Atlanta, GA) by eluting with the specified solvents. ¹H and ¹³C NMR spectra were recorded at 400, 500, 600 MHz and 100, 125, 150 MHz with a Varian Gemini Spectrometer. Chemical shifts are reported as parts per million (ppm) relative to deuteriochloroform (CDCl₃), 7.26 ppm for ¹H NMR and 77.16 ppm for ¹³C NMR, respectively. Visualization in TLC was effected with *p*-phosphoryl molybdic acid

solution. MS analyses were carried out at the Mass Spectrometry and Proteomics Facility of the Department of Chemistry, University of Texas at Austin.

2.2.2. Cloning and Expression of Micromonospora echinospora GenK

The gentamicin producer *M. echinospora* (NRRL 2953) was obtained from the Agricultural Research Service of the US Department of Agriculture as a lyophilized sample. It was inoculated into 10 mL of tryptic soybroth (TSB) starter culture,¹⁵¹ which was incubated overnight at 30 °C with shaking at 250 rpm. The following day, 0.1 mL of starter culture was inoculated into 10 mL of fresh TSB, and the resulting culture was inoculated at 30 °C for 72 hr. Chromosomal DNA was extracted from 1.5 mL of this culture using a Qiagen DNeasy tissue kit. The *genK* gene was PCR amplified from the isolated *M. echinospora* DNA with the following primers:

5'-CAACATATGAACGCGCTGGTGGCAGC (forward),

5'-TAACGAATTCAGTGGGAAACCGCCTCGG (reverse),

and cloned into a pCR:blunt vector (Invitrogen) prior to excision with *NdeI* and *EcoRI* and ligation into the pET24 vector (Invitrogen). The resulting expression plasmid, pET24:*genK*, was used to transform *E. coli* Rosetta DE3 cells (Novagen) for expression of native and, recombinant GenK. Transformants were grown in 3 L of LB medium at 18 °C with shaking at 200 rpm. GenK expression was induced with 1 mM IPTG when the cells had reached an optical density of 0.6 at 600 nm. Cells were harvested by centrifugation 20 h after induction (5000 × *g*, 15 min), resuspended in 50 mM Tris•HCl buffer (pH 8.0) containing 10% glycerol, and 1 mM PMSF, and lysed by sonication. Cell lysate was centrifuged at 25,000 × *g* for 30 min to pellet the insoluble material. Preliminary protein expression analysis using SDS-PAGE revealed that GenK, although abundant, was expressed exclusively as inclusion bodies. Therefore, the resulting

supernatant was discarded and the pelleted inclusion bodies were washed three times in lysis buffer and dissolved in 20 mL of 50 mM Tris•HCl buffer (pH 8.0) containing 20% glycerol, 1 mM PMSF and 5 M urea. This material was centrifuged at 4,000 × g for 10 min to remove insoluble material.

2.2.3. Refolding of GenK from Inclusion Bodies and Reconstitution with Iron and Sulfide

The solublized GenK inclusion bodies, processed as described above, were added dropwise to 200 mL of refolding buffer composed of 50 mM Tris•HCl (pH 8.0), 1 mM MgSO₄, 5 mM reduced glutathione, 0.5 mM glutathione disulfide, 5 mM DTT, and 10% glycerol. This mixture was kept refrigerated for 24 h with gentle stirring and was then concentrated to 7 mL in an Amicon YM-10 centrifugal concentrator. The protein was dialyzed overnight against 800 mL of 50 mM Tris•HCl (pH 8.0) containing 0.1 M NaCl, 1 mM DTT, and 20% glycerol. Afterward, the protein was centrifuged to remove precipitated material and the supernatant was reconstituted with iron and sulfide in an anaerobic glovebox according to an established procedure.¹⁵² Iron content in the reconstituted GenK was assessed using the ferrozine assay¹⁵³ and the sulfide content was determined by the method devised by Helmut Bienert.¹⁵⁴ The iron and sulfide content reported in the chapter is an average of three measurements.

2.2.4. Enzymatic Synthesis and Purification of S-Adenosyl-L-Methionine (SAM)

The gene encoding *E. coli* SAM synthetase (*metK*) was cloned into pET28b(+) and the resulting construct was used to transform BL21(DE3) for expression of recombinant His₆-tagged protein. An incubation of 6 L of *E. coli* containing pET28b(+):*metK* in LB media was induced with 0.1 mM IPTG when the culture had

reached an optical density of 0.5 at 600 nm, overexpressed overnight at 25 °C, harvested by centrifugation ($5000 \times g$, 15 min), and resuspended in 100 mL of 50 mM Tris-HCl (pH 8.0), 1 M NaCl, 1 mM β-mercaptoethanol, 10 mM imidazole, and 10% glycerol (lysis buffer). Resuspended cells were lysed by sonication and the lysate was centrifuged at $27,200 \times g$ for 20 min. Cleared lysate was mixed with 10 mL of Ni²⁺-nitrilotriacetic acid (NTA) resin (Qiagen) pre-equilibrated with wash buffer (lysis buffer containing 20 mM imidazole) for 90 min. The Ni²⁺-NTA was poured into a column casing and rinsed with 100 mL of the wash buffer. MetK was eluted by the addition of elution buffer (lysis buffer containing 0.25 M imidazole). The pooled MetK was dialyzed against a 1 L solution of 50 mM Tris•HCl (pH 8.0), 0.15 M NaCl, 1 mM β-mercaptoethanol, and 10% glycerol. The final yield of MetK was 60 mL of 9.5 mg/mL. MetK was used to prepare SAM enzymatically from L-methionine and ATP and purified as described previously.¹⁵⁵ ¹³CD₃-methyl-SAM was prepared from ¹³CD₃-methyl-L-methionine (purchased from SigmaAldrich) using the same procedure. The purity of the enzymatically prepared SAM was assessed by high performance liquid chromatography (HPLC) using a Dionex CarboPac PA1 (4×250 mm) column with an isocratic 0.5 M ammonium acetate elution at 1 mL/min. The purified SAM was lyophilized, resuspended in 2 mL H₂O, divided into 50 μ L aliquots, and stored at -80 °C until use.

2.2.5. GenK Activity Assays

All GenK activity assays were conducted at ambient temperature under anaerobic conditions in a Coy anaerobic chamber. The reaction mixture contained the following: 50 mM Tris•HCl (pH 8.0), 10 mM dithiothreitol (DTT), 1 mM methyl viologen (MV), 4 mM nicotinamide adenine dinucleotide phosphate (NADPH), 4 mM enzymatically prepared SAM, 1 mM gentamicin X_2 (Gen X_2) (7), 1 mM cobalamin (Cbl), and 5 μ M

GenK (unless otherwise noted). Assay reactions were initiated by the addition of GenX₂, and ran for 8 h unless otherwise specified. Reaction aliquots designated for HPLC analysis of adenosylated products, *S*-adenosylhomocysteine (SAH) and 5'deoxyadenosine (5'-dAdo), were quenched by adding a volume of 30% trichloroacetic acid (TCA) equal to one tenth of the analyte volume. Reaction aliquots designated for analysis of aminoglycosides (30 μ L) were quenched with a mixture containing 112 μ L of methanol, 6 μ L of 0.5 M NaOH, and 1.8 μ L of 8.4 M 1-fluoro-2,4-dinitrobenzene (FDNB) to give final concentrations of 20 mM NaOH, 75% methanol, and 0.1 M FDNB. These were heated at 80 °C for 5 min to promote derivatization of primary amines with FDNB so they could be detected by UV absorbance at 340 nm.¹⁵⁶ Reaction aliquots analyzed by mass spectroscopy were quenched by passage through YM-10 Microcon centrifugal filters to remove GenK.

2.2.6. HPLC Detection of SAH and 5'-dAdo

A 25 μ L aliquot of a TCA-quenched GenK reaction described above was analyzed by HPLC equipped with a Varian Microsorb-MV 100-5 C18 (4.6 × 250 mm) column pre-equilibrated in H₂O containing 0.1% trifluoroacetic acid (solvent A). After sample loading, the column was eluted with a linear gradient from 0–20% acetonitrile containing 0.1% trifluoroacetic acid (solvent B) over 30 min to resolve 5'-dAdo and SAH. The UV detector was set at 260 nm. A blend of authentic, commercially obtained 5'-dAdo and SAH was injected as a retention time standard for each. Quantitation of 5'dAdo and SAH produced during assays was achieved by comparison of analyte peak areas with those of 5'-dAdo and SAH standards of known concentration.

2.2.7. HPLC Detection of Aminoglycoside

A 10 μ L aliquot of a FDNB-derivatized GenK assay mixtures was prepared as described in Section 2.2.5, loaded onto a Varian Microsorb-MV 100-5 C18 (4.6 × 250 mm) column and eluted with a gradient from 60:40 \rightarrow 50:50 water:acetonitrile over 20 min. The UV detector was set at 340 nm. A blend of authentic, commercially obtained GenX₂ and G418 was injected as a retention time standard for each. Quantitation of the conversion of GenX₂ to G418 was achieved by comparing the peak areas for each.

2.2.8. Mass Spectroscopic Analysis of GenK Activity

Assay mixtures were diluted in methanol and infused directly into an Agilent 6530 Accurate Mass QTOF-MS at 0.15 mL/min. The analyte was ionized by electrospray ionization and detected in positive ion mode.

2.2.9. Synthesis of 2-Steptamine Derivative (4).

The overall synthetic scheme is shown in Scheme 2-7. Compound **4** was prepared as previously described.¹⁵⁷



Scheme 2-7. Synthetic scheme for the preparation of 5,6-di-*O*-acetyl-1,3-diazido-1,2,3-trideoxy-myo-inositol (4).

2.2.10. Synthesis of Xylose Derivative (8)

The overall synthetic scheme is shown in Scheme 2-8. Compound 8 was prepared as previously described.¹⁵⁸ $H_{Q} \longrightarrow O$ Ac_2O, Pyr Ac_2O, Pyr A



Scheme 2-8. Synthetic scheme for the preparation of (*3R*, *3S*, *5R*)-3,4,5-tris(benzyloxy)-2-(phenylthio)tetrahydro-2*H*-pyran (**8**).

2.2.11. Synthesis of Gentamicin A₂

The overall synthetic scheme is shown in Scheme 2-9.



Scheme 2-9. Synthetic scheme for the preparation of gentamicin A₂.

(2*R*,3*S*,4*R*,5*R*,6*S*)-5-azido-2-(((tert-butyldiphenylsilyl)oxy)methyl)-6-(phenylthio)tetrahydro-2*H*-pyran-3,4-diol (13): Compound 12 was prepared as previously described.¹⁵⁹ To a solution of compound 12 (2.0 g, 6.73 mmol) in anhydrous DMF (60 mL), imidazole (0.916 g, 13.45 mmol) was added. The mixture was stirred for 30 min at room temperature. The mixture was cooled to 0 °C and *tert*-butyldiphenylchlorosilane (2.1 mL, 8.07 mmol) was added. The reaction was then allowed to stand for 10 h at room temperature. The mixture was treated with methanol (10 mL) at 0 °C and all the solvent was evaporated under reduced pressure. The residue was diluted with dichloromethane (80 mL) and washed with saturated aqueous solution of ammonium chloride (70 mL).
The aqueous layer was extracted with dichloromethane (70 mL \times 2). The combined organic extracts were washed with brine, dried over magnesium sulfate, filtered and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 1:1) to afford compound **13** (3.4 g, 94 %) as a colorless oil. All NMR spectra were consistent with previously reported data.¹⁶⁰ HRMS (ESI, positive) calculated for C₂₈H₃₃N₃O₄SSi⁺ [M + Na]⁺ 558.1853, found 558.1870.

(((2R,3S,4R,5R,6S)-5-azido-3,4-bis(benzyloxy)-6-(phenylthio)tetrahydro-2H-pyran-

2-yl)methoxy)(tert-butyl)diphenylsilane (14): Sodium hydride (60% dispersion in mineral oil, 0.558 g, 13.96 mmol) and benzyl bromide (1.96 mL, 16.5 mmol) was added to a solution of compound **13** (3.4 g, 6.35 mmol) in DMF (60 mL) at 0 °C. After stirring the mixture for 8 h at room temperature, methanol (20 mL) was added at 0 °C. The mixture was stirred for an additional 20 min before concentration under reduced pressure. The residue was diluted with dichloromethane (80 mL) and washed with water (60 mL) and brine, dried over magnesium sulfate, filtered and concentrated *in vacuo*. The crude compound **14** (4.311 g, 95%) as a clear oil. All NMR spectra were consistent with previously reported data.¹⁶⁰ HRMS (ESI, positive) calculated for C₄₂H₄₅N₃O₄SSi⁺ [*M* + *Na*]⁺ 739.2792, found 738.2793.

((2R,3S,4R,5R,6S)-5-azido-3,4-bis(benzyloxy)-6-(phenylthio)tetrahydro-2H-pyran-2yl)methanol (15): Tetrabutylammonium fluoride (1 M solution in THF, 12.04 mL, 12. 04 mmol) was added to a solution of compound 14 (4.311 g, 6.02 mmol) in THF (60 mL). The reaction mixture was stirred at room temperature for 2 h, and quenched by adding a saturated solution of aqueous ammonium chloride (40 mL). The aqueous layer was extracted with ethyl acetate (60 mL \times 2). The combined organic extracts were washed with brine, dried over magnesium sulfate, filtered and concentrated under reduced

pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:1) to give compound **15** (2.847 g, 99%) as a white solid. All NMR spectra were consistent with previously reported data.¹⁶¹ HRMS (ESI, positive) calculated for $C_{42}H_{45}N_3O_4SSi^+$ [M + Na]⁺ 739.2792, found 738.2793.

(2S,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-(((4-methoxybenzyl)oxy)methyl)-2-

(phenylthio)tetrahydro-2H-pyran (16): sodium hydride (60% in mineral oil, 0.289 g, 7.23 mmol) and p-methoxybenzyl chloride (0.98 mL, 7.23 mmol) were added to a solution of compound 15 (2.847 g, 5.96 mmol) in anhydrous DMF (60 mL) at 0 °C. The reaction mixture was stirred at room temperature for 8 h. The mixture was treated with methanol (10 mL) at 0 °C and all the solvent was evaporated under reduced pressure. The residue was diluted with dichloromethane (60 mL) and washed with water (40 mL). The aqueous layer was extracted with dichloromethane (60 mL \times 2), and the combined organic extracts were washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 9:1) to afford compound 16 (2.951 g, 82%) as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 7.68–7.64 (m, 2H, Ph), 7.40–7.29 (m, 13H, Ph), 7.26–7.23 (m, 2H, Ph), 6.95–6.90 (m, 2H, Ph), 4.90 (dd, J = 14.0, 10.6 Hz, 2H, PMB), 4.83 (d, J = 10.9 Hz, 1H, Bn), 4.62 (d, J = 10.9 Hz, 1H, Bn), 4.61 (d, J = 11.6 Hz, 1H, Bn), 4.52 (d, J = 11.6 Hz, 1H, Bn), 4.47 (d, J = 10.1 Hz, 1H, H-1), 3.84 (s, 3H, -OMe), 3.82-3.74 (m, 2H, H-6, H-6'), 3.66 (dd, J = 9.8, 8.9 Hz, 1H, H-4), 3.56 (t, J = 9.1Hz, 1H, H-3), 3.53-3.49 (m, 1H, H-5), 3.41 (dd, J = 10.1, 9.2 Hz, 1H, H-2). ¹³C NMR (100 MHz, CDCl₃) δ 159.3, 134.9, 137.7, 133.6, 131.4, 130.3, 129.34, 129.03, 128.55, 128.48, 128.35, 128.30, 128.22, 128.03, 127.90, 127.86, 113.8, 86.0, 85.1, 79.4, 77.61, 75.88, 75.04, 73.1, 68.4, 55.3. HRMS (ESI, positive) calculated for $C_{42}H_{45}N_3O_4SSi^+$ [M + Na]⁺ 739.2792, found 738.2793.

(1S,2S,3R,4S,6R)-4,6-diazido-3-(((2S,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-(((4-

methoxybenzyl)oxy)methyl)tetrahydro-2H-pyran-2-yl)oxy)cyclohexane-1,2-diyl

diacetate (17): Compound 4 (2.023 g, 6.78 mmol) and compound 16 (4.865g, 8.139 mmol) were co-evaporated from dry toluene three times and further dried under high vacuum overnight. A solution of compound 4 and 16 in anhydrous diethyl ether (60 mL) and anhydrous dichloromethane (20 mL) was then added to preactivated 4 Å molecular sieves. After stirring for 30 min at room temperature, the mixture was cooled to -40 °C. N-Iodosuccinimide (3.968 g, 17.64 mmol) was added, and the reaction mixture was stirred for 20 min. Trifluoromethanesulfonic acid (0.30 mL, 3.39 mmol) was slowly added, and the reaction was warmed to -20 °C and kept stirring for 30 min. Sodium bisulfite (1 g), sodium bicarbonate (1 g), and water (10 mL) were added at 0 °C, and the mixture was stirred for 10 min at room temperature. The reaction mixture was diluted with dichloromethane (50 mL), filtered through a Celite pad, and washed with a saturated solution of aqueous sodium bicarbonate (100 mL). The aqueous layers were extracted with dichloromethane (100 mL \times 3), and the combined organic phase was washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:2) to give compound **17** (3.465 g, 65%) as a clear oil. ¹H NMR (500 MHz, CDCl₃) δ 7.33–7.23 (m, 12H, Ph), 7.13-7.11 (m, 2H, Ph), 6.85-6.83 (m, 2H, Ph), 5.15 (t, J = 9.8 Hz, 1H, H-5), 5.14 (d, J = 3.9 Hz, 1H, H-1'), 4.93 (t, J = 10.0 Hz, 1H, H-6), 4.85 (dd, J = 30.6, 10.6 10.85 Hz, 1H, Bn), 4.41 (d, J = 11.8 Hz, 1H, Bn), 4.15 (td, J = 8.1, 2.4 Hz, 1H, H-5'), 3.95 (dd, J = 10.5, 9.0 Hz, 1H, H-3'), 3.81 (dd, J = 10.8, 2.8 Hz, 1H, H-6'), 3.77 (dd, J = 10.8, 1H, H-6'), 3.77 (dd, J9.9, 9.3 Hz, 1H, H-4'), 3.76 (s, 3H, -OCH₃), 3.67–3.59 (m, 3H, H-1, H-6', H-4), 3.43– 3.38 (m, 1H, H-3), 3.36 (dd, J = 10.5, 3.8 Hz, 1H, H-2'), 2.40 (dt, J = 13.4, 4.6 Hz, 1H,

H-2), 2.094 (s, 3H, -OAc), 2.085 (s, 3H, -OAc), 1.60 (q, J = 12.5 Hz, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 169.8, 169.5, 159.3, 137.9, 137.8, 129.8, 129.7, 128.45, 128.38, 127.97, 127.86, 127.75, 127.67, 113.8, 99.3, 79.7, 78.29, 78.11, 75.3, 74.9, 74.1, 73.6, 73.2, 71.7, 67.4, 63.1, 58.7, 57.7, 55.2, 31.8, 20.64, 20.57. HRMS (ESI, positive) calculated for C₄₂H₄₅N₃O₄SSi⁺ [M + Na]⁺ 739.2792, found 738.2793.

(1S,2R,3R,4S,6R)-4,6-diazido-3-(((2S,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-

(((4-methoxybenzyl)oxy)methyl)tetrahydro-2H-pyran-2-yl)oxy)cyclohexane-1,2-diol

(18) : Sodium methoxide (4.37 M in methanol, 0.19 mL, 8.14 mmol) was added dropwise to a solution of compound 17 (3.2 g, 4.072 mmol) in methanol (40 mL). The reaction mixture was stirred at room temperature for 8 h and then neutralized with Amberlite IR-120 (H⁺ form) resin, filtered and concentrated. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:2) to give compound **18** (2.347 g, 82%) as a pale yellow oil. ¹H NMR (500 MHz, CDCl₃) δ 7.35–7.24 (m, 10H, Ph), 7.13–7.11 (m, 2H, Ph), 6.84–6.82 (m, 2H, Ph), 5.14 (d, J = 3.7 Hz, 1H, H-1'), 4.88 (dd, J = 37.2, 10.6 10.8 Hz, 1H, Bn), 4.40 (d, J = 11.8 Hz, 1H, Bn), 4.21 (s, 1H, OH), 4.08 (td, J = 8.0, 2.6Hz, 1H, H-5'), 3.96 (dd, J = 10.2, 9.1 Hz, 1H, H-3'), 3.78 (dd, J = 10.7, 3.2 Hz, 1H, H-6'), 3.75 (s, 3H, OCH₃), 3.75 (m, 1H, H-4'), 3.66-3.63 (m, 2H, H-2', H-6'), 3.50-3.38 (m, 3H, H-4, H-5, H-1), 3.28–3.22 (m, 2H, H-3, H-6), 2.33–2.27 (m, 1H, H-2), 1.53–1.46 (m, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 159.3, 137.8, 137.5, 129.74, 129.70, 128.49, 128.43, 128.05, 127.98, 127.85, 127.72, 113.8, 99.6, 84.0, 80.9, 78.1, 75.82, 75.63, 75.32, 74.97, 73.2, 71.6, 67.5, 64.3, 59.6, 58.7, 55.2, 32.1. HRMS (ESI, positive) calculated for $C_{42}H_{45}N_{3}O_{4}SSi^{+}[M + Na]^{+}$ 739.2792, found 738.2793.

(*1S*,*2R*,*3S*,*5R*,*6S*)-3,5-diazido-2-(((*2S*,*3R*,*4R*,*5S*,*6R*)-3-azido-4,5-bis(benzyloxy)-6-(((4-methoxybenzyl)oxy)methyl)tetrahydro-2*H*-pyran-2-yl)oxy)-6-(((*2R*,*3R*,*4S*,*5R*)-3,4,5-

tris(benzyloxy)tetrahydro-2H-pyran-2-yl)oxy)cyclohexan-1-ol (19): Compound 8 (2.352 g, 4.59 mmol) and compound 18 (2.925 g, 4.171 mmol) were co-evaporated from dry toluene three times and further dried under high vacuum overnight. A solution of compound 8 and 18 in anhydrous dichloromethane (50 mL) was added to a preactivated 4 Å molecular sieves and N-iodosuccinimide (1.502 g, 6.67 mmol). After stirring for 30 min at room temperature, the mixture was cooled to -78 °C. Trifluoromethanesulfonic acid (0.11 mL, 1.25 mmol) was slowly added, and the reaction was warmed to 0 °C slowly. The reaction was quenched by adding solid sodium sulfite (0.5 g), sodium bicarbonate (0.5 g) and water (100 mL) at 0 °C. The reaction mixture was diluted with dichloromethane, filtered, and washed with a saturated solution of sodium bicarbonate and brine. The residue was dried over sodium sulfate, filtered, concentrated in reduced pressure, and purified by flash column chromatography on silica gel (Hex:EtOAc = 3:2) to give compound **19** (3.746 g, 81%) as a clear oil. ¹H NMR (500 MHz, CDCl₃) δ 7.44-7.24 (m, 25H, Ph), 7.14 (m, 2H, Ph), 6.89–6.80 (m, 2H, Ph), 5.47 (d, J = 3.7 Hz, 1H, H-1'), 5.08 (d, J = 3.6 Hz, 1H, H-1"), 4.95–4.84 (m, 4H, Bn), 4.81–4.72 (m, 4H, Bn), 4.63 (dd, J = 13.7, 11.7 Hz, 2H, Bn), 4.49 (d, J = 10.9 Hz, 1H, Bn), 4.42 (d, J = 11.8 Hz, 1H, J)Bn), 4.37 (d, J = 2.1 Hz, 1H, OH), 4.16–4.10 (m, 1H, H-5'), 3.97 (dtd, J = 18.4, 9.6, 9.2, 6.9 Hz, 3H, H-5, H-3', H-3"), 3.84-3.69 (m, 6H, H-5", H-4', H-6', OMe), 3.65 (ddd, J =9.1, 5.3, 2.2 Hz, 2H, H-5, H-6'), 3.62-3.54 (m, 1H, H-4"), 3.54-3.48 (m, 3H, H-1, H-2', H-2"), 3.43–3.36 (m, 1H, H-4), 3.36–3.27 (m, 2, H-3, H-6), 2.40–2.31 (m, 1H, H-2), 1.54 (q, J = 12.7 Hz, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 159.40, 138.89, 138.21, 138.17, 138.14, 137.97, 129.99, 129.82, 128.61, 128.59, 128.56, 128.52, 128.49, 128.40, 128.20, 128.17, 128.15, 128.11, 128.05, 128.01, 127.98, 127.92, 127.85, 127.79, 127.73, 113.89, 99.11, 98.21, 83.41, 81.60, 80.90, 80.52, 79.28, 78.18, 77.75, 75.76, 75.64, 75.25, 75.02, 73.77, 73.72, 73.28, 71.54, 67.68, 63.88, 61.23, 59.81, 58.83, 55.34, 32.39, 29.84.

HRMS (ESI, positive) calculated for $C_{60}H_{65}N_9O_{12}^+$ [*M* + *Na*]⁺ 1126.4645, found 1126.4643.

(1S,2R,3S,5R,6S)-3,5-diazido-2-(((2S,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-(hydroxymethyl)tetrahydro-2H-pyran-2-yl)oxy)-6-(((2R,3R,4S,5R)-3,4,5-tris(benzyloxy-)tetrahydro-2H-pyran-2-yl)oxy)cyclohexan-1-ol (20): 2,3-dichloro-5,6-dicyano-1,4benzoquinone (1.183 g, 5.21 mmol) was added to a solution of compound 19 (3.746g, 2.47 mmol) in dichloromethane (50 mL) and water (5 mL). The reaction mixture was stirred for 4 h at room temperature, and then quenched by adding a saturated solution of aqueous sodium bicarbonate (30 mL). The aqueous layer was extracted with dichloromethane (40 mL \times 3), and the combined organic phase was washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:2) and recrystallization with Hex/EtOAc (10:1) to give compound 20 (2.330 g, 70%) as a white solid. ¹H NMR (500 MHz, CDCl₃) δ 7.58–7.27 (m, 25H, Bn), 5.50 (d, J = 3.7 Hz, 1H, H-1'), 5.07 (d, J = 3.6 Hz, 1H, H-1"), 5.00–4.85 (m, 5H, Bn), 4.84–4.70 (m, 4H, Bn), 4.66 (d, J = 11.5 Hz, 1H, Bn), 4.44 (d, J = 2.1 Hz, 1H, OH), 4.10-4.02 (m, 2H, H-3', H-5'),4.01-3.91 (m, 2H, H-3", H-5"), 3.87 (dd, J = 12.2, 2.5 Hz, 1H, H-6'), 3.80 (dd, J = 12.3, 3.4 Hz, 1H, H-6'), 3.75 (dd, J = 11.1, 5.8 Hz, 1H, H-5"), 3.69 (m, 2H, H-4', H-5'), 3.61 (ddd, J = 10.5, 8.6, 5.6 Hz, 1H, H-4"), 3.53 (m, 2H, H-2", H-1), 3.44 (dd, J = 10.3, 3.7)Hz, 1H, H-2'), 3.36 (m, 3H, H-4, H-3, H-6), 2.36 (dt, J = 13.0, 4.3 Hz, 1H, H-2), 1.53 (q, J = 12.5 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 138.82, 138.15, 138.11, 138.00, 137.89, 128.58, 128.54, 128.50, 128.41, 128.16, 128.07, 128.03, 127.98, 127.95, 127.92, 127.67, 98.76, 98.20, 83.90, 81.05, 80.74, 79.21, 77.83, 77.60, 75.63, 75.58, 75.08, 73.67, 73.65, 72.18, 63.79, 61.36, 61.27, 59.60, 58.94, 32.36. HRMS (ESI, positive) calculated for $C_{52}H_{57}N_9O_{11}^+ [M + Na]^+ 1006.4070$, found 1006.4076.

Gentamicin A₂ (21): Trimethylphosphine (1 M solution in THF, 1.96 mL, 1.96 mmol) was added to a solution of compound 20 (0.476 g, 0.50 mmol) in THF (12 mL) and 0.1 M NaOH (3 mL), and the reaction mixture was stirred at 50 °C for 2 h. After cooling to room temperature, the solution was evaporated under reduced pressure. After added in a mixture of acetic acid (3 mL), water (9 mL) and methanol (9 mL), the solution was deaerated by evacuating the air in the flask and refilling it with argon five times. Palladium hydroxide on carbon (20%, Degussa type, 0.2 g) was added and the solution was charged with hydrogen gas. The reaction mixture was then stirred at room temperature under hydrogen gas for 24 h. The solution was then filtered through a Celite pad, washed with water and methanol, and concentrated under reduced pressure. The residue was purified by chromatography on Amberlite CG-50 resin (0-5% concentrated ammonium hydroxide in water), acidified with hydrochloric acid (pH ca. 4), and lyophilized to provide compound **21** (0.204 g, 90%) as an oily solid. ¹H NMR (500 MHz, $D_{2}O(35.61 \text{ (d}, J = 4.0 \text{ Hz}, 1\text{H}, \text{H}-1'), 4.96 \text{ (d}, J = 2.5 \text{ Hz}, 1\text{H}, \text{H}-1''), 3.89-3.82 \text{ (m}, 3\text{H}, 1.00 \text{ Hz})$ H-3', H-5', H-6'), 3.81-3.75 (m, 2H, H-4', H-5), 3.72-3.63 (m, 4H, H-5", H-6', H-5", H-3"), 3.63-3.48 (m, 5H, H-6, H-2", H-4", H-1, H-3'), 3.46-3.42 (m, 1H, H-4), 3.42-3.38 (m, 1H, H-2'), 2.47 (dt, J = 12.6, 4.3 Hz, 1H, H-2), 1.85 (q, J = 12.7 Hz, 1H, H-2). ¹³C NMR (125 MHz, D₂O) δ 101.8, 97.1, 83.4, 80.4, 73.72, 73.61, 72.91, 69.36, 68.99, 68.88, 62.4, 60. 4, 53.9, 49.6, 48.7, 27.9 HRMS (ESI, positive) calculated for $C_{17}H_{33}N_3O_{11}^+$ [M] + Na]⁺ 478.2007, found 478.2012.

2.2.12. Synthesis of 6'-Amino-6'-dehydroxy-gentamicin A2

The overall synthetic scheme is shown in Scheme 2-10.



Scheme 2-10. Synthetic scheme for the preparation of 6'-amino-6'-deoxy-gentamicin A₂.

((2R, 3S, 4R, 5R, 6S)-5-azido-3, 4-bis(benzyloxy)-6-(((1R, 2S, 3S, 4R, 6S)-4, 6-diazido-2hydroxy-3-(((2R, 3R, 4S, 5R)-3, 4, 5-tris(benzyloxy)tetrahydro-2*H*-pyran-2-yl)oxy)cyclo-hexyl)oxy)tetrahydro-2*H*-pyran-2-yl)methyl 4-methylbenzenesulfonate (22): *p*-Toluenesulfonyl chloride (0.417 g, 2.19 mmol) was added to a solution of compound 21 (0.420 g, 0.44 mmol) in pyridine (20 mL) at 0 °C. The reaction mixture was stirred at room temperature for 8 h. Methanol (5 mL) was then added to the solution at 0 °C, and the mixture was stirred at room temperature for 30 min. The solution was concentrated under reduced pressure, and co-evaporated with toluene twice. The residue was diluted with dichloromethane (30 mL), and washed with a saturated solution of aqueous sodium bicarbonate (20 mL). The aqueous layer was extracted with dichloromethane (30 mL × 2), and the combined organic phase was washed with brine, dried over magnesium sulfate, filtered and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 4:1) to give the tosylate **22** (0.481 g, 97%) as a clear oil. ¹H NMR (500 MHz, CDCl₃) δ 7.84–7.78 (m, 2H, Bn), 7.44–7.27 (m, 25H, Bn), 7.22 (dd, J = 7.5, 2.0 Hz, 2H, Bn), 5.46 (d, J = 3.7 Hz, 1H, H-1'), 5.04 (d, J = 3.5 Hz, 1H, H-1"), 4.95–4.87 (m, 3H, Bn), 4.86 (d, J = 5.3 Hz, 1H, Bn), 4.82 (d, J = 6.2 Hz, 1H, Bn), 4.81–4.73 (m, 3H, Bn), 4.65 (d, J = 11.5 Hz, 1H, Bn), 4.54 (d, J = 10.8 Hz, 1H, Bn), 4.41–4.32 (m, 2H, H-6', H-6'), 4.31–4.17 (m, 2H, H-5", H-5'), 4.04–3.87 (m, 3H, H-3", H-3', H-5"), 3.75 (dd, J = 11.2, 5.7 Hz, 1H, H-5"), 3.70–3.56 (m, 3H, H-4', H-4", H-5), 3.56–3.46 (m, 2H, H-2", H-1), 3.40–3.26 (m, 4H, H-2', H-4, H-3, H-6), 2.42 (s, 3H, Ts), 2.34 (dt, J = 13.4, 4.3 Hz, 1H, H-2), 1.50 (q, J = 12.5 Hz, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 144.90, 138.77, 138.11, 138.09, 138.07, 137.70, 137.60, 132.93, 129.85, 128.52, 128.51, 128.48, 128.38, 128.37, 128.04, 128.02, 128.00, 127.97, 127.94, 127.92, 127.89, 127.82, 127.80, 127.64, 98.43, 98.26, 84.28, 80.67, 80.51, 80.11, 79.14, 77.51, 77.36, 77.31, 77.10, 77.06, 76.85, 76.81, 75.60, 75.50, 75.28, 75.05, 73.68, 73.64, 69.68, 68.21, 63.47, 61.29, 59.51, 58.68, 32.20, 21.66. HRMS (ESI, positive) calculated for C₅₉H₆₃N₉O₁₃S⁺ [M + Na]⁺ 1160.4158, found 1160.4164.

(*1S*,2*R*,3*S*,5*R*,6*S*)-3,5-diazido-2-(((2*R*,3*R*,4*R*,5*R*,6*R*)-3-azido-6-(azidomethyl)-4,5-bis-(benzyloxy)tetrahydro-2*H*-pyran-2-yl)oxy)-6-(((2*R*,3*R*,4*S*,5*R*)-3,4,5-tris(benzyloxy)tetrahydro-2*H*-pyran-2-yl)oxy)cyclohexan-1-ol (23): Sodium azide (0.137 g, 2.11 mmol) and tetrabutylammonium iodide (1.5 mg, 4.3 mmol) was added to a solution of the previously synthesized tosylate 22 (0.481 g, 0.42 mmol) in DMF (20 mL). The reaction mixture was stirred at 60 °C for 7 h. After cooling to room temperature, the reaction solution was evaporated under reduced pressure, and the residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 5:1) to give compound 23 (0.327 g, 77%) as a colorless oil. ¹H NMR (500 MHz, CDCl₃) δ 7.38–7.28 (m, 25H, Bn), 5.53 (d, *J* = 3.7 Hz, 1H, H-1'), 5.02 (d, *J* = 3.5 Hz, 1H, H-1"), 4.93–4.71 (m, 8H, Bn), 4.66 (m, 2H, Bn), 4.37 (d, *J* = 2.1 Hz, 1H, OH), 4.19 (ddd, *J* = 10.0, 4.4, 2.5 Hz, 1H, H-5'), 3.99 (dd, *J* = 10.4, 8.8 Hz, 1H, H-3'), 3.97–3.92 (m, 1H, H-3"), 3.92-3.86 (m, 1H, H-5"), 3.72 (dd, *J* = 11.1, 5.6 Hz, 1H, H-5"), 3.66 (td, J = 9.0, 2.0 Hz, 1H, H-5), 3.61-3.47 (m, 5H, H-4, H-4', H-4", H-2", H-1), 3.44 (dd, J = 6.9, 3.5 Hz, 1H, H-2'), 3.42-3.33 (m, 3H, H-6', H-6', H-3), 3.30 (t, J = 9.5 Hz, 1H, H-6), 2.35 (dt, J = 13.2, 4.4, 1H, H-2), 1.54–1.48 (m, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 138.74, 138.07, 138.03, 137.70, 128.52, 128.50, 128.48, 128.34, 128.06, 128.01, 127.99, 127.98, 127.95, 127.89, 127.87, 127.80, 127.60, 98.44, 98.27, 84.26, 80.65, 80.64, 80.10, 79.11, 78.65, 77.47, 75.58, 75.54, 75.33, 75.17, 73.64, 73.63, 71.07, 63.66, 61.28, 59.54, 58.81, 51.04, 32.30. HRMS (ESI, positive) calculated for C₅₂H₅₆N₁₂O₁₀⁺ [M + Na]⁺ 1031.4135, found 1031.4142.

6'-amino-6'-dehydroxy-GenA₂ (24): Trimethylphosphine (1 M solution in THF, 1.29 mL, 1.29 mmol) was added to a solution of compound 23 (0.327 g, 0.324 mmol) in THF (12 mL) and 0.1 M NaOH (4 mL). The reaction mixture was stirred at 50 °C for 2 h, cooled to room temperature, and concentrated *in vacuo*. The material was dissolved in a mixture of acetic acid (3 mL), water (9 mL) and methanol (9 mL). The solution was deaerated by evacuating the air inside of the flask and refilling it with nitrogen five times. Palladium hydroxide on carbon (20%, Degussa type, 0.1 g) was added and the solution was charged with hydrogen gas. The reaction mixture was stirred at room temperature under hydrogen gas for 24 hr. The solution was filtered through Celite pad, washed with water and methanol, and concentrated under reduced pressure. The residue was purified by chromatography on Amberlite CG-50 resin (NH4⁺ form) (5-10% concentrated ammonium hydroxide in water), acidified with hydrochloric acid (pH ca. 4), and lyophilized to provide compound 24 (0.13 g, 88%) as an oily solid. ¹H NMR (500 MHz, D_2O) δ 6.02 (d, J = 3.9 Hz, 1H, H-1'), 5.07 (d, J = 2.5 Hz, 1H, H-1"), 4.12–4.01 (m, 3H, H-5', H-3', H-5), 3.93 (t, J = 9.1 Hz, 1H, H-4'), 3.85–3.50 (m, 8H, H-5", H-3", H-5", H-6, H-2", H-4", H1, H3), 3.60–3.50 (m, 3H, H-6', H-4, H-2'), 3.36 (dd, J = 13.6, 6.7 Hz, 1H, H-6'), 2.65–2.56 (m, 1H, H-2), 2.01 (q, J = 12.7 Hz, 1H, H-2). ¹³C NMR (125 MHz,

D₂O) δ 101.9, 96.2, 83.6, 77.51, 74.11, 72.88, 71.63, 70.58, 69.2, 68.83, 68.19, 62.5, 53.4, 49.7, 48.2, 40.2, 27.9. HRMS (ESI, positive) calculated for C₁₇H₃₄N₄O₁₀⁺ [*M* + *Na*]⁺ 477.2167, found 477.2167.

2.2.13. Synthesis of 2'-Deamino-2'-hydroxy-gentamicin A2

The overall synthetic scheme is shown in Scheme 2-11.



Scheme 2-11. Synthetic scheme for the preparation of 2'-deamino-2'-hydroxy-GenA₂.

(*2R*,*3R*,*4S*,*5R*,*6S*)-2-(acetoxymethyl)-6-(phenylthio)tetrahydro-2*H*-pyran-3,4,5-triyl triacetate (27): Compound 26 was prepared as previously described.¹⁶² Imidazole (4.167

g, 36.72 mmol) and tert-butyldiphenylsilyl chloride (5.73 mL, 22.03 mmol) was added to a solution of compound 26 (5.00 g,18.36 mmol) in dry DMF (150 mL) at 0 °C, and the reaction mixture was stirred at room temperature for 12 h. The solution was cooled to 0 °C, and methanol (30 mL) was added to quench the reaction. The mixture was concentrated and diluted with dichloromethane (100 mL). The solution was then washed with a saturated solution of aqueous ammonium chloride (80 mL). The aqueous wash was then extracted with dichloromethane (100 mL \times 2). The combined organic layers were washed with brine, dried over magnesium sulfate, filtered and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 1:2) to give compound 27 (8.533 g, 91 %) as a white solid. ¹H NMR (500 MHz, CDCl₃) § 7.75–7.70 (m, 4H, Ph), 7.56–7.52 (m, 2H, Ph), 7.46–7.35 (m, 6H, Ph), 7.27–7.21 (m, 3H, Ph), 4.53 (d, J = 9.7 Hz, 1H, C-1), 4.00–3.88 (m, 2H, C-6, C-6), 3.69– 3.56 (m, 2H, C-3, C-4), 3.50–3.42 (m, 1H, C-5), 3.36 (t, J = 8.8 Hz, 1H, C-2), 1.07 (s, 9H, TBDPS). ¹³C NMR (125 MHz, CDCl₃) δ 135.65, 135.60, 132.91, 132.80, 132.53, 131.91, 129.87, 129.86, 129.00, 128.00, 127.82, 127.81, 87.86, 79.01, 77.78, 71.62, 71.14, 64.32, 26.81, 19.23. HRMS (ESI, positive) calculated for $C_{28}H_{34}O_5SSi^+[M+Na]^+$ 533.1788, found 533.1796.

(2*R*,3*S*,4*S*,5*R*,6*S*)-2-(hydroxymethyl)-6-(phenylthio)tetrahydro-2*H*-pyran-3,4,5-triol (28): Sodium hydride (60% dispersion in mineral oil, 1.039 g, 25.96 mmol) and benzyl bromide (3.11 mL, 25.96 mmol) was added to a solution of compound 27 (3.4 g, 6.35 mmol) in DMF (40 mL) at 0 °C. After stirring the mixture for 8 h at room temperature, methanol (40 mL) was added at 0 °C. The mixture was stirred for an additional 20 min before concentration under reduced pressure. The residue was diluted with dichloromethane (100 mL) and washed with water (80 mL). The remaining aqueous layer was extracted with dichloromethane (80 mL \times 2). The combined organic layers were washed with brine, dried over magnesium sulfate, filtered and concentrated *in vacuo*. The residue was purified by flash column chromatography (Hex/EtOAc = 10:1) to give compound **28** (4.311 g, 95%) as a clear oil. ¹H NMR (500 MHz, CDCl₃) δ 7.91–7.85 (m, 2H, Ph), 7.84–7.79 (m, 2H, Ph), 7.74–7.67 (m, 2H, Ph), 7.54–7.21 (m, 24H, Ph), 5.04–4.94 (m, 4H, Bn), 4.84 (d, *J* = 10.3 Hz, 1H, Bn), 4.80 (d, *J* = 10.3 Hz, 2H, H-1, Bn), 4.10 (dd, *J* = 11.4, 1.8 Hz, 1H, H-6), 4.04 (dd, *J* = 11.4, 3.7 Hz, 1H, H-6), 3.91 (t, *J* = 9.4 Hz, 1H, H-4), 3.83 (t, *J* = 9.0 Hz, 1H, H-3), 3.66 (dd, *J* = 9.8, 8.7 Hz, 1H, H-2), 3.51 (ddd, *J* = 9.6, 3.7, 1.8 Hz, 1H, H-5), 1.20 (s, 9H, TBDPS). ¹³C NMR (125 MHz, CDCl₃) δ 138.43, 138.25, 138.16, 135.96, 135.71, 134.25, 133.54, 132.97, 131.74, 129.76, 129.72, 129.00, 128.59, 128.52, 128.51, 128.24, 128.06, 128.05, 127.95, 127.94, 127.92, 127.88, 127.85, 127.77, 127.34, 87.58, 86.96, 80.87, 80.05, 77.51, 77.40, 77.14, 76.89, 76.10, 75.49, 75.23, 62.76, 26.97, 19.40. HRMS (ESI, positive) calculated for C₄₉H₅₂O₅SSi⁺ [*M* + *Na*]⁺ 803.3197, found 803.3204.

(2R,3S,4S,5R,6S)-2-(((tert-butyldiphenylsilyl)oxy)methyl)-6-(phenylthio)tetrahydro-

2*H*-pyran-3,4,5-triol (29): Tetrabutylammonium fluoride (1.0 M solution in THF, 25.6 mL) was added to a solution of compound 28 (10 g, 12.8 mmol) in THF (120 mL) was added and the reaction mixture was stirred at room temperature for 2 hr. The reaction was quenched by adding a saturated solution of aqueous ammonium chloride (80 mL) at 0 °C. The aqueous layer was extracted with ethyl acetate (80 mL × 2). The combined organic layers were dried over magnesium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 2:1) to give compound 29 (6.253 g, 90 %) as a white solid. ¹H NMR (500 MHz, CDCl₃) δ 7.57–7.50 (m, 2h, Ph), 7.43–7.38 (m, 2H, Ph), 7.38–7.27 (m, 16H, Ph), 4.97–4.84 (m, 4H, Bn), 4.78 (d, *J* = 10.2 Hz, 1H, Bn), 4.74 (d, *J* = 9.9 Hz, 1H, H-1), 4.67 (d, *J* = 11.0 Hz, 1H Bn), 3.89 (dd, *J* = 12.0, 2.7 Hz, 1H, H-6), 3.80–3.66 (m, 2H, H-3, H-

6), 3.60 (t, J = 9.4 Hz, 1H, H-4), 3.51 (dd, J = 9.8, 8.7 Hz, 1H, H-2), 3.41 (ddd, J = 9.8, 4.9, 2.7 Hz ,1H, H-5). ¹³C NMR (125 MHz, CDCl₃) δ 138.31, 137.90, 137.84, 133.48, 131.85, 129.06, 128.54, 128.49, 128.46, 128.23, 128.05, 128.05, 127.99, 127.93, 127.81, 127.77, 127.70, 87.55, 86.57, 81.12, 79.33, 77.60, 75.85, 75.57, 75.14, 62.15. HRMS (ESI, positive) calculated for C₃₃H₃₄O₅S⁺ [M + Na]⁺ 565.2019, found 565.2040.

tert-butyldiphenyl(((2R,3R,4S,5R,6S)-3,4,5-tris(benzyloxy)-6-(phenylthio)tetrahydro-2H-pyran-2-yl)methoxy)silane (30): Sodium hydride (60% dispersion in mineral oil, 0.552 g, 13.8 mmol) was added to a solution of compound **29** (6.24 g, 11.5 mmol) in dry DMF (100 mL) at 0 °C, and the reaction mixture was stirred at room temperature for 30 min. The solution was cooled to 0 °C, and p-methoxybenzyl chloride (1.87 mL, 13.8 mmol) was added. The reaction mixture was stirred at room temperature for 8 h, quenched by adding methanol (30 mL) at 0 °C, and concentrated under reduced pressure. The residue was diluted with dichloromethane (80 mL) and washed with a saturated solution of aqueous ammonium chloride (60 mL). The remaining aqueous solution was extracted with dichloromethane (60 mL \times 2). The combined organic layers were washed with brine, dried over magnesium sulfate, filtered, and concentrated. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 4:1) to provide compound **30** (6.326 g, 83%) as a white solid. ¹H NMR (500 MHz, CDCl₃) δ 7.73–7.62 (m, 2H, Ph), 7.52–7.22 (m, 20H, Ph), 7.01–6.88 (m, 2H, Ph), 5.02–4.97 (m, 2H, Bn), 4.94 (d, J = 11.0 Hz, 1H, Bn), 4.91 (d, J = 10.9 Hz, 1H, Bn), 4.83 (d, J = 10.3 Hz, 1H, Bn)4.77 (d, J = 9.8 Hz, 1H, H-1), 4.70–4.62 (m, 2H, Bn), 4.56 (d, J = 11.6 Hz, 1H, Bn), 3.86 (s, 4H, -OMe, H-6), 3.83-3.77 (m, 2H, H-3, H-6), 3.74 (t, J = 9.3 Hz, 1H, H-4), 3.65-3.56 (m, 2H, H-2, H-5). ¹³C NMR (125 MHz, CDCl₃) & 159.25, 138.50, 138.15, 138.13, 134.00, 131.95, 130.41, 129.45, 128.98, 128.53, 128.50, 128.48, 128.30, 127.99, 127.94, 127.86, 127.78, 127.48, 113.83, 87.56, 86.82, 80.94, 79.19, 77.90, 75.89, 75.50, 75.10, 73.14, 68.69, 55.33. HRMS (ESI, positive) calculated for $C_{41}H_{42}O_6S^+ [M + Na]^+$ 685.2594, found 685.2607.

(1S,2S,3R,4S,6R)-4,6-diazido-3-(((2S,3R,4S,5R,6R)-3,4,5-tris(benzyloxy)-6-(((4-methoxybenzyl)oxy)methyl)tetrahydro-2H-pyran-2-yl)oxy)cyclohexane-1,2-diyl diacetate (31): Compound 4 (2.283 g, 7.65 mmol) and compound 30 (6.088 g, 9.18 mmol) were co-evaporated from dry toluene three times under reduced pressure and further dried under high vacuum 12 h. A solution of compound 4 and 29 in anhydrous diethyl ether (90 mL) and anhydrous dichloromethane (30 mL) was added to preactivated 4 Å molecular sieves. After stirring for 30 min at room temperature, the mixture was cooled to -40 °C. N-Iodosuccinimide (4.478 g, 19.9 mmol) was added, and the reaction mixture was stirred for 20 min. Trifluoromethanesulfonic acid (0.34 mL, 3.83 mmol) was slowly added, and the reaction was warmed to -20 °C and kept stirring for 30 min. The reaction solution was diluted with dichloromethane, and filtered through a Celite pad. The filtered solution was washed with 10% aqueous sodium bisulfite solution (80 mL), and a saturated aqueous solution of sodium bicarbonate solution (80 mL). The combined aqueous solution was extracted with dichloromethane (100 mL x 2). The combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:2) to give compound **31** (4.3 g, 66%) as a clear oil. ¹H NMR (500 MHz, CDCl₃) § 7.37–7.22 (m, 15H, Ph), 7.14–7.09 (m, 2H, Ph), 6.88–6.81 (m, 2H, Ph), 5.23– 5.15 (m, 2H, H-5), 4.95 (t, J = 10.0 Hz, 1H, H-6, H-1'), 4.85 (d, J = 10.8 Hz, 1H, Bn), 4.82–4.77 (m, 2H, Bn), 4.71 (d, J = 11.8 Hz, 1H, Bn), 4.65 (d, J = 11.8 Hz, 1H, Bn), 4.59 (d, J = 11.8 Hz, 1H, Bn), 4.43 (d, J = 11.0 Hz, 1H, Bn), 4.39 (d, J = 11.8 Hz, 1H, Bn),4.06 (dt, J = 10.1, 2.5 Hz, 1H, H-5'), 3.99–3.92 (m, 1H, H-3'), 3.79–3.71 (m, 4H, H-6'), 3.68 (dd, J = 10.1, 9.1 Hz, 1H, H-4'), 3.66–3.60 (m, 2H, H-6', H-4), 3.60–3.47 (m, 3H, H-

3, H-2', H-1), 2.35 (dt, J = 13.4, 4.5 Hz, 1H, H-2), 2.09 (s, 3H, OAc), 1.87 (s, 3H), OAc), 1.53 (dt, J = 13.5, 12.5 Hz, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 169.97, 169.74, 159.40, 138.75, 138.53, 137.98, 130.05, 129.84, 128.61, 128.50, 128.39, 128.24, 128.05, 127.73, 127.72, 127.66, 113.89, 98.10, 81.79, 79.71, 77.72, 77.62, 77.36, 75.77, 74.97, 74.21, 73.76, 73.50, 73.24, 71.55, 67.73, 60.13, 57.92, 55.34, 32.55, 20.90, 20.75. HRMS (ESI, positive) calculated for C₄₅H₅₀N₆O₁₁⁺ [M + Na]⁺ 873.3430, found 873.3441.

(1S,2R,3R,4S,6R)-4,6-diazido-3-(((2S,3R,4S,5R,6R)-3,4,5-tris(benzyloxy)-6-(((4-methoxybenzyl)oxy)methyl)tetrahydro-2H-pyran-2-yl)oxy)cyclohexane-1,2-diol (32): Sodium methoxide (5.4 M solution in methanol, 0.19 mL, 1.01 mmol) was added to a solution of compound 31 (4.3 g, 5.05 mmol) in methanol (50 mL), stirred at room temperature for 8 h, and neutralized with Amberlite IR-120 (H⁺ form). The mixture was filtered through a Celite pad, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc =1:1) to give compound **32** (3.139 g, 81 %) as a clear liquid. ¹H NMR (500 MHz, CDCl₃) δ 7.40–7.23 (m, 15H, Ph), 7.12-7.07 (m, 2H, Ph), 6.86-6.79 (m, 2H, Ph), 4.94 (d, J = 3.8 Hz, 1H, H-1'), 4.91 (d, J = 2.8 Hz, 2H, Bn), 4.88 (d, J = 11.7 Hz, 1H, Bn), 4.79–4.70 (m, 3H, Bn), 4.60 (d, J = 11.9 Hz, 1H, Bn), 4.44 (d, J = 10.6 Hz, 1H, Bn), 4.38 (d, J = 11.8 Hz, 1H), 4.05–3.98 (m, 2H, H-5', H-3'), 3.80–3.68 (m, 5H, OMe, H-6', H-4'), 3.66–3.59 (m, 2H, H-6', H-2'), 3.48-3.36 (m, 3H, H-1, H-5, H-6), 3.28-3.13 (m, 2H, H-3, H-4), 2.27 (dt, J =13.3, 4.4 Hz, 1H, H-2), 1.46 (dt, J = 13.4, 12.1 Hz, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) § 159.29, 138.43, 138.03, 136.84, 129.81, 129.79, 128.70, 128.53, 128.48, 128.43, 128.36, 127.79, 127.73, 127.71, 127.66, 113.76, 101.39, 85.45, 82.38, 79.13, 77.71, 75.57, 75.55, 75.39, 75.05, 74.58, 73.17, 71.46, 67.52, 59.67, 59.43, 55.20, 32.34. HRMS (ESI, positive) calculated for $C_{41}H_{46}N_6O_9^+$ [M + Na]⁺ 789.3218, found 789.3232.

(1R,2R,3S,5R,6S)-3,5-diazido-2-(((2S,3R,4S,5R,6R)-3,4,5-tris(benzyloxy)-6-(((4-methoxybenzyl)oxy)methyl)tetrahydro-2H-pyran-2-yl)oxy)-6-(((2R,3R,4S,5R)-3,4,5-tris-(benzyloxy)tetrahydro-2H-pyran-2-yl)oxy)cyclohexan-1-ol (33): Compound 8 (2.668) g, 5.210 mmol) and compound 32 (3.629 g, 4.73 mmol) were co-evaporated from dry toluene under reduced pressure three times and further dried under high vacuum for 8 hr. A solution of compound 8 and 32 in anhydrous dichloromethane (150 mL) was added to preactivated 4 Å molecular sieves and *N*-iodosuccinimide (1.704 g, 7.57 mmol). After stirring for 30 min at room temperature, the reaction mixture was cooled to -78 °C. Trifluoromethanesulfonic acid (0.13 mL, 1.42 mmol) was added dropwise, and the reaction was warmed to 0 °C slowly. The reaction was quenched by adding solid sodium sulfite (0.5 g), sodium bicarbonate (0.5 g), and a few drops of water. The reaction mixture was diluted with dichloromethane (150 mL), filtered, and washed with a saturated aqueous solution of sodium bicarbonate (100 mL \times 2) and brine. The residue was dried over sodium sulfate, filtered, concentrated under reduced pressure, and purified by flash column chromatography on silica gel (Hex:EtOAc = 3:2) to give compound **32** (3.621 g, 81%) as a clear oil. ¹H NMR (500 MHz, CDCl₃) δ 7.44–7.27 (m, 29H, Ph), 7.22–7.18 (m, 1H, Ph), 7.14 (m, 2H, Ph), 6.90–6.81 (m, 2H, Ph), 5.32 (d, J = 3.7 Hz, 1H, H-1'), 5.18 (d, J = 3.6 Hz, 1H, H-1"), 4.96 (dd, J = 11.0, 7.6 Hz, 2H, Bn), 4.90 (d, J = 11.0 Hz, 2H, Bn), 4.80 (td, J = 12.3, 9.1 Hz, 6H, Bn), 4.65 (dd, J = 13.4, 11.7 Hz, 2H, Bn), 4.59–4.53 (m, 1H, OH), 4.48 (d, J = 10.7 Hz, 1H, Bn), 4.43 (d, J = 11.8 Hz, 1H, Bn), 4.13 (dt, J = 10.1, 2.5 Hz, 1H, H-5'), 4.07-3.98 (m, 2H, H-3', H-5"), 3.95 (dd, J = 9.6, 8.2 Hz, 1H, H-3"), 3.86-3.78 (m, 1H, H-6'), 3.78-3.71 (m, 4H, H-4', OMe), 3.71-3.63 (m, 3H, H-5, H-2', H-6'), 3.63–3.51 (m, 4H, H-5", H-4", H-2", H-1), 3.42–3.35 (m, 1H, H-6), 3.35–3.25 (m, 2H, H-4, H-3), 2.36 (dt, J = 12.9, 4.3 Hz, 1H, H-2), 1.57 (g, J = 12.5 Hz, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 159.27, 138.87, 138.70, 138.32, 138.24, 138.15, 137.34,

129.96, 129.76, 128.55, 128.42, 128.41, 128.38, 128.35, 128.33, 128.30, 128.08, 128.00, 127.95, 127.81, 127.78, 127.74, 127.65, 127.60, 127.57, 113.77, 99.95, 98.03, 83.58, 82.07, 81.67, 80.99, 79.87, 79.37, 77.88, 77.66, 75.63, 75.57, 75.02, 74.76, 74.14, 73.60, 73.42, 73.14, 71.22, 67.70, 60.86, 60.13, 58.83, 55.18, 32.29. HRMS (ESI, positive) calculated for $C_{67}H_{72}N_6O_{13}^+$ [M + Na]⁺ 1191.5050, found 1191.5050.

2'-deamino-2'-hydroxy-gentamicin A₂ (34): Trimethylphosphine (1 M solution in THF, 0.08 mL, 0.80 mmol) was added to a solution of compound 33 (0.466 g, 0.39 mmol) in THF (9 mL) and 0.1 M NaOH (3 mL). The reaction mixture was stirred at 50 °C for 2 h, cooled to room temperature, and concentrated under reduced pressure. The residue was dissolved in a mixture of acetic acid (3 mL), water (9 mL) and methanol (9 mL). The solution was deaerated by evacuating the air inside of the flask and refilling it with nitrogen five times. Palladium hydroxide on carbon (20%, Degussa type, 0.1 g) was added and the solution was charged with hydrogen gas. The reaction mixture was stirred at room temperature under hydrogen gas for 24 hr. The solution was then filtered through a Celite pad, washed with water and methanol, and concentrated under reduced pressure. The residue was purified by chromatography on Amberlite CG-50 resin (NH_4^+ form) (1-5% concentrated ammonium hydroxide in water), concentrated under reduced pressure, acidified with hydrochloric acid (pH ca. 4), and lyophilized to provide compound 34 (0.167 g, 92%) as an oily solid. ¹H NMR (500 MHz, D₂O) δ 5.59 (d, J = 3.9 Hz, 1H, H-1'), 5.04 (d, J = 3.0 Hz, 1H, H-1"), 4.05 (ddd, J = 10.1, 8.3, 3.2 Hz, 1H, H-5'), 4.00–3.90 (m, 2H, H-5, H-3"), 3.80 (t, J = 9.5 Hz, 1H, H-3'), 3.76–3.55 (m, 9H, H-4", H-2', H-4, H-1, H-5", H-6, H-5", H-2", H-3), 3.47 (dd, J = 13.4, 3.2 Hz, 1H, H-6'), 3.40 (t, J = 9.5 Hz, 1H, H-4'), 3.20 (dd, J = 13.3, 8.3 Hz, 1H, H-6'), 2.57 (dt, J = 12.6, 4.2 Hz, 1H, H-2), 1.96 (q, J = 12.4 Hz, 1H, H-2). ¹³C NMR (125 MHz, D₂O) δ 101.84, 95.44, 83.82, 78.05,

72.95, 72.11, 71.98, 71.73, 70.80, 70.68, 68.90, 68.63, 62.20, 49.95, 47.54, 40.30, 27.52. HRMS (ESI, positive) calculated for $C_{17}H_{32}N_2O_{12}^{+}[M + Na]^{+}$ 479.1847, found 479.1854.

2.2.14. Synthesis of 2'-Deamino-2'-hydroxy-6'-amino-6'-dehydroxy-gentamicin A2

The overall synthetic scheme is shown in Scheme 2-12.



Scheme 2-12. Synthetic scheme for the preparation of 2'-deamino-2'-hydroxy-6'-amino-6'-dehydroxy-gentamicin A₂.

(*1R*,2*R*,3*S*,5*R*,6*S*)-3,5-diazido-2-(((2*S*,3*R*,4*S*,5*R*,6*R*)-3,4,5-tris(benzyloxy)-6-(hydroxylmethyl)tetrahydro-2*H*-pyran-2-yl)oxy)-6-(((2*R*,3*R*,4*S*,5*R*)-3,4,5-tris(benzyloxy)tetrahydro-2*H*-pyran-2-yl)oxy)cyclohexan-1-ol (35): 2,3-Dichloro-5,6-dicyano-1,4benzoquinone (0.636 g, 2.802 mmol) was added to a solution of compound 33 (2.184 g,

1.87 mmol) in dichloromethane (40 mL) and water (4 mL). The reaction mixture was stirred for 4 h at room temperature and quenched by adding a saturated aqueous solution of sodium bicarbonate (40 mL). The aqueous layer was extracted with dichloromethane (60 mL \times 3), and the combined organic phase was washed with brine, dried over magnesium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:2) to give compound **35** (1.251 g, 64%) as a clear liquid. ¹H NMR (500 MHz, CDCl₃) δ 7.36–7.27 (m, 29H, Ph), 7.21-7.16 (m, 1H, Ph), 5.27 (d, J = 3.7 Hz, 1H, H-1'), 5.12 (d, J = 3.6 Hz, 1H, H-1"), 5.00–4.82 (m, 6H, Bn), 4.82 - 4.72 (m, 6H, Bn), 4.66 (dd, J = 23.4, 11.3 Hz, 2H, Bn), 4.07–3.95 (m, 3H, H-5', H-3', H-5"), 3.91 (dd, J = 9.6, 8.2 Hz, 1H, H-3"), 3.84 (dd, J = 11.9, 2.6 Hz, 1H, H-6'), 3.76 (dd, J = 12.0, 3.6 Hz, 1H, H-6'), 3.64-3.49 (m, 7H, 10.00)H-5, H-4, H-4", H-2', H-5", H-1, H-2"), 3.39-3.23 (m, 3H, H-6, H-3, H-4), 2.37 (dt, J =13.3, 4.4 Hz, 1H, H-2), 1.54 (q, J = 12.5 Hz, 2H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 138.79, 138.58, 138.25, 138.09, 138.06, 137.29, 128.56, 128.49, 128.43, 128.39, 128.34, 128.11, 128.01, 128.00, 127.94, 127.89, 127.81, 127.79, 127.75, 127.64, 127.58, 99.57, 98.17, 83.23, 82.06, 81.87, 80.88, 79.95, 79.27, 77.75, 77.22, 75.61, 75.11, 74.84, 74.12, 73.61, 73.46, 71.78, 61.64, 60.95, 60.02, 58.88, 32.34. HRMS (ESI, positive) calculated for $C_{59}H_{64}N_6O_{12}^+$ $[M + Na]^+$ 1071.4474, found 1071.4477.

(*1R*,2*R*,3*S*,5*R*,6*S*)-3,5-diazido-2-(((2*R*,3*R*,4*S*,5*R*,6*R*)-6-(azidomethyl)-3,4,5-tris(benzyloxy)tetrahydro-2*H*-pyran-2-yl)oxy)-6-(((2*R*,3*R*,4*S*,5*R*)-3,4,5-tris(benzyloxy)tetrahydro-2*H*-pyran-2-yl)oxy)cyclohexan-1-ol (36): *p*-Toluenesulfonyl chloride (1.137 g, 5.96 mmol) was added to a solution of compound 35 (1.251 g, 1.19 mmol) in pyridine (30 mL) at 0 °C. The reaction mixture was stirred at room temperature for 9 hr. After the addition of methanol (10 mL) at 0 °C, the mixture was stirred at room temperature for 30 min. The solution was concentrated under reduced pressure, and co-evaporated with

toluene twice. The residue was diluted with dichloromethane (50 mL), and washed with a saturated aqueous solution of sodium bicarbonate (40 mL). The aqueous layer was extracted with dichloromethane (50 mL \times 2), and the combined organic phase was washed with brine, dried over magnesium sulfate, filtered and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:1) to give tosylate **36** (0.766 g, 56%) as a clear oil. ¹H NMR (500 MHz, $CDCl_3$) δ 7.79 (d, J = 8.2 Hz, 2H, Ph), 7.41–7.27 (m, 29H, Ph), 7.23–7.13 (m, 3H, Ph), 5.27 (d, J = 3.6 Hz, 1H, H-1'), 5.08 (d, J = 3.6 Hz, 1H, H-1"), 4.93 (dd, J = 20.2, 11.0 Hz, 2H, Bn), 4.88-4.79 (m, 3H, Bn), 4.79-4.70 (m, 5H, Bn), 4.63 (d, J = 11.5 Hz, 1H, Bn), 4.48 (d, J = 10.6 Hz, 1H, Bn), 4.43 (s, 1H, OH), 4.35 (dd, J = 10.6, 3.2 Hz, 1H, H-6'), 4.21 (dd, J = 10.6, 2.1 Hz, 1H, H-6'), 4.15 (dt, J = 10.1, 2.6 Hz, 1H, H-5'), 4.03–3.83 (m, 3H, H-5", H-3', H-3"), 3.63–3.46 (m, 7H, H-5, H-4", H-4', H-5", H-2', H-1, H-2"), 3.36– 3.23 (m, 3H, H-6, H-3, H-4), 2.40 (s, 3H, Me), 2.37–2.26 (m, 1H, H-2), 1.54–1.42 (m, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 144.81, 138.77, 138.47, 138.21, 138.08, 137.78, 137,26, 132.84, 129.80, 128.55, 128.43, 128.40, 128.34, 128.29, 128.09, 128.04, 127.99, 127.93, 127.90, 127.82, 127.81, 127.64, 127.59, 99.02, 98.29, 82.72, 82.43, 81.68, 80.80, 79.64, 79.23, 77.64, 77.22, 76.83, 75.60, 75.55, 75.07, 74.95, 73.89, 73.62, 73.49, 69.43, 68.38, 61.02, 59.91, 58.66, 32.18, 21.64. HRMS (ESI, positive) calculated for $C_{66}H_{70}N_6O_{14}S^+ [M + Na]^+$ 1225.4563, found 1225.4569.

(1R,2R,3S,5R,6S)-3,5-diazido-2-(((2R,3R,4S,5R,6R)-6-(azidomethyl)-3,4,5-tris(benzyloxy)tetrahydro-2*H*-pyran-2-yl)oxy)-6-(((2R,3R,4S,5R)-3,4,5-tris(benzyloxy)tetrahydro-2*H*-pyran-2-yl)oxy)cyclohexan-1-ol (37): Sodium azide (0.215 g, 3.30 mmol) and tetrabutylammonium iodide (0.005 g, 0.013 mmol) was added to a solution of the previously synthesized tosylate 36 (0.795 g, 0.66 mmol) in DMF (20 mL). The reaction mixture was stirred at 60 °C for 8 h. After cooling to room temperature, the reaction solution was evaporated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:1) to give compound **37** (0.511 g, 72%) as a colorless oil. ¹H NMR (500 MHz, CDCl₃) δ 7.42–7.27 (m, 29H, Ph), 7.23–7.17 (m, 1H, Ph), 5.39 (d, *J* = 3.7 Hz, 1H, H-1'), 5.10 (d, *J* = 3.6 Hz, 1H, H-1"), 4.98 (d, *J* = 10.9 Hz, 1H, Bn), 4.92 (d, *J* = 11.1 Hz, 2H, Bn), 4.86 (d, *J* = 10.9 Hz, 2H, Bn), 4.82–4.73 (m, 5H, Bn), 4.64 (dd, *J* = 11.3, 6.1 Hz, 2H, Bn), 4.47 (d, *J* = 1.8 Hz, 1H, OH), 4.18 (ddd, *J* = 10.0, 4.3, 2.5 Hz, 1H, H-5'), 4.02 (t, *J* = 9.4 Hz, 1H, H-3'), 3.98–3.86 (m, 2H, H-5", H-3"), 3.66–3.50 (m, 8H, H-5, H-2', H-5", H-4", H-6', H-4', H-1, H-2"), 3.44 (dd, *J* = 13.2, 4.3 Hz, 1H, H-6'), 3.40–3.31 (m, 3H, H-6, H-3, H-4), 2.38 (dt, *J* = 13.5, 4.0 Hz, 1H, H-2), 1.63–1.46 (m, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 138.79, 138.52, 138.23, 138.10, 137.95, 137.36, 128.55, 128.49, 128.44, 128.42, 128.36, 128.29, 128.08, 128.01, 127.94, 127.93, 127.82, 127.75, 127.67, 127.60, 98.96, 98.33, 82.91, 82.44, 81.69, 80.79, 79.96, 79.25, 78.18, 77.63, 77.24, 75.61, 75.21, 75.04, 73.91, 73.62, 73.49, 70.76, 61.05, 59.94, 58.82, 51.26, 32.31. HRMS (ESI, positive) calculated for C₅₉H₆₃N₉O₁₁⁺ [*M* + *Na*]⁺ 1096.4539, found 1096.4550.

2'-deamino-2'-hydroxy-6'-amino-6'-deoxy-gentamicin A_2 (**38**): Trimethylphosphine (1 M solution in THF, 1.40 mL, 1.40 mmol) was added to a solution of compound **37** (0.5 g, 0.465 mmol) in THF (9 mL) and 0.1 M NaOH (3 mL). The reaction mixture was stirred at 50 °C for 2 h, cooled to room temperature, and concentrated under reduced pressure. The residue was dissolved in a mixture of acetic acid (3 mL), water (9 mL) and methanol (9 mL). The solution was deaerated by evacuating the air inside of the flask and refilling it with nitrogen five times. Palladium hydroxide on carbon (20%, Degussa type, 0.2 g) was added and the solution was charged with hydrogen gas. The reaction mixture was filtered through a Celite pad, washed with water and methanol, and concentrated under reduced reduced reduced through a Celite pad, washed with water and methanol, and concentrated under reduced methanol.

pressure. The residue was purified by chromatography on Amberlite CG-50 resin (NH₄⁺ form) (5-10% concentrated ammonium hydroxide in water), concentrated under reduced pressure, acidified with hydrochloric acid (pH ca. 4), and lyophilized to provide compound **38** (0.188 g, 89%) as an oily solid. ¹H NMR (500 MHz, D₂O) δ 5.62 (d, *J* = 3.8 Hz, 1H, H-1'), 5.07 (d, *J* = 3.2 Hz, 1H, H-1''), 4.08 (ddd, *J* = 10.0, 8.3, 3.2 Hz, 1H, H-5'), 4.02–3.95 (m, 2H, H-5, H-3''), 3.86–3.81 (m, 1H, H-3'), 3.81–3.55 (m, 9H, H-4'', H-5'', H-4, H-5'', H-1, H-6, H-2', H-2'', H-3), 3.50 (dd, *J* = 13.3, 3.2 Hz, 1H, H-6'), 3.47 – 3.38 (m, 1H, H-4'), 3.23 (dd, *J* = 13.4, 8.3 Hz, 1H, H-6'), 2.64 – 2.55 (m, 1H, H-2), 1.99 (q, *J* = 12.6 Hz, 1H, H-2). ¹³C NMR (125 MHz, D₂O) δ 101.84, 95.47, 83.82, 78.07, 72.97, 72.13, 72.02, 71.76, 70.84, 70.71, 68.93, 68.65, 62.23, 49.98, 47.59, 40.34, 27.54. HRMS (ESI, positive) calculated for C₁₇H₃₃N₃O₁₁⁺ [*M* + *Na*]⁺ 478.2007, found 478.2009.

2.2.15. Synthesis of Garosamine Derivative (44)

The overall synthetic scheme is shown in Scheme 2-13.



Scheme 2-13. Synthetic scheme for the preparation of garosamine derivative (44).

(3aR,6S,7R,7aR)-7-(benzyloxy)-1,3a-dimethyl-6-(p-tolylthio)hexahydro-2H-pyrano-

[4,3-d]oxazol-2-one (44): Compound 43 was prepared as previously described.¹⁶³ Sodium hydride (60% in mineral oil, 1.022 g, 25.56 mmol) was added to a solution of compound 43 (7.19 g, 23.24 mmol) in DMF (200 mL) at 0 °C under a nitrogen atmosphere. The reaction mixture was stirred for 30 min at room temperature. After the reaction solution was cooled to 0 °C, benzyl bromide (3.32 mL, 27.88 mmol) was added, and the mixture was stirred at room temperature for 3 h. Methanol (30 mL) was added at 0 °C to quench the reaction. After all solvents were evaporated under reduced pressure, the residue was diluted with dichloromethane (200 mL) and water (150 mL). The aqueous layer was extracted with dichloromethane (150 mL \times 3). The combined organic layers were washed with brine, dried over magnesium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 2:1) to give compound 44 (8.81 g, 95%) as a clear solid. ¹H NMR (500 MHz, CDCl₃) δ 7.46–7.28 (m, 7H, Ph), 7.13 (d, J = 7.7 Hz, 2H, Ph), 5.26 (d, J = 3.3 Hz, 1H, H-1), 4.84 (d, J = 11.7 Hz, 1H, Bn), 4.54 (d, J = 11.2 Hz, 1H, Bn), 4.16 (d, J = 12.3 Hz, 1H, H-5), 3.81 (t, J = 3.6 Hz, 1H, H-2), 3.61 (d, J = 12.3 Hz, 1H, H-5),3.44 (d, J = 4.1 Hz, 1H, H-3), 2.86 (s, 3H, NMe), 2.34 (s, 3H, PhMe), 1.52 (s, 3H, Me). ¹³C NMR (125 MHz, CDCl₃) δ 157.59, 138.10, 137.00, 132.25, 130.34, 129.97, 128.69, 128.29, 127.84, 85.96, 75.97, 75.59, 72.93, 66.46, 64.25, 30.36, 24.09, 21.20. HRMS (ESI, positive) calculated for $C_{22}H_{25}NO_4S^+[M+Na]^+$ 422.1397, found 422.1411.

2.2.16. Synthesis of Gentamicin B

The overall synthetic scheme is shown in Scheme 2-14.



Scheme 2.14. Synthetic scheme for the preparation of gentamicin B.

(*3aR,6R,7R,7aR*)-7-(benzyloxy)-6-(((*1S,2R,3R,4S,6R*)-4,6-diazido-3-(((*2R,3R,4S,5R, 6R*)-6-(azidomethyl)-3,4,5-tris(benzyloxy)tetrahydro-2*H*-pyran-2-yl)oxy)-2-hydroxycyclohexyl)oxy)-1,3a-dimethylhexahydro-2*H*-pyrano[4,3-d]oxazol-2-one (46): Compound 45 was prepared as previously described.¹⁵⁷ Compound 44 (0.263 g, 0.659 mmol) and compound 45 (0.402 g, 0.599 mmol) were co-evaporated from dry toluene under reduced pressure three times and further dried under high vacuum for overnight. A solution of compound 44 and 45 in anhydrous dichloromethane (15 mL) was added to preactivated 4 Å molecular sieves and *N*-iodosuccinimide (0.216 g, 0.959 mmol) were added. After stirring for 30 min at room temperature, the reaction mixture was cooled to -78 °C. Trifluoromethanesulfonic acid (0.016 mL, 0.179 mmol) was added dropwise over 10 min, and the reaction was warmed to 0 °C slowly. The reaction was quenched by adding solid sodium sulfite (0.2 g), sodium bicarbonate (0.2 g), and a few drops of water. The reaction mixture was diluted with dichloromethane (20 mL), filtered, and washed with a saturated aqueous solution of sodium bicarbonate (60 mL × 2) and brine. The residue was dried over sodium sulfate, filtered, concentrated in reduced pressure, and purified by flash column chromatography on silica gel (Hex:EtOAc = 3:2) to give compound **46** (0.329 g, 58%) as a clear oil. ¹H NMR (500 MHz, CDCl₃) δ 7.39–7.28 (m, 20H, Ph), 5.43 (d, *J* = 3.7 Hz, 1H, H-1'), 5.18 (d, *J* = 3.3 Hz, 1H, H-1"), 4.99 (d, *J* = 10.9 Hz, 1H, Bn), 4.94–4.85 (m, 3H, Bn), 4.85–4.76 (m, 2H, Bn), 4.64 (d, *J* = 11.0 Hz, 1H, Bn), 4.59 (d, *J* = 11.5 Hz, 1H, Bn), 4.38 (d, *J* = 1.6 Hz, 1H, OH), 4.22–4.09 (m, 2H, H-5', H-5"), 4.03 (t, *J* = 9.3 Hz, 1H, H-3'), 3.86 (dd, *J* = 5.4, 3.3 Hz, 1H, H-2"), 3.71–3.59 (m, 3H, H-2', H-5, H-5"), 3.58–3.36 (m, 8H, H-6', H-4', H-1, H-6, H-3", H-6, H-4, H-3), 2.86 (s, 3H, NMe), 2.36 (dt, *J* = 13.2, 4.1 Hz, 1H, H-2), 1.56 (q, *J* = 12.2 Hz, 1H, H-2), 1.30 (s, 3H, Me). ¹³C NMR (125 MHz, CDCl₃) δ 157.36, 138.55, 138.00, 137.53, 137.38, 128.68, 128.58, 128.57, 128.53, 128.28, 128.23, 128.15, 128.00, 127.93, 127.82, 127.77, 98.89, 96.15, 82.72, 82.47, 81.77, 80.07, 78.30, 77.57, 75.69, 75.29, 74.83, 74.68, 73.90, 70.87, 65.59, 62.77, 60.01, 58.81, 51.32, 32.12, 30.47, 23.33. HRMS (ESI, positive) calculated for C₄₈H₅₄N₁₀O₁₁⁺ [*M* + *Na*]⁺ 969.3866, found 969.3864.

(3R,4R,5R,6R)-5-(benzyloxy)-6-(((1S,2R,3R,4S,6R)-4,6-diazido-3-(((2R,3R,4S,5R,6R)-6-(azidomethyl)-3,4,5-tris(benzyloxy)tetrahydro-2H-pyran-2-yl)oxy)-2-hydroxycyclohexyl)oxy)-3-methyl-4-(methylamino)tetrahydro-2H-pyran-3-ol (47): To a solution of compound 46 (0.292 g, 0.309 mmol) in absolute ethanol (5 mL) was added 2N NaOH (2 mL). The reaction mixture was stirred at 90 °C for 10 h, and cooled to room temperature. The solution was diluted with ethyl acetate (20 mL) and washed with H₂O (20 mL). The aqueous layer was extracted with ethyl acetate (20 mL × 3). The combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (4% MeOH in DCM) to give compound 47 (0.231 g, 81%) as a clear oil. ¹H NMR (500 MHz, CDCl₃) δ 7.40–7.30 (m, 20H, Ph), 5.42 (d, *J* = 3.7 Hz, 1H, H-1'), 5.31 (d, *J* = 3.4 Hz, 1H, H-1"), 5.00 (d, *J* = 11.0 Hz, 1H, Bn), 4.94 (d, *J* = 11.0 Hz, 1H, Bn), 4.89 (d, *J* = 11.0 Hz, 1H, Bn), 4.86–4.76 (m, 3H, Bn), 4.66 (d, *J* = 11.0 Hz, 1H, Bn), 4.56 (s, 1H, OH), 4.50 (d, *J* = 11.8 Hz, 1H, Bn), 4.21 (ddd, *J* = 10.0, 4.3, 2.5 Hz, 1H, H-5'), 4.05 (t, *J* = 9.3 Hz, 1H, H-3'), 3.91 (d, *J* = 12.3 Hz, 1H, H-5"), 3.70–3.49 (m, 6H, H-5, H-2', H-6', H-4', H-2", H-1), 3.48–3.32 (m, 6H, H-6', H-3", H-4, H-6, H-3, H-5"), 2.64 (d, *J* = 10.3 Hz, 1H, NH), 2.53 (s, 3H, NMe), 2.40–2.35 (m, 1H, H-2), 1.66–1.52 (m, 1H, H-2), 1.11 (s, 3H, Me). ¹³C NMR (125 MHz, CDCl₃) δ 138.50, 137.95, 137.77, 137.44, 128.65, 128.52, 128.47, 128.18, 128.13, 128.11, 128.09, 127.95, 127.76, 127.71, 99.06, 96.92, 82.63, 82.45, 81.75, 80.13, 78.27, 76.04, 75.63, 75.24, 75.19, 73.98, 72.19, 70.98, 70.82, 66.25, 63.01, 60.03, 58.83, 51.27, 38.67, 32.21, 24.57. HRMS (ESI, positive) calculated for C₄₇H₅₆N₁₀O₁₀⁺ [*M* + *Na*]⁺ 943.4073, found 943.4062.

Gentamicin B (48): Trimethylphosphine (1 M solution in THF, 1.22 mL, 1.22 mmol) was added to a solution of compound 47 (0.188 g, 0.204 mmol) in THF (6 mL) and 0.1 M NaOH (2 mL) was added. The reaction mixture was stirred at 50 °C for 2 h, cooled to room temperature, and concentrated under reduced pressure. The residue was dissolved in a mixture of acetic acid (1.5 mL), water (4.5 mL) and methanol (4.5 mL). The solution was deaerated by evacuating the air inside of the flask and refilling it with nitrogen five times. Palladium hydroxide on carbon (20%, Degussa type, 0.1 g) was added and the solution was charged with hydrogen gas. The reaction mixture was stirred at room temperature under hydrogen gas for 24 hr. The solution was filtered through a Celite pad, washed with water and methanol, and concentrated under reduced pressure. The residue was purified by chromatography on Amberlite CG-50 resin (NH₄⁺ form) (5-10% concentrated ammonium hydroxide in water), concentrated under reduced pressure, acidified with hydrochloric acid (pH ca. 4), and lyophilized to provide compound **48**

(0.086 g, 88%) as an oily solid. ¹H NMR (500 MHz, D₂O) δ 5.58 (d, *J* = 3.9 Hz, 1H, H-1'), 5.12 (d, *J* = 3.7 Hz, 1H, H-1''), 4.23 (dd, *J* = 10.9, 3.7 Hz, 1H, H-2''), 4.09–3.99 (m, 2H, H-5', H-5''), 3.98–3.90 (m, 2H, H-5, H-4), 3.82–3.73 (m, 2H, H-3', H-6), 3.69 (dd, *J* = 9.9, 3.9 Hz, 1H, H-2'), 3.67–3.58 (m, 2H, H-1, H-3), 3.52 (dd, *J* = 25.2, 11.8 Hz, 2H, H-3'' H-5''), 3.46 (dd, *J* = 13.4, 3.3 Hz, 1H, H-6'), 3.40 (dd, *J* = 10.0, 9.0 Hz, 1H, H-4'), 3.20 (dd, *J* = 13.4, 8.0 Hz, 1H, H-6'), 2.94 (s, 3H, NMe), 2.57 (dt, *J* = 12.5, 4.2 Hz, 1H, H-2), 1.95 (q, *J* = 12.5 Hz, 1H, H-2), 1.36 (s, 3H, Me). ¹³C NMR (125 MHz, D₂O) δ 101.14, 95.76, 83.79, 78.19, 72.28, 72.07, 70.69, 70.65, 69.85, 68.64, 67.68, 66.27, 63.24, 49.72, 47.52, 40.22, 34.42, 27.50, 20.85. HRMS (ESI, positive) calculated for C₁₉H₃₈N₄O₁₀⁺ [*M* + *Na*]⁺ 505.2480, found 505.2480.

2.2.17. Synthesis of JI-20A

The overall synthetic scheme is shown in Scheme 2-15.



Scheme 2.15. Synthetic scheme for the preparation of JI-20A.

(1S,2S,3R,4S,6R)-4,6-diazido-3-(((2S,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-(hydroxymethyl)tetrahydro-2*H*-pyran-2-yl)oxy)cyclohexane-1,2-diyl diacetate (49): 2,3-Dichloro-5,6-dicyano-1,4-benzoquinone (0.251 g, 1.109 mmol) was added to a stirring solution of compound 17 (0.726 g, 0.925 mmol) in dichloromethane (20 mL) and water (2 mL). The reaction mixture was stirred at room temperature for 3 h and quenched by adding a saturated aqueous solution of sodium bicarbonate (40 mL). The aqueous solution was extracted with dichloromethane (50 mL × 3), and the combined organic layers were washed with brine. The organic layers were dried over magnesium sulfate, filtered, and concentrated under reduced pressure. The crude product was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:2) to provide compound **49** (0.442 g, 72%) as a clear oil. ¹H NMR (400 MHz, CDCl₃) δ 7.42–7.26 (m, 10H, Ph), 5.14 (t, *J* = 9.8 Hz, 1H, H-5), 5.08 (d, *J* = 3.8 Hz, 1H, H-1'), 4.97–4.81 (m, 4H, H-6, Bn), 4.72 (d, *J* = 11.0 Hz, 1H, Bn), 4.08 (dt, *J* = 10.1, 2.9 Hz, 1H, H-5'), 3.98 (dd, *J* = 10.4, 8.9 Hz, 1H, H-3'), 3.88–3.76 (m, 2H, H-6', H-6'), 3.68 (dd, *J* = 10.1, 8.9 Hz, 1H, H-4'), 3.64–3.54 (m, 2H, H-1, H-4), 3.40 (ddd, *J* = 12.4, 9.9, 4.6 Hz, 1H, H-3), 3.32 (dd, *J* = 10.4, 3.8 Hz, 1H, H-2'), 2.36 (dt, *J* = 13.3, 4.6 Hz, 1H, H-2), 2.09 (s, 6H, OAc), 1.58 (dt, *J* = 13.3, 12.5 Hz, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 169.93, 169.58, 137.77, 137.71, 128.59, 128.53, 128.06, 128.05, 127.98, 127.94, 99.48, 79.53, 78.67, 77.74, 75.41, 75.08, 74.07, 73.43, 72.33, 63.23, 61.10, 58.70, 57.63, 31.78, 20.67, 20.60. HRMS (ESI, positive) calculated for C₃₀H₃₅N₉O₉⁺ [*M* + *Na*]⁺ 688.2450, found 688.2466.

(15,25,3R,4S,6R)-4,6-diazido-3-(((25,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-((tosyloxy)methyl)tetrahydro-2*H*-pyran-2-yl)oxy)cyclohexane-1,2-diyl diacetate (50): *p*-Toluenesulfonyl chloride (0.304 g, 1.595 mmol) was added to a solution of compound **49** (0.213 g, 0.319 mmol) in pyridine (20 mL) at 0 °C, and the reaction mixture was stirred at room temperature for 15 h, and methanol (5 mL) was added at 0 °C to quench the reaction. The reaction solution was concentrated under reduced pressure, and coevaporated with toluene twice. The residue was diluted with dichloromethane (50 mL) and washed with a saturated aqueous solution of ammonium chloride (40 mL × 3). The organic layer was washed with brine, dried over magnesium sulfate, filtered, and concentrated under reduced pressure. The crude product was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:1) to provide compound **50** (0.238 g, 91%) as a clear liquid. ¹H NMR (500 MHz, CDCl₃) δ 7.80–7.78 (m, 2H, Ph), 7.35–7.30 (m, 10H, Ph), 7.21–7.15 (m, 2H, Ph), 5.11 (t, J = 9.7 Hz, 1H, H-5), 5.03 (d, J = 3.6 Hz, 1H, H-1'), 4.90 (t, J = 9.8 Hz, 1H, H-6), 4.87–4.81 (m, 2H, 2×Bn), 4.80 (d, J = 10.6 Hz, 1H, Bn), 4.51 (d, J = 10.7 Hz, 1H, Bn), 4.35 (dd, J = 10.8, 3.0 Hz, 1H, H-6'), 4.23–4.19 (m, 2H, H-5', H-6), 3.93 (dd, J = 10.4, 8.8 Hz, 1H, H-3'), 3.65–3.58 (m, 2H, H-4', H-1), 3.55 (t, J = 9.6 Hz, 1H, H-4), 3.44–3.34 (m, 1H, H-3), 3.28 (dd, J = 10.4, 3.6 Hz, 1H, H-2'), 2.41 (s, 3H, Me), 2.39–2.32 (m, 1H, H-2), 2.08 (s, 3H, OAc), 2.08 (s, 3H, OAc), 1.61–1.52 (m, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 169.95(OC(=O)), 169.57 (O(C=O)), 145.11 (Ph), 137.61 (Ph), 137.47 (Ph), 132.88 (Ph), 129.99 (Ph), 128.63 (Ph), 128.62 (Ph), 128.13 (Ph), 128.10 (Ph), 128.01 (Ph), 127.95 (Ph), 99.18 (C-1'), 79.58 (C-3'), 78.36 (C-4), 77.50 (C-4'), 75.46 (Bn), 75.26 (Bn), 74.17 (C-6), 73.52 (C-5), 69.98 (C-5'), 68.02 (C-6'), 63.06 (C-2'), 58.61 (C-3), 57.77 (C-1), 31.81 (C-2), 21.77 (Me), 20.76 (Ac), 20.70 (Ac). HRMS (ESI, positive) calculated for C₃₇H₄₁N₉O₁₁S⁺ [M + Na]⁺ 842.2538, found 842.2529.

(15,25,3R,45,6R)-4,6-diazido-3-(((2R,3R,4R,5R,6R)-3-azido-6-(azidomethyl)-4,5-bis-(benzyloxy)tetrahydro-2*H*-pyran-2-yl)oxy)cyclohexane-1,2-diyl diacetate (51): Sodium azide (0.155 g, 2.385 mmol) and tetrabutylammonium iodide (0.02 g, 0.059 mmol) were added to a stirring solution of compound **50** (0.925 g, 1.192 mmol) in DMF (30 mL). The reaction mixture was stirred at 60 °C for 12 h, and cooled to room temperature before concentration under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:1) to provide compound **51** (0.734 g, 89%) as a clear liquid. ¹H NMR (400 MHz, CDCl₃) δ 7.39–7.26 (m, 10H, Ph), 5.15 (t, *J* = 9.8 Hz, 1H, H-5), 5.11 (d, *J* = 3.8 Hz, 1H, H-1'), 4.93 (t, *J* = 10.0 Hz, 1H, H-6), 4.90–4.81 (m, 3H, Bn), 4.62 (d, *J* = 11.1 Hz, 1H, Bn), 4.23 (dt, *J* = 10.0, 3.3 Hz, 1H, H-5'), 3.95 (dd, *J* = 10.4, 8.8 Hz, 1H, H-3'), 3.67–3.55 (m, 3H, H-4', H-1, H-4), 3.55–3.49 (m, 1H, H-6'), 3.48–3.37 (m, 2H, H-6', H-3), 3.34 (dd, *J* = 10.4, 3.8 Hz, 1H, H-2'), 2.42 (dt, J = 13.4, 4.6 Hz, 1H, H-2), 2.09 (d, J = 2.7 Hz, 6H, OAc), 1.69–1.57 (m, 1H, H-2). ¹³C NMR (100 MHz, CDCl₃) δ 170.00, 169.67, 137.67, 128.72, 128.67, 128.24, 128.15, 127.97, 99.27, 79.62, 99.25, 79.62, 78.80, 78.60, 75.57, 75.42, 74.24, 73.60, 71.53, 63.29, 58.79, 57.83, 50.94, 31.97, 20.81, 20.73. HRMS (ESI, positive) calculated for $C_{30}H_{34}N_{12}O_8^+ [M + Na]^+$ 713.2515., found 713.2510.

(1S,2R,3R,4S,6R)-4,6-diazido-3-(((2R,3R,4R,5R,6R)-3-azido-6-(azidomethyl)-4,5bis(benzyloxy)tetrahydro-2H-pyran-2-yl)oxy)cyclohexane-1,2-diol (52): Sodium methoxide (5 M solution in methanol, 0.05mL, 0.241 mmol) was added to a solution of 51 (0.834 g, 1.205 mmol) in methanol. The reaction mixture was stirred at room temperature for 12 h. The basic solution was neutralized with Amberlite IR-120 resin (H⁺ form), filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 2:1) to give compound 52 (0.665 g, 91%) as a clear liquid. ¹H NMR (400 MHz, CDCl₃) δ 7.39–7.26 (m, 10H, Bn), 5.17 (d, J = 3.7 Hz, 1H, H-1'), 4.95–4.85 (m, 3H, Bn), 4.64 (d, J = 11.0 Hz, 1H, Bn), 4.16-4.08 (m, 1H, H-5'), 4.06 (d, J = 2.1 Hz, 1H, OH), 3.99 (dd, J = 10.2, 8.9 Hz, 1H, H-3'), 3.67–3.59 (m, 2H, OH, H-6'), 3.56 (dd, J = 13.3, 2.5 Hz, 1H, H-5), 3.53–3.38 (m, 4H, H-5, H-6', H-1, H-4), 3.32 - 3.25 (m, 2H, H-6, H-3), 2.37-2.28 (m, 1H, H-2), 1.55-1.44 (m, 1H, H-2). ¹³C NMR (100 MHz, CDCl₃) δ 137.60, 137.46, 128.74, 128.69, 128.27, 128.24, 128.21, 127.94, 99.56, 84.04, 80.94, 78.80, 76.08, 75.87, 75.46, 75.43, 71.50, 64.37, 59.81, 58.93, 51.04, 32.17. HRMS (ESI, positive) calculated for $C_{26}H_{30}N_{12}O_6^+$ [M +Na]⁺ 629.2303, found 629.2311.

(3aR, 6R, 7R, 7aR) - 7 - (benzyloxy) - 6 - (((1S, 2S, 3R, 4S, 6R) - 4, 6 - diazido - 3 - (((2R, 3R, 4R, 5R, 6 - R) - 3 - azido - 6 - (azidomethyl) - 4, 5 - bis(benzyloxy) tetrahydro - 2H - pyran - 2 - yl)oxy) - 2 - hyd-roxycyclohexyl)oxy) - 1, 3a - dimethylhexahydro - 2H - pyrano[4, 3 - d]oxazol - 2 - one (53): Compound 44 (0.550 g, 1.378 mmol) and compound 52 (0.698 g, 1.148 mmol) were co-

evaporated with dry toluene under reduced pressure three times and further dried under high vacuum for overnight. A solution of compound 44 and 52 in anhydrous ether (16 mL) and anhydrous dichloromethane (4 mL) was added to a preactivated 4 Å molecular sieves and N-iodosuccinimide (0.5167 g, 2.297 mmol). After stirring for 30 min at room temperature, the reaction mixture was cooled to -48 °C. Silver trifluoromethanesulfonate (0.590 g, 2.297 mmol) was added, and the reaction was warmed to -10 °C to stand for 3 h. The reaction mixture was diluted with ethyl acetate (20 mL) and filtered through a Celite pad. The solution was washed with 10% aqueous solution of sodium bisulfite (50 mL) and a saturated aqueous solution of sodium bicarbonate (50 mL). The combined aqueous layers were extracted with ethyl acetate (50 mL \times 3). The combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 1.2: 1) to provide compound 53 (0.567g, 56%) as a clear liquid. ¹H NMR (600 MHz, CDCl₃) δ 7.41–7.26 (m, 15H, Ph), 5.62 (d, J = 3.8 Hz, 1H, H-1'), 5.02 (d, J = 3.2 Hz, 1H, H-1"), 4.94–4.84 (m, 4H, Bn), 4.62 (dd, J = 13.2, 11.4 Hz, 2H, Bn), 4.23-4.15 (m, 2H, H-5', H-5"), 4.00 (dd, J = 10.3, 8.9 Hz, 1H, H-3'), 3.95 (dd, J = 10.3, 10.3 Hz, 1H, H-3'), 3.95 (dd, J = 10.3, 10.3 Hz, 1H, H-3'), 3.95 (dd, J = 10.3 Hz, 10.34.3, 3.1 Hz, 1H, H-2"), 3.80 (d, J = 12.4 Hz, 1H, H-5"), 3.68 (td, J = 8.9, 1.5 Hz, 1H, H-5), 3.57 (dd, *J* = 10.0, 8.8 Hz, 1H, H-4'), 3.53 (dd, *J* = 13.2, 2.5 Hz, 1H, H-6'), 3.51–3.44 (m, 3H, H-1, H-4, H-3"), 3.46–3.35 (m, 4H, H-2', H-6, H-6', H-3), 2.83 (s, 3H, NMe), 2.33 (dt, J = 13.3, 4.4 Hz, 1H, H-2), 1.57–1.45 (m, 1H, H-2), 1.41 (s, 3H,Me). ¹³C NMR (150 MHz, CDCl₃) δ 157.27, 137.85, 137.83, 137.42, 128.75, 128.66, 128.63, 128.42, 128.16, 128.13, 128.07, 128.04, 127.95, 98.29, 97.12, 84.86, 80.32, 80.14, 78.79, 75.62, 75.31, 75.20, 74.69, 73.35, 71.21, 67.00, 63.67, 63.10, 59.80, 58.85, 51.15, 32.16, 30.32, 23.67. HRMS (ESI, positive) calculated for $C_{41}H_{47}N_{13}O_{10}^+ [M + Na]^+ 904.3461$, found 904.2459.

(3R,4R,5R,6R)-5-(benzyloxy)-6-(((1S,2S,3R,4S,6R)-4,6-diazido-3-(((2R,3R,4R,5R,6R)-3-azido-6-(azidomethyl)-4,5-bis(benzyloxy)tetrahydro-2H-pyran-2-yl)oxy)-2-hydroxycyclohexyl)oxy)-3-methyl-4-(methylamino)tetrahydro-2H-pyran-3-ol (54): A 2 N aqueous sodium hydroxide solution (4 mL) was added to a solution of compound 53 (0.4385 g, 0.497 mmol) in ethanol (12 mL). The reaction mixture was stirred at 90 °C for 12 h, and cooled to room temperature. The solution was diluted with ethyl acetate (20 mL) and washed with water (20 mL). The residue was washed with brine, dried over sodium sulfate, filtered, and concentrated. The crude product was purified by flash column chromatography on silica gel (4% methanol in dichloromethane) to give compound **54** (0.344 g, 81%) as a clear oil. ¹H NMR (400 MHz, CDCl₃) δ 7.41–7.27 (m, 15H), 5.57 (d, J = 3.8 Hz, 1H, H-1'), 5.20 (d, J = 3.4 Hz, 1H, H-1"), 4.91–4.84 (m, 3H, Bn), 4.78 (d, J = 11.7 Hz, 1H, Bn), 4.63 (d, J = 11.2 Hz, 1H, Bn), 4.52 (d, J = 11.7 Hz, 1H, Bn), 4.45 (d, J = 1.9 Hz, 1H, OH), 4.20 (ddd, J = 10.0, 4.4, 2.5 Hz, 1H, H-5'), 3.98 (dd, J = 10.3, 8.8 Hz, 1H, H-3'), 3.89 (d, J = 12.2 Hz, 1H, H-5''), 3.73-3.64 (m, 1H, H-5),3.62 (m, 1H, H-2"), 3.59–3.46 (m, 3H, H-4', H-5", H-6'), 3.46–3.32 (m, 4H, H-2', H-6', H-4, H-3), 3.30 (t, J = 9.4 Hz, 1H, H-6), 2.69 (d, J = 10.3 Hz, 1H, H-3"), 2.54 (s, 3H, NMe), 2.38–2.29 (m, 1H, H-2), 1.52 (q, J = 12.7 Hz, 1H, H-2), 1.19 (s, 3H, Me). ¹³C NMR (125 MHz, CDCl₃) δ 137.85, 137.84, 137.78, 128.82, 128.69, 128.65, 128.38, 128.28, 128.18, 128.16, 128.11, 127.97, 98.54, 97.40, 84.79, 80.65, 80.24, 78.81, 76.13, 75.85, 75.67, 75.35, 72.65, 71.23, 71.04, 66.92, 63.76, 63.47, 59.71, 59.01, 51.20, 38.63, 32.37, 25.25. HRMS (ESI, positive) calculated for $C_{40}H_{49}N_{13}O_9^+ [M + Na]^+ 878.3668$, found 878.3649. JI-20A (55): Trimethylphosphine (1 M solution in THF, 1.08 mL, 1.08 mmol) was added to a solution of compound 54 (0.231 g, 0.270 mmol) in THF (9 mL) and 0.1 M NaOH (3 mL). The reaction mixture was stirred at 50 °C for 2 h, cooled to room temperature, and concentrated under reduced pressure. The residue was dissolved in a mixture of acetic

acid (2 mL), water (6 mL) and methanol (6 mL). The solution was degassed by evacuating the air inside of the flask and refilling it with nitrogen five times. Palladium hydroxide on carbon (20%, Degussa type, 0.1 g) was added and the solution was charged with hydrogen gas. The reaction mixture was stirred at room temperature under hydrogen gas for 24 h. The solution was filtered through a Celite pad, washed with water and methanol, and concentrated under reduced pressure. The residue was purified by chromatography on Amberlite CG-50 resin (NH_4^+ form) (5-10% concentrated ammonium) hydroxide in water), concentrated under reduced pressure, acidified with hydrochloric acid (pH ca. 4), and lyophilized to provide compound 55 (0.110 g, 85%) as an oily solid. ¹H NMR (500 MHz, D_2O) δ 5.99 (d, J = 3.9 Hz, 1H, H-1'), 5.14 (d, J = 3.7 Hz, 1H, H-1"), 4.25 (dd, J = 10.9, 3.7 Hz, 1H, H-2"), 4.10–4.02 (m, 4H, H-4, H-5', H-3', H-5"), 3.93 (t, J = 9.1 Hz, 1H, H-5), 3.88-3.81 (m, 1H, H-6), 3.72-3.59 (m, 2H, H-1, H-3), 3.59-3.47 (m, 1H, H-6), 3.72-3.59 (m, 2H, H-1, H-3), 3.59-3.47 (m, 1H, H-6), 3.72-3.59 (m, 2H, H-1, H-3), 3.59-3.47 (m, 1H, H-6), 3.72-3.59 (m, 2H, H-1, H-3), 3.59-3.47 (m, 1H, H-6), 3.72-3.59 (m, 2H, H-1, H-3), 3.59-3.47 (m, 1H, H-6), 3.72-3.59 (m, 2H, H-1, H-3), 3.59-3.47 (m, 1H, H-6), 3.72-3.59 (m, 2H, H-1, H-3), 3.59-3.47 (m, 1H, H-6), 3.72-3.59 (m, 2H, H-1, H-3), 3.59-3.47 (m, 1H, H-6), 3.72-3.59 (m, 2H, H-1, H-3), 3.59-3.47 (m, 1H, H-6), 3.72-3.59 (m, 2H, H-1, H-3), 3.59-3.47 (m, 1H, H-6), 3.72-3.59 (m, 2H, H-1, H-3), 3.59-3.47 (m, 1H, H-6), 3.72-3.59 (m, 2H, H-1, H-3), 3.59-3.47 (m, 1H, H-6), 3.72-3.59 (m, 2H, H-1, H-3), 3.59-3.47 (m, 1H, H-6), 3.72-3.59 (m, 2H, H-1, H-3), 3.59-3.47 (m, 1H, H-6), 3.72-3.59 (m, 2H, H-1, H-3), 3.59-3.47 (m, 1H, H-6), 3.72-3.59 (m, 2H, H-1, H-3), 3.59-3.47 (m, 1H, H-6), 3.72-3.59 (m, 2H, H-1, H-3), 3.59-3.47 (m, 1H, H-6), 3.72-3.59 (m, 2H, H-1, H-3), 3.59-3.47 (m, 1H, H-6), 3.72-3.59 (m, 2H, H-1, H-3), 3.59-3.47 (m, 1H-1, H-2), 3.59-3.47 (m, 1H-1, H-3), 3.59-3.47 (m, 1H-3), 3.59-3.47 (m, 1H-3), 3.59(m, 5H, H-3", H-4', H-5", H-2', H-6'), 3.32 (dd, J = 13.6, 6.9 Hz, 1H, H-6'), 2.95 (s, 3H, NMe), 2.57 (dt, J = 12.6, 4.3 Hz, 1H, H-2), 2.07–1.90 (m, 1H, H-2), 1.38 (s, 3H, Me). ¹³C NMR (125 MHz, D₂O) δ 101.12, 95.90, 83.63, 77.54, 74.22, 70.60, 69.85, 69.13, 68.19, 67.85, 66.17, 63.17, 53.43, 49.57, 48.20, 40.07, 34.47, 28.00, 20.90. HRMS (ESI, positive) calculated for $C_{19}H_{39}N_5O_9^+$ $[M + Na]^+$ 504.2640, found 504.2639.

2.2.18. Synthesis of 2'-Deamino-2'-hydroxy-gentamicin X₂

The overall synthetic scheme is shown in Scheme 2-16.



Scheme 2.16. Synthetic scheme for the preparation of 2'-deamino-2'-hydroxy-GenX₂.

(*3aR,6R,7R,7aR*)-7-(benzyloxy)-6-(((*1S,2R,3R,4S,6R*)-4,6-diazido-2-hydroxy-3-(((*2S,-3R,4S,5R,6R*)-3,4,5-tris(benzyloxy)-6-((benzyloxy)methyl)tetrahydro-2*H*-pyran-2-yl-)oxy)cyclohexyl)oxy)-1,3a-dimethylhexahydro-2*H*-pyrano[4,3-d]oxazol-2-one (57): Compound 56 was prepared as previously described.¹⁵⁷ Compound 44 (0.4478 g, 1.121 mmol) and compound 56 (0.688 g, 0.934 mmol) were co-evaporated with dry toluene under reduced pressure three times and further dried under high vacuum overnight. A solution of compound 44 and 56 in anhydrous ether (12 mL) and anhydrous dichloromethane (3 mL) was added to preactivated 4 Å molecular sieves and *N*-iodosuccinimide (0.4204 g, 1.868 mmol). After stirring for 30 min at room temperature, the reaction mixture was cooled to -40 °C, silver trifluoromethanesulfonate (0.480 g, 1.868 mmol) was added, and the reaction was warmed to -10 °C to stand for 3 h. The reaction mixture was diluted with ethyl acetate (20 mL) and filtered through a Celite pad. The solution was washed with a 10% aqueous solution of sodium bisulfite (50 mL) and a
saturated aqueous solution of sodium bicarbonate solution (50 mL). The combined aqueous layers were extracted with ethyl acetate (50 mL \times 3), and the combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 1.5 : 1) to provide compound 57 (0.576g, 61%) as a clear liquid. ¹H NMR (500 MHz, CDCl₃) δ 7.38–7.26 (m, 23H, Ph), 7.14–7.12 (m, 2H, Ph), 5.31 (d, J = 3.6 Hz, 1H, H-1'), 5.24 (d, J = 3.7 Hz, 1H, H-1"), 4.95 (d, J = 10.9 Hz, 1H, Bn), 4.91–4.85 (m, 2H, 2×Bn), 4.82 (d, J = 11.1 Hz, 2H, 2×Bn), 4.76 (d, J = 11.4 Hz, 1H, Bn), 4.65 (d, J = 12.1 Hz, 1H, Bn), 4.57 (d, J = 11.3 Hz, 1H, Bn), 4.49 (dd, J = 22.7, 10.99.4 Hz, 1H, H-3'), 3.83-3.80 (m, 2H, H-6', H-2"), 3.74 (dd, J = 10.1, 9.0 Hz, 1H, H-4'), $3.69 \text{ (dd, } J = 10.6, 1.9 \text{ Hz}, 1\text{H}, \text{H-6'}, 3.67-3.56 \text{ (m, 3H, H-2', H-5, H-5'')}, 3.52-3.40 \text{ (m, 3H, H-5'')}, 3.52-3.40 \text{ (m,$ 3H, H-1, H-6, H-3"), 3.35-3.29 (m, 2H, H-4, H-3), 2.87 (s, 3H, NMe), 2.34 (d, J = 14.5Hz, 1H, H-2), 1.58–1.51 (m, 1H, H-2), 1.26 (s, 3H, Me). ¹³C NMR (125 MHz, CDCl₃) δ 157.50 (N(C=O)O), 138.69 (Ph), 138.28 (Ph), 138.04 (Ph), 137.50 (Ph), 137.41 (Ph), 128.71 (Ph), 128.63 (Ph), 128.55 (Ph), 128.49 (Ph), 128.32 (Ph), 128.29 (Ph), 128.23 (Ph), 128.11 (Ph), 127.99 (Ph), 127.97 (Ph), 127.84 (Ph), 127.82 (Ph), 127.81 (Ph), 127.77 (Ph), 100.00 (C-1'), 95.75 (C-1"), 83.63 (C-4), 82.20 (C-3'), 81.61 (C-6), 80.04 (C-2'), 77.88 (C-4'), 77.83 (C-4''), 75.72 (Bn), 75.43 (C-2''), 75.20 (Bn), 74.69 (C-5), 74.24 (Bn), 73.69 (Bn), 73.63 (Bn), 71.39 (C-5'), 68.36 (C-6'), 65.07 (C-5"), 62.76 (C-3"), 60.22 (C-1), 58.91 (C-3), 32.19 (C-2), 30.70 (NMe), 23.31 (Me). HRMS (ESI, positive) calculated for $C_{55}H_{61}N_7O_{12}^+$ $[M + Na]^+$ 1034.4270, found 1034.4252.

(3R,4R,5R,6R)-5-(benzyloxy)-6-(((1S,2R,3R,4S,6R)-4,6-diazido-2-hydroxy-3-(((2S,3-R,4S,5R,6R)-3,4,5-tris(benzyloxy)-6-((benzyloxy)methyl)tetrahydro-2H-pyran-2-yl)-oxy)cyclohexyl)oxy)-3-methyl-4-(methylamino)tetrahydro-2H-pyran-3-ol (58): A

solution of 2 N aqueous sodium hydroxide (5 mL) was added to a solution of compound 57 (0.446 g, 0.441 mmol) in ethanol (15 mL). The reaction mixture was stirred at 90 °C for 10 h, and cooled to room temperature. The solution was diluted with ethyl acetate (25 mL) and washed with water (25 mL). The residue was washed with brine, dried over sodium sulfate, filtered, and concentrated. The crude product was purified by flash column chromatography on silica gel (4% methanol in dichloromethane) to give compound **58** (0.369 g, 85%) as a clear oil. ¹H NMR (600 MHz, CDCl₃) δ 7.38–7.24 (m, 23H, Ph), 7.13–7.12(m, 2H, Ph), 5.33 (d, J = 3.5 Hz, 1H, H-1"), 5.27 (d, J = 3.7 Hz, 1H, H-1'), 4.94 (d, J = 11.0 Hz, 1H, Bn), 4.88 (d, J = 11.0 Hz, 1H, Bn), 4.85–4.79 (m, 3H, Bn), 4.75 (d, J = 11.3 Hz, 1H, Bn), 4.65 (d, J = 12.2 Hz, 1H, Bn), 4.53 (d, J = 1.7 Hz, 1H, OH), 4.50 (dd, J = 13.3, 11.4 Hz, 2H, Bn), 4.45 (d, J = 11.9 Hz, 1H, Bn), 4.10 (dt, Hz, Hz), 4.10 (dt, Hz, Hz), 4.10 (dt, Hz), 4.1 10.3, 2.5 Hz, 1H, H-5'), 4.02 (dd, J = 9.8, 9.1 Hz, 1H, H-3'), 3.94 (d, J = 12.4 Hz, 1H, H-5"), 3.81 (dd, J = 10.7, 3.1 Hz, 1H, H-6'), 3.74 (dd, J = 10.2, 9.0 Hz, 1H, H-4'), 3.69 (dd, J = 10.7, 2.1 Hz, 1H, H-6'), 3.64 (dd, J = 9.8, 3.7 Hz, 1H, H-2'), 3.62–3.57 (m, 1H, H-5), 3.56–3.47 (m, 2H, H-1, H-2"), 3.39 (t, J = 9.5 Hz, 1H, H-4), 3.31–3.27 (m, 3H, H-6, H-3, H-5"), 2.59 (d, J = 10.4 Hz, 1H, H-3"), 2.49 (s, 3H, NMe), 2.36 (dt, J = 12.9, 4.3 Hz, 1H, H-2), 1.56 (q, J = 12.6 Hz, 1H, H-2), 1.05 (s, 3H, Me). ¹³C NMR (150 MHz, CDCl₃) δ 137.89, 128.79, 128.65, 128.58, 128.57, 128.52, 128.33, 128.30, 128.17, 128.14, 128.00, 127.86, 127.79, 100.26, 96.70, 83.98, 82.29, 81.14, 80.23, 77.90, 77.32, 76.13, 75.75, 75.24, 75.01, 74.43, 73.73, 72.23, 71.42, 71.15, 68.41, 66.15, 63.10, 60.42, 58.99, 32.37, 24.77. HRMS (ESI, positive) calculated for $C_{54}H_{63}N_7O_{11}^+$ $[M + H]^+$ 986.4658, found 986.4660.

2'-deamino-2'-hydroxy-gentamicin X₂ (**59**): Trimethylphosphine (1 M solution in THF, 0.66 mL, 0.66 mmol) was added to a solution of compound **58** (0.164 g, 0.166 mmol) in THF (6 mL) and 0.1 M NaOH (2 mL). The reaction mixture was stirred at 50 °C for 2 h,

cooled to room temperature, and concentrated under reduced pressure. The residue was dissolved in a mixture of acetic acid (1 mL), water (3 mL) and methanol (3 mL). The solution was deaerated by evacuating the air inside of the flask and refilling it with nitrogen five times. Palladium hydroxide on carbon (20%, Degussa type, 0.05 g) was added and the solution was charged with hydrogen gas. The reaction mixture was stirred at room temperature under hydrogen gas for 24 h. The solution was filtered through a Celite pad, washed with water and methanol, and concentrated under reduced pressure. The residue was purified by chromatography on Amberlite CG-50 resin (NH_4^+ form) (1-5% concentrated ammonium hydroxide in water), concentrated under reduced pressure, acidified with hydrochloric acid (pH ca. 4), and lyophilized to provide compound 59 (0.067 g, 84%) as an oily solid. ¹H NMR (500 MHz, D₂O) δ 5.41 (d, J = 4.0 Hz, 1H, H-1'), 5.10 (d, J = 3.8 Hz, 1H, H-1"), 4.22 (dd, J = 10.9, 3.7 Hz, 1H, H-2"), 4.02 (d, J = 12.9 Hz, 1H, H-5"), 3.96–3.88 (m, 2H, H-6', H-5), 3.88–3.80 (m, 2H, H-5', H-6), 3.80–3.73 (m, 2H, H-3', H-4), 3.73–3.66 (m, 2H, H-2', H-6'), 3.66–3.57 (m, 2H, H-1, H-3), 3.54 (d, J = 10.8 Hz, 1H, H-3"), 3.49 (d, J = 12.9 Hz, 1H, H-5"), 3.39 (dd, J = 10.1, 9.0 Hz, 1H, H-4'), 2.93 (s, 3H, NMe), 2.55 (dt, J = 12.6, 4.3 Hz, 1H, H-2), 1.93 (q, J = 12.6 Hz, 1H, H-2), 1.35 (s, 3H, Me). ¹³C NMR (125 MHz, D₂O) δ 100.99, 98.48, 83.06, 80.71, 73.06, 72.90, 72.50, 71.09, 69.88, 69.47, 67.60, 66.30, 63.30, 60.64, 49.71, 48.30, 34.46, 27.62, 20.87. HRMS (ESI, positive) calculated for $C_{19}H_{37}N_3O_{11}^+ [M + Na]^+ 506.2320$, found 506.2328.

2.3. RESULTS AND DISCUSSION

2.3.1. Gene Analysis

GenK is assigned to be a Cbl-dependent radical SAM enzyme based on gene analysis of *Micromonospora echinospora*.⁷⁷ A sequence analysis and comparison with other putative Cbl-dependent radical SAM *sp*³-carbon methyltransferases is shown in Figure 2-1. The absence of a conserved histidine residue within the Cbl-binding domain indicates that the corrin cobalt is probably coordinated to dimethylbenzimidazole. This is in contrast to methionine synthase, methyl malonyl-CoA mutase, glutamate mutase and others in which histidine in a conserved DXHX₂G motif serves as the lower axial ligand for the corrinoid cobalt.¹²¹ The sequence alignment also suggests that Cbl-dependent radical SAM *sp*³ *C*-methyltransferases harbor a single [4Fe-4S] cluster per monomer as only the cysteine residues comprising the CX₃CX₂C motif (highlighted in grey) are universally conserved in this class of enzymes.

	117	CMFTPYYESAYELARMAKRVLPNAKVIVGGQHGTVAFP-HVLEVPEVDAVMLGEAEVTTV
Fms7	110	CMFTPYYEPAYELGRLAKQILPQARVILGGQHPTVAHP-HALAEEAFDALVLGEAEANVV
Fom3	124	SIFSNQADNVHHLLKLADLVAPEAVTSIGGAHARYFPK-ACLDDPNLDAVFLGEGEMTFL
ThnK	73	CYV-WNFRROMKVARLVKERHPGMLVVAGGPHVPDRPGDFFARHPYVDVLVHGEGETAFR
ThnL	68	IPYTTSVRVSRDVTHOARRLWPGTPIVLGGHHPTVSAEWLTGFAADWIVAGEGGGPLA
ThnP	78	VLT-SSLKNGIKLASEVRHRHPNALTVLGGVGASPIAR-KLIEENAADVVVRGEGEYSFS
P+mT.	96	TMYDLHTVDAVRLLRCVRKADPSVFVATGGAFCTYNAKLTAERTPEADCVAFGEGELTVE
P+mM	96	TMYDLHTVDAVRLLRCVRAADDSVFVATGGAFCTYNGKLTAERTPEADCVAFGEGELTVE
P+mN	102	VI.G-WNFRAFGTI.AETFKOUNPDGWUTFGGNHUAHOAERVFRMFPOUDVUUNGEGELUFR
CndT	137	TTFTVCEPWLRALCHVTREVLPSTKTTMCGYYYAVNVKKFLALDADTFCVCECEORLP
Swb9	78	VI.G-WNIREFGALAETFKOLNPRGLVVFCCTHVANOAERTFRMFDDVDVIVNGEGDLVLP
BchO	52	SRT-TEATRAYETADEFRKRGKTVVI.GGI.HTSFNPEEAAAHADCIVVGEADNI.WT
Doing	51	• ** * **
GenK	176	ALLDAFATGRPLTELLGVAFRCGEGLCECATPGTPHIRPRAPEVA
Fms7	159	EIVEALAAGRSLRGMPGL/TFRCGTGLCDCPRPSGVHIOPRAEFLO
Fom 3	173	LWLEHLNGNVREDEVHGTAWRDRDCKTOTKPELPLTSSMRPEGPEOGKSSPMLSMAG
ThnK	122	FILITERIADHDDVTRVDCVSVRHCTFA_VDCRDAFRID
Thn.	116	
ThnD	126	
D+mT	1/6	
P CIIIL D+mM	140	
P CIIIM D±mN	1 5 1	
CodT	105	
Crubo	100	
SwD9	127	
BGUÕ	96	TLLDDVANNRLKERYDSKDFPPVKAIT
		•
Conv	221	
Genk Emg 7	104	
rms/	194	DEDGLALPAVDELDMGSIDET-ATEITSRGUPTSUSFUTV
	220	ELDUTDEDAWUUV NWEWVEETVAVOCDVEVOCDV COLVECDOEDUU
Fom3	220	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT
Fom3 ThnK	220 158	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT RRIETPSPYLLGVMDGAVATCRQRDLRFYALWETNRGCPYSCAFCD-
Fom3 ThnK ThnL	220 158 142	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT RRIETPSPYLLGVMDGAVATCRQRDLRFYALWETNRGCPYSCAFCD- ALDDLPMPDRTRLAHRGRYFHSIYRPV-ALINFTAGCPYTCKFCSL
Fom3 ThnK ThnL ThnP	220 158 142 155	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT RRIETPSPYLLGVMDGAVATCRQRDLRFYALWETNRGCPYSCAFCD- ALDDLPMPDRTRLAHHRGRYFHSIYRPV-ALIRFTAGCPYTCKFCSL NLDKLPKPARDLADLDLYRRISRGRS-GNLVTSRGCSYACAYCYS
Fom3 ThnK ThnL ThnP PtmL	220 158 142 155 171	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT RRIETPSPYLLGVMDGAVATCRQRDLRFYALWETNRGCPYSCAFCD- ALDDLPMPDRTRLAHHRGRYFHSIYRPV-ALIRFTAGCPYTCKFCSL NLDKLPKPARDLADLDLYRRISRGRS-GNLVTSRGCSYACAYCYS DLHKQAWPARDLLVHHRGAGIPTPV-ASTYTSRGCHAKCTFCYV
Fom3 ThnK ThnL ThnP PtmL PtmM	220 158 142 155 171 171	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT RRIETPSPYLLGVMDGAVATCRQRDLRFYALWETNRGCPYSCAFCD- ALDDLPMPDRTRLAHHRGRYFHSIYRPV-ALIRFTAGCPYTCKFCSL NLDKLPKPARDLADLDLYRRISRGRS-GNLVTSRGCSYACAYCYS DLHKQAWPARDLLVHHRGAGIPTPV-ASTYTSRGCHAKCTFCYA DLSKQAWPARDVLIHHREAGIPTPR-ASTYTSRGCHAKCTFCYA
Fom3 ThnK ThnL ThnP PtmL PtmM PtmM	220 158 142 155 171 171 171	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT RRIETPSPYLLGVMDGAVATCRQRDLRFYALWETNRGCPYSCAFCD- ALDDLPMPDRTRLAHHRGRYFHSIYRPV-ALIRFTAGCPYTCKFCSL NLDKLPKPARDLADLDLYRRISRGRS-GNLVTSRGCSYACAYCYS DLHKQAWPARDLLVHHRGAGIPTPV-ASTYTSRGCHAKCTFCYV DLSKQAWPARDVLIHHREAGIPTPR-ASTYTSRGCHAKCTFCYA DLEILPSPILTGAIPLA-DSQGRFLYDYAIMETNRGCPYKCAFCY-
Fom3 ThnK ThnL ThnP PtmL PtmM PtmM CndI	220 158 142 155 171 171 171 211	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT RRIETPSPYLLGVMDGAVATCRQRDLRFYALWETNRGCPYSCAFCD- ALDDLPMPDRTRLAHHRGRYFHSIYRPV-ALIRFTAGCPYTCKFCSL NLDKLPKPARDLADLDLYRRISRGRS-GNLVTSRGCSYACAYCYS DLHKQAWPARDLLVHHRGAGIPTPV-ASTYTSRGCHAKCTFCYV DLSKQAWPARDVLIHHREAGIPTPR-ASTYTSRGCHAKCTFCYA DLEILPSPILTGAIPLA-DSQGRFLYDYAIMETNRGCPYKCAFCY- DMNELPIVDWSLSTRVE-PPIDPIATPVATWVETQRGCVFSCEFCDY
Fom3 ThnK ThnL ThnP PtmL PtmM PtmN CndI Swb9	220 158 142 155 171 171 171 211 157	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT RRIETPSPYLLGVMDGAVATCRQRDLRFYALWETNRGCPYSCAFCD- ALDDLPMPDRTRLAHHRGRYFHSIYRPV-ALIRFTAGCPYTCKFCSL NLDKLPKPARDLADLDLYRRISRGRS-GNLVTSRGCYACAYCYS DLHKQAWPARDLLVHHRGAGIPTPV-ASTYTSRGCHAKCTFCYV DLSKQAWPARDVLIHHREAGIPTPR-ASTYTSRGCHAKCTFCYA DLEILPSPILTGAIPLA-DSQGRFLYDYAIMETNRGCPYKCAFCY- DMNELPIVDWSLSTRVE-PPIDPIATPVATWVETQRGCVFSCEFCDY DLDVIPSPFLTGAIPLLDDNDRFRYDVALMETNRGCPYKCSFCY-
Fom3 ThnK ThnL ThnP PtmL PtmM PtmN CndI Swb9 BchQ	220 158 142 155 171 171 171 211 157 123	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT RRIETPSPYLLGVMDGAVATCRQRDLRFYALWETNRGCPYSCAFCD- ALDDLPMPDRTRLAHHRGRYFHSIYRPV-ALIRFTAGCPYTCKFCSL NLDKLPKPARDLADLDLYRRISRGRS-GNLVTSRGCYACAYCYS DLHKQAWPARDLLVHHRGAGIPTPV-ASTYTSRGCHAKCTFCYV DLSKQAWPARDVLIHHREAGIPTPR-ASTYTSRGCHAKCTFCYA DLEILPSPILTGAIPLA-DSQGRFLYDYAIMETNRGCPYKCAFCY- DMNELPIVDWSLSTRVE-PPIDPIATPVATWVETQRGCVFSCEFCDY DLDVIPSPFLTGAIPLL-DDNDRFRYDVALMETNRGCPYKCSFCY- PLDYARIAKASKRTKVDGTKSIPIYVTRGCPFNCSFCVT
Fom3 ThnK ThnL ThnP PtmL PtmM PtmN CndI Swb9 BchQ	220 158 142 155 171 171 171 211 157 123	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT RRIETPSPYLLGVMDGAVATCRQRDLRFYALWETNRGCPYSCAFCD- ALDDLPMPDRTRLAHHRGRYFHSIYRPV-ALINFTAGCPYTCKFCSL NLDKLPKPARDLADLDLYRRISPTPV-ASTYTSRGCHAKCTFCYV DLSKQAWPARDLLVHHRGAGIPTPV-ASTYTSRGCHAKCTFCYA DLEILPSPILTGAIPLA-DSQGRFLYDYAIMETNRGCPYKCAFCY- DMNELPIVDWSLSTRVE-PPIDPIATPVATWVETQRGCVFSCEFCDY DLDVIPSPFLTGAIPLL-DDNDRFRYDVALMETNRGCPYKCSFCY- PLDYARIAKASKRTKVDGTKSIPIYVTRGCPFNCSFCVT ** * :*
Fom3 ThnK ThnL ThnP PtmL PtmM PtmM CndI Swb9 BchQ	220 158 142 155 171 171 171 211 157 123	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT RRIETPSPYLLGVMDGAVATCRQRDLRFYALWETNRGCPYSCAFCD- ALDDLPMPDRTRLAHHRGRYFHSIYRPV-ALIKFTAGCPYTCKFCSL NLDKLPKPARDLADLDLYRRISPTPV-ASTYTSRGCHAKCTFCYV DLSKQAWPARDLLVHHRGAGIPTPV-ASTYTSRGCHAKCTFCYV DLSKQAWPARDVLIHHREAGIPTPR-ASTYTSRGCHAKCTFCYA DLEILPSPILTGAIPLA-DSQGFLYDYAIMETNRGCPYKCAFCY- DMNELPIVDWSLSTRVE-PPIDPIATPVATWVETQRGCVFSCEFCDY DLDVIPSPFLTGAIPLLDDNDRFRYDVALMETNRGCPYKCSFCY- PLDYARIAKASKRTKVDGTKSIPIYVTRGCPFNCSFCVT ** * :*
Fom3 ThnK ThnL ThnP PtmL PtmM PtmN CndI Swb9 BchQ GenK	220 158 142 155 171 171 171 171 157 123 259	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT RRIETPSPYLLGVMDGAVATCRQRDLRFYALWETNRGCPYSCAFCD- ALDDLPMPDRTRLAHHRGRYFHSIYRPV-ALIRFTAGCPYTCKFCSL NLDKLPKPARDLADLDLYRRISRGRS-GNLVTSRGCYACAYCYS DLHKQAWPARDLLVHHRGAGIPTPV-ASTYTSRGCHAKCTFCYV DLSKQAWPARDVLIHHREAGIPTPR-ASTYTSRGCHAKCTFCYA DLEILPSPILTGAIPLADSQGRFLYDYAIMETNRGCPYKCAFCY- DMNELPIVDWSLSTRVE-PPIDPIATVATWVETQRGCVFSCEFCDY DLDVIPSPFLTGAIPLLDDNDRFRYDVALMETNRGCPYKCSFCY- PLDYARIAKASKRTKVDGTKSIPIVVTRGCPFNCSFCVT ** * :*
Fom3 ThnK ThnL ThnP PtmL PtmM PtmN CndI Swb9 BchQ GenK Fms7	220 158 142 155 171 171 171 171 157 123 259 223	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT RRIETPSPYLLGVMDGAVATCRQRDLRFYALWETNRGCPYSCAFCD- ALDDLPMPDRTRLAHHRGRYFHSIYRPV-ALIRFTAGCPYTCKFCSL NLDKLPKPARDLADLDLYRRISRGRS-GNLVTSRGCYACAYCYS DLHKQAWPARDLLVHHRGAGIPTPV-ASTYTSRGCHAKCTFCYV DLSKQAWPARDVLIHHREAGIPTPR-ASTYTSRGCHAKCTFCYA DLEILPSPILTGAIPLA-DSQGRFLYDYAIMETNRGCPYKCAFCY- DMNELPIVDWSLSTRVE-PPIDPIAPVATWVETQRGCVFSCEFCDY DLDVIPSPFLTGAIPLL-DDNDRFRYDVALMETNRGCPYKCSFCY- PLDYARIAKASKRTKVDGTKSIPIVVTRGCPNCSFCVT ** *:*
Fom3 ThnK ThnL ThnP PtmL PtmM PtmN CndI Swb9 BchQ GenK Fms7 Fom3	220 158 142 155 171 171 171 211 157 123 259 223 262	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT RRIETPSPYLLGVMDGAVATCRQRDLRFYALWETNRGCPYSCAFCD- ALDDLPMPDRTRLAHHRGRYFHSIYRPV-ALWETNRGCPYSCAFCSL NLDKLPKPARDLAVHLRGAGIPTPV-ASTYTSRGCHAKCTFCYV DLSKQAWPARDVLIHHREAGIPTPV-ASTYTSRGCHAKCTFCYA DLEILPSPILTGAIPLA-DSQGRFLYDYALMETNRGCPYKCAFCY- DMNELPIVDWSLSTRVE-PPIDPIATPVATWVETQRGCVFSCEFCDY DLDVIPSPFLTGAIPLL-DDNDRFRYDVALMETNRGCPYKCSFCY- PLDYARIAKASKRTKVDGTKSIPIVVTRGCPKCSFCY- ** * ::* HATVGKQFRARDPQRVVDEIEHYVNVHGVRFLVEDDNFTFDIER HATVGKKFRARAPENVVDEIEHYVDEYGIDEFHIQDDNITNDMDH
Fom3 ThnK ThnL ThnP PtmL PtmM PtmM CndI Swb9 BchQ GenK Fms7 Fom3 ThnK	220 158 142 155 171 171 171 171 1211 157 123 259 223 262 194	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT RRIETPSPYLLGVMDGAVATCRQRDLRFYALWETNRGCPYSCAFCD- ALDDLPMPDRTRLAHHRGRYFHSIYRPV-ALIRFTAGCPYTCKFCSL NLDKLPKPARDLADLDLYRRISRGRS-GNLVTSRGCYACAYCYS DLHKQAWPARDLLVHHRGAGIPTPV-ASTYTSRGCHAKCTFCYV DLSKQAWPARDVLIHHREAGIPTPR-ASTYTSRGCHAKCTFCYA DLEILPSPILTGAIPLA-DSQGRFLYDYAIMETNRGCPYKCAFCY- DMNELPIVDWSLSTRVE-PPIDPIATPVATWVETQRGCVFSCEFCDY DLDVIPSPFLTGAIPLLDDNDRFRYDVALMETNRGCPYKCAFCY- PLDYARIAKASKRTKVDGTKSIPIVVTRGCPFNCSFCVT ** :: HATVGKQFRARDPQRVVDEIEHYVNVHGVRRFLVEDDNFTFDIER HATVGKKFRARAPENVVDEIEHYVDEYGIDEFHIQDDNITNDMDH WGSATMSALRLFDAERLQEEIEW-FAEHDVEDLFVCDANFGILPR
Fom3 ThnK ThnL ThnP PtmL PtmM PtmN CndI Swb9 BchQ GenK Fms7 Fom3 ThnK ThnL	220 158 142 155 171 171 171 1211 157 123 259 223 262 194 178	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT RRIETPSPYLLGVMDGAVATCRQRDLRFYALWETNRGCPYSCAFCD- ALDDLPMPDRTRLAHHRGRYFHSIYRPV-ALIRFTAGCPYTCKFCSL NLDKLPKPARDLADLDLYRRISRGRS-GNLVTSRGCYACAYCYS DLHKQAWPARDLLVHHRGAGIPTPV-ASTYTSRGCHAKCTFCYV DLSKQAWPARDVLIHHREAGIPTPR-ASTYTSRGCHAKCTFCYA DLEILPSPILTGAIPLA-DSQGRFLYDYAIMETNRGCPYKCAFCY- DMNELPIVDWSLSTRVE-PPIDPIATPVATWVETQRGCVFSCEFCDY DLDVIPSPFLTGAIPLLDDNDRFRYDVALMETNRGCPYKCSFCY- PLDYARIAKASKRTKVDGTKSIPIVVTRGCPFNCSFCVT ** *:* HATVGKQFRARDPQRVVDEIEHYVNVHGVRRFLVEDDNFTFDIER HATVGKKFRARAPENVVDEIEHYVDEYGIDEFHIQDDNITNDMDH WGSATMSALRLFDAERLQEEIEW-FAEHDVEDLFVCDANFGILPR WRMTDRRYLVKDIDRVLAEIADIDGDNLYVVDDEAFIQPVR
Fom3 ThnK ThnL ThnP PtmL PtmM PtmN CndI Swb9 BchQ GenK Fms7 Fom3 ThnK ThnL ThnP	220 158 142 155 171 171 171 171 1211 157 123 259 223 262 194 178 189	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT RRIETPSPYLLGVMDGAVATCRQRDLRFYALWETNRGCPYSCAFCD- ALDDLPMPDRTRLAHHRGRYFHSIYRPV-ALIRFTAGCPYTCKFCSL NLDKLPKPARDLADLDLYRRISRGRS-GNLVTSRGCPYACAYCYS DLHKQAWPARDLLVHHRGAGIPTPV-ASTYTSRGCHAKCTFCYV DLSKQAWPARDVLIHHREAGIYALMETNRGCPYKCAFCY- DMNELPIVDWSLSTRVE-PPIDPIATPVATWVETQRGCVFSCEFCDY DLDVIPSPFLTGAIPLLDDNDRFRYDVALMETNRGCPYKCSFCY- PLDYARIAKASKRTKVDGTKSIPIYVTRGCPFNCSFCVT ** *:* HATVGKKFRARAPENVVDEIEHYVNVHGVRRFLVEDDNFTFDIER HATVGKKFRARAPENVVDEIEHYVTEHGIRRFFIEDDNFTFDIAR THFWGQKLRRSVDNVVNEVLRLRDEYGIDEFHIQDDNITNDMDH WGSATMSALRLFDAERLQEEIEW-FAEHDVEDLFVCDANFGILPRR KHQWGVGQRRHSAARVVDEIRELVEVYGFDRIRIEDDDFVEDVPR
Fom3 ThnK ThnL PtmL PtmM PtmN CndI Swb9 BchQ GenK Fms7 Fom3 ThnK ThnL ThnP PtmL	220 158 142 155 171 171 171 1211 157 123 259 223 262 194 178 189 204	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT RRIETPSPYLLGVMDGAVATCRQRDLRFYALWETNRGCPYSCAFCD- ALDDLPMPDRTRLAHHRGRYFHSIYRPV-ALINFTAGCPYTCKFCSL NLDKLPKPARDLADLDLYRRISRGRS-GNLVTSRGCSYACAYCYS DLHKQAWPARDLLVHHRGAGIPTPV-ASTYTSRGCHAKCTFCYV DLSKQAWPARDVLIHHREAGIYAIMETNRGCPYKCAFCY- DMNELPIVDWSLSTRVE-PPIDYAIMETNRGCPYKCAFCY- DMNELPIVDWSLSTRVE-PPIDVAIMETNRGCPYKCSFCY- PLDYARIAKASKRTKVDGTKSIPIYVTRGCPFNCSFCY- ** *:* HATVGKKFRARAPENVVDEIEHYVNVHGVRRFLVEDDNFTFDIER HATVGKKFRARAPENVVDEIEHYVDEYGIDEFHIQDDNITNDMDH WGSATMSALRLFDAERLQEEIEW-FARIAKASKRTKVDDAFGILPR WRMTDRRYLVKDIDRVLAEIADIDGDNLYVVDDEAFIQPVR KHQWGVGQRRHSAARVVDEIRELVREFGTRFLWFNDDNFGGAFQDGYNH
Fom3 ThnK ThnL ThnP PtmL PtmM PtmN CndI Swb9 BchQ GenK Fms7 Fom3 ThnK ThnL ThnP PtmL PtmM	220 158 142 155 171 171 171 157 123 259 223 262 194 178 189 204 204	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT RRIETPSPYLLGVMDGAVATCRQRDLRFYALWETNRGCPYSCAFCD- ALDDLPMPDRTRLAHHRGRYFHSIYRPV-ALIRFTAGCPYTCKFCSL NLDKLPKPARDLADLDLYRRISRGRS-GNLVTSRGCSYACAYCYS DLHKQAWPARDLLVHHRGAGIPTPV-ASTYTSRGCHAKCTFCYV DLSKQAWPARDVLIHHREAGIYAIMETNRGCPYKCAFCY- DMNELPIVDWSLSTRVE-PPIDYAIMETNRGCPYKCAFCY- DMNELPIVDWSLSTRVE-PPIDVAIMETNRGCPYKCSFCY- PLDYARIAKASKRTKVDGTKSIPIYVTRGCPFNCSFCY- PLDYRIAKASKRTKVDGTKSIPIYVTRGCPFNCSFCV ** * :: HATVGKQFRARDPQRVVDEIEHYVNVHGVRFFLVEDDNFTFDIER HATVGKKFRARAPENVVDEIEHYVDEYGIDEFHIQDDNITNDMDH WGSATMSALRLFDAERLQEEIEW-FDEYGIDEFHIQDDNITNDMDH RHAC
Fom3 ThnK ThnL PtmL PtmM PtmN CndI Swb9 BchQ GenK Fms7 Fom3 ThnK ThnL ThnP PtmL PtmM PtmM	220 158 142 155 171 171 171 157 123 262 194 178 189 204 204 204 205	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT RRIETPSPYLLGVMDGAVATCRQRDLRFYALWETNRGCPYSCAFCD- ALDDLPMPDRTRLAHHRGRYFHSIYRPV-ALWETNRGCPYSCAFCD- NLDKLPKPARDLADLDLYRRISRGRS-GNLVTSRGCSYACAYCYS DLHKQAWPARDLLVHHRGAGIPTPV-ASTYTSRGCHAKCTFCYV DLSKQAWPARDVLIHHREAGIYALMETNRGCPYKCAFCY- DMNELPIVDWSLSTRVE-PPIDYALMETNRGCPYKCAFCY- DMNELPIVDWSLSTRVE-PPIDVALMETNRGCPYKCAFCY- PLDYVALMETNRGCPYKCAFCY- PLDYARIAKASKRTKVDGTKSIPIVVTRGCPYKCSFCY ** * * :* HATVGKQFRARDPQRVVDEIEHYVNVHGVRFLVEDDNFTFDIER HATVGKKFRARAPENVVDEIEHYVTEHGIRRFFIEDDNFTFDIAR THFWGQKLRRSVDNVNNEVLRLRDEYGIDEFHIQDDNITNDMDH WGSATMSALRLFDAERLQEEIEW-FBHDVEDLFVCDANFGILPR WRMTDRYLVKDIDRVLAEIADIDGDNLYVVDDEAFIQPVR PRAPGVTAGNAWRVRSPVDVVDEIEFLQREFGTRFLWFNDDNFGGAFQDGYNH PRQPGVENG-PWRVRPIGDAVDEIEFLQ
Fom3 ThnK ThnL PtmL PtmM PtmN CndI Swb9 BchQ GenK Fms7 Fom3 ThnK ThnL ThnP PtmL PtmM PtmN CndI	220 158 142 155 171 171 171 1211 157 123 262 194 178 189 204 204 204 204 205 247	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT RRIETPSPYLLGVMDGAVATCRQRDLRFYALWETNRGCPYSCAFCD- ALDDLPMPDRTRLAHHRGRYFHSIYRPV-ALIKFTAGCPYTCKFCSL NLDKLPKPARDLAVHRGAGIRGRS-GNLVTSRGCHAKCTFCYV DLSKQAWPARDVLVHHRGAGIPTPV-ASTYTSRGCHAKCTFCYV DLSKQAWPARDVLIHHREAGIPTPR-ASTYTSRGCHAKCTFCYV DLSKQAWPARDVLIHHREAGIYALMETNRGCPYKCAFCY- DMNELPIVDWSLSTRVE-PPIDPIATPVATWVETQRGCVFSCEFCDY DLDVIPSPFLTGAIPLLDDNDRFRYDVALMETNRGCPYKCAFCY- PLDYARIAKASKRTKVDGTKSIPIVVTRGCPFNCSFCVT ** ::: HATVGKQFRARDPQRVVDEIEHYVNVHGVRRFLVEDDNFTFDIER HATVGKKFRARAPENVVDEIEHYVTEHGIRRFFIEDDNFTFDIAR THFWGQKLRRSVDNVNNEVLRLRDEYGIDEFHIQDDNITNDMDH WGSATMSALRLFDAERLQEEIEW-FAEHDVEDLFVCDANFGILPR WRMTGQRRHSAARVVDEIRELVREFGTRFLWFNDDNFGGAFQDGYNH PRQPGVENG-PWRVRFIGDAVDEIFLQREFGTRFLWFNDDNFGGAFQDGYNH PRQPGVENG-PWRVRFIGDAVDEIFLQ
Fom3 ThnK ThnL ThnP PtmL PtmM CndI Swb9 BchQ GenK Fms7 Fom3 ThnK ThnL ThnP PtmL PtmM PtmN CndI Swb9	220 158 142 155 171 171 171 121 121 123 259 223 262 194 178 189 204 204 204 204 204 204 191	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT RRIETPSPYLLGVMDGAVATCRQRDLRFYALWETNRGCPYSCAFCD- ALDDLPMPDRTRLAHHRGRYFHSIYRPV-ALIRFTAGCPYTCKFCSL NLDKLPKPARDLADLDLYRRISRGRS-GNLVTSRGCYACAYCYS DLHKQAWPARDLLVHHRGAGIPTPV-ASTYTSRGCHAKCTFCYV DLSKQAWPARDVLIHHREAGIPTPR-ASTYTSRGCHAKCTFCYA DLEILPSPILTGAIPLA-DSQGRFLYDYALMETNRGCPYKCAFCY- DMNELPIVDWSLSTRVE-PPIDPIATPVATWVETQRGCVFSCEFCDY DLDVIPSPFLTGAIPLLDDNDRFRYDVALMETNRGCPYKCAFCY- PLDYARIAKASKRTKVDGTKSIPIVVTRGCPFNCSFCVT ** ** HATVGKKFRARAPENVVDEIEHYVNVHGVRRFLVEDDNFTFDIER HATVGKKFRARAPENVVDEIEHYVDEYGIDEFHIQDDNITNDMDH WGSATMSALRLFDAERLQEEIEW-FAEHDVEDLFVCDANFGILPR WRMTDRRYLVKDIDRVLAEIADIDGDNLYVVDDEAFIQPVR KHQWGVGQRRHSAARVVDEIRELVREFGTRFLWFNDDNFGGAFQDGYNH PRQPGVENG-PWRVRPIGDAVDEIEFLQREFGTRFLWFNDDNFGGAFQDGYHH WGGATGQKMRAFSRERLREELDV-LGRHGAEILMLADSNFGLLRQ RTIQTPAVMTTDRAAEAILA-AGVSPRGSVRITDSTATFPHKR WGGAVGQRVQSFSRARLRAELEL-FARLKVHTIVLCDANFGMLRA
Fom3 ThnK ThnL ThnP PtmL PtmM PtmN CndI Swb9 BchQ GenK Fms7 Fom3 ThnK ThnL ThnP PtmN PtmN CndI Swb9 BchQ BchQ BchQ BchQ	220 158 142 155 171 171 171 171 121 157 123 262 194 178 189 204 204 204 204 205 2247 191 142	ELDHIPFPAWHHYNMEKYFEIKAYQSPYTVGSRV-GQLYTSRGCTAHCTFCTT RRIETPSPYLLGVMDGAVATCRQRDLRFYALWETNRGCPYSCAFCD- ALDDLPMPDRTRLAHHRGRYFHSIYRPV-ALIRFTAGCPYTCKFCSL NLDKLPKPARDLADLDLYRRISRGRS-GNLVTSRGCYACAYCYS DLHKQAWPARDLLVHHRGAGIPTPV-ASTYTSRGCHAKCTFCYV DLSKQAWPARDVLIHHREAGIPTPR-ASTYTSRGCHAKCTFCYA DLEILPSPILTGAIPLADSQGRFLYDYAIMETNRGCPYKCAFCY- DMNELPIVDWSLSTRVE-PPIDPIATPVATWVETQRGCVFSCEFCDY DLDVIPSPFLTGAIPLLDDNDRFRYDVALMETNRGCPYKCSFCY- PLDYARIAKASKRTKVDGTKSIPIVVTRGCPFNCSFCVT ** *:* HATVGKKFRARAPENVVDEIEHYVNVHGVRRFLVEDDNFTFDIAR HATVGKKFRARAPENVVDEIEHYVDEYGIDEFHIQDDNITNDMDH WGSATMSALRLFDAERLQEEIEW-FAEHDVEDLFVCDANFGILPR WRMTDRRYLVKDIDRVLAEIADIDGDNLYVVDDEAFIQPVR KHQWGVGQRRISAARVVDEIRELVRFGTRFLWFNDDNFGGAFQDGYHH PRQPGVERG-PWRVRPIGDAVDEIEFLQRFGTRFLWFNDDNFGGAFQDGYHH PRQPGVTAGNAWRVRSPVDVVDEIEFLQRFGTRFLWFNDDNFGGAFQDGYHH PRQPGVTAGNAWRVRSPVDVVDEIEFLQRFGTRFLWFNDDNFGGAFQDGYHH PRQPGVTAGNAWRVRSPVDVVDEIEFLQRFGTRFLWFNDDNFGGAFQDGYHH WGGATGQKMRAFSRERLREELDV-LGRHGAEILMLADSNFGLLRQ RTIQTPAV

Figure 2-1. Partial amino acid sequence alignment of demonstrated and putative Cbldependent radical SAM *sp*³-carbon methyltransferases. GenK (gi|85814024) from *Micromonospora echinospora*, Fms7 (gi|1125024) from *Micromonospora olivasterospora*, Fom3 (gi|196166519) from *Streptomyces fradiae*, ThnK (gi|30577688), ThnL (gi|30577689), and ThnP (gi|30577693) from *Streptomyces cattleya*, PtmL (gi|212379254), PtmM (gi|212379255), and PtmN (gi|212379256) from *Streptomyces pactum*, CndI (gi|223940938) from *Chondromyces crocatus*, Swb9 (gi|283131230) from *Streptomyces* sp. SNA15896, and BchQ (gi|21674591) from *Chlorobium tepidum* TLS are involved in the biosynthesis of gentamicin, fortimicin, fosfomycin, thienamycin, pactamycin, chondrochlorens, quinomycin, 2-methylhopanoid, and bacteriochlorophyll biosynthesis, respectively.

2.3.2. Purification of GenK

The *genK* gene was amplified from the genomic DNA of *M. echinospora* using polymerase chain reaction (PCR) and was cloned into a pET24 vector. The recombinant *genK* was heterologously overexpressed in *Escherichia coli*. Although the expression of GenK was quite good, the protein was obtained only as insoluble inclusion bodies. It was therefore subjected to denaturation with 5 M urea and the refolding protocol described in Section 2.2.3. The refolded GenK was then incubated with FeCl₃ and Na₂S under anaerobic conditions in order to reconstituted [4Fe-4S] cluster unless otherwise stated. In some cases, MeCbl was added during reconstitution. Unbound iron and sulfide was removed by gel filtration. The purity of refolded, reconstituted GenK was assessed by SDS-PAGE (Figure 2-2).



Figure 2-2. UV-visible absorbance spectra and SDS PAGE of isolated GenK. The green, blue, and red spectra correspond to as-isolated GenK (8 mM), reconstituted GenK (15 mM), and reconstituted GenK (15 mM) treated with 1 mM sodium dithionite. The bleaching of the absorbance shoulder at 420 nm is characteristic of a bound iron-sulfur cluster. The calculated molecular weight of GenK is 69.3 kDa.

The UV-visible spectrum of reconstituted GenK and the bleaching of the characteristic absorbance shoulder at 420 nm in the presence of sodium dithionite are consistent with the presence of a bound [4Fe-4S] cluster. Further analysis revealed that 3.5 ± 0.9 equivalents of iron and 3.8 ± 1.1 equivalents of sulfide are bound to each GenK monomer, implying that GenK contains a single [4Fe-4S] cluster as predicted by the amino acid sequence alignment.

2.3.3. In vitro Activity Assays of GenK

The activity of GenK was assayed using 1 mM commercially available Gentamicin X_2 as the substrate in 50 mM Tris•HCl buffer (pH 8.0) containing 10 mM DTT, 4 mM SAM, and 1 mM cobalamin (Cbl). NADPH (4 mM) and methyl viologen (MV, 1mM) were employed as the electron source and mediator, respectively, to reduce the GenK [4Fe-4S] cluster to the active +1 redox state. Methylcobalamin (MeCbl) or hydroxocobalamin (HOCbl) were usually present in the GenK assay, but other cobalamins also supported GenK activity. The incubation time was usually 8 h, however, assays to check for the presence of active enzyme or assess cobalamin dependency had different times as discussed below. The assay concentration of GenK was typically 0.01 mM unless otherwise states.

2.3.4. Products of GenK Catalysis

The GenK-catalyzed reaction was monitored using reversed-phased HPLC and two different elution programs: one to detect the formation of 5'-dAdo and SAH and another to detect the conversion of GenX₂ to G418. As aminoglycosides do not have any UV-active chromophore, the amino moieties of GenX₂ and G418 were derivatized with 1-fluoro-2,4-dinitorobenzene (FDNB) allowing detection at 340 nm. HPLC traces monitoring the formation of the conversion of GenX₂ to G418 catalyzed by GenK are shown in Figure 2-3. HPLC traces monitoring the formation of 5'-dAdo and SAH formation as co-product during turnover are shown in Figure 2-4. Control assays without cobalamin, without substrate, without enzyme, or with non-reconstituted enzyme were also conducted. In the presence of either MeCbl or HOCbl (but not without Cbl), GenK catalyzes methylation of GenX₂ to produce G418 (traces b and c in Figure 2-3) as well as the formation of 5'-Ado and SAH from SAM (traces b and c in Figure 2-4). Reconstitution of GenK with iron and sulfide is required to obtain activity (trace h in Figure 2-3 and Figure 2-4). The presence of SAH in assays without GenK indicates that a notable extent of Cbl methylation takes place non-enzymatically under the conditions employed even when MeCbl is used as substrate (traces d, e, and f in Figure 2-4).¹⁶⁴ This suggests that the C-Co bond of MeCbl is prone to non-enzymatic cleavage during the assay (see Section 3.3.4).



Figure 2-3. HPLC traces showing conversion of $GenX_2$ to G418 in the presence of GenK. The small peaks at ~16.6 min in traces d-g and ~15.4 min in trace f are unrelated to product and substrate. Trace (a) is a standard composed of derivatized, authentic GenX₂ and G418, reaction mixtures. The other HPLC traces correspond to reaction mixtures containing (b) MeCbl, (c) HOCbl, (d) MeCbl but no GenK, (e) HOCbl but no GenK, (f) MeCbl but no GenX₂, (g) no Cbl, (h) MeCbl and non-reconstituted GenK.



Figure 2-4. HPLC traces showing the production of both 5'-dAdo and SAH during the GenK reaction. Trace (a) is a standard composed of authentic 5'-dAdo and SAH. The other HPLC traces correspond to reaction mixtures containing (b) MeCbl, (c) HOCbl, (d) MeCbl but no GenK, (e) HOCbl but no GenK, (f) MeCbl but no GenX₂, (g) no Cbl, (h) MeCbl and non-reconstituted GenK. Detector was set at 260 nm.

2.3.5. Reducing Systems for GenK

In order to check which reducing systems can reduce the [4Fe-4S]²⁺ cluster of GenK to active [4Fe-4S]¹⁺ state, various reducing systems including *E. coli* flavodoxin (fld)/flavodoxin reductase (fpr)/NADPH, methyl viologen (MV)/NADPH, benzyl viologen (BV)/ NADPH, dithionite, dithionite/MV, and dithionite/BV were investigated (Figure 2-5). The reaction was monitored by the formation of SAH and 5'-dAdo (Although SAH is present in all traces, separate experiments demonstrated that it is formed non-enzymatically due to the methylation of reduced cobalamin under the reaction conditions employed). Among the experimental conditions, only MV and NADPH or the flavodoxin/flavodoxin reductase/NADPH (fpr/fld/NADPH) system were

capable of activating GenK. Although dithionite is commonly used to activate radical SAM enzymes for *in vitro* studies, it is no effective for GenK. Neither NADPH and benzyl viologen (BV), dithionite and MV, nor dithionite and BV are capable of activating GenK. The coordination of SO_2^- to the corrinoid cobalt may compromise the ability of dithionite to support multiple turnovers with Cbl-dependent radical SAM enzymes.



Figure 2-5. Effectiveness of various reducing systems for the activation of GenK. HPLC traces with UV detection at 260 nm for observation of SAH and 5'-dAdo are shown.

2.3.6. Stoichiometry of Products Catalyzed by GenK

In order to measure the quantity of 5'-dAdo and SAH produced by GenK, assays with different concentrations of the enzyme (2, 4, 6, 8, and 10 μ M) over a time course (8, 12, 16, and 20 h) were conducted. Control reactions without GenK were used to assess the quantity of non-enzymatically produced SAH and this amount was subtracted out of

the enzyme assay results. The results indicate that that 5'-dAdo and SAH are produced in equal proportions (Figure 2-6). Also they show a linear relationship between the enzyme concentration and extent of product formation. Based on 8 h incubation, the average rate of both 5'-dAdo and SAH production was estimated to be approximately 0.02 min⁻¹.



Figure 2-6. 5'-dAdo and SAH production vs GenK concentration at different times.

Figure 2-7 indicates the stoichiometry of product formation at 0.01 mM GenK over a time course. This figure also demonstrates that 5'-dAdo and SAH are produced in equal proportion. The relative amounts of 5'-dAdo and G-418 produced by GenK reaction were also measured. Quantification of 5'-dAdo from trace b and c in figure 2-4, and G418 in traces b and c in Figure 2-3, gives 0.49, 0.32, 0.49, and 0.40 mM, respectively,

demonstrating that 5'-dAdo and G418 are produced at a ratio close to 1:1. Combined with the previous data, 5'-dAdo, SAH and G418 are each produced in equivalent amounts.



Figure 2-7. Stoichiometry of 5'-dAdo and SAH production during the GenK reaction

2.3.7. Methyl Transfer Manner of GenK Using Labeled SAM

In order to further investigate the role of cobalamin and the origin of the methyl group during the methyl transfer reaction, assays were conducted in the presence of either HOCbl or MeCbl and unlabeled SAM or ¹³CD₃-*methyl*-SAM followed by mass spectroscopic analysis of G418 and Cbl. The results are shown in Figures 2-8, 2-9, and 2-10. Figure 2-8 shows the mass spectra limited to the regions for substrate and product. And Figure 2-9 and Figure 2-10 display mass regions for the $[M + 2H]^{2+}$ and $[M + H]^+$ ions, respectively, of the cobalamin complexes. Mass spectra of cobalamin complexes appeared clearly $[M + 2H]^{2+}$ region rather than $[M + H]^+$ region. When unlabeled SAM was used as a substrate in the GenK reaction with HOCbl (trace a in Figure 2-8, 2-9 and 2-10) or MeCbl (trace e in Figure 2-8, 2-9 and 2-10), unlabeled G418 ($[M + H]^+$ m/z

497.28), MeCbl²⁺ ($[M + H]^{2+}$ m/z 672.80), and MeCbl⁺ ($[M + H]^{+}$ m/z 1344.58) were detected in the mass spectra. When ¹³CD³-methyl-SAM was used with HOCbl, ¹³CD₃labeled products were detected in both G418 ($[M + H]^+$ m/z 501.30) (trace b in Figure 2.8) and Cbl ($[M + H]^{2+}$ m/z 674.81, $[M + H]^{+}$ m/z 1348.57) (trace b in Figure 2-9 and 2-10). When ¹³CD₃-methyl-SAM was used with MeCbl (f), both ¹³CD₃-labeled and unlabeled G418 (trace f in Figure 2-8) and ¹³CD₃-labeled Cbl and unlabeled Cbl (trace f in Figure 2-9 and 2-10) were detected. As control reactions, assays without GenK were conducted (c, d, g, and h in Figure 2-8, 2-9 and 2-10). As expected, G418 was not obtained at all (trace c, d, g, and h in Figure 2-8), and ¹³CD₃-labelled or unlabeled MeCbl or HOCbl appeared (trace c, d, g, and h in Figure 2-9 and 2-10). These observations indicate that methylcobalamin can be generated from SAM and cobalamin and with or without the presence of the enzyme. Second, SAM is the methyl source for methylcobalamin is SAM, that is, the S-methyl substituent in SAM is transferred to Cbl. Third, combined with the previous results that methylation at GenX₂ does not occur without cobalamin (trace g in Figure 2-3 and Figure 2-4), the methyl group cannot be directly transferred from SAM to substrate to generate product. Fourth, the methyl group of product is from methylcobalamin because methylcobalamin was generated and the methyl group was transferred to GenX₂ to generate G418 when hydroxocobalamin instead of methylcobalamin was used in the GenK assay (trace a in Figure 2-8, 2-9, and 2-10). Altogether, methyl group is transferred from SAM to Cbl, and subsequently to $Gen X_2$ to give G418.



Figure 2-8. Mass spectra of GenX₂ and G418. The spectra correspond to assays with (a) HOCbl and unlabeled SAM, (b) HOCbl and ¹³CD₃-*methyl*-SAM, (c) HOCbl and unlabeled SAM without GenK, (d) HOCbl and ¹³CD₃-*methyl*-SAM without GenK, (e) MeCbl and unlabeled SAM, (f) MeCbl and ¹³CD₃-*methyl*-SAM, (g) MeCbl and unlabeled SAM without GenK, (h) MeCbl and ¹³CD₃-*methyl*-SAM without GenK.



Figure 2-9. Mass spectra of the Cbl $[M + H]^{2+}$ region. Traces are described in the legend of Figure 2-10.



Figure 2-10. Mass spectra of the Cbl $[M + H]^+$ region. Traces are described in the legend of Figure 2-10.

2.3.8. Cobalamin Dependence of GenK Activity

In order to test whether any form of cobalamin will support GenK activity or only methylcobalamin is required for methylation reaction, methyl-, hydroxo-, cyano-, and adenosyl-cobalamin were considered in the assay. The formation of 5'-dAdo and SAH were monitored by HPLC (Figure 2-11). Assays with different types of cobalamin showed the formation of 5'-dAdo and SAH (traces c, d, e, and f in Figure 2-8). Combined with the previous results in Section 2.3.4 which the formation of 5'-dAdo is related to methylation on GenX₂ (Figure 2-3 and 2-4), these results indicates that any of these complexes can support GenK activity. However, there was no observable formation of 5'-dAdo and SAH in absence of any cobalamin (trace b in Figure 2-11).



Figure 2-11. HPLC traces of GenK reaction in the presence of cobalamin. Trace (a) is a standard composed of authentic 5'-dAdo and SAH. The other HPLC traces correspond to reaction mixtures containing (b) no Cbl, (c) MeCbl, (d) HOCbl, (e) CNCbl, and (f) AdoCbl. Detector was set at 260 nm.

In order to check whether cobalamin is a cofactor and MeCbl is regenerated via methylation by SAM, assays with different concentrations of MeCbl were conducted at various time points, 4, 8, 12 and 16 h (Figure 2-12). Considered that the formations of a product, G418, and 5'-dAdo are related (Section 2.3.6), G418 production is nearly maximized in the presence of any cobalamin (as same amount as the substrate GenX₂). This observation shows that cobalamin is used as a co-substrate for the GenK reaction.



Figure 2-12. GenK activity vs concentrations of [MeCbl].

2.3.9. Syntheses of Putative GenK Substrates

As shown in Scheme 2-9 in section 2.2.11 and Scheme 2-10 in section 2.2.12, syntheses of gentamicin A_2 and 6'-amino-6'-dehydroxy-gentamicin A_2 began from D-glucosamine. Acetylation of free alcohols and amino groups with acetic anhydride on pyridine, thiophenyl group introduction using boron trifluoride diethyletherate and basic hydrolysis with 2 N aqueous sodium hydroxide solution gave the triol **11**. Azide

interconversion from the amine used the diazotransfer reaction with triflic azide and copper sulfate in methanol and dichloromethane.¹⁶⁵ The primary alcohol was selectively protected with *tert*-butyl diphenylsilyl (TBDPS) group and secondary alcohols were protected with benzyl group. TBDPS group removal using tetrabutyl ammonium fluoride and *p*-methoxybenzyl (PMB) group protection on the primary alcohol gave compound **16**. Coupling with 2-streptamine derivative **4** with *N*-iodosuccinimide and trifluoromethanesulfonic acid in ether and dichloromethane at -20 °C gave pseudodisaccharide **17**. After basic hydrolysis of the acetyl group using sodium methoxide, another coupling reaction with the xylose derivative **8** provided the pseudotrisaccharide **19**. PMB group deprotection using 2,3-dichloro-5,6-dicyano-1,4-benzoquinone gave the alcohol **20**. The azide groups were converted to amines using the Staudinger reaction.¹⁵⁷ Final deprotection of benzyl group by hydrolysis provided gentamicin A₂ in 2.5% yield over 14 steps.

From the alcohol **20**, an azide group was introduced via $S_N 2$ type reaction using sodium azide and tetrabutylammonium iodide after tosylate group was introduced. Staudinger reaction and hydrogenation gave 6'-amino-6'-dehydroxy-gentamicin A_2 in 66% yield over 4 steps from the previous synthesized compound **20**.

As shown in Scheme 2-11 in Section 2.2.13 and Scheme 2-12 in Section 2.2.14, the syntheses of 2'-deamino-2'-hydroxy-gentamicin A_2 and 2'-deamino-2'-hydroxy-6'amino-6'-dehydroxy-gentamicin A_2 started from β -D-glucose pentaacetate. The following steps in the syntheses were similar to the syntheses of gentamicin A_2 and 6'-amino-6'dehydroxy-gentamicin A_2 . The synthesis of 2'-deamino-2'-hydroxy-gentamicin A_2 was finished in 7.8% yield over 11 steps. The synthesis of 2'-deamino-2'-hydroxy-6'-amino-6'dehydroxy-gentamicin A_2 was accomplished with 23% yield over 5 steps from the previously synthesized compound **33**. As shown in Scheme 2-13 in Section 2.2.15, the synthesis of garosamine derivative **44** began from acidic hydrolysis using anhydrous hydrochloric acid (acetyl chloride in methanol) of gentamicin sulfate (a mixture of gentamicin C_{1A} , gentamicin C_1 and gentamicin C_2). The secondary amine was selectively protected with trichloroethyloxycarbonyl (Troc) group. Selective benzoylation of the secondary alcohol and replacement of the anomeric methoxy group with thioglycoside followed. Oxazolidinone ring formation via treatment with methylamine (33% MeNH₂ in EtOH) resulted in simultaneous removal of the benzoate to provide the compound **43**. Benzyl protection with benzyl bromide and sodium hydroxide in DMF gave the oxazolidinone donor **44** in 17.7% yield over 6 steps.

As shown in Scheme 2-14 in Section 2.2.16, the synthesis of gentamicin B started from the previously synthesized compound **45**. After coupling reaction with garosamine derivative **44** and *N*-iodosuccinimide and trifluoromethanesulfonic acid in dichloromethane under -78 °C, basic hydrolysis using 2 N aqueous sodium hydroxide to remove oxazolidinone ring followed. Staudinger reaction and hydrogenation gave gentamicin B in 41.3% yield over 4 steps.

As shown in Scheme 2-15 in Section 2.2.17, the synthesis of JI-20A started from pseudodisaccharide **17**. After PMB deprotection using DDQ, azide introduction, basic hydrolysis, coupling with garosamine derivative **44** with *N*-iodosuccinimide and silver trifluoromethanesulfonate in ether and dichloromethane at -10 °C gave pseudotrisaccharide **53**. Basic hydrolysis and final deprotections of azide and benzyl groups using the Staudinger reaction and hydrogenation furnished JI-20A in 21.2% yield over 8 steps.

As shown in Scheme 2-16 in Section 2.2.18, the synthesis of 2'-deamino-2'hydroxy-gentamicin X_2 began from previously synthesized pseudodisaccharide **56**. The following steps followed the same reactions as the synthesis of gentamicin B. The synthesis of 2'-deamino-2'-hydroxy-gentamicin X_2 was finished in 43.6% over 4 steps.



Figure 2-13. Early biosynthetic pathway for gentamicins.

2.3.10. Substrates Catalyzed by GenK

Members of class B RSMTs such as $TsrM^{128}$ and $ThnK^{132}$ can accept other compounds besides the intermediates in their biosynthetic pathway. Also previous experiments using mutation and reconstitution for target genes showed that GenK may catalyze methylation with gentamicin A₂ and gentamicin A.⁸¹

In order to probe the substrate specificity of GenK catalysis, nine intermediates from earlier steps in the biosynthetic pathway for gentamicin, paromamine, gentamicin A_2 , 6'-amino-6'-dehydroxy-gentamicin A_2 , 2'-deamino-2'-hydroxy-6'-amino-6'dehydroxy-gentamicin A_2 , 2'-deamino-2'-hydroxy-gentamicin A_2 , gentamicin A, JI-20A, gentamicin B, 2'-deamino-2'-hydroxy-gentamicin X_2 (boxed compounds in Figure 2-13), and neamine were selected. Paromamine¹⁶⁶ and neamine¹⁶⁷ were obtained by acidic hydrolysis of paromomycin and neomycin.

GenK assays with various substrate homologues were conducted with 0.02 mM of GenK and 1 mM of substrate for 18 h (other conditions were the same as before). Formation of 5'-dAdo was followed by HPLC to check whether 5'-dAdo• can abstract a hydrogen atom from the putative substrates (Figure 2-14). HPLC traces of the incubation using pseudodisaccharides, paromamine and neamine, in the GenK assay did not show the appearance of the 5'-dAdo peak, indicating that GenK does not accept disaccharide as a substrate (trace b and c in Figure 2-14). GenK assays with eight pseudotrisaccharides showed the formation of 5'-dAdo, even though the majority of these compounds gave only small amounts of 5'-dAdo detected by HPLC (traces d-m in Figure 2-15). The relative proportions of 5'-dAdo formation compared to consumption of GenX₂ were 3.0% (gentamicin A2), 6.5% (6'-amino-6'-dehydroxy-gentamicin A2), 2.1% (2'-deamino-2'hydroxy-6'-amino-6'-dehydroxy-gentamicin 7.6% (2'-deamino-2'-hydroxy-A₂), gentamicin A₂), 80.8% (gentamicin A), 29.0% (JI-20A), 3.1% (gentamicin B), and 6.5%

(2'-deamino-2'-hydroxy-gentamicin X_2). Except gentamicin A and JI-20A, other compounds showed very slow formation of 5'-dAdo.

Mass spectroscopic data shown in Figure 2-15 indicated that all eight compounds were converted to their methylated products by GenK. Even though the observed molecular ion peaks of these methylated products were weak, their masses were confirmed by HRMS (gentamicin A₂: calc. $[M + H]^+$ 470.2344, observed 470.2340, 6'- amino-6'-dehydroxy-gentamicin A₂: calc. $[M + H]^+$ 469.2504, observed 469.2502, 2'- deamino-2'-hydroxy-6'-amino-6'-dehydroxy-gentamicin A₂: calc. $[M + H]^+$ 471.2185, observed 471.2163, 2'-deamino-2'-hydroxy-gentamicin A₂: calc. $[M + H]^+$ 470.2344, observed 470.2343, Gentamicin A: calc. $[M + Na]^+$ 505.2480, observed 505.2491, JI-20A: calc. $[M + Na]^+$ 518.2796, observed 518.2794, gentamicin B: calc. $[M + H]^+$ 497.2817, observed 497.2816, 2-deamino-2'-hydroxy-gentamicin X₂: calc. $[M + H]^+$ 498. 2657, observed 497.2654). These data indicate that GenK can accept a variety of alternative substrates but substitutions at C2' or C6', or garosamine moiety have significant influence on GenK activity. GenX₂ remains the substrate-choice for GenK.



Figure 2-14. HPLC traces of GenK reaction with the following putative substrates; a) is corresponding to SAH and 5'-dAdo standard b) paromamine, c) neamine, d) GenA₂, e) 6'-amino-6'-dehydroxy-GenA₂, f) 2'-deamino-2'-hydroxy-6'-amino-6'-dehydroxy-GenA₂, g) 2'-deamino-2'-hydroxy-GenA₂, h) GenA, i) JI-20A, j) GenB, k) 2'-deamino-2'-hydroxy-GenX₂, l) without any substrate, and m) GenX₂.



Figure 2-15. Mass spectroscopic data of GenK reaction with A) GenA₂, B) 6'-amino-6'dehydroxy-GenA₂, C) 2'-deamino-2'-hydroxy-6'-amino-6'-dehydroxy-GenA₂, D) 2'deamino-2'-hydroxy-GenA₂, E) GenA, F) JI-20A, G) GenB, H) 2'-deamino-2'-hydroxy-GenX₂.

2.4. CONCLUSION

The existence of cobalamin (Cbl)-dependent enzymes that are members of the radical S-adenosyl-L-methionine (SAM) superfamily was previous predicted on the basis of bioinformatics analysis. A number of these are Cbl-dependent methyltransferases. In this Chapter, in vitro activity of GenK, a Cbl-dependent radical SAM enzyme that methylates an unactivated sp^3 carbon during the biosynthesis of gentamicin, an aminoglycoside antibiotic, was demonstrated. Experiments to investigate the stoichiometry of the GenK reaction revealed that 1 equiv each of 5'-deoxyadenosine and S-adenosyl-homocysteine are produced for each methylation reaction catalyzed by GenK. Furthermore, isotope-labeling experiments demonstrated that the S-methyl group from SAM is transferred to Cbl and the aminoglycoside product during the course of the reaction. The catalytic cycle of GenK, and likely other Cbl-dependent radical SAM methyltransferases, involves the consumptions of two SAM molecules and two reducing equivalents. The first molecule of SAM acts as the source of 5'-dAdo•, which abstracts a hydrogen atom from C6' of GenX₂ to give 5'-dAdo and the GenX₂ radical. The second SAM is used to methylate Cbl(I) during turnover to regenerate Me-Cbl for a subsequent round of catalysis. GenK does not only catalyze methylation of GenX₂, but also catalyzes the methylation of many aminoglycoside intermediates in the biosynthetic pathway of gentamicin even though the methylation rate is slower compared to the original substrate, GenX₂. These results demonstrate the promiscuity of GenK catalysis. Efforts to further

elucidate the chemical mechanism of this intriguing enzyme will be discussed in Chapter 3. Although it is a technical challenge to study Cbl-dependent radical SAM enzymes, this emerging class of biocatalysts offers considerable potential for the discovery of new and unprecedented enzyme chemistry.

Chapter 3: Mechanistic Studies of GenK

3.1. INTRODUCTION

Most methylation reactions in nature involve the transfer of the *S*-adenosyl-Lmethionine (SAM) (**1**) methyl group to the target molecule through an S_N2 type reaction with a carbon, nitrogen, oxygen or sulfur anion serving as the nucleophile.¹⁶⁸ However, cases of enzyme-catalyzed methylation reaction involving a single electron transfer mechanism, referred to as radical SAM methyltransferases, are increasing.¹¹⁵ In the introduction to Chapter 2, several examples of these enzymes were examined. The methyl transfer reaction for most radical SAM methyltransferases is initiated by single electron transfer from the reduced ion-sulfur cluster to the positively charged sulfonium moiety of SAM accompanied by homolysis of the C5'-S bond to generate methionine (**2**) and a 5'deoxyadenosyl radical (5'-dAdo•) (**3**) (Figure 3-1).¹⁰¹ This 5'-dAdo radical can then induce the subsequent methylation reaction by abstracting a hydrogen atom from a substrate to generate substrate radical, which then reacts with the methyl donor.¹⁰¹

The class B radical SAM methyltransferases contain a cobalamin (Cbl)-binding domain at the *N*-terminus and a characteristic Cx_3Cx_2C or similar sequence at the *C*-terminus, which are important for their activities.¹¹⁵ Members of this class whose activity has been demonstrated *in vitro* include PhpK,¹³³⁻¹³⁵ TsrM,¹²⁷⁻¹²⁹ Fom3,¹³⁰⁻¹³¹ GenD1,⁸¹ and ThnK.¹³² Recent research has provided some insights into their mechanisms.



Figure 3-1. Initial reaction of radical SAM enzymes.

The mechanism of PhpK, the *P*-methyltransferase responsible for the formation of the second C-P bond in L-phosphinotricin, was explored by Wang *et al.*¹³³⁻¹³⁵ They used a two-dimensional ${}^{1}\text{H}{-}^{31}\text{P}$ gradient heteronuclear single quantum correlation (gHSQC) nuclear magnetic resonance (NMR) experiment to determine that methylcobalamin (MeCbl) is a direct methyl donor.¹³³ They also performed a ${}^{13}\text{C}{-}^{31}\text{P}$ multiple quantum ${}^{1}\text{H}{-}^{13}\text{C}{-}^{31}\text{P}(\text{HCP})$ experiment in ${}^{1}\text{H}{-}^{31}\text{P}$ two-dimensional mode directly on a PhpK-catalyzed reaction mixture using ${}^{13}\text{CH}_{3}$ -labeled methylcobalamin as the methyl group donor. The observation of a 3:1:1:3 multiplet in the spectrum suggested direct bond formation between ${}^{13}\text{CH}_{3}$ and ${}^{31}\text{P}{.}^{135}$ These results indicated that the primary methyl donor is MeCbl in the reaction. However, they did not report the isolation of 5'-deoxyadenosine (5'-dAdo) (4) and/or other SAM cleavage products. Figure 3-2 shows two proposed mechanisms for PhpK catalysis.¹³⁵

In Figure 3-2A, after hydrogen abstraction by 5'-dAdo•, the substrate radical obtains a methyl radical from MeCbl to yield the methylated product. The oxidation state of cobalamin is changed from Cbl(III) to Cbl(II). In Figure 3-2B, after reductive cleavage of SAM followed by reduction of phosphinate radical with one electron and one proton, a nucleophilic attack by the phosphorus anion on MeCbl(III) provides the methylated intermediate and Cbl(I). A final deprotonation reaction furnishes the product.



Figure 3-2. Proposed mechanisms for PhpK catalysis.

The mechanistic studies of tryptophan methyltransferase TsrM, involved in the biosynthesis of the thiopeptide antibiotic thiostrepton, have been done by Berteau *et al.*¹²⁷⁻¹²⁸ Isotope-labeling experiments using [methyl-²H₃]-SAM as the substrate showed the generation of [methyl-²H₃]-2-methyltryptophan as a product, even when unlabeled MeCbl was used. These results suggest that the enzyme initially transfers a methyl group from SAM to cobalamin in the reaction. UV-visible spectroscopic monitoring of the reaction showed a slight increase in absorption between 400 and 476 nm and a decrease at 520 nm, which are indicative of the formation of Cbl(II). Unlike a number of radical

SAM enzymes, the TsrM reaction did not produce 5'-dAdo and did not require an external electron donor like dithionite. A mutant TsrM which lacks the canonical CxxxCxxC motif did not catalyze methylation of tryptophan, but it produced large amounts of *S*-adenosylhomocysteine (SAH) (**5**) and MeCbl. These results indicate that radical SAM chemistry was not required to initiate the methyl transfer reaction. The proposed mechanism for TsrM is shown in Figure 3-3.¹²⁷ Following transfer of the methyl radical to C2 of tryptophan, deprotonation occurs to give the observed methylated product. A one-electron transfer from the radical intermediate to the [4Fe-4S]²⁺ cluster is proposed, and the electron is then transferred from the reduced [4Fe-4S]¹⁺ cluster to Cbl(II) to generate Cbl(I). The generated Cbl(I) attacks the electrophilic methyl group of SAM to regenerate MeCbl and form SAH (**5**).



Figure 3-3. Proposed mechanism for TsrM catalysis.

Mechanistic studies concerning Fom3, a putative Cbl-dependent radical SAM enzyme which catalyzes the methylation of 2-hydroxyethylphosphonate (2-HEP) to produce 2-hydroxypropylphosphonate (2-HPP), have not yielded much insight. Wang *et*

al demonstrated through spectroscopic evidence using electron paramagnetic resonance (EPR) that Fom3 catalyzes methyl group transfer to 2-HEP.¹³⁰ The proposed mechanism of Fom3 is shown in Figure 3-4. Fom3 uses radical SAM chemistry to generate 5'-dAdo• (3) which abstracts a hydrogen atom from the C-2 position of 2-HEP to produce 5'-dAdo (4) and a substrate radical. The C-2 substrate radical reacts with MeCbl(III) to yield 2-HPP and Cbl(II).



Figure 3-4. Hypothesized mechanism for Fom3.

The catalytic mechanism of GenD1, a Cbl-dependent methyltransferase which uses gentamicin A (GenA) to generate gentamicin X_2 (Gen X_2) in the biosynthetic pathway of gentamicin C₁, has not yet been proposed. Leadlay and Sun *et al* detected the formation of the product and 5'-dAdo using reconstituted GenD1.⁸¹ One result they found was that assays using hydroxocobalamin (HOCbl) instead of MeCbl also showed GenD1 activity, presumably demonstrating that MeCbl is regenerated by a methyltransfer process during catalysis. However, the formation of SAH was not observed. The mechanism of GenD1 proposed in Figure 3-5 is based on these results.



Figure 3.5. Proposed mechanism of GenD1.

ThnK, a Cbl-dependent methyltransferase which catalyzes the methylation required to form the C6-ethyl side chain in a stereocontrolled manner in thienamycin biosynthesis, was studied by Townsend *et al.*¹³² They found near-equimolar amounts of SAH (**5**) and 5'-dAdo (**4**) during ThnK catalysis. Early labeling studies showed that the C-6 ethyl side chain of thienamycin originates from methionine.¹⁶⁹ An experiment using a chiral [¹H, ²H, ³H-methyl]-L-methionine showed that the absolute configuration of the terminal methyl in the C6 side chain is retained.¹⁶⁹ This result is consistent with an overall methyl transfer processes involving two inversions and a MeCbl intermediate. The proposed mechanism of ThnK is shown in Figure 3.6.



Figure 3-6. Proposed mechanism of ThnK.

Class B radical SAM methyltransferases potentially utilizes two molecules of SAM. The 5'-dAdo• (3) from one SAM initiates the methylation reaction, and the other SAM acts as the methyl source. Cobalamin is an intermediate methyl group carrier between SAM and the substrate. The general mechanism of class B radical SAM methyltransferases is shown in Figure 3-7.



Figure 3-7. Proposed mechanism of Class B radical SAM methyltransferases.

Several mechanisms can be envisioned for the GenK-catalyzed conversion of $GenX_2$ (6) to G418 (7). In this chapter, possible GenK reaction mechanisms are proposed,

and several attempts to elucidate the chemical mechanism of this enzyme are reported. These experiments involve analysis by UV-visible spectroscopy, assays using isotopelabeled compounds, and mechanistic investigation using various substrate analogs.

3.2. EXPERIMENTAL PROCEDURES

3.2.1. Materials and Equipment

All chemicals were purchased from Fisher Science (Pittsburgh, PA) or Sigma-Aldrich (St. Louise, MO) and used without further purification unless otherwise specified. DNA gel extraction and spin minipreps were obtained from Qiagen (Valencia, CA). All reagents for sodium dodecyl sulfate polyacrylamide gel electrophoresis (SDS-PAGE) and Amicon and Microcon YM-10 filtration products were purchased from Bio-Rad (Hercules, CA) and Millipore (Billerica, MA), respectively. *Escherichia coli* DH5a cells were bought from Bethesda Research Laboratories (Muskegon, MI). The vector pET24b (+) and enzyme KOD DNA polymerase were purchased from Novagen (Madison, WI). DNA modifying enzymes (for restriction digestion and ligation), PCR primers, and the overexpression host E. coli BL21 star (DE3) were acquired from Invitrogen (Carlsbad, CA) and New England Biolabs (NEB, Beverly, MA). Luria Bertani (LB) media are products of Difco (Detroit, MI) or Fisher Scienctific (Pittsburgh, PA). Pre-stained protein markers were purchased from NEB. Protein concentrations were determined by Bradford Assay¹⁴⁹ using bovine serum albumin as the standard. The relative molecular mass and purity of enzyme samples were determined using SDS-PAGE. The general methods and protocols for recombinant DNA manipulations are as described by Sambrook *et al.*¹⁵⁰ DNA sequencing was performed at the Core Facilities of the Institute of Cellular and Molecular Biology, University of Texas at Austin. All
reactions involving GenK were performed in a Coy Anaerobic Chamber (glovebox) under an atmosphere of approximately 97.5% N₂ and 2.5% H₂ with less than 1 ppm O₂. All solvents used for GenK assays were deaerated by bubbling with nitrogen gas before they were transferred into the glovebox. The solvents in the glove box were stirred open to the anaerobic atmosphere overnight to allow equilibration before use. Gentamicin X_2 , G-418, and GenA were purchased from TOKU-E (Bellingham, WA). Tetrahydrofuran (THF) was distilled from sodium benzophenone and dichloromethane (CH₂Cl₂) was distilled from calcium hydride under a nitrogen atmosphere. Other anhydrous solvents were purchased from Acros Organics (Pittsburgh, PA). Analytic thin layer chromatography (TLC) was performed on pre-coated TLC glass plates (Silica gel, grade 60, F254, 0.25 mm layer thickness) purchased from EMD chemicals (Madison, WI). Flash column chromatography was carried out on silica gel (230-400 mesh, grade 60) from Sorbent Technologies (Atlanta, GA) by eluting with the specific solvents. ¹H and ¹³C NMR spectra were recorded at 400, 500, 600 MHz and 100, 125, 150 MHz with a Varian Gemini spectrometer. Chemical shifts are reported as parts per million (ppm) relative to the deuteriochloroform (CDCl₃), 7.26 ppm for ¹H NMR and 77.16 ppm for ¹³C NMR, respectively. Visualization was effected with *p*-phosphoryl molybdic acid solution. The MS analyses were carried out at the Mass Spectrometry and Proteomics Facility of the Department of Chemistry, University of Texas at Austin.

3.2.2. Cloning, Expression, Refolding and Reconstitution of GenK

Cloning, expression, refolding and reconstitution of GenK were identical to the previous methods in Sections 2.2.2 and 2.2.3.

3.2.3. GenK Activity Assays, HPLC detection of SAH, 5'-dAdo, Aminoglycosides, and Mass Spectroscopic Analysis of GenK Activity

GenK activity assays, HPLC detection of SAH, 5'-dAdo and aminoglycosides, and mass spectroscopic analysis of GenK activity were conducted by following the previous methods outlined in Sections 2.2.5, 2.2.6, 2.2.7, and 2.2.8.

3.2.4. Synthesis of [6'S-²H]-Gentamicin X₂ (20)

The overall synthetic scheme is shown in Scheme 3-1.



Scheme 3-1. Synthetic scheme for the preparation of $[6'S^{-2}H]$ -GenX₂ (20).

Methyl(2S,3S,4R,5R,6S)-5-azido-3,4-bis(benzyloxy)-6-(phenylthio)tetrahydro-2H-pyran-2-carboxylate (9): (2,2,6,6,-Tetramethylpiperidin-1-yl)oxyl (TEMPO) (0.025 g, 0.16 mmol) and (diacetoxyiodo)benzene (0.646 g, 2.01 mmol) were added to a solution of compound 8 (0.384 g, 0.80 mol) in dichloromethane (10 mL) and water (10 mL). The reaction mixture was stirred at room temperature for 2 h, and 10% aqueous solution of sodium sulfite (30 mL) was added to quench the reaction. The two layers were separated, and the aqueous layer was extracted with ethyl acetate (30 mL \times 3). The combined organic layers were washed with brine, dried over magnesium sulfate, filtered, and under reduced The concentrated pressure. crude acid was dissolved in dimethylformamide (20 mL). Potassium carbonate (0.333 g, 2.41 mmol) and iodomethane (0.10 mL, 1.61 mmol) were added to the solution. The reaction mixture was stirred at room temperature for 12 h, and concentrated under reduced pressure. The residue was diluted with dichloromethane (30 mL) and washed with saturated aqueous solution of ammonium chloride (30 mL \times 3). The combined aqueous solution was extracted with dichloromethane (40 mL \times 3). The combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The crude product was purified by flash column chromatography on silica gel (Hex/EtOAc = 4:1 ~ 2:1) to provide compound 9 (0.357 g, 88%) as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 7.71–7.63 (m, 2H, Ph), 7.45–7.33 (m, 11H, Ph), 7.32–7.29 (m, 2H, Ph), 4.91 (d, J = 1.3 Hz, 2H, 2×Bn), 4.82 (d, J = 10.9 Hz, 1H, Bn), 4.66 (d, J =10.9 Hz, 1H, Bn), 4.50 (d, J = 10.1 Hz, 1H, C-1), 3.97 (d, J = 9.8 Hz, 1H, H-5), 3.82 (dd, J = 9.8, 9.0 Hz, 1H, H-4), 3.79 (s, 3H, OMe), 3.59 (t, J = 9.2 Hz, 1H, H-3), 3.43 (dd, J =10.1, 9.3 Hz, 1H, H-2). ¹³C NMR (100 MHz, CDCl₃) δ 168.29, 137.54, 137.40, 133.84, 130.57, 129.15, 128.70, 128.53, 128.46, 128.22, 128.10, 127.98, 127.87, 86.64, 84.17,

79.08, 78.06, 75.92, 75.09, 64.59, 52.57. HRMS (ESI, positive) calculated for $C_{27}H_{27}N_3O_5S^+[M+Na]^+$ 528.1564, found 528.1564.

((2R,3S,4R,5R,6S)-5-azido-3,4-bis(benzyloxy)-6-(phenylthio)tetrahydro-2H-pyran-2vl)methan-d2-ol (10): Lithium aluminum deuteride (0.315 g, 7.52 mmol) was added to a solution of compound 9 (2.923 g, 5.78 mmol) in tetrahydrofuran (50 mL) at - 78 °C. The reaction mixture was stirred for 1 h and ethyl acetate (20 mL) was added. The solution was warmed to room temperature and water (10 mL) was added to quench the reaction. Diethyl ether (20 mL) and 10% aqueous solution of sodium hydroxide (5 mL) were added. The mixtures were filtered through a Celite pad and washed with diethyl ether (50 mL). The filtered solution was dried over sodium sulfate, filtered again, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 2:1) to provide compound 10 (1.385 g, 50%) as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 7.58–7.49 (m, 2H, Ph), 7.38–7.27 (m, 13H, Ph), 4.91–4.79 (m, 3H, $3 \times Bn$), 4.64 (d, J = 11.0 Hz, 1H, Bn), 4.46 (dd, J = 10.2, 1H, H-1'), 3.58–3.46 (m, 2H, H-4, H-5), 3.40–3.29 (m, 2H, H-2, H-3). ¹³C NMR (100 MHz, CDCl₃) δ 137.75, 137.63, 133.64, 131.17, 129.29, 128.72, 128.69, 128.36, 128.22, 128.07, 86.23, 85.06, 79.61, 76.06, 75.27, 65.41. HRMS (ESI, positive) calculated for $C_{26}H_{25}D_2N_3O_4S^+$ [M + Na]⁺ 502.1740, found 502.1741.

(2S,3S,4R,5R,6S)-5-azido-3,4-bis(benzyloxy)-6-(phenylthio)tetrahydro-2H-pyran-2carbaldehyde-d (11): Dimethyl sulfoxide (8.04 mL, 57.72 mmol) was added dropwise to a solution of oxalyl chloride (1.48 mL, 17.31 mmol) in dichloromethane (80 mL) at -78 °C. After stirring for 15 min, compound 10 (5.534 g, 11.54 mmol) in dichloromethane (30 mL) was added dropwise. After stirring for 45 min, trimethylamine (8.04 mL, 57.72 mmol) was added slowly. After stirring for 20 min, the solution was warmed to room temperature, and the reaction mixture was stirred for additional 1 h. The reaction was quenched by adding H_2O (100 mL). The aqueous layer was extracted with dichloromethane (80 mL × 3). The combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The crude product was not purified and used for the next reaction.

(S)-((2R,3S,4R,5R,6S)-5-azido-3,4-bis(benzyloxy)-6-(phenylthio)tetrahydro-2H-pyr-

an-2-vl)methan-d-ol (12): Compound 11 was dissolved in dichloromethane (120 mL). (R)-Alpine-Borane[®] (0.5 M solution in THF, 25.39 mL, 12.70 mmol) was added to the stirring solution and the reaction mixture was stirred for 24 h. Acetaldehyde (20 mL) was added to the solution and the mixture was stirred for 1 h. The solution was concentrated under reduced pressure. The residue was diluted with THF (110 mL) and 3 N aqueous solution of sodium hydroxide (80 mL) and 30% hydrogen peroxide (80 mL) was added at 0 °C. The solution was stirred for 1 h, and concentrated under reduced pressure until all organic solvents disappeared. The remaining aqueous residue was extracted with ethyl acetate (100 mL \times 3). The combined organic layers were washed with brine, dried over magnesium sulfate, filtered, and concentrated under reduced pressure. The crude product was purified by flash column chromatography on silica gel (Hex/EtOAc = 2:1) to provide compound **12** (3.809 g, 69%) as a white solid. ¹H NMR (500 MHz, CDCl₃) δ 7.57–7.52 (m, 2H, Ph), 7.39-7.27 (m, 13H, Ph), 4.91-4.81 (m, 3H, $3 \times Bn$), 4.64 (d, J = 11.0 Hz, 1H, Bn), 4.46 (d, J = 10.1 Hz, 1H, H-1), 3.67 (d, J = 4.7 Hz, 1H, H-6'), 3.57–3.47 (m, 2H, H-4, H-5), 3.41–3.29 (m, 2H, H-2, H-3). ¹³C NMR (125 MHz, CDCl₃) δ 137.78, 137.67, 133.64, 131.23, 129.29, 128.71, 128.69, 128.36, 128.21, 128.08, 86.28, 85.09, 79.71, 75.27, 65.47. HRMS (ESI, positive) calculated for $C_{26}H_{26}DN_3O_4S^+$ $[M + Na]^+$ 501.1677, found 501.1679.

(2S,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-((S)-(benzyloxy)methyl-d)-2-(phenylthio)tetrahydro-2H-pyran (13): Sodium hydride (60% dispersion in mineral oil, 0.177 131

g, 4.20 mmol) was added to a solution of compound 12 (1.321 g, 2.80 mmol) in dimethylformamide (30 mL) at 0 °C. The solution was warmed to room temperature, and the reaction mixture was stirred for 30 min. The temperature was cooled to 0 °C, and benzyl bromide (0.49 mL, 4.20 mmol) was added. The reaction mixture was stirred at room temperature for 13 h. At 0 °C, methanol (10 mL) was added slowly and the reaction solution was concentrated under reduced pressure. The residue was diluted with dichloromethane (70 mL), and washed with water (70 mL \times 2). The combined aqueous layers were extracted with dichloromethane (70 mL \times 2). The combined organic layers were washed with brine, dried over magnesium sulfate, filtered, and concentrated under reduced pressure. The crude product was purified by flash column chromatography on silica gel (Hex/EtOAc = 15:1) to provide compound **13** (1.194 g, 75%) as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 7.64–7.60 (m, 2H, Ph), 7.40–7.16 (m, 18H, Ph), 4.89–4.82 (m, 2H, $2 \times Bn$), 4.80 (d, J = 10.9 Hz, 1H, Bn), 4.65–4.57 (m, 2H, $2 \times Bn$), 4.55 (d, J = 12.09.7, 9.0 Hz, 1H, H-4), 3.52 (t, J = 9.1 Hz, 1H, H-3), 3.48 (dd, J = 9.7, 4.4 Hz, 1H, H-5), 3.36 (dd, J = 10.1, 9.2 Hz, 1H, H-2). ¹³C NMR (100 MHz, CDCl₃) δ 138.33, 137.98, 137.74, 133.75, 131.30, 129.10, 128.63, 128.58, 128.48, 128.31, 128.12, 128.00, 127.95, 127.72, 127.68, 86.07, 85.19, 79.40, 77.65, 75.99, 75.15, 73.51, 65.19. HRMS (ESI, positive) calculated for $C_{33}H_{32}DN_3O_4S^+[M+Na]^+591.2147$, found 591.2148.

(1S,2S,3R,4S,6R)-4,6-diazido-3-(((2S,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-((S)-(benzyloxy)methyl-d)tetrahydro-2H-pyran-2-yl)oxy)cyclohexane-1,2-diyl diacetate (15): Compound 13 (0.812 g, 1.43 mmol) and compound 14 (0.327 g, 1.10 mmol) were co-evaporated from dry toluene three times and further dried under high vacuum overnight. A solution of compound 13 and 14 in anhydrous diethyl ether (15 mL) and anhydrous dichloromethane (3 mL) was then added to preactivated 4 Å molecular sieves.

After stirring for 30 min at room temperature, the mixture was cooled to -40 °C. N-Iodosuccinimide (0.641 g, 2.85 mmol) was added, and the reaction mixture was stirred for 20 min. Trifluoromethanesulfonic acid (0.07 mL, 0.77 mmol) was slowly added, and the reaction was warmed to -20 °C and kept stirring for 30 min. The reaction solution was diluted with ether and filtered. The residue was washed with 10% aqueous solution of sodium bisulfite (20 mL) and saturated aqueous solution of sodium bicarbonate (20 mL). The combined aqueous solutions were extracted with dichloromethane (30 mL \times 3). The combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 2:1) to provide compound 15 (0.506 g, 61%) as a clear oil. ¹H NMR (400 MHz, CDCl₃) & 7.46–7.26 (m, 13H, Ph), 7.18–7.15 (m, 2H, Ph), 5.19–5.10 (m, 2H, H-1', H-5), 4.92–4.88 (dd, J = 19.7, 10.1 Hz, 2H, Bn, H-6), 4.86– 4.77 (m, 2H, $2 \times Bn$), 4.65 (d, J = 12.1 Hz, 1H, Bn), 4.55 (d, J = 10.9 Hz, 1H, Bn), 4.50 (d, J = 12.1 Hz, 1H, Bn), 4.16 (dd, J = 10.1, 2.9 Hz, 1H, H-5'), 3.96 (dd, J = 10.4, 8.9 Hz, 1H, H-3'), 3.85-3.73 (m, 2H, H-6', H-4'), 3.69-3.54 (m, 2H, H-4, H-1), 3.45-3.31 (m, 2H, H-3, H-2'), 2.39 (dt, J = 13.3, 4.6 Hz, 1H, H-2), 2.09 (s, 3H, OAc), 2.09 (s, 3H, OAc), 1.59 (q, J = 12.7 Hz, 1H, H-2'). ¹³C NMR (100 MHz, CDCl₃) δ 169.99, 169.65, 137.97, 137.89, 137.85, 128.58, 128.54, 128.52, 128.11, 128.02, 127.92, 127.87, 127.84, 99.48, 79.68, 78.39, 78.20, 75.47, 75.10, 74.20, 73.66, 71.71, 63.22, 58.75, 57.80, 31.90, 20.77, 20.70. HRMS (ESI, positive) calculated for $C_{37}H_{40}DN_9O_9^+$ $[M + Na]^+$ 779.2982, found 779.2980.

(1S,2R,3R,4S,6R)-4,6-diazido-3-(((2S,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-((S)-(benzyloxy)methyl-d)tetrahydro-2H-pyran-2-yl)oxy)cyclohexane-1,2-diol (16): Sodium methoxide (4.37 M in methanol, 0.06 mL, 0.24 mmol) was added dropwise to a solution of compound 15 (0.610 g, 0.81 mmol) in methanol (9 mL) and tetrahydrofuran 133

(1 mL). The reaction mixture was stirred at room temperature for 6 h and then neutralized with Amberlite IR-120 (H⁺ form) resin, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 2:1) to give compound 16 (0.493 g, 91%) as a pale yellow oil. ¹H NMR (600 MHz, CDCl₃) δ 7.38–7.27 (m, 13H, Ph), 7.21–7.12 (m, 2H, Ph), 5.17 (d, J = 3.6 Hz, 1H, H-1'), 4.93 (d, J = 10.6 Hz, 1H, Bn), 4.86 (d, J = 10.7 Hz, 1H, Bn), 4.79 (d, J = 10.8Hz, 1H, Bn), 4.64 (d, J = 12.0 Hz, 1H, Bn), 4.55 (d, J = 10.9 Hz, 1H, Bn), 4.50 (d, J =12.0 Hz, 1H, Bn), 4.10 (dd, J = 10.1, 3.3 Hz, 1H, H-5'), 3.99 (dd, J = 10.2, 9.0 Hz, 1H, H-3'), 3.81-3.76 (m, 2H, H-6', H-4'), 3.64 (dd, J = 10.2, 3.6 Hz, 1H, H-2'), 3.52-3.46 (m, 1H, H-5), 3.44 (t, J = 9.1 Hz, 1H, H-6), 3.42–3.37 (m, 1H, H-1), 3.30–3.22 (m, 2H, H-4, H-3), 2.29 (dt, J = 12.9, 4.0 Hz, 1H, H-2), 1.54–1.44 (m, 1H, H-2). ¹³C NMR (150 MHz, CDCl₃) § 137.91 (Ph), 137.87 (Ph), 137.68 (Ph), 128.62 (Ph), 128.59 (Ph), 128.54 (Ph), 128.19 (Ph), 128.11 (Ph), 128.08 (Ph), 128.00 (Ph), 127.91 (Ph), 127.89 (Ph), 99.66 (C-1'), 83.98 (C-4), 81.08 (C-3'), 78.29 (C-4'), 76.01 (C-5), 75.77 (Bn), 75.47 (C-6), 75.15 (Bn), 73.71 (Bn), 71.77 (C-5'), 64.40 (C-2'), 59.78 (C-1), 58.90 (C-3), 32.22 (C-2). HRMS (ESI, positive) calculated for $C_{33}H_{36}DN_9O_7^+$ $[M + Na]^+$ 695.2771, found 695.2768.

(*3aR*,*6R*,*7R*,*7aR*)-7-(benzyloxy)-6-(((*1S*,*2S*,*3R*,*4S*,*6R*)-4,6-diazido-3-(((*2S*,*3R*,*4R*,*5S*,*6-R*)-3-azido-4,5-bis(benzyloxy)-6-((*S*)-(benzyloxy)methyl-d)tetrahydro-2*H*-pyran-2-yl)oxy)-2-hydroxycyclohexyl)oxy)-1,3a-dimethylhexahydro-2*H*-pyrano[4,3-d]oxazol-2-one (18): Compound 16 (0.369 g, 0.549 mmol) and compound 17 (0.263 g, 0.659 mmol) were co-evaporated from dry toluene three times and further dried under high vacuum overnight. A solution of compound 16 and 17 in anhydrous diethyl ether (5 mL) and anhydrous dichloromethane (1 mL) was added to a preactivated 4 Å molecular sieves and *N*-iodosuccinimide (0.247 g, 1.098 mmol). After stirring 30 min at room temperature,

the mixture was cooled to -40 °C. Silver trifluoromethanesulfonate (0.282 g, 1.098 mmol) was added, and the reaction was warmed to -10 °C to stand for 12 h. The reaction mixture was diluted with ethyl acetate (20 mL) and filtered through a Celite pad. The solution was washed with 10% aqueous solution of sodium bisulfite (20 mL) and saturated aqueous solution of sodium bicarbonate (20 mL). The combined aqueous layers were extracted with ethyl acetate (50 mL \times 3). The combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 1.5 : 1) to provide compound **18** (0.390 g, 75%) as a clear liquid. ¹H NMR ¹H NMR (500 MHz, CDCl₃) δ 7.38–7.25 (m, 18 H, Bn), 7.21–7.13 (m, 2H, Bn), 5.57 (d, J = 3.8 Hz, 1H, H-1'), 5.05 (d, J = 3.2 Hz, 1H, H-1"), 4.90–4.88 (m, 3H, Bn), 4.80 (d, J = 10.9 Hz, 1H, Bn), 4.63 (d, J = 12.1 Hz, 1H, Bn), 4.59 (d, J = 11.6 Hz, 1H, Bn), 4.55 (d, J = 11.0 Hz, 1H, Bn), 4.49 (d, J = 12.1 Hz, 1H, Bn), 4.24–4.09 (m, 3H, H-5", H-5', OH), 4.00 (t, J = 9.6 Hz, 1H, H-3'), 3.93 (t, J = 3.9 Hz, 1H, H-2"), 3.84–3.71 (m, 4H, H-6', H-6, H-5", H-4'), 3.66 (t, J = 9.0 Hz, 1H, H-5), 3.50–3.43 (m, 4H, H-4, H-1, H-3", H-2'), 3.41–3.26 (m, 2H, H-6, H-3), 2.83 (s, 3H, NMe), 2.30 (dt, J = 13.4, 4.6 Hz, 1H, H-2), 1.48 (d, J = 12.7Hz, 1H, H-2), 1.38 (s, 3H, Me). ¹³C NMR (125 MHz, CDCl₃) δ 157.22, 138.07, 137.98, 137.93, 137.37, 128.62, 128.47, 128.41, 128.25, 128.04, 127.92, 127.89, 127.86, 127.73, 127.70, 98.60, 96.71, 84.06, 80.62, 80.26, 78.14, 77.24, 75.44, 75.02, 74.95, 74.38, 73.55, 73.50, 71.39, 66.54, 63.64, 62.87, 59.75, 58.72, 32.01, 30.24, 23.47. HRMS (ESI, positive) calculated for $C_{48}H_{53}DN_{10}O_{11}^{+}[M+Na]^{+}970.3929$, found 970.3925.

(3R,4R,5R,6R)-5-(benzyloxy)-6-(((1S,2S,3R,4S,6R)-4,6-diazido-3-(((2S,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-((S)-(benzyloxy)methyl-d)tetrahydro-2H-pyran-2-yl)oxy)-2-hydroxycyclohexyl)oxy)-3-methyl-4-(methylamino)tetrahydro-2H-pyran-3-ol (19): A 2 N aqueous solution of sodium hydroxide (4 mL) was added to a solution of 135

compound 11 (0.302 g, 0.329 mmol) in ethanol (12 mL). The reaction mixture was stirred at 90 °C for 12 h, and cooled to room temperature. The solution was diluted with ethyl acetate (15 mL) and washed with water (15 mL). The residue was washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The crude product was purified by flash column chromatography on silica gel (5% methanol in dichloromethane) to provide compound **19** (0.261 g, 89%) as a clear oil. ¹H NMR (500 MHz, CDCl₃) δ 7.44–7.23 (m, 18H, Bn), 7.18–7.16 (m, 2H, Bn), 5.52 (d, J = 3.7 Hz, 1H, H-1'), 5.25 (d, J = 3.3 Hz, 1H, H-1"), 4.93–4.84 (m, 2H, Bn), 4.81 (dd, J = 11.3, 2.8 Hz, 2H, Bn), 4.65 (d, J = 12.1 Hz, 1H, Bn), 4.55 (d, J = 10.9 Hz, 1H, Bn), 4.52 (d, J = 6.3 Hz, 1H, Bn), 4.50 (d, J = 6.6 Hz, 1H, Bn), 4.15 (dd, J = 10.1, 3.2 Hz, 1H, H-5'), 4.00 (dd, J =10.3, 8.9 Hz, 1H, H-3'), 3.95 (d, J = 12.3 Hz, 1H, H-5"), 3.84–3.72 (m, 2H, H-6', H-4'), 3.68 (t, J = 9.0 Hz, 1H, H-5), 3.61 (dd, J = 10.2, 3.4 Hz, 1H, H-2"), 3.56–3.46 (m, 3H, H-5", H-1, H-2'), 3.42 (dd, J = 10.0, 8.9 Hz, 1H, H-4), 3.38–3.25 (m, 2H, H-3, H-6), 2.68 (d, J = 10.3 Hz, 1H, H-3"), 2.54 (s, 3H, NMe), 2.34 (dt, J = 13.2, 4.5 Hz, 1H, H-2), 1.52 (q, J = 12.7 Hz, 1H, H-2), 1.18 (s, 3H, Me). ¹³C NMR (125 MHz, CDCl₃) δ 138.11, 138.02, 137.96, 137.81, 128.74, 128.56, 128.51, 128.48, 128.28, 128.22, 128.13, 128.00, 127.97, 127.85, 127.83, 127.81, 98.97, 97.11, 83.77, 81.23, 80.47, 78.20, 76.14, 75.61, 75.59, 75.06, 73.61, 72.46, 71.49, 71.05, 66.68, 63.81, 63.27, 59.79, 58.90, 38.67, 32.29, 24.55. HRMS (ESI, positive) calculated for $C_{47}H_{55}DN_{10}O_{10}^+$ [*M* + *Na*]⁺ 944.4136, found 944.4142.

[6'S-²H]-Gentamicin X_2 (20): Trimethylphosphine (1 M solution in THF, 0.34 mL, 0.34 mmol) was added to a solution of compound 19 (0.63 mg, 0.068 mmol) in THF (3 mL) and 0.1 M aqueous solution of sodium hydroxide (1 mL). The reaction mixture was stirred at 50 °C for 2 h, cooled to room temperature, and concentrated under reduced pressure. The residue was dissolved in a mixture of acetic acid (0.5 mL), water (1.5 mL),

and methanol (1.5 mL). The solution was deaerated by evacuating the air inside of the flask and refilling it with nitrogen five times. Palladium hydroxide on carbon (20%, Degussa type, 0.05 g) was added and the solution was charged with hydrogen gas. The reaction mixture was stirred at room temperature under hydrogen gas for 24 h. The solution was filtered through a Celite pad, washed with water and methanol, and concentrated under reduced pressure. The residue was purified by chromatography on Amberlite CG-50 resin (NH_4^+ form) (5-10% concentrated ammonium hydroxide in water), concentrated under reduced pressure, acidified with hydrochloric acid (pH ca. 4), and lyophilized to provide compound **20** (0.03 g, 92%) as an oily solid. ¹H NMR (500 MHz, D_2O) δ 5.75 (d, J = 4.0 Hz, 1H, H-1'), 5.17 (d, J = 3.7 Hz, 1H, H-1"), 4.26 (dd, J =11.0, 3.7 Hz, 1H, H-2"), 4.07 (d, J = 12.8 Hz, 1H, H-5"), 4.02–3.95 (m, 2H, H-3', H-4), 3.89-3.76 (m, 4H, H-4', H-5, H-6', H-6), 3.70-3.60 (m, 2H, H-1, H-3), 3.59-3.46 (m, 4H, H-2', H-3', H-5', H-5'''), 2.96 (s, 3H, NMe), 2.60 (dt, J = 12.7, 4.3 Hz, 1H, H-2), 2.00 (q, J = 12.6 Hz, 1H, H-2), 1.39 (s, 3H. Me). ¹³C NMR (125 MHz, D₂O) δ 101.03, 96.88, 83.33, 80.11, 73.75, 73.57, 69.88, 69.35, 68.93, 67.77, 66.20, 63.26, 53.93, 49.50, 48.65, 34.50, 27.79, 20.90. HRMS (ESI, positive) calculated for $C_{19}H_{37}DN_4O_{10}^+$ $[M + H]^+$ 484.2723, found 484.2723.

3.2.5. Synthesis of [6'*R*-²H]-Gentamicin X₂ (29)

The overall synthetic scheme is shown in Scheme 3-2.



Scheme 3-2. Synthetic scheme for the preparation of $[6'R^{-2}H]$ -GenX₂ (29).

(S)-((2R,3S,4R,5R,6S)-5-azido-3,4-bis(benzyloxy)-6-(phenylthio)tetrahydro-2H-pyran-2-yl)methyl-d 4-methylbenzenesulfonate (21): p-Toluenesulfonyl chloride (0.457 g, 2.40 mmol) was added to a solution of 12 (0.574 g, 1.20 mmol) in pyridine (12 mL). The reaction mixture was stirred at room temperature for 24 h, and concentrated under reduced pressure. The residue was diluted with dichloromethane (40 mL) and washed with saturated aqueous solution of ammonium chloride (30 mL). The remaining aqueous layer was washed with dichloromethane (30 mL \times 3). The combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The crude product was purified by flash column chromatography on silica gel (Hex/EtOAc = 4:1) to provide compound **21** (0.689 g, 91%) as an oil. ¹H NMR (400 MHz, CDCl₃) δ 7.81–7.79 (m, 2H, Ph), 7.58–7.45 (m, 2H, Ph), 7.41–7.24 (m, 13H, Ph), 7.20–7.18 (m, 2H, Ph), 4.86 (d, *J* = 10.6 Hz, 1H, Bn), 4.83–4.73 (m, 2H, 2×Bn), 4.50 (d, *J* = 10.7 Hz, 1H, Bn), 4.33 (d, *J* = 10.1 Hz, 1H, H-1), 4.15 (d, *J* = 4.0 Hz, 1H, H-6), 3.50–3.38 (m, 3H, H-3, H-4, H-5), 3.25 (t, *J* = 9.5 Hz, 1H, H-2'), 2.40 (s, 3H, OMe). ¹³C NMR (100 MHz, CDCl₃) δ 145.15, 137.50, 137.39, 134.03, 132.93, 130.69, 130.05, 129.17, 128.70, 128.27, 128.16, 128.12, 85.97, 84.98, 76.04, 75.28, 64.98, 21.79. HRMS (ESI, positive) calculated for C₃₃H₃₂DN₃O₆S₂⁺ [*M* + *Na*]⁺ 655.1766, found 655.1764.

(R)-((2R,3S,4R,5R,6S)-5-azido-3,4-bis(benzyloxy)-6-(phenylthio)tetrahydro-2H-pyran-2-yl)methyl-d benzoate (22): Sodium benzoate (0.388 g, 3.40 mmol) was added to a solution of compound 21 (0.717 g, 1.13 mmol) in dimethylformamide (11mL). The reaction mixture was stirred at 100 °C for 12 h, cooled to room temperature, and concentrated under reduced pressure. The residue was diluted with dichloromethane (40 mL) and washed with water (30 mL \times 3). The combined aqueous layers were extracted with dichloromethane (40 mL \times 3). The combined organic layers were washed with brine, dried over magnesium sulfate, filtered, and concentrated under reduced pressure. The crude product was purified by flash column chromatography on silica gel (Hex/EtOAc = 4:1) to provide compound 22 (0.582 g, 89%) as a clear oil. ¹H NMR (400 MHz, CDCl₃) δ 8.05–7.99 (m, 2H, Ph), 7.67–7.59 (m, 1H, Ph), 7.59–7.53 (m, 2H, Ph), 7.50–7.45 (m, 2H, Ph), 7.40–7.22 (m, 10H, Ph), 7.17–7.10 (m, 2H, Ph), 4.92 (d, J = 10.310.9 Hz, 1H, Bn), 4.45 (d, J = 10.1 Hz, 1H, H-1), 3.70–3.53 (m, 3H, H-3, H-4, H-5), 3.40–3.29 (m, 1H, H-2). ¹³C NMR (100 MHz, CDCl₃) δ 166.10, 137.41, 137.38, 134.30, 133.37, 130.47, 129.91, 129.86, 129.04, 128.77, 128.71, 128.69, 128.58, 128.50, 128.35,

128.28, 128.16, 85.72, 85.21, 77.27, 77.22, 76.26, 75.35, 64.99. HRMS (ESI, positive) calculated for $C_{33}H_{30}DN_3O_5S^+[M+Na]^+605.1939$, found 605.1947.

(*R*)-((*2R*,*3S*,*4R*,*5R*,*6S*)-5-azido-3,4-bis(benzyloxy)-6-(phenylthio)tetrahydro-2*H*-pyran-2-yl)methan-d-ol (23): 2 N aqueous solution of sodium hydroxide (2.52 mL, 5.03 mmol) was added to a solution of 22 (0.733 g, 1.26 mmol) in methanol (12 mL). The reaction mixture was stirred at room temperature for 5 h, neutralized with Amberlite IR-120 (H⁺ form) resin, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 2:1) to give compound 23 (0.560 g, 93%) as a white solid. ¹H NMR (500 MHz, CDCl₃) δ 7.57–7.52 (m, 2H, Ph), 7.37–7.27 (m, 13H), 4.91–4.81 (m, 3H, 3×Bn), 4.64 (d, *J* = 11.0 Hz, 1H, Bn), 4.46 (d, *J* = 10.2 Hz, 1H, H-1), 3.86 (d, *J* = 2.6 Hz, 1H, H-6), 3.57–3.49 (m, 2H, H-4, H-5), 3.39 – 3.29 (m, 2H, H-2, H-3). ¹³C NMR (125 MHz, CDCl₃) δ 137.79, 137.68, 133.64, 131.24, 129.29, 128.71, 128.69, 128.36, 128.21, 128.07, 86.29, 85.09, 75.27, 65.47. HRMS (ESI, positive) calculated for C₂₆H₂₆DN₃O₄S⁺ [*M* + *Na*]⁺ 501.1677, found 501.1679.

(2S,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-((R)-(benzyloxy)methyl-d)-2-(phenylthio)tetrahydro-2H-pyran (24): Sodium hydride (60% dispersion in mineral oil, 0.072 g, 1.71 mmol) was added to a solution of compound 23 in dimethylformamide (9 mL) at 0 °C. The reaction mixture was stirred at room temperature for 30 min. Benzyl bromide (0.20 mL, 1.71 mmol) was added at 0 °C, and the reaction mixture was stirred at room temperature for 12 h. At 0 °C, methanol (5 mL) was added slowly, and the solution was concentrated under reduced pressure. The residue was diluted with dichloromethane (50 mL) and washed with water (30 mL \times 2). The combined aqueous layers were extracted with dichloromethane (40 mL \times 3). The combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The crude product was purified by flash column chromatography on silica gel (Hex/EtOAc = 15:1) to provide compound **24** (0.364 g, 75%) as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 7.67–7.59 (m, 2H, Ph), 7.39–7.19 (m, 18H, Ph), 4.90–4.82 (m, 2H, 2×Bn), 4.80 (d, *J* = 10.9 Hz, 1H, Bn), 4.66–4.58 (m, 2H, Bn), 4.55 (d, *J* = 12.0 Hz, 1H, Bn, 4.43 (d, *J* = 10.1 Hz, 1H, H-1), 3.77 (d, *J* = 1.9 Hz, 1H, H-6), 3.63 (dd, *J* = 9.7, 9.0 Hz, 1H, H-4), 3.52 (t, *J* = 9.1 Hz, 1H, H-3), 3.48 (dd, *J* = 9.7, 1.9 Hz, 1H, H-5), 3.36 (dd, *J* = 10.1, 9.3 Hz, 1H, H-2). ¹³C NMR (100 MHz, CDCl₃) δ 138.32, 137.98, 137.72, 133.74, 131.28, 129.08, 128.62, 128.57, 128.47, 128.30, 128.11, 127.98, 127.93, 127.71, 127.67, 86.06, 85.18, 79.39, 77.64, 75.98, 75.14, 73.51, 65.17. HRMS (ESI, positive) calculated for C₃₃H₃₂DN₃O₄S⁺ [*M* + *Na*]⁺ 591.2147, found 591.2146.

(15,25,3R,45,6R)-4,6-diazido-3-(((2S,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-((R)-(benzyloxy)methyl-d)tetrahydro-2H-pyran-2-yl)oxy)cyclohexane-1,2-diyl diacetate (25): Compound 24 (0.956 g, 1.68 mmol) and compound 14 (0.385 g, 1.29 mmol) were co-evaporated from dry toluene three times and further dried under high vacuum overnight. A solution of compound 24 and 14 in anhydrous diethyl ether (15 mL) and anhydrous dichloromethane (3 mL) was then added to preactivated 4 Å molecular sieves. After stirring for 30 min at room temperature, the mixture was cooled to -40 °C. *N*lodosuccinimide (0.756 g, 3.36 mmol) was added, and the reaction mixture was stirred for 30 min. Trifluoromethanesulfonic acid (0.08 mL, 0.90 mmol) was slowly added, and the reaction was warmed to -20 °C and kept stirring for 30 min. The reaction solution was diluted with ether (15 mL) and filtered. The residue was washed with 10% aqueous solution of sodium bisulfite (15 mL) and saturated aqueous solution of sodium bicarbonate (15 mL). The combined aqueous solutions were extracted with dichloromethane (40 mL \times 3). The combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:1) to provide compound **25** (0.595 g, 61%) as a clear oil. ¹H NMR (400 MHz, CDCl₃) δ 7.43–7.27 (m, 13H, Ph), 7.18–7.15 (m, 2H, Ph), 5.21–5.09 (m, 2H, H-1', H-5), 4.95–4.88 (dd, *J* = 20.3, 10.3 Hz, 2H, Bn, H-6), 4.85–4.77 (m, 2H, 2×Bn), 4.65 (d, *J* = 12.1 Hz, 1H, Bn), 4.54 (d, *J* = 10.9 Hz, 1H, Bn), 4.52–4.46 (m, 1H, Bn), 4.16 (dd, *J* = 10.1, 2.0 Hz, 1H, H-5'), 3.96 (dd, *J* = 10.4, 8.9 Hz, 1H, H-3'), 3.79 (dd, *J* = 10.1, 9.0 Hz, 1H, H-4'), 3.70–3.56 (m, 3H, H-4, H-1, H-6'), 3.46–3.40 (m, 1H, H-3), 3.36 (dd, *J* = 10.4, 3.9 Hz, 1H, H-2'), 2.39 (dt, *J* = 13.4, 4.6 Hz, 1H, H-2), 2.09 (s, 3H, OAc), 2.08 (s, 3H, OAc), 1.59 (q, *J* = 12.7 Hz, 1H, H-2). ¹³C NMR (100 MHz, CDCl₃) δ 169.99, 169.66, 138.00, 137.91, 137.86, 128.59, 128.55, 128.53, 128.12, 128.03, 127.93, 127.89, 127.84, 99.49, 79.70, 75.47, 75.11, 74.21, 73.68, 73.64, 71.73, 63.24, 58.77, 57.81, 31.92, 20.78, 20.71. HRMS (ESI, positive) calculated for C₃₇H₄₀DN₉O₉⁺ [*M* + *Na*]⁺779.2982, found 779.2996.

(*1S*,2*R*,3*R*,4*S*,6*R*)-4,6-diazido-3-(((*2S*,3*R*,4*R*,5*S*,6*R*)-3-azido-4,5-bis(benzyloxy)-6-((*R*-)-(benzyloxy)methyl-d)tetrahydro-2*H*-pyran-2-yl)oxy)cyclohexane-1,2-diol (26): Sodium methoxide (4.37 M in methanol, 0.08 mL, 0.35 mmol) was added dropwise to a solution of compound **25** (0.874 g, 1.15 mmol) in methanol (10 mL). The reaction mixture was stirred at room temperature for 1 h and then neutralized with Amberlite IR-120 (H⁺ form) resin, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 2:1) to give compound **26** (0.738 g, 95%) as a pale yellow oil. ¹H NMR (500 MHz, CDCl₃) δ 7.38–7.28 (m, 13H, Ph), 7.21–7.15 (m, 2H, Ph), 5.22 (d, *J* = 3.7 Hz, 1H, H-1'), 4.94 (d, *J* = 10.7 Hz, 1H, Bn), 4.87 (d, *J* = 10.7 Hz, 1H, Bn), 4.81 (d, *J* = 10.9 Hz, 1H, Bn), 4.66 (d, *J* = 12.1 Hz, 1H, Bn), 4.57 (d, *J* = 10.9 Hz, 1H, Bn), 4.51 (d, *J* = 12.0 Hz, 1H, Bn), 4.12 (dd, *J* = 10.0, 2.1 Hz, 1H, H-5'), 4.01 (dd, *J* = 10.2, 9.0 Hz, 1H, H-3'), 3.80 (dd, *J* = 10.1, 9.0 Hz, 1H, H-4'), 3.69 (d, *J* = 2.0 Hz, 1H, H-6'), 3.63 (dd, *J* = 10.2, 3.6 Hz, 1H, H-2'), 142 3.50 (t, J = 8.7 Hz, 1H, H-5), 3.47–3.35 (m, 2H, H-6, H-1), 3.34–3.22 (m, 2H, H-4, H-3), 2.28 (dt, J = 12.9, 4.2 Hz, 1H, H-2), 1.48 (dt, J = 13.0, 11.7 Hz, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 137.92 (Ph), 137.86 (Ph), 137.69 (Ph), 128.57 (Ph), 128.54 (Ph), 128.50 (Ph), 128.15 (Ph), 128.04 (Ph), 127.93 (Ph), 127.87 (Ph), 127.83 (Ph), 99.51 (C-1'), 83.56 (C-4), 80.96 (C-3'), 78.27 (C-4'), 75.69 (C-5), 75.46 (C-6), 75.09 (Bn), 73.65 (Bn), 71.70 (C-5'), 67.90 (C-6'), 64.30 (C-2'), 59.77 (C-1), 58.89 (C-3), 32.15 (C-2). HRMS (ESI, positive) calculated for C₃₃H₃₆DN₉O₇⁺ [M + Na]⁺ 695.2771, found 695.2770.

(3aR,6R,7R,7aR)-7-(benzyloxy)-6-(((1S,2S,3R,4S,6R)-4,6-diazido-3-(((2S,3R,4R,5S,6-

R)-3-azido-4,5-bis(benzyloxy)-6-((R)-(benzyloxy)methyl-d)tetrahydro-2H-pyran-2-

yl)oxy)-2-hydroxycyclohexyl)oxy)-1,3a-dimethylhexahydro-2H-pyrano[4,3-d]oxazol-2-one (27): Compound 26 (0.721 g, 1.07 mmol) and compound 17 (0.471 g, 1.18 mmol) were co-evaporated from dry toluene three times and further dried under high vacuum overnight. A solution of compound 26 and 17 in anhydrous diethyl ether (10 mL) and anhydrous dichloromethane (2 mL) was added to a preactivated 4 Å molecular sieves and N-iodosuccinimide (0.482 g, 2.14 mmol). After stirring 30 min at room temperature, the mixture was cooled to -40 °C. Silver trifluoromethanesulfonate (0.551 g, 2.14 mmol) was added, and the reaction was warmed to -10 °C to stand for 3 h. The reaction mixture was diluted with ethyl acetate (10 mL) and filtered through a Celite pad. The solution was washed with 10% aqueous solution of sodium bisulfite (15 mL) and saturated aqueous solution of sodium bicarbonate (15 mL). The combined aqueous layers were extracted with ethyl acetate (20 mL \times 3). The combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:2) to provide compound **27** (0.589 g, 58%) as a clear liquid. ¹H NMR (500 MHz, CDCl₃) δ 7.42–7.26 (m, 18H, Bn), 7.19–7.14 (m, 2H, Bn), 5.57 (d, J = 3.7 Hz, 1H, H-1'), 5.06 (d, J = 3.2 Hz, 143

1H, H-1"), 4.92–4.87 (m, 3H, Bn), 4.81 (d, J = 10.9 Hz, 1H, Bn), 4.65 (d, J = 12.1 Hz, 1H, Bn), 4.61 (d, J = 11.6 Hz, 1H, Bn), 4.55 (d, J = 11.0 Hz, 1H, Bn), 4.50 (d, J = 12.2 Hz, 1H, Bn), 4.24–4.12 (m, 3H, H-5", H-5', OH), 4.00 (dd, J = 10.3, 8.9 Hz, 1H, H-3'), 3.94 (dd, J = 4.5, 3.2 Hz, 1H, H-2"), 3.85–3.73 (m, 3H, H-6, H-5", H-4'), 3.70–3.63 (m, 2H, H-5, H-6'), 3.48 (ddd, J = 8.7, 5.7, 4.1 Hz, 4H, H-4, H-1, H-3", H-2'), 3.45–3.37 (m, H, H-6), 3.37–3.29 (m, 1H, H-3), 2.84 (s, 3H, NMe), 2.32 (dt, J = 13.2, 4.5 Hz, 1H, H-2), 1.56–1.44 (m, 1H, H-2), 1.40 (s, 3H, Me). ¹³C NMR (100 MHz, CDCl₃) δ 157.35, 138.13, 138.03, 137.98, 137.41, 128.74, 128.59, 128.53, 128.39, 128.16, 128.03, 128.00, 127.86, 98.77, 96.81, 84.15, 80.85, 80.41, 78.21, 77.35, 75.60, 75.09, 74.50, 73.69, 73.63, 71.50, 66.64, 63.75, 63.01, 59.88, 58.79, 32.14, 30.37, 23.60. HRMS (ESI, positive) calculated for C₄₈H₅₃DN₁₀O₁₁⁺ [M + Na]⁺970.3929, found 970.3925.

(*3R*,*4R*,*5R*,*6R*)-5-(benzyloxy)-6-(((*1S*,*2S*,*3R*,*4S*,*6R*)-4,6-diazido-3-(((*2S*,*3R*,*4R*,*5S*,*6R*)-3-azido-4,5-bis(benzyloxy)-6-((*R*)-(benzyloxy)methyl-d)tetrahydro-2*H*-pyran-2-yl)oxy)-2-hydroxycyclohexyl)oxy)-3-methyl-4-(methylamino)tetrahydro-2*H*-pyran-3-ol

(28): A 2 N aqueous solution of sodium hydroxide (10 mL) was added to a solution of compound 27 (0.592 g, 0.625 mmol) in ethanol (30 mL). The reaction mixture was stirred at 90 °C for 14 h, and cooled to room temperature. The solution was diluted with ethyl acetate (30 mL) and washed with water (30 mL). The residue was washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The crude product was purified by flash column chromatography on silica gel (5% methanol in dichloromethane) to provide compound 28 (0.437 g, 76%) as a clear oil. ¹H NMR (500 MHz, CDCl₃) δ 7.41–7.24 (m,18, Bn), 7.17–7.15 (m, 2H, Bn), 5.51 (d, *J* = 3.7 Hz, 1H, H-1'), 5.25 (d, *J* = 3.3 Hz, 1H, H-1''), 4.88 (d, *J* = 3.3 Hz, 2H, Bn), 4.83–4.76 (m, 2H, Bn), 4.65 (d, *J* = 12.1 Hz, 1H, Bn), 4.56–4.47 (m, 3H, Bn), 4.42 (s, 1H, OH), 4.13 (dd, *J* = 10.1, 2.1 Hz, 1H, H-5'), 3.99 (dd, *J* = 10.3, 9.0 Hz, 1H, H-3'), 3.94 (d, *J* = 12.3 Hz, 1H, 144

H-5"), 3.77 (dd, J = 10.1, 8.9 Hz, 1H, H-4'), 3.70–3.64 (m, 3H, H-5, H-6', H-4'), 3.64– 3.58 (m, 1H, H-2"), 3.56–3.45 (m, 3H, H-1, H-5", H-2'), 3.41 (dd, J = 10.1, 8.9 Hz, 1H, H-4), 3.37–3.28 (m, 2H, H-6, H-3), 2.68 (d, J = 10.3 Hz, 1H, H-3"), 2.54 (s, 3H, NMe), 2.34 (dt, J = 13.2, 4.4 Hz, 1H, H-2), 1.51 (q, J = 12.7 Hz, 1H, H-2), 1.18 (s, 3H, Me). ¹³C NMR (125 MHz, CDCl₃) δ 138.14, 138.04, 137.98, 137.81, 128.77, 128.60, 128.54, 128.52, 128.32, 128.27, 128.16, 128.03, 128.00, 127.89, 127.85, 99.02, 97.11, 83.75, 81.30, 80.51, 78.23, 77.36, 76.13, 75.64, 75.62, 75.09, 73.66, 72.50, 71.53, 71.07, 63.85, 63.35, 59.83, 58.92, 38.66, 32.32, 24.54. HRMS (ESI, positive) calculated for C₄₇H₅₅DN₁₀O₁₁⁺ [M+Na]⁺944.4136, found 944.4106.

[6'R-²H]-Gentamicin X₂ (29): Trimethylphosphine (1 M solution in THF, 0.84 mL, 0.84 mmol) was added to a solution of compound 28 (0.129 mg, 0.14 mmol) in THF (4.8 mL) and 0.1 M aqueous solution of sodium hydroxide (1.6 mL). The reaction mixture was stirred at 50 °C for 2 h, cooled to room temperature, and concentrated under reduced pressure. The residue was dissolved in a mixture of acetic acid (1.2 mL), water (3.6 mL) and methanol (3.6 mL). The solution was degassed by evacuating the air inside of the flask and refilling it with nitrogen five times. Palladium hydroxide on carbon (20%, Degussa type, 0.1 g) was added and the solution was charged with hydrogen gas. The reaction mixture was stirred at room temperature under hydrogen gas for 24 h. The solution was filtered through a Celite pad, washed with water and methanol, and concentrated under reduced pressure. The residue was purified by chromatography on Amberlite CG-50 resin (NH_4^+ form) (5% concentrated ammonium hydroxide in water), concentrated under reduced pressure, acidified with hydrochloric acid (pH ca. 4), and lyophilized to provide compound **29** (0.057 g, 84%) as an oily solid. ¹H NMR (500 MHz, D_2O) δ 5.70 (d, J = 4.0 Hz, 1H, H-1'), 5.13 (d, J = 3.7 Hz, 1H, H-1"), 4.24 (dd, J = 10.9, 3.7 Hz, 1H, H-2"), 4.04 (d, J = 12.8 Hz, 1H, H-5"), 3.99–3.90 (m, 3H, H-6', H-3', H-4), 145

3.90–3.79 (m, 3H, H-4', H-5, H-6), 3.68–3.58 (m, 2H, H-1, H-3), 3.57–3.44 (m, 4H, H-5", H-3", H-5', H-2'), 2.94 (s, 3H, NMe), 2.57 (dt, J = 12.6, 4.3 Hz, 1H, H-2), 1.95 (q, J = 12.7 Hz, 1H, H-2), 1.37 (s, 3H, Me). ¹³C NMR (125 MHz, D₂O) δ 101.05, 97.01, 83.36, 80.22, 73.69, 73.59, 69.83, 69.27, 68.92, 67.78, 66.17, 63.20, 53.86, 49.43, 48.57, 34.45, 27.81, 20.87. HRMS (ESI, positive) calculated for C₁₉H₃₇DN₄O₁₀⁺ [M + H]⁺ 484.2723, found 484.2725.

3.2.6. Synthesis of [6'-F]-gentamicin X₂ (35)

The overall synthetic scheme is shown in Scheme 3-3.



Scheme 3-3. Synthetic scheme for the preparation of [6'-F]-GenX₂ (35)

(1S,2S,3R,4S,6R)-4,6-diazido-3-(((2S,3R,4R,5S,6S)-3-azido-4,5-bis(benzyloxy)-6-(fluoromethyl)tetrahydro-2*H*-pyran-2-yl)oxy)cyclohexane-1,2-diyl diacetate (31): Diethylaminosulfur trifluoride (DAST) (0.20 mL, 1.53 mmol) was added to a solution of compound **30** (0.511 g, 0.767 mmol) in dichloromethane (10 mL) at 0 °C. The reaction mixture was stirred at 0 °C for 30 min and at room temperature for 12 h. The reaction was quenched by adding saturated aqueous solution of sodium bicarbonate (20 mL). The aqueous layer was extracted with dichloromethane (30 mL \times 3). The combined organic layers were washed with brine, dried over magnesium sulfate, filtered, and concentrated. The crude product was purified by flash column chromatography on silica gel (Hex/EtOAc = 4:1) to provide compound **31** (0.307 g, 60%) as a clear liquid. ¹H NMR (500 MHz, CDCl₃) δ 7.40–7.34 (m, 10H, Ph), 5.19–5.15 (m, 2H, H-1', H-5), 4.99–4.87 (m, 4H, H-6, $3 \times Bn$), 4.82–4.55 (m, 3H, Bn, H-6', H-6'), 4.22 (dd, J = 31.2, 10.2 Hz, 1H, H-5'), 4.02 (t, J = 9.6 Hz, 1H, H-3'), 3.72 (t, J = 9.6 Hz, 1H, H-4'), 3.66 (t, J = 9.8 Hz, 1H, H-4), 3.64-3.54 (m, 1H, H-1), 3.41-3.36 (m, 2H, H-2', H-3), 2.35 (dt, J = 13.4, 4.7 Hz, 1H, H-2), 2.12 (s, 1H, Ac), 2.09 (s, 1H, Ac), 1.62 (q, J = 12.7 Hz, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) & 169.75 (Ac), 169.46 (Ac), 137.59 (Ph), 137.52 (Ph), 128.50 (Ph), 128.45 (PH), 127.98 (Ph), 127.96 (Ph), 127.92 (Ph), 127.78 (Ph), 99.29 (C-1'), 81.43 (d, J = 173.4, C-6', 79.44 (C-3'), 78.40 (C-4), 77.30 (d, J = 5.1 Hz, C-4'), 75.29 (Bn), 75.17 (Bn), 73.94 (C-6), 73.31 (C-5), 71.08 (d, J = 18.2 Hz, C-5'), 62.93 (C-2'), 58.42 (C-3), 57.53(C-1), 31.44 (C-2), 20.52 (OAc), 20.48 (OAc). ¹⁹F NMR (375 MHz, CDCl₃) δ -234.52 (td, J = 47.7, 31.3 Hz). HRMS (ESI, positive) calculated for $C_{30}H_{34}FN_9O_8^+$ [M +Na]⁺ 690.2407, found 690.2414.

(1S,2R,3R,4S,6R)-4,6-diazido-3-(((2S,3R,4R,5S,6S)-3-azido-4,5-bis(benzyloxy)-6-(fluoromethyl)tetrahydro-2H-pyran-2-yl)oxy)cyclohexane-1,2-diol (32): Sodium methoxide (4.37 M in methanol, 0.7 mL, 0.33 mmol) was added to a solution of compound 31 147 (0.735 g, 1.10 mmol) in tetrahydrofuran (2 mL) and methanol (18 mL). The reaction mixture was stirred at room temperature for 12 h, neutralized with Amberlite IR-120 (H⁺ form) resin, filtered, and concentrated under reduced pressure. The crude product was purified by flash column chromatography on silica gel (Hex/EtOAc = 2:1) to give compound **32** (0.426 g, 73%) as a clear oil. ¹H NMR (500 MHz, CDCl₃) δ 7.37–7.30 (m, 10H, Bn), 5.17 (d, *J* = 3.7 Hz, 1H, H-1'), 4.95 (d, *J* = 10.6 Hz, 1H, Bn), 4.91–4.86 (m, 2H, Bn), 4.76–4.52 (m, 3H, Bn, H-6', H-6'), 4.18–3.97 (m, 2H, H-5', H-3'), 3.71 (dd, *J* = 10.2, 9.0 Hz, 1H, H-4'), 3.63 (dd, *J* = 10.2, 3.6 Hz, 1H, H-2'), 3.55–3.36 (m, 3H, H-4, H-5, H-1), 3.32–3.19 (m, 2H, H-6, H-3), 3.03 (s, 1H, OH), 2.31 (dt, *J* = 12.9, 3.9 Hz, 1H, H-2), 1.58–1.43 (m, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 137.59 (Ph), 137.49 (Ph), 128.73 (Ph), 128.67 (Ph), 128.23 (Ph), 128.21 (Ph), 127.93 (Ph), 99.77 (C-1'), 84.09 (C-6), 81.63 (d, *J* = 172.1, C-6'), 80.94 (C-3'), 77.46 (d, *J* = 5.4, C-4'), 76.07 (Bn), 75.87 (C-4), 75.43 (Bn, C-5), 71.35 (d, *J* = 18.2, C-5'), 64.28 (C-2'), 59.78 (C-1), 58.91 (C-3), 32.12 (C-2). ¹⁹F NMR (470 MHz, CDCl₃) δ -235.11 (td, *J* = 47.8, 30.9 Hz). HRMS (ESI, positive) calculated for C₂₆H₃₀FN₉O₆⁺ [*M* + *Na*]⁺606.2195, found 606.2202.

(3aR, 6R, 7R, 7aR)-7-(benzyloxy)-6-(((1*S*, 2*S*, 3*R*, 4*S*, 6*R*)-4,6-diazido-3-(((2*S*, 3*R*, 4*R*, 5*S*, 6-*S*)-3-azido-4,5-bis(benzyloxy)-6-(fluoromethyl)tetrahydro-2*H*-pyran-2-yl)oxy)-2-hydroxycyclohexyl)oxy)-1,3a-dimethylhexahydro-2*H*-pyrano[4,3-d]oxazol-2-one (33): Compound 32 (0.543 g, 0.808 mmol) and compound 17 (0.355 g, 0.888 mmol) were coevaporated from dry toluene three times and further dried under high vacuum overnight. A solution of compound 32 and 17 in anhydrous diethyl ether (10 mL) and anhydrous dichloromethane (2 mL) was added to a preactivated 4 Å molecular sieves and *N*iodosuccinimide (0.364 g, 1.616 mmol). After stirring 30 min at room temperature, the mixture was cooled to -40 °C. Silver trifluoromethanesulfonate (0.415 g, 1.616 mmol) was added, and the reaction was warmed to -10 °C to stand for 3 h. The reaction mixture

was diluted with ethyl acetate (10 mL) and filtered through a Celite pad. The solution was washed with 10% aqueous solution of sodium bisulfite (20 mL) and saturated aqueous solution of sodium bicarbonate (25 mL). The combined aqueous layers were extracted with ethyl acetate (40 mL \times 3). The combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:2) to provide compound **33** (0.365 g, 53%) as a clear liquid. ¹H NMR (500 MHz, CDCl₃) δ 7.42–7.28 (m, 15H, Ph), 5.60 (d, J = 3.8 Hz, 1H, H-1'), 5.03 (d, J = 3.2 Hz, 1H, H-1"), 4.92–4.87 (m, 4H, 4×Bn), 4.74–4.52 (m, 4H, 2×Bn, H-6', H-6'), 4.22–4.08 (m, 3H, H-5", OH, H-5'), 4.03 (dd, J = 10.3, 8.9 Hz, 1H, H-3'), 3.95 (dd, J = 4.4, 3.1 Hz, 1H, H-2"), 3.80 (d, J =12.4 Hz, 1H, H-5"), 3.70-3.65 (m, 2H, H-4', H-5), 3.51-3.38 (m, 5H, H-1, H-3", H-4, H-2', H-6), 3.35 (ddd, J = 12.4, 10.0, 4.4 Hz, 1H, H-3), 2.83 (s, 3H, NMe), 2.33 (dt, J =13.2, 4.4 Hz, 1H, H-2), 1.51 (q, J = 12.6 Hz, 1H, H-2), 1.41 (s, 3H, Me). ¹³C NMR (125) MHz, CDCl₃) δ 157.28 (N(C(=O)O), 137.85 (Ph), 137.82 (Ph), 137.41 (Ph), 128.73 (Ph), 128.64 (Ph), 128.61 (Ph), 128.39 (Ph), 128.15 (Ph), 128.08 (Ph), 128.05 (Ph), 128.02 (Ph), 127.91 (Ph), 98.54 (C-1'), 97.00 (C-2"), 84.60 (C-6), 81.76 (d, J = 173.3 Hz, C-6'), 80.50 (C-4), 80.15 (C-3'), 77.36 (C-4'), 77.21 (C-4"), 75.63 (Bn), 75.33 (Bn), 75.12 (C-5), 74.63 (Bn), 73.42 (C-2"), 71.06 (d, J = 18.2 Hz, C-5'), 66.87 (C-5"), 63.58 (C-2'), 63.04 (C-3"), 59.78 (C-1), 58.77 (C-3), 32.08 (C-2), 30.32 (NMe), 23.63 (Me). ¹⁹F NMR (375 MHz, CDCl₃) δ -234.59 (td, J = 47.8, 30.7 Hz). HRMS (ESI, positive) calculated for $C_{41}H_{47}FN_{10}O_{10}^{+}[M+Na]^{+}881.3353$, found 881.3368.

(3R,4R,5R,6R)-5-(benzyloxy)-6-(((1S,2S,3R,4S,6R)-4,6-diazido-3-(((2S,3R,4R,5S,6S)-3-azido-4,5-bis(benzyloxy)-6-(fluoromethyl)tetrahydro-2H-pyran-2-yl)oxy)-2-hydroxycyclohexyl)oxy)-3-methyl-4-(methylamino)tetrahydro-2H-pyran-3-ol (34): A 2 N aqueous solution of sodium hydroxide (5 mL) was added to a solution of compound 33 149

(0.365 g, 0.425 mmol) in ethanol (15 mL). The reaction mixture was stirred at 90 °C for 10 h, and cooled to room temperature. The solution was diluted with ethyl acetate (20 mL) and washed with water (20 mL). The residue was washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The crude product was purified by flash column chromatography on silica gel (5-8% methanol in dichloromethane) to provide compound **34** (0.251 g, 71%) as a clear oil. ¹H NMR (500 MHz, CDCl₃) δ 7.39–7.30 (m, 15H, Ph), 5.56 (d, J = 3.7 Hz, 1H, H-1'), 5.23 (d, J = 3.3Hz, 1H, H-1"), 4.89 (d, J = 8.7 Hz, 3H, 3×Bn), 4.78 (d, J = 11.6 Hz, 1H, Bn), 4.75–4.64 (m, 2H, Bn H-6'), 4.63-4.50 (m, 2H, Bn H-6'), 4.20-4.09 (m, 1H, H-5'), 4.03 (dd, J =10.3, 9.0 Hz, 1H, H-3'), 3.92 (d, J = 12.2 Hz, 1H, H-5"), 3.74–3.64 (m, 3H, H-2", H-5, H-4'), 3.57–3.47 (m, 2H, H-5", H-1), 3.47–3.39 (m, 2H, H-2', H-4), 3.39–3.27 (m, 2H, H-3, H-6), 2.72 (d, J = 10.3 Hz, 1H, H-3"), 2.56 (s, 3H, NMe), 2.34 (dt, J = 13.2, 4.5 Hz, 1H, H-2), 1.52 (q, J = 12.7 Hz, 1H, H-2), 1.21 (s, 3H, Me). ¹³C NMR (125 MHz, CDCl₃) δ 137.80 (Ph), 137.78 (Ph), 137.74 (Ph), 128.72 (Ph), 128.63 (Ph), 128.59 (Ph), 128.28 (Ph), 128.23 (Ph), 128.14 (Ph), 128.08 (Ph), 128.04 (Ph), 127.90 (Ph), 98.76 (C-1'), 97.27 (C-1''), 84.42 (C-6), 81.74 (d, J = 173.3 Hz, C-6'), 80.81 (C-4), 80.21 (C-3'), 77.37 (C-4'), 77.33 (C-4"), 76.12 (C-2"), 75.71 (C-5), 75.63 (Bn), 75.32 (Bn), 72.54 (Bn), 71.04 (d, J= 18.1 Hz, C-5'), 67.07 (C-5"), 63.59 (C-2'), 63.44 (C-3"), 59.65 (C-1), 58.86 (C-3), 38.41 (C-2), 32.21 (NMe), 24.26 (Me). ¹⁹F NMR (376 MHz, Chloroform-d) δ -234.58 (td, J = 47.7, 30.8 Hz). HRMS (ESI, positive) calculated for $C_{40}H_{49}FN_{10}O_9^+ [M + Na]^+ 855.3560$, found 855.3570.

[6'-F]-gentamicin X_2 (35): Trimethylphosphine (1 M solution in THF, 1.89 mL, 1.89 mmol) was added to a solution of compound 34 (0.165 mg, 0.198 mmol) in THF (3 mL) and 0.1 M aqueous solution of NaOH (1 mL). The reaction mixture was stirred at 50 °C for 2 h, cooled to room temperature, and concentrated under reduced pressure. The 150

residue was dissolved in a mixture of acetic acid (2 mL), water (6 mL) and methanol (6 mL). The solution was degassed by evacuating the air inside of the flask and refilling it with nitrogen five times. Palladium hydroxide on carbon (20%, Degussa type, 0.1 g) was added and the solution was charged with hydrogen gas. The reaction mixture was stirred at room temperature under hydrogen gas for 24 h. The solution was filtered through a Celite pad, washed with water and methanol, and concentrated under reduced pressure. The residue was purified by chromatography on Amberlite CG-50 resin (NH_4^+ form) (5%) concentrated ammonium hydroxide in water), concentrated under reduced pressure, acidified with hydrochloric acid (pH ca. 4), and lyophilized to provide compound 35 (0.080 g, 84%) as an oily solid. ¹H NMR (500 MHz, D₂O) δ 5.79 (d, J = 4.0 Hz, 1H, H-1'), 5.14 (d, J = 3.7 Hz, 1H, H-1"), 4.85–4.82 (m, 1H, H-6'), 4.74–4.67 (m, 1H, H-6'), 4.25 (dd, J = 10.9, 3.7 Hz, 1H, H-2"), 4.11–4.01 (m, 2H, H-5', H-5"), 4.00–3.95 (m, 2H, H-3', H-4), 3.90 (t, J = 9.0 Hz, 1H, H-5), 3.83 (dd, J = 10.2, 8.9 Hz, 1H, H-6), 3.69–3.59 (m, 3H, H-1, H-3, H-4'), 3.56 (d, J = 10.9 Hz, 1H, H-3"), 3.54 (d, J = 12.8 Hz, 1H, H-5"), 3.51 (dd, J = 10.9, 4.0 Hz, 1H, H-2'), 2.94 (s, 3H, NMe), 2.58 (dt, J = 12.6, 4.2 Hz, 1H, 1)H-2), 1.96 (q, J = 12.6 Hz, 1H, H-2), 1.38 (s, 3H, Me). ¹³C NMR (125 MHz, D₂O) δ 101.07 (C-1"), 96.85 (C-1'), 83.43 (C-6), 82.21 (d, J = 167.3 Hz, C-6'), 79.52 (C-4), 73.84 (C-5), 72.22 (d, J = 17.6 Hz, C-5'), 69.85 (C-4"), 68.77 (C-3'), 68.19 (d, J = 7.2 Hz, C-4'), 67.83 (C-5"), 66.17 (C-2"), 63.19 (C-3"), 53.70 (C-2'), 49.46 (C-1), 48.37 (C-3), 34.48 (NMe), 27.80 (C-2), 20.90 (Me). ¹⁹F NMR (565 MHz, D₂O) δ -233.97 (td, J = 47.1, 25.3 Hz). HRMS (ESI, positive) calculated for $C_{19}H_{37}FN_4O_9^+$ $[M + H]^+$ 485.2617, found 485.2628.

3.2.7. Synthesis of [6',6' 6'-²H, ²H,F]-Gentamicin X₂ (43)

The overall synthetic scheme is shown in Scheme 3-4.



Scheme 3-4. Synthetic scheme for the preparation of $[6', 6', 6'-{}^{2}H, {}^{2}H, F]$ -GenX₂ (43).

(2S,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-(((4-methoxybenzyl)oxy)methyl-d2)-2-(phenylthio)tetrahydro-2*H*-pyran (36): Sodium hydride (60% dispersion in mineral oil, 0.125 g, 2.97 mmol) and 4-methoxybenzyl chloride (0.47 mL, 3.43 mmol) were added to a solution of compound **10** (1.096 g, 2.29 mmol) in dimethylformamide (20 mL) at 0 °C. The reaction mixture was stirred at room temperature for 12 h, and concentrated. The residue was diluted with dichloromethane (80 mL) and washed with water (60 mL). The aqueous layer was extracted with dichloromethane (50 mL × 3). The combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The crude product was purified by flash column chromatography on silica gel (Hex/EtOAc = 10:1) to give compound **36** (1.109 g, 81%) as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 7.68–7.63 (m, 2H, Ph), 7.40–7.28 (m, 13H, Ph), 7.25–7.21 (m, 2H, Ph), 6.94–6.88 (m, 2H, Ph), 4.93–4.85 (m, 2H, PMB), 4.82 (d, *J* = 10.9 Hz, 1H, Bn), 4.61 (d, *J* = 11.3 Hz, 2H, Bn), 4.51 (d, *J* = 11.6 Hz, 1H), Bn, 4.46 (d, *J* = 10.1 Hz, 1H, H-1), 3.83 (s, 3H, -OMe), 3.64 (dd, *J* = 9.7, 8.9 Hz, 1H, H-4), 3.56 (d, *J* = 9.2 Hz, 1H), H-3, 3.49 (d, *J* = 9.6 Hz, 1H, H-5), 3.40 (dd, *J* = 10.1, 9.2 Hz, 1H, H-2). ¹³C NMR (100 MHz, CDCl₃) δ 159.29, 137.97, 137.73, 133.64, 131.39, 130.36, 129.37, 129.07, 128.58, 128.52, 128.40, 128.26, 128.07, 127.94, 127.90, 113.86, 86.09, 85.14, 79.35, 77.63, 75.92, 75.07, 73.12, 65.19, 55.34. HRMS (ESI, positive) calculated for C₁₄H₃₃D₂N₃O₅S⁺[*M* + *Na*]⁺622.2315, found 622.2317.

(1S,2S,3R,4S,6R)-4,6-diazido-3-(((2S,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-(((4methoxybenzyl)oxy)methyl-d2)tetrahydro-2H-pyran-2-yl)oxy)cyclohexane-1,2-diyl diacetate (37): Compound 36 (1.040 g, 1.73 mmol) and compound 14 (0.431 g, 1.44 mmol) were co-evaporated from dry toluene three times and further dried under high vacuum overnight. A solution of compound 36 and 14 in anhydrous diethyl ether (18 mL) and anhydrous dichloromethane (6 mL) was then added to preactivated 4 Å molecular sieves. After stirring for 30 min at room temperature, the mixture was cooled to -40 °C. *N*-Iodosuccinimide (0.845 g, 3.75 mmol) was added, and the reaction mixture was stirred for 30 min. Trifluoromethanesulfonic acid (0.08 mL, 0.87 mmol) was slowly added, and the reaction was warmed to -20 °C and kept stirring for 30 min. The reaction solution was diluted with ether and filtered. The residue was washed with 10% aqueous solution

of sodium bisulfite (20 mL) and saturated aqueous solution of sodium bicarbonate (20 mL). The combined aqueous solutions were extracted with dichloromethane (40 mL \times 3). The combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:1) to provide compound **37** (0.682 g, 60%) as a clear oil. ¹H NMR (500 MHz, CDCl₃) δ 7.41–7.23 (m, 12H, Ph), 7.15–7.13 (m, 2H, Ph), 6.86-6.84 (m, 2H, Ph), 5.18-5.11 (m, 2H, H-5, H-1'), 4.94 (t, J = 10.0 Hz, 1H, H-6), 4.89 (d, J = 10.5 Hz, 1H, PMB), 4.83 (d, J = 10.7 Hz, 1H, PMB), 4.77 (d, J = 10.9 Hz, 1H, Bn), 4.61 (d, J = 11.8 Hz, 1H, Bn), 4.49 (d, J = 10.9 Hz, 1H, Bn), 4.42 (d, J = 11.8Hz, 1H, Bn), 4.15 (d, J = 10.1 Hz, 1H, H-5'), 3.96 (dd, J = 10.5, 8.9 Hz, 1H, H-3'), 3.84– 3.72 (m, 4H, -OCH', H-4'), 3.68–3.56 (m, 2H, H-1, H-4), 3.46–3.33 (m, 2H, H-3, H-2'), 2.40 (dt, J = 13.3, 4.5 Hz, 1H, H-2), 2.09 (d, J = 3.6 Hz, 6H, OAc), 1.60 (q, J = 12.6 Hz, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 169.86, 169.52, 159.31, 137.92, 137.80, 129.82, 129.67, 128.48, 128.47, 128.45, 128.39, 128.38, 127.99, 127.88, 127.77, 127.68, 127.63, 113.80, 99.40, 79.56, 78.28, 78.08, 75.33, 74.93, 74.11, 73.54, 73.15, 71.55, 63.12, 58.68, 57.71, 55.21, 31.82, 20.66, 20.60. HRMS (ESI, positive) calculated for $C_{38}H_{41}D_2N_9O_{10}^+$ [*M* + *Na*]⁺ 810.3151, found 810.3144.

(1S,2S,3R,4S,6R)-4,6-diazido-3-(((2S,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-(hydroxymethyl-d2)tetrahydro-2H-pyran-2-yl)oxy)cyclohexane-1,2-diyl diacetate (38): 2,3-Dichloro-5,6-dicyano-1,4-benzoquinone (0.250 g, 1.10 mmol) was added to a solution of compound 37 (0.5784 g, 0.734 mmol) in dichloromethane (20 mL) and water (2 mL). The reaction mixture was stirred for 4 h at room temperature and quenched by adding saturated aqueous solution of sodium bicarbonate (20 mL). The aqueous layer was extracted with dichloromethane (30 mL \times 3), and the combined organic phase was washed with brine, dried over magnesium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:2) to give compound **38** (0.367 g, 75%) as a clear liquid. ¹H NMR (500 MHz, CDCl₃) δ 7.37–7.28 (m, 10H, Ph), 5.14 (t, J = 9.8 Hz, 1H, H-5), 5.08 (d, J = 3.9 Hz, 1H, H-1'), 4.92 (dd, J = 14.0, 10.3 Hz, 2H, H-6, Bn), 4.89–4.82 (m, 2H, Bn), 4.70 (d, J = 11.1 Hz, 1H, Bn), 4.06 (dd, J = 10.1, 0.5 Hz, 1H, H-5'), 3.98 (dd, J = 10.4, 8.9 Hz, 1H, H-3'), 3.68–3.57 (m, 3H, H-4', H-1, H-4), 3.42 (ddd, J = 12.4, 9.9, 4.6 Hz, 1H, H-3), 3.32 (dd, J = 10.4, 3.8 Hz, 1H, H-2'), 2.40 (dt, J = 13.4, 4.6 Hz, 1H, H-2), 2.09 (s, 6H, OAc), 1.59 (dt, J = 13.5, 12.5 Hz, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 170.02, 169.65, 137.85, 137.80, 128.70, 128.63, 128.17, 128.16, 128.08, 128.06, 99.58, 79.67, 78.81, 77.83, 75.55, 75.21, 74.23, 73.58, 72.30, 63.41, 58.90, 57.80, 32.02, 20.80, 20.72. HRMS (ESI, positive) calculated for C₃₀H₃₃D₂N₉O₉⁺ [M + Na]⁺ 690.2575, found 690.2578.

(*1S*,*2S*,*3R*,*4S*,*6R*)-4,6-diazido-3-(((*2S*,*3R*,*4R*,*5S*,*6S*)-3-azido-4,5-bis(benzyloxy)-6-(fluoromethyl-d2)tetrahydro-2*H*-pyran-2-yl)oxy)cyclohexane-1,2-diyl diacetate (39): Diethylaminosulfur trifluoride (DAST) (0.104 mL, 0.786 mmol) was added to a solution of compound **38** (0.262 g, 0.393 mmol) in dichloromethane (5 mL) at 0 °C. The reaction mixture was stirred at 0 °C for 30 min and at room temperature for 12 h. The reaction was quenched by adding saturated aqueous solution of sodium bicarbonate (10 mL). The aqueous layer was extracted with dichloromethane (20 mL × 3). The combined organic layers were washed with brine, dried over magnesium sulfate, filtered, and concentrated. The crude product was purified by flash column chromatography on silica gel (Hex/EtOAc = 4:1) to provide compound **39** (0.171 g, 65%) as a clear liquid. ¹H NMR (500 MHz, CDCl₃) δ 7.39–7.28 (m, 10H, Bn), 5.19–5.09 (m, 2H, H-1', H-5), 4.98–4.82 (m, 4H, H-6, 3×Bn), 4.67 (d, *J* = 11.0 Hz, 1H, Bn), 4.16 (dd, *J* = 31.2, 10.2 Hz, 1H, H-5'), 4.02–3.94 (m, 1H, H-3'), 3.72–3.58 (m, 3H, H-4', H-4, H-1), 3.41 (ddd, *J* = 12.4, 10.0, 155

4.5 Hz, 1H, H-2'), 3.35 (dd, J = 10.4, 3.8 Hz, 1H, H-3), 2.40 (dt, J = 13.4, 4.6 Hz, 1H, H-2), 2.09 (d, J = 0.8 Hz, 6H, OAc), 1.61 (q, J = 12.7 Hz, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 169.96 (Ac), 169.63 (Ac), 137.70 (Ph), 137.65 (Ph), 128.69 (Ph), 128.63 (Ph), 128.17 (Ph), 128.12 (Ph), 128.11 (Ph), 127.93 (Ph), 99.47 (C-1'), 79.61 (C-3'), 78.60 (C-4), 77.48 (C-4'), 75.56 (Bn), 75.40 (Bn), 74.19 (C-6), 73.55 (C-5), 71.20 (d, J = 18.2 Hz, C-5'), 63.19 (C-2'), 58.72 (C-3), 57.79 (C-1), 31.86 (C-2), 20.77 (Ac), 20.70 (Ac). ¹⁹F NMR (375 MHz, CDCl₃) δ -236.30 (d, J = 31.3 Hz). HRMS (ESI, positive) calculated for C₃₀H₃₂D₂FN₉O₈⁺ [M + Na]⁺ 692.2532, found 692.2525.

(1S,2R,3R,4S,6R)-4,6-diazido-3-(((2S,3R,4R,5S,6S)-3-azido-4,5-bis(benzyloxy)-6-(fluoromethyl-d2)tetrahydro-2H-pyran-2-yl)oxy)cyclohexane-1,2-diol (40): Sodium methoxide (4.37 M in methanol, 0.004 mL, 0.019 mmol) was added to a solution of compound **39** (0.043 g, 0.065 mmol) in tetrahydrofuran (0.5 mL) and methanol (4.5 mL). The reaction mixture was stirred at room temperature for 5 h, neutralized with Amberlite IR-120 (H⁺ form) resin, filtered, and concentrated under reduced pressure. The crude product was purified by flash column chromatography on silica gel (Hex/EtOAc = 2:1) to give compound 40 (0.033 g, 85%) as a clear oil. ¹H NMR (500 MHz, CDCl₃) δ 7.41–7.29 (m, 10H, Bn), 5.19 (d, J = 3.7 Hz, 1H, H-1'), 4.95 (d, J = 10.6 Hz, 1H, Bn), 4.92–4.85 (m, 2H, Bn), 4.69 (d, J = 10.9 Hz, 1H, Bn), 4.17 (s, 1H, OH), 4.13–3.98 (m, 2H, H-5', H-3'), 3.71 (dd, J = 10.2, 9.0 Hz, 1H, H-4'), 3.63 (dd, J = 10.2, 3.6 Hz, 1H, H-2'), 3.55–3.36 (m, 3H, H-4, H-5, H-1), 3.35-3.22 (m, 2H), H-6, H-3, 3.17 (s, 1H, OH), 2.31 (dt, J = 12.9, 4.0 Hz, 1H, H-2), 1.57–1.42 (m, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 137.58 (Ph), 137.48 (Ph), 128.71 (Ph), 128.65 (Ph), 128.20 (PH), 128.19 (Ph), 127.92 (Ph), 99.73 (C-1'), 83.97 (C-6), 80.90 (C-3'), 77.45 (C-4'), 76.09 (Bn), 75.84 (C-4), 75.42 (Bn), 75.40 (C-5), 71.21 (d, J = 18.1 Hz, C-5'), 64.24 (C-2'), 59.78 (C-1), 58.90 (C-3), 32.10 (C-2).

¹⁹F NMR (375 MHz, CDCl₃) δ -235.84 (d, J = 31.1 Hz). HRMS (ESI, positive) calculated for C₂₆H₂₈D₂FN₉O₆⁺ [M + Na]⁺ 608.2321, found 608.2313.

(*3aR*, *6R*, *7R*, *7aR*)-7-(benzyloxy)-6-(((*1S*, *2S*, *3R*, *4S*, *6R*)-4, 6-diazido-3-(((2S, 3R, 4R, 5S, 6-S)-3-azido-4, 5-bis(benzyloxy)-6-(fluoromethyl-d2)tetrahydro-2H-pyran-2-yl)oxy)-2-hydroxycyclohexyl)oxy)-1, 3a-dimethylhexahydro-2H-pyrano[4, 3-d]oxazol-2-one

(41): Compound 40 (0.418 g, 0.714 mmol) and compound 17 (0.342 g, 0.856 mmol) were co-evaporated from dry toluene three times and further dried under high vacuum overnight. A solution of compound 40 and 17 in anhydrous diethyl ether (10 mL) and anhydrous dichloromethane (2 mL) was added to a preactivated 4 Å molecular sieves and *N*-iodosuccinimide (0.321 g, 1.427 mmol). After stirring 30 min at room temperature, the mixture was cooled to -40 °C. Silver trifluoromethanesulfonate (0.367 g, 1.427 mmol) was added, and the reaction was warmed to -10 °C to stand for 12 h. The reaction mixture was diluted with ethyl acetate (20 mL) and filtered through a Celite pad. The solution was washed with 10% aqueous solution of sodium bisulfite (20 mL) and saturated aqueous solution of sodium bicarbonate (20 mL). The combined aqueous layers were extracted with ethyl acetate (30 mL \times 3). The combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:2) to provide compound 41 (0.387 g, 63%) as a clear liquid. ¹H NMR (500 MHz, $CDCl_3$) δ 7.40–7.29 (m, 15H, Ph), 5.60 (d, J = 3.7 Hz, 1H, H-1'), 5.03 (d, J = 3.2 Hz, 1H, H-1"), 4.92–4.86 (m, 4H, 4×Bn), 4.68 (d, J = 11.1 Hz, 1H, Bn), 4.61 (d, J = 11.6 Hz, 1H, Bn), 4.22-4.17 (m, 2H, H-5", OH), 4.13 (dd, J = 30.6, 10.2 Hz, 1H, H-5'), 4.07-4.00 (m, 1H, H-3'), 3.95 (dd, J = 4.4, 3.2 Hz, 1H, H-2"), 3.80 (d, J = 12.4 Hz, 1H, H-5"), 3.73–3.63 (m, 2H, H-4', H-5), 3.53-3.38 (m, 5H, H-1, H-3", H-4, H-2', H-6), 3.35 (ddd, J = 12.4, 10.0, 4.4 Hz, 1H, H-3), 2.83 (s, 3H, NMe), 2.33 (dt, J = 13.2, 4.5 Hz, 1H. H-2), 1.51 (q, J = 12.6 Hz, 1H, H-2), 1.41 (s, 3H, Me). ¹³C NMR (125 MHz, CDCl₃) δ 157.28 (N(C(=O)O), 137.85 (Ph), 137.82 (Ph), 137.41 (Ph), 128.73 (Ph), 128.64 (Ph), 128.61 (Ph), 128.39 (Ph), 128.15 (Ph), 128.08 (Ph), 128.05 (Ph), 128.02 (Ph), 128.01 (Ph), 127.91 (Ph), 98.54 (C-1'), 97.00 (C-1"), 84.60 (C-6), 80.50 (C-4), 80.15 (C-3'), 77.34 (C-4'), 77.22 (C-4"), 75.63 (Bn), 75.32 (Bn), 75.12 (C-5), 74.62 (Bn), 73.41 (C-2"), 70.93 (d, J = 18.2 Hz, C-5'), 66.87 (C-5"), 63.57 (C-2"), 63.03 (C-3"), 59.78 (C-1), 58.76 (C-3), 32.08 (C-2), 30.32 (NMe), 23.63 (Me). HRMS (ESI, positive) calculated for C₄₁H₄₅D₂FN₁₀O₁₀⁺ [M + H]⁺ 883.3478, found 883.3499.

(3R,4R,5R,6R)-5-(benzyloxy)-6-(((1S,2S,3R,4S,6R)-4,6-diazido-3-(((2S,3R,4R,5S,6S)-3-azido-4,5-bis(benzyloxy)-6-(fluoromethyl-d2)tetrahydro-2H-pyran-2-yl)oxy)-2-hydroxycyclohexyl)oxy)-3-methyl-4-(methylamino)tetrahydro-2H-pyran-3-ol (42): A 2 N aqueous solution of sodium hydroxide (5 mL) was added to a solution of compound 41 (0.303 g, 0.352 mmol) in ethanol (15 mL). The reaction mixture was stirred at 90 °C for 10 h, and cooled to room temperature. The solution was diluted with ethyl acetate (15 mL) and washed with water (15 mL). The residue was washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The crude product was purified by flash column chromatography on silica gel (5-6% methanol in dichloromethane) to provide compound 42 (0.221 g, 75%) as a clear oil. ¹H NMR (500 MHz, CDCl₃) δ 7.40–7.30 (m, 15H, Ph), 5.55 (d, J = 3.7 Hz, 1H, H-1'), 5.22 (d, J = 3.3 11.0 Hz, 1H, Bn), 4.51 (d, J = 11.7 Hz, 1H, Bn), 4.13 (dd, J = 30.7, 10.2 Hz, 1H, H-5'), 4.02 (dd, J = 10.2, 9.0 Hz, 1H, H-3'), 3.91 (d, J = 12.3 Hz, 1H, H-5"), 3.72–3.58 (m, 3H, H-5, H-2", H-4), 3.56–3.48 (m, 2H, H-5", H-1), 3.47–3.39 (m, 2H, H-2', H-4), 3.37–3.26 (m, 2H, H-3, H-6), 2.68 (d, J = 10.3 Hz, 1H, H-3"), 2.53 (s, 3H, NMe), 2.34 (dt, J = 13.2, 4.5 Hz, 1H, H-2), 1.52 (q, J = 12.7 Hz, 1H, H-2), 1.18 (s, 3H, Me). ¹³C NMR (125 MHz, 158

CDCl₃) δ 137.82 (Ph), 137.80 (Ph), 137.79 (Ph), 128.76 (Ph), 128.66 (Ph), 128.62 (Ph), 128.31 (Ph), 128.22 (Ph), 128.16 (Ph), 128.11 (Ph), 128.07 (Ph), 127.92 (Ph), 98.78 (C-1'), 97.30 (C-1''), 84.38 (C-6), 80.87 (C-4), 80.24 (C-3'), 77.34 (C-4'), 76.14 (C-2''), 75.71 (C-5), 75.67 (Bn), 75.34 (Bn), 72.54 (Bn), 70.87 (d, *J* = 18.3, C-5'), 66.80 (C-5''), 63.63 (C-2'), 63.30 (C-3''), 59.69 (C-1), 58.88 (C-3), 38.66 (C-2), 32.26 (NMe), 24.53 (Me). ¹⁹F NMR (375 MHz, CDCl₃) δ -235.80 (d, *J* = 30.7 Hz). HRMS (ESI, positive) calculated for C₄₀H₄₇D₂FN₁₀O₉⁺ [*M* + *H*]⁺ 837.3686, found 857.3686.

[6',6',6'-²H, ²H,F]-gentamicin X₂ (43): Trimethylphosphine (1 M solution in THF, 0.8 mL, 0.789 mmol) was added to a solution of compound 42 (0.109 mg, 0.132 mmol) in THF (4 mL) and 0.1 M aqueous solution of NaOH (1.3 mL). The reaction mixture was stirred at 50 °C for 2 h, cooled to room temperature, and concentrated under reduced pressure. The residue was dissolved in a mixture of acetic acid (2 mL), water (6 mL) and methanol (6 mL). The solution was degassed by evacuating the air inside of the flask and refilling it with nitrogen five times. Palladium hydroxide on carbon (20%, Degussa type, 0.05 g) was added and the solution was charged with hydrogen gas. The reaction mixture was stirred at room temperature under hydrogen gas for 24 h. The solution was filtered through a Celite pad, washed with water and methanol, and concentrated under reduced pressure. The residue was purified by chromatography on Amberlite CG-50 resin (NH₄⁺ form) (5% concentrated ammonium hydroxide in water), concentrated under reduced pressure, acidified with hydrochloric acid (pH ca. 4), and lyophilized to provide compound **43** (0.049 g, 76%) as an oily solid. ¹H NMR (500 MHz, D₂O) δ 5.79 (d, J = 4.0 Hz, 1H, H-1'), 5.14 (d, J = 3.6 Hz, 1H, H-1"), 4.25 (dd, J = 10.9, 3.7 Hz, 1H, H-2"), 4.09–4.02 (m, 2H, H-5', H-5"), 4.01–3.96 (m, 2H, H-3', H-4), 3.90 (t, J = 9.1 Hz, 1H, H-5), 3.84 (dd, J = 10.2, 8.9 Hz, 1H, H-6), 3.69–3.59 (m, 3H, H-1, H-3, H-4'), 3.56 (d, J =11.0 ppm, 1H, H-3"), 3.55 (d, J = 12.9 Hz, 1H, H-5"), 3.51 (dd, J = 10.9, 4.1 Hz, 1H, H-159

2'), 2.95 (s, 3H, NMe), 2.59 (dt, J = 12.5, 4.2 Hz, 1H, H-2), 1.97 (q, J = 12.6 Hz, 1H, H-2), 1.38 (d, J = 1.5 Hz, 3H, Me). ¹³C NMR (125 MHz, D₂O) δ 101.08 (C-1"), 96.85 (C-1'), 83.44 (C-6), 79.52 (C-4), 73.84 (C-5), 72.10 (d, J = 17.8 Hz, C-5'), 69.85 (C-4"), 68.77 (C-3'), 68.17 (d, J = 7.4 Hz, C-4'), 66.17 (C-2"), 63.19 (C-3"), 53.69 (C-2'), 49.46 (C-1), 48.36 (C-3), 34.47 (NMe), 27.81 (C-2), 20.89 (Me). ¹⁹F NMR (565 MHz, D₂O) δ - 234.65 (d, J = 25.0 Hz). HRMS (ESI, positive) calculated for C₁₉H₃₅D₂FN₄O₉⁺ [M + H]⁺ 487.2743, found 487.2742.

3.2.8. Synthesis of [6'-OMe]-Gentamicin X₂ (49)

The overall synthetic scheme is shown in Scheme 3-5.



Scheme 3-5. Synthetic scheme for the preparation of [6'-OMe]-GenX₂ (49).

(2S, 3R, 4R, 5S, 6R)-3-azido-4,5-bis(benzyloxy)-6-(methoxymethyl)-2-(phenylthio)-tetrahydro-2H-pyran (44): Sodium hydride (60% dispersion in mineral oil, 0.267 g, 6.34 mmol) was added to a solution of compound 8 (1.515 g, 3.17 mmol) in dimethylformamide (30 mL) at 0 °C. The reaction mixture was stirred at room temperature for 1 h. Iodomethane (0..40 mL, 6.34 mmol) was added at 0 °C and the reaction mixture was stirred at room temperature for 12 h. At 0 °C, methanol (10 mL) was added slowly and the reaction mixture was stirred at room temperature for additional 30 min. After being concentrated under reduced pressure, the mixture was diluted with dichloromethane (70 mL) and water (70 mL). The aqueous layer was extracted with dichloromethane (60 mL \times 3). The combined organic layers were washed with brine, dried over magnesium sulfate, filtered, and concentrated under reduced pressure. The crude product was purified by flash column chromatography on silica gel (Hex/EtOAc = 4:1) to give compound 44 (1.17 g, 75%) as a clear liquid. ¹H NMR (500 MHz, CDCl₃) δ 7.75–7.70 (m, 2H, Ph), 7.48–7.36 (m, 13H, Ph), 5.00–4.89 (m, 3H, $3 \times Bn$), 4.75 (d, J =11.0 Hz, 1H, Bn), 4.53 (d, J = 10.2 Hz, 1H, H-1), 3.75 (d, J = 3.1 Hz, 2H, H-6), 3.71 (t, J = 9.4 Hz, 1H, H-4), 3.62 (t, J = 9.1 Hz, 1H, H-3), 3.54–3.45 (m, 5H, H-5, OMe, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 137.89 (Ph), 137.60 (Ph), 133.25 (Ph), 131.51 (Ph), 128.90 (Ph), 128.40 (Ph), 128.39 (Ph), 128.16 (Ph), 128.06 (Ph), 127.87 (Ph), 127.80 (Ph), 127.72 (Ph), 86.17 (C-1), 84.93(C-3), 79.16 (C-5), 77.37 (C-4), 75.68 (Bn), 74.87 (Bn), 70.96 (C-6), 65.11 (C-2), 59.30 (OMe). HRMS (ESI, positive) calculated for $C_{27}H_{29}N_{3}O_{4}S^{+}[M+Na]^{+}514.1771$, found 514.1784.

(1S,2S,3R,4S,6R)-4,6-diazido-3-(((2S,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-(methoxymethyl)tetrahydro-2*H*-pyran-2-yl)oxy)cyclohexane-1,2-diyl diacetate (45): Compound 44 (1.413 g, 2.83 mmol) and compound 14 (0.714 g, 2.39 mmol) were coevaporated from dry toluene three times and further dried under high vacuum overnight. A solution of compound 44 and 14 in anhydrous diethyl ether (30 mL) and anhydrous dichloromethane (10 mL) was then added to preactivated 4 Å molecular sieves. After stirring for 30 min at room temperature, the mixture was cooled to -40 °C. N-Iodosuccinimide (1.401 g, 6.23 mmol) was added, and the reaction mixture was stirred for 30 min. Trifluoromethanesulfonic acid (0.11 mL, 0.120 mmol) was slowly added, and the reaction was warmed to -20 °C and kept stirring for 30 min. The reaction solution was diluted with diethyl ether (40 mL) and filtered. The residue was washed with 10% aqueous solution of sodium bisulfite (50 mL) and saturated aqueous solution of sodium bicarbonate (50 mL). The combined aqueous solutions were extracted with dichloromethane (60 mL \times 3). The combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:1) to provide compound **45** (1.009 g, 62%) as a clear oil. ¹H NMR (500 MHz, CDCl₃) δ 7.38–7.27 (m, 10, Ph), 5.18-5.10 (m, 2H, H-5, H-1'), 4.98-4.88 (m, 2H, Bn, H-6), 4.88-4.80 (m, 2H, $2 \times Bn$, 4.66 (d, J = 11.0 Hz, 1H, Bn), 4.13 (dt, J = 10.2, 2.3 Hz, 1H, H-5'), 3.96 (dd, J = 10.210.4, 8.9 Hz, 1H, H-3'), 3.79-3.69 (m, 2H, H-4', H-6'), 3.67-3.55 (m, 3H, H-4, H-1, H-6'), 3.46-3.32 (m, 5H, H-3, OMe, H-2'), 2.39 (dt, J = 13.4, 4.6 Hz, 1H, H-2), 2.09 (s, 3H, OAc), 2.08 (s, 3H, OAc), 1.60 (q, J = 12.8 Hz, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 169.88 (OAc), 169.56 (OAc), 138.02 (Ph), 137.84 (Ph), 128.55 (Ph), 128.50 (Ph), 128.03 (Ph), 127.93 (Ph), 127.92 (Ph), 127.84 (Ph), 99.51 (C-1'), 79.56 (C-3'), 78.45 (C-4), 78.08 (C-4'), 75.35 (Bn), 75.09 (Bn), 74.13 (C-6), 73.55 (C-5), 71.57 (C-5'), 70.36 (C-6'), 63.15
(C-2'), 59.26 (OMe), 58.70 (C-3), 57.73 (C-1), 31.82 (C-2), 20.67(OAc), 20.62 (OAc). HRMS (ESI, positive) calculated for $C_{31}H_{37}N_9O_9^+$ $[M + Na]^+$ 702.2606, found 702.2604. (1S,2R,3R,4S,6R)-4.6-diazido-3-(((2S,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-(methoxymethyl)tetrahydro-2H-pyran-2-yl)oxy)cyclohexane-1,2-diol (46): Sodium methoxide (4.37 M in methanol, 0.04 mL, 0.086 mmol) was added to a solution of compound 45 (0.585 g, 0.861 mmol) in methanol (20 mL). The reaction mixture was stirred at room temperature for 5 h, neutralized with Amberlite IR-120 (H⁺ form) resin. filtered, and concentrated under reduced pressure. The crude product was purified by flash column chromatography on silica gel (Hex/EtOAc = 1:1) to give compound 46 (0.468 g, 91%) as a clear oil. ¹H NMR (500 MHz, CDCl₃) δ 7.37–7.29 (m, 10H, Ph), 5.16 (d, J = 3.7 Hz, 1H, H-1'), 4.94 (d, J = 10.6 Hz, 1H, Bn), 4.86 (d, J = 10.6 Hz, 1H, Bn),4.85 (d, J = 11.0 Hz, 1H, Bn), 4.66 (d, J = 11.0 Hz, 1H, Bn), 4.25 (d, J = 2.3 Hz, 1H, OH), 4.07 (ddd, J = 10.1, 2.9, 1.9 Hz, 1H, H-5'), 3.99 (dd, J = 10.2, 9.0 Hz, 1H, H-3', H-3'), 3.75 (dd, J = 10.1, 9.0 Hz, 1H, H-4'), 3.71 (dd, J = 10.7, 3.2 Hz, 1H, H-6'), 3.63 (dd, J = 10.2, 3.7 Hz, 1H, H-2', H-2'), 3.59 (dd, J = 10.7, 2.0 Hz, 1H, H-6'), 3.52–3.38 (m, 3H, H-5, H-6, H-1), 3.37 (s, 3H, OMe), 3.28–3.25 (m, 3H, H-3, H-4, OH), 2.33–2.25 (m, 1H, H-2), 1.56 – 1.43 (m, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 137.98 (Ph), 137.67 (Ph), 128.63 (Ph), 128.57 (Ph), 128.16 (Ph), 128.05 (Ph), 128.02 (Ph), 127.89 (Ph), 99.74 (C-1'), 83.96 (C-4), 80.98 (C-3'), 78.15 (C-4'), 76.05 (C-5), 75.70 (Bn), 75.42 (C-6), 75.17 (Bn), 71.60 (C-5'), 70.56 (C-6'), 64.31 (C-2'), 59.75 (C-1), 59.32 (OMe), 58.92 (C-3), 32.18 (C-2). HRMS (ESI, positive) calculated for $C_{27}H_{33}N_9O_7^+$ $[M + Na]^+$ 618.2395, found 618.2396.

(*3aR*, *6R*, *7R*, *7aR*)-7-(benzyloxy)-6-(((*1S*, *2S*, *3R*, *4S*, *6R*)-4,6-diazido-3-(((*2S*, *3R*, *4R*, *5S*, *6-R*)-3-azido-4,5-bis(benzyloxy)-6-(methoxymethyl)tetrahydro-2*H*-pyran-2-yl)oxy)-2hydroxycyclohexyl)oxy)-1,3a-dimethylhexahydro-2*H*-pyrano[4,3-d]oxazol-2-one

(47): Compound 46 (0.432 g, 0.726 mmol) and compound 17 (0.319 g, 0.798 mmol) were co-evaporated from dry toluene three times and further dried under high vacuum overnight. A solution of compound 46 and 17 in anhydrous dichloromethane (10 mL) was added to a preactivated 4 Å molecular sieves and N-iodosuccinimide (0.261 g, 1.161 mmol). After stirring 30 min at room temperature, the mixture was cooled to -78 °C. Trifluoromethanesulfonic acid (0.02 mL, 0.217 mmol) was added slowly, and the reaction was warmed to 0 °C during 1 h. Sodium sulfite (0.1 g), sodium bicarbonate (0.1 g), and few drops of water were added to the reaction solution. The reaction mixture was diluted with dichloromethane (20 mL) and filtered through a Celite pad. The solution was washed with saturated aqueous solution of sodium bicarbonate (30 mL). The combined aqueous layers were extracted with dichloromethane (30 mL \times 3). The combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 1:1) to provide compound 47 (0.480 g, 76%) as a clear solid. ¹H NMR (500 MHz, CDCl₃) δ 7.39–7.29 (m, 15H, Ph), 5.53 (d, J = 3.7 Hz, 1H, H-1'), 5.07 $(d, J = 3.2 \text{ Hz}, 1\text{H}, \text{H-1''}), 4.93-4.83 \text{ (m, 4H, 4×Bn)}, 4.66 \text{ (d, } J = 11.1 \text{ Hz}, 1\text{H}, \text{Bn}), 4.60 \text{ (d, } J = 11.1 \text{ Hz}, 1\text{H}, 1\text$ (d, J = 11.6 Hz, 1H, Bn), 4.21 (d, J = 12.4 Hz, 1H, H-5''), 4.17 (d, J = 1.6 Hz, 1H, OH),4.13-4.07 (m, 1H, H-5'), 4.00 (dd, J = 10.3, 9.0 Hz, 1H, H-3'), 3.93 (dd, J = 4.6, 3.2 Hz, 1H, H-2"), 3.79 (d, J = 12.6 Hz, 1H, H-5"), 3.76–3.63 (m, 3H, H-4', H-6', H-5), 3.58 (dd, J = 10.6, 2.0 Hz, 1H, H-6'), 3.53–3.39 (m, 5H, H-1, H-3", H-6, H-2', H-4), 3.40–3.30 (m, 4H, OMe, H-3), 2.84 (s, 3H, NMe), 2.33 (dt, J = 13.2, 4.5 Hz, 1H, H-2), 1.60–1.44 (m, 1H, H-2), 1.40 (s, 3H, Me). ¹³C NMR (125 MHz, CDCl₃) δ 157.30 (N(C=O)O), 138.20 (Ph), 138.00(Ph), 137.41 (Ph), 128.70 (Ph), 128.57 (Ph), 128.53 (Ph), 128.34 (Ph), 128.11 (Ph), 128.00 (Ph), 127.93 (Ph), 127.91 (Ph), 127.88 (Ph), 98.83 (C-1'), 96.77 (C-1"), 84.07 (C-6), 80.97 (C-4), 80.29 (C-3'), 78.12 (C-4'), 77.30 (C-4"), 75.51 (Bn), 75.11

(Bn), 75.05 (C-5), 74.45 (Bn), 73.70 (C-2"), 71.37 (C-5'), 70.66 (C-6'), 66.59 (C-5"), 63.71 (C-3"), 62.98 (C-2'), 59.86 (C-1), 59.32 (OMe), 58.77 (C-3), 32.11 (C-2), 30.33 (NMe), 23.56 (Me). HRMS (ESI, positive) calculated for $C_{42}H_{50}N_{10}O_{11}^{+}$ [M + Na]⁺ 893.3553, found 893.3556.

(3R,4R,5R,6R)-5-(benzyloxy)-6-(((1S,2S,3R,4S,6R)-4,6-diazido-3-(((2S,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-(methoxymethyl)tetrahydro-2H-pyran-2-yl)oxy)-2-hydroxycyclohexyl)oxy)-3-methyl-4-(methylamino)tetrahydro-2*H*-pyran-3-ol (48): A 2 N aqueous solution of sodium hydroxide (8 mL) was added to a solution of compound 47 (0.431 g, 0.495 mmol) in ethanol (20 mL). The reaction mixture was stirred at 90 °C for 10 h, and cooled to room temperature. The solution was diluted with ethyl acetate (20 mL) and washed with water (20 mL). The residue was washed with brine, dried over magnesium sulfate, filtered, and concentrated under reduced pressure. The crude product was purified by flash column chromatography on silica gel (5% methanol in dichloromethane) to provide compound **48** (0.355 g, 85%) as a white solid. ¹H NMR (500 MHz, CDCl₃) δ 7.39–7.29 (m, 15H, Ph), 5.48 (d, J = 3.8 Hz, 1H, H-1'), 5.25 (d, J = 3.4 Hz, 1H, H-1"), 4.90 (d, J = 10.7 Hz, 1H, Bn), 4.86 (d, J = 10.8 Hz, 1H, Bn), 4.85 (d, J = 10.8 Hz, 1H, Bn), 4.85 (d, J = 10.7 Hz, 1H, Bn), 4.85 (d, J = 10.8 Hz, 11.1 Hz, 1H, Bn) 4.80 (d, J = 11.7 Hz, 1H, Bn), 4.65 (d, J = 11.1 Hz, 1H), 4.49 (d, J = 11.1 11.8 Hz, 1H, Bn), 4.45–4.39 (m, 1H, OH), 4.10 (dt, J = 10.1, 2.5 Hz, 1H, H-5'), 3.99 (dd, J = 10.3, 9.0 Hz, 1H, H-3'), 3.94 (d, J = 12.3 Hz, 1H, H-5"), 3.77–3.71 (m, 1H, H-4'), 3.71-3.63 (m, 2H, H-6', H-5), 3.60-3.49 (m, 4H, H-2", H-6', H-5", H-1), 3.46 (dd, J =10.3, 3.8 Hz, 1H, H-2'), 3.40 (dd, J = 10.1, 8.8 Hz, 1H, H-4), 3.37–3.30 (m, 5H, OMe, H-6, H-3), 2.66 (d, J = 10.3 Hz, 1H, H-3"), 2.52 (s, 3H, NMe), 2.35 (dt, J = 13.1, 4.4 Hz, 1H, H-2), 1.53 (dt, J = 13.2, 12.4 Hz, 1H, H-2), 1.16 (s, 3H, Me). ¹³C NMR (125 MHz, CDCl₃) § 138.21 (Ph), 137.99 (Ph), 137.82 (Ph), 128.76 (Ph), 128.61 (Ph), 128.57 (Ph), 128.31 (Ph), 128.23 (Ph), 128.15 (Ph), 127.98 (Ph), 127.96 (Ph), 127.91 (Ph), 99.10 (C-165

1'), 97.11 (C-1"), 83.66 (C-6), 81.46 (C-4), 80.43 (C-3'), 78.14 (C-4'), 76.17 (C-2"), 75.58 (C-5, Bn), 75.15 (Bn), 72.47 (Bn), 71.39 (C-5'), 71.08 (C-4"), 70.69 (C-6'), 66.64 (C-5"), 63.81 (C-2'), 63.24 (C-3"), 59.84 (C-1), 59.35 (OMe), 58.90 (C-3), 38.73 (NMe), 32.32 (C-2), 24.63 (Me). HRMS (ESI, positive) calculated for $C_{41}H_{52}N_{10}O_{10}^{+}$ [M + Na]⁺ 867.3760, found 867.3764.

[6'-OMe]-Gentamicin X₂ (49): Trimethylphosphine (1 M solution in THF, 1.01 mL, 1.01 mmol) was added to a solution of compound 48 (0.143 mg, 0.169 mmol) in THF (6 mL) and 0.1 M aqueous solution of sodium hydroxide (2 mL). The reaction mixture was stirred at 50 °C for 2 h, cooled to room temperature, and concentrated under reduced pressure. The residue was dissolved in a mixture of acetic acid (2 mL), water (6 mL) and methanol (6 mL). The solution was deaerated by evacuating the air inside of the flask and refilling it with nitrogen five times. Palladium hydroxide on carbon (20%, Degussa type, 0.1 g) was added and the solution was charged with hydrogen gas. The reaction mixture was stirred at room temperature under hydrogen gas for 24 h. The solution was filtered through a Celite pad, washed with water and methanol, and concentrated under reduced pressure. The residue was purified by chromatography on Amberlite CG-50 resin (NH₄⁺ form) (5% concentrated ammonium hydroxide in water), concentrated under reduced pressure, acidified with hydrochloric acid (pH ca. 4), and lyophilized to provide compound **49** (0.068 g, 81%) as an oily solid. ¹H NMR (500 MHz, D₂O) δ 5.63 (d, J = 4.0 Hz, 1H, H-1'), 5.12 (d, J = 3.7 Hz, 1H, C-1"), 4.22 (dd, J = 10.9, 3.7 Hz, 1H, H-2"), 4.04 (d, J = 12.8 Hz, 1H, H-5"), 3.98–3.94 (m, 1H, H-5'), 3.94–3.88 (m, 1H, H-3'), 3.86– 3.82 (m, 2H, H-4, H-5), 3.82–3.75 (m, 2H, H-6', H-6), 3.66 (dd, J = 10.9, 6.6 Hz, 1H, H-6'), 3.62-3.47 (m, 5H, H-1, H-3", H-5", H-4', H-3), 3.44 (dd, J = 10.8, 4.0 Hz, 1H, H-2'), 3.41 (s, 3H, OMe), 2.93 (s, 3H, NMe), 2.50 (dt, J = 12.9, 4.3 Hz, 1H, H-2), 1.86 (q, J =12.6 Hz, 1H, H-2), 1.36 (s, 3H, Me). ¹³C NMR (125 MHz, D₂O) δ 100.90 (C-1"), 97.13

(C-1'), 83.66 (C-6), 81.34 (C-4), 73.73 (C-5), 72.22 (C-5'), 70.75 (C-6'), 69.84 (C-4"), 69.34 (C-4'), 69.13 (C-3'), 67.71 (C-5"), 66.19 (C-2"), 63.26 (C-3"), 58.72 (OMe), 53.94 (C-2'), 49.58 (C-1), 48.60 (C-3), 34.51 (NMe), 28.68 (C-2), 20.89 (Me). HRMS (ESI, positive) calculated for $C_{20}H_{40}N_4O_{10}^+$ [M + Na]⁺ 519.2637, found 519.2642.

3.3. RESULTS AND DISCUSSION

3.3.1. Proposed Mechanism

As discussed in Chapter 2, three mechanisms have been proposed for the GenK catalyzed methylation of GenX₂ (6) to form G418 (7) (Scheme 3-6). In the first mechanism (Scheme 3-6A), the 5'-dAdo• (3) is generated via reductive cleavage of SAM (1) and abstracts a hydrogen atom from C-6' of the substrate (6). The resulting radical (50) then accepts a methyl radical from Me-Cbl to generate Cbl(II) and the G418 product (7). A similar mechanism has been proposed for most class B radical SAM methyltransferases, such as Fom3,¹³⁰ GenD1,⁸¹ and ThnK¹³² as mentioned in Section 3.1. In this mechanism, MeCbl could be regenerated after each turnover via reduction of Cbl(II) to Cbl(I), followed by methylation by SAM (1).

Given that the transfer of a methyl radical is heretofore unprecedented among the Cbl-dependent methyltransferases, mechanisms involving the transfer of a methyl cation from Me-Cbl were also considered (Scheme 3-6B, C). The early steps of these alternative mechanism are identical to those of Scheme 3-6A. However, after hydrogen atom abstraction, the radical intermediate (**50**) is deprotonated to form a ketyl radical **51/52**, similar to those in the reactions catalyzed by DesII^{152, 170–176} and (*R*)-2-hydroxyacyl-CoA dehydratase.^{177–178} Nucleophilic attack by the ketyl radical anion (**52**) on Me-Cbl leads to the transfer of an electrophilic methyl cation, yielding a Cbl(I) intermediate (as in

traditional Me-Cbl chemistry)¹⁷⁹⁻¹⁸⁰, and an alkoxy product radical (**53**). The alkoxy radical (**53**) could then be quenched by an external electron and a proton (Scheme 3-6B), or via hydrogen atom transfer from 5'-dAdo (**4**) (Scheme 3-6C). In this latter mechanism, SAM is regenerated while restoring the [4Fe-4S] cluster to the +1 redox state. The resultant SAM (**1**) could then methylate Cbl(I), releasing SAH (**5**) as the sole adenosylated product. Thus, in this mechanism, the input of exogenous electrons is not required and only a single equivalent of SAM (**1**) is consumed during each catalytic cycle. This is in contrast to mechanisms A and B, in which two equivalents of SAM (**1**) (in addition to two reducing equivalents) are consumed for each methyl transfer event catalyzed by GenK.





Scheme 3-6. Possible GenK reaction mechanisms. (A) GenX₂ radical (50) is quenched by the transfer of methyl radical from Me-Cbl(III) to give Cbl(II), which is reduced to Cbl(I) in order to accept a new methyl group from SAM (1). (B) Transfer of methyl cation to GenX₂ ketyl radical (52) followed by reduction and protonation of product radical (53). (C) Product radical (53) is quenched by a hydrogen from 5'-dAdo (4) to give 5'-dAdo• (3), which can be re-incorporated into SAM (1). In this mechanism, a single equivalent of SAM (1) can serve as a source of both 5'-dAdo• (3) and CH_3^+ .

3.3.2. Syntheses of [6'S-²H]-GenX₂ (20) and [6'R-²H]-GenX₂ (29)

To characterize the initial hydrogen abstraction by $5'-dAdo \cdot (3)$ and determine the stereochemistry of the methyltransfer reaction from $Gen X_2$ (6) to produce G418 (7), two isotope-labeled compounds, $[6'S^{-2}H]$ -GenX₂ (20) and $[6'R^{-2}H]$ -GenX₂ (29), were examined. In the mechanism proposed for GenK in Section 3.3.1, hydrogen atom abstraction by 5'-dAdo• (3) occurs at the C6' position of $Gen X_2$ (6). The syntheses of the two stereochemically labeled compounds began from the alcohol (8) which was used in the synthesis of gentamicin A₂ in Section 2.2.11. Scheme 3-1 in Section 3.2.4 and Scheme 3-2 in Section 3.2.5 show the overall schemes to synthesize $[6'S^{-2}H]$ -GenX₂ (20) [6'R-²H]-GenX₂ (29) respectively. Oxidation using 2,2,6,6-tetramethyl-1and piperidinyloxy (TEMPO) and (diacetoxyiodo)benzene (BAIB) generated 6'-carboxylic acid,¹⁸¹ and the following methylation reaction with iodomethane and mildly basic potassium carbonate converted the carboxylic acid to a methyl ester to provide compound 9. The C6'-dideuterated alcohol (10) was then generated by the hydride reduction of the ester (9) with lithium aluminum tetradeuteride. The resulting dideuterated O-6 primary alcohol of compound 10 was oxidized to an aldehyde with one deuterium by Swern oxidation. Selective reduction with (R)-(+)-Alpine-Borane[®] gave [6'-S-²H]-substrate (12).¹⁷⁶ The synthesis of $[6'-R-^2H]$ -substrate (23) was attempted using (S)-(-)-Alpine-Borane[®], however the diastereomer, $[6'-S^{-2}H]$ -substrate (12), was also produced. In order to produce one stereospecific compound with (R) stereochemistry, an $S_N 2$ type conversion was used instead to invert the (S)-isomer. From compound 12, the formation of tosylate, followed by nucleophilic benzoate substitution, and basic hydrolysis with sodium methoxide was performed to get $[6'-R-^2H]$ -substrate (23). The benzyl protection reaction on the primary alcohol gave tribenzylated compounds, 13 and 24. NMR spectra in Figure 3-8 shows tribenzylated compounds, 13 and 24, with varying stereochemisty. The compound 12 in Figure 3-8A was obtained from using (R)-(+)-Alpine-Borane[®].

NMR spectrum displays only one stereoisomer (doublet at 3.72 ppm (J = 4.4 Hz) for the pro-(S) proton). The compounds in Figure 3-8B was acquired using (S)-(–)-Alphine-Borane[®], and two diastereomers were found (doublet in 3.76 ppm (J = 1.9 Hz) for the pro-(R) proton and 3.72 ppm (J = 4.4 Hz) for the pro-(S) proton) with a ratio of S isomer/R isomer = 0.65/ 0.23 based on NMR integration. The compound in Figure 3-8C was obtained by the stereochemical inversion through S_N2 reaction, and its NMR spectrum indicates the presence of only a single diastereomer (doublet at 3.75 ppm (J = 1.9 Hz) having the pro-(R) proton at C6'.

After coupling to the 2-strepatamine derivative **14** and basic hydrolysis with sodium methoxide, another coupling reaction of compounds **16** and **26** with the garosamine derivative **17** gave pseudotrisaccharides **19** and **27** respectively. Basic hydrolysis, and a final deprotection reaction using a Staudinger reaction and hydrogenation using palladium hydroxide on carbon gave $[6'S^{-2}H]$ -GenX₂ (**20**) and $[6'R^{-2}H]$ -GenX₂ (**29**). The synthesis of $[6'S^{-2}H]$ -GenX₂ (**20**) was completed with a 7.1% yield over 10 steps from the known compound **8**. The synthesis of $[6'-R^{-2}H]$ -GenX₂ (**29**) was completed with a 3.4% yield over 13 steps from the intermediate **12** in the synthesis of $[6'S^{-2}H]$ -GenX₂ (**20**).



Figure 3-8. NMR spectra of two labeled compounds, **20** and **29** obtained from A) (*R*)-(+)-Alpine-Borane[®], B) (*S*)-(–)-Alphine-Borane[®], and C) (*R*)-(+)-Alpine-Borane[®] and stereochemical inversion through $S_N 2$ reaction.

3.3.3. Stereospecific Hydrogen Abstraction

GenK assays with 6'-deuterated compounds, **20** and **29**, were conducted to determine the stereochemical course of the hydrogen abstraction step. Assay conditions are the same as the previously described GenK assays. Three substrates, GenX₂ (**6**), [6'S- 2 H]-GenX₂ (**20**) and [6'R- 2 H]-GenX₂ (**29**) were used in the assays. Detection of 5'-dAdo (**4**) by mass spectroscopy was conducted after collecting the corresponding fractions by HPLC (Figure 3-9).



Figure 3-9. HPLC traces showing the production of 5'-dAdo (4) and SAH (5) during GenK assays with 6'-labeled compounds. Trace (a) a standard composed of authentic 5'-dAdo (4) and SAH (5). The remaining traces result from reaction mixtures containing (b) GenX₂ (6), (c) [6'S⁻²H]-GenX₂ (20), and (c) [6'R⁻²H]-GenX₂ (29)

Figure 3-10 shows the mass spectroscopic data used to identify 5'-dAdo (4; m/z calc. 252.2 [M + H]⁺) when the three previously mentioned substrates were used (GenX₂ (6) in 3-10A, [6'S-²H]-GenX₂ (20) in 3-10B, and [6'R-²H]-GenX₂ (21) in 3-10C). When GenX₂ (6) and [6'S-²H]-GenX₂ (20) were used for the GenK reaction, an m/z 252 peak was observed. However, when [6'R-²H]-GenX₂ (29) was used, a m/z 253 peak was identified, indicating that deuterium from the substrate is incorporated into 5'-dAdo• (3) to form the deuterated 5'-dAdo (54; calc. [M + H]⁺ 253.2 m/z). The small peak around 252 in Figure 3-10C might be due to uncoupled quenching of one electron reduction followed by protonation. These results demonstrate that the pro-*R* hydrogen at the C6' position of the substrate is abstracted by 5'-dAdo• (4) in the beginning of the GenK reaction.



Figure 3-10. Electrospray ionization (ESI) mass data of resulting 5'-dAdo (4) generated in the assays of GenK with A) GenX₂ (6), B) $[6'S^{-2}H]$ -GenX₂ (20), and C) $[6'R^{-2}H]$ -GenX₂ (29).

In order to examine mechanism C outlined in Section 3.3.1 where SAM is regenerated and SAH is released at the end of each catalytic cycle, the reaction mixture was analyzed by ESIMS to check for SAH (5; calc. $[M + H]^+ = 385.1 \text{ m/z}$) and SAM (1; calc. $[M + H]^+ = 399.1 \text{ m/z}$) in the assays using GenX₂ (6), $[6'S^{-2}H]$ -GenX₂ (20), and $[6'R^{-2}H]$ -GenX₂ (29) as substrates (Figures 3-11 and 3-12). All three mass spectra appear to be the same in Figures 3-11 and 3-12, and there is no deuterium incorporation in SAH (5) and SAM (1). These results combined with the observation that 5'-dAdo (4), SAH (5), and G418 (7) are each produced in equivalent amounts (See Section 2.3.6) indicate that mechanism C can be ruled out. Two SAM molecules are consumed per reaction. The first molecule of SAM (1) acts as the source of the 5'-dAdo• (3), which abstracts a hydrogen atom from C-6' of GenX₂ (6) to give 5'-dAdo (4) and the GenX₂ radical (3). The second 174

SAM (1) is used to re-methylate Cbl(I) during turnover to regenerate Me-Cbl for the subsequent round of catalysis.



Figure 3-11. ESI Mass data of resulting SAH (5; calc. $[M + H]^+ = 385.1 \text{ m/z}$) in the assays using A) GenX₂ (6), B) [6'S⁻²H]-GenX₂ (20), and C) [6'R⁻²H]-GenX₂ (29).



Figure 3-12. ESI mass data of resulting SAM (1; calc. $[M + H]^+ = 399.1 \ m/z)$ in the assays using A) GenX₂ (**6**), B) [6'S⁻²H]-GenX₂ (**20**), and C) [6'R⁻²H]-GenX₂ (**29**).

Figure 3-13 shows mass spectra of the product in the assays using GenX₂ (6), $[6'S^{-2}H]$ -GenX₂ (20), and $[6'R^{-2}H]$ -GenX₂ (29). When unlabeled substrate (6; $[M + H]^{+} = 483.27 \text{ m/z}$) and $[6'R^{-2}H]$ -GenX₂ (29; $[M + H]^{+} = 484.28 \text{ m/z}$) were used, unlabeled G418 (7; $[M + H]^{+} = 497.27 \text{ m/z}$) was produced as a result of GenK catalysis. However, when

 $[6'S^{-2}H]$ -GenX₂ (20) was utilized as a substrate, labeled G418 (55; $[M + H]^+ = 498.28$ *m/z*) was detected.



Figure 3-13. ESI mass data of resulting G418 in the assays using A) GenX₂ (6), B) [6'S- 2 H]-GenX₂ (20), and C) [6'R- 2 H]-GenX₂ (29).

In addition, products in the assays using GenX₂ (**6**), $[6'S^{-2}H]$ -GenX₂ (**20**), and $[6'R^{-2}H]$ -GenX₂ (**29**) were collected by HPLC after derivatizations with 1-fluoro-2,4dinitrobenzene (FDNB) to add a UV-active chromophore which would otherwise be absent in the aminoglycosides. FDNB-derivatized G418 (**56**; $[M + H]^+ = 995.3 m/z$) was detected at $[M+H]^+$ 995.3 *m/z* when unlabeled GenX₂ (**6**) and $[6'R^{-2}H]$ -GenX₂ (**29**) were used (Figure 3-14A). However, FDNB-derivatized G418 with deuterium (**57**; $[M+H]^+$ 996.4 *m/z*) was produced when $[6'S^{-2}H]$ -GenX₂ (**20**) was treated in the assay. These results also show that deuterium in $[6'R^{-2}H]$ -GenX₂ (**29**) was incorporated into 5'-dAdo (4) and the methyl group from Me-Cbl was transferred at C6' position with retention of configuration. As expected, in the GenK reaction with $[6'S^{-2}H]$ -GenX₂ (20), deuterium was retained in the product, producing deuterated G418 (55).



Figure 3-14. Chemical ionization (CI) mass data of 1-fluoro-2,4-dinitrobenzene (FDNB)derivatized G418 in the assays using A) GenX₂(**6**), B) [6'*S*-²H]-GenX₂ (**20**), and C) [6'*R*-²H]-GenX₂ (**29**).

Combined with the results concerning 5'-dAdo (4) formation in the assays using $[6'S^{-2}H]$ -GenX₂ (20), and $[6'R^{-2}H]$ -GenX₂ (29) and the absolute stereochemistry of G418 (7) obtained by X-ray crystallography,¹⁴ it can be concluded that the pro-*R* hydrogen at the C6' position of GenX₂ (6) is abstracted by 5'-dAdo• (3) and without any rotation, the

methyl group is transferred from Me-Cbl to the exact same position of pro-R hydrogen. The overall stereochemistry in GenK catalysis is maintained from the substrate to the product.

3.3.4. UV-Vis Spectroscopic Experiments

In order to investigate the redox states of cobalamin during GenK catalysis, the reaction was monitored by UV-visible spectroscopy under anaerobic conditions for 1 h and the spectrum of the reaction mixture was recorded every 5 min. (Figure 3-15). This experiment was conducted using this assay conditions containing GenK (0.005 mM), DTT (1 mM), MeCbl (0.1 mM), NADPH (0.4 mM), MV (0.1 mM), SAM (0.4 mM), GenX₂ (6) (0.1 mM) in Tris·HCl buffer (5 mM; pH 8.0) under anaerobic condition. The spectra in Figure 3-15 correspond to assays resulting from (A) the reaction mixture, (B) no GenX₂ (6), (C) no GenK, and (D) no GenK and no SAM. In Figure 3-15A, the peak around 520 nm corresponding to Cbl(III)¹⁸³ was found to decrease, while the peak around 400 nm $(Cbl(I))^{184}$ increased. In Figures 3-15B and 3-15C, a clear shift from ~520 nm to ~476 nm and an increase in the absorbance between 400 nm and 450 nm were observed. Both are characteristics of MeCbl(III) conversion into Cbl(II).¹⁸⁴ It was also noted that without SAM or GenK (Figure 3-15D), the decrease of the peak at ~520 nm was not as significant as those observed in other spectra, but increases of the peak around 476 and 400 nm were detected, indicating that Cbl(II) and Cbl(I) were generated from MeCbl in the presence of the reducing system (methyl viologen and NADPH). Thus, based on the combined spectroscopic data, it can be concluded that the Me-Cbl(III) bond of methylcobalamin is easily cleaved in the presence of SAM and MV/NADPH to give Cbl(I) regardless whether GenK is present or absent. However, in the absence of SAM, the cleavage rate of Me-Cbl bond was much slower when compared to other conditions.

The formation of Cbl(II) is not obvious in the presence of GenK and GenX₂ (6). The peak around 476 nm was not obvious in the assay of all reaction mixtures (Figure 3-15A). In the absence of substrate, SAM, or GenK, the peak about 476 nm corresponding to Cbl(II) was discerned clearly (Figures 3-15B, 3-15C, and 3-15D). In the presence of GenK and GenX₂ (Figure 3-15A), Cbl(II) might exist too transiently to be detected by UV-vis spectrometry, or Cbl(II) might not exist during the GenK reaction. From UV-vis spectroscopy, it was not possible to differentiate between mechanism A and mechanism B described in Section of 3.3.1.



Figure 3-15. UV-visible analysis of GenK assay. The traces (A-D) are from the reaction mixtures containing GenK (0.05 mM), DTT (10 mM), MeCbl (1 mM), NADPH (4 mM), MV (1 mM), SAM (4 mM), GenX₂ (1 mM) in Tris·HCl buffer (50 mM; pH 8.0) under anaerobic condition with the following modifications: A) no changes, B) no GenX₂, C) no GenK, and D) no GenK and no SAM. Measurements were performed every 5 min from 0 min to 30 min and every 10 min from 30 min to 60 min (from blue to red).

3.3.5. Syntheses of [6'-F]-GenX₂ (35), [6',6',6'-²H,²H,F]-GenX₂ (43), and [6'-OMe]-GenX₂ (49)

To distinguish between mechanisms A and B in Section 3.3.1, [6'-F]-GenX₂ (**35**) and [6',6',6'- 2 H, 2 H,F]-GenX₂ (**43**) were designed as a probe because an alkoxy radical could not be generated from [6'-F]-GenX₂ (**35**) and the GenK reaction could only proceed via mechanism A if GenK can accept it as a substrate. Using [6',6',6'- 2 H, 2 H,F]-GenX₂ (**43**) as a probe may give clues to distinguish the mechanism of the reaction. The synthesis of [6'-F]-GenX₂ (**35**) began with the alcohol **30** which was used for the synthesis of JI-20A in Section 2.2.17. The fluoro group was introduced through an S_N2 type reaction using diethylaminosulfur trifluoride (DAST) in dichloromethane to provide compound **31**. After basic hydrolysis using sodium methoxide, coupling of compound **32** with garosamine derivative **17** gave the pseudotrisaccharide **33**. After basic hydrolysis of the carbamate, final deprotection using the Staudinger reaction, and hydrogenation with palladium hydroxide on carbon, [6'-F]-GenX₂ (**35**) was obtained with a 14% yield over 6 steps from the previous synthesized compound **30**. The synthesis of [6',6',6'-²H,²H,F]-GenX₂ (**43**) was conducted using the similar methods as the synthesis of [6'-F]-GenX₂ (**35**) with 7% yield over 9 steps from the previously synthesized dideuterated alcohol **10**.

Another possible substrate to distinguish between mechanisms A and B in the Section 3.3.1 is [6'-OMe]-GenX₂ (49), because deprotection of the initially formed C6' radical (50) by a base cannot occur and thus the alkoxy radical species (51) cannot be produced. If the reaction proceeds via mechanism B, no reaction is expected. The synthesis of [6'-OMe]-GenX₂ (49) began from the alcohol 8 which was prepared in the synthesis of gentamicin A₂ in Section 2.2.11. A methoxy group was introduced with sodium hydride and methyl iodide in dimethylformamide (DMF) to provide compound

44. The following steps were similar to the other substrate syntheses. The overall yield of the synthesis was 22% over 7 steps from the previously synthesized compound **8**.

3.3.6. Assays with GenX₂ Derivatives

Assays with GenX₂ derivatives, [6'-F]-GenX₂ (**35**), [6',6',6'- 2 H,²H,F]-GenX₂ (**43**), and [6'-OMe]-GenX₂ (**49**) were conducted under the same conditions as before, and 5'dAdo (**4**) formation was monitored by HPLC as the evidence of radical reaction initiation (Figure 3-16). For comparison, the assay with the original substrate GenX₂ (**6**) was also performed. Neither [6'-F]-GenX₂ (**35**), [6',6',6'- 2 H,²H,F]-GenX₂ (**43**), nor [6'-OMe]-GenX₂ (**49**) showed any reaction with GenK. Different reducing systems like MV/NADPH, BV/NADPH, dithionite, fld/fdr/NADPH, dithionite/MV and dithionite/BV were all tested with [6'-F]-GenX₂ (**35**), but none of them resulted in the formation of 5'dAdo (**4**). These three GenX₂ derivatives which were designed as mechanistic probes to determine the mechanism of GenK are not recognized by GenK.



Figure 3-16. HPLC traces with GenK reaction with the following GenX₂ derivatives; a) SAH and 5'-dAdo standard, b) [6'-F]-GenX₂ (**35**), c) [6',6',6'- 2 H, 2 H,F]-GenX₂ (**43**), and d) [6'-OMe]-GenX₂ (**49**)

Whether [6'-F]-GenX₂ (**35**) is an inhibitor for GenK was also tested because it has a similar structure to the natural substrate. If [6'-F]-GenX₂ (**35**) could compete with GenX₂ (**6**), it may be considered as an inhibitor competing for the active site of the enzyme even though GenK does not catalyze the radical formation reaction. Inhibition assays were conducted under the same conditions as before, but the incubation time was 20 h. When the concentration of GenX₂ (**6**) was fixed at 1 mM and lower concentrations of [6'-F]-GenX₂ (**35**) were used (0.05, 0.1, 0.2, 0.5, and 1 mM) in the GenK assay, the amounts of 5'-dAdo (**4**) and SAH (**5**) formations were the same as when GenX₂ (**6**) was used alone (Figure 3-17). Even in the GenK assays where the enzyme was pre-incubated with [6'-F]-GenX₂ (**35**) for 4 h and GenX₂ (**6**) was added later, the results were similar as in the absence of pre-incubation time. Experiments with higher concentrations of GenX₂ (**6**) compared to [6'-F]-GenX₂ (**35**) did not provide any evidences that the fluoridesubstrate binds to the active site of GenK and inhibits methylation of the enzyme.



Figure 3-17. Formations of A) 5'-dAdo and B) SAH from the co-incubation of [6'-F]-GenX₂ (**35**) and GenX₂ (**6**) in the GenK reaction for 20 h. Concentration of GenX₂ (6) was fixed as 1 mM, and concentration of [6'-F]-GenX₂ (**35**) was changed from 0 to 1 mM.

Assays with excess [6'-F]-GenX₂ (**35**) compared to GenX₂ (**6**) (1 mM) showed that although the fluoro-substrate inhibits the activity of GenK, it is a very weak inhibitor (Figure 3-18). Assuming that it competes with GenX₂ (**6**) for the active site, it does not display a strong affinity. Combined with the results of the [6'-OMe]-GenX₂ (**49**) assay, it may be concluded that the 6'-hydroxy group plays a role in binding GenX₂ to GenK.



Figure 3-18. Formations of A) 5'-dAdo and B) SAH from the co-incubation of [6'-F]-GenX₂ (**35**) and GenX₂ (**6**) in the GenK reaction for 20 h. Concentration of GenX₂ (6) was fixed as 1 mM, and concentration of [6'-F]-GenX₂ (**35**) was changed from 1 to 20 mM.

3.4. CONCLUSION

The mechanism of methyl transfer by Class B radical SAM methyltransferase has not yet been elucidated. In this chapter, three mechanisms were considered for the GenKcatalyzed conversion of GenX₂ to G418. In the first mechanism, the 5'-dAdo• generated via the reductive cleavage of SAM abstracts a hydrogen atom from C-6' of the substrate. 183 The resulting substrate radical accepts a methyl radical from Me-Cbl to generate Cbl(II) and the G418 product. Other mechanisms, which involve the transfer of a methyl cation from Me-Cbl, were also considered. After hydrogen atom abstraction, the radical intermediate is deprotonated to form a ketyl radical. Nucleophilic attack by an anion at the Me-Cbl leads to methyl transfer, yielding a Cbl(I) intermediate and an alkoxide product radical. The radical could then be quenched by an external electron and proton or via hydrogen atom transfer from 5'-dAdo. In this latter mechanism, SAM is regenerated while restoring the [4Fe-4S] cluster to the +1 redox state.

The observation that 5'-dAdo, SAH and G418 are each produced in equivalent amounts is consist with mechanisms A and B, but not C, because in mechanisms A and B two equivalents of SAM are consumed for each methyl transfer event (one to initiate the radical reaction and the other as a methyl donor), but in mechanism C the input of exogenous electrons is not required and only a single equivalent of SAM is necessary for each catalytic cycle.

Incorporation of deuterium into 5'-dAdo and loss of deuterium from $GenX_2$ were observed by mass spectra, which showed that the reaction with (*R*)-deuterated $GenX_2$ gave deuterated 5'-dAdo and non-deuterated product, and that the reaction with (*S*)deuterated $GenX_2$ gave the opposite results. These results indicate that the 6'-pro-*R* hydrogen is abstracted by the 5'-dAdo radical. Combined with the reported structure of G418, it was ascertained that methylation occurs with retention of configuration at C6'.

To distinguish the manner of methyl transfer, three $GenX_2$ derivatives were tested. However, none of them showed turnover upon incubation with GenK. The fluoride substrate did inhibit the activity of GenK very weakly. Assuming that it competes with GenX₂ for the active site, then it must not have a strong affinity for the active site. The 6'-hydroxyl group apparently plays a key role in binding of $GenX_2$ to GenK.

Efforts are currently underway to further elucidate the chemical mechanism of this intriguing enzyme. Another substrate [5'-F]-GenX₂ would be a possible substrate to distinguish the two mechanisms. If the methyl transfer reaction proceeds via methyl radical reaction (mechanism A), this compound would produce [5'-F]-G418. However, if an anionic mechanism is operative (mechanism B), and both the leaving group and the anion have an anti-orientation, the product would not contain fluoride because an anion at the C6' position would result in the elimination of fluoride. The synthesis of [5'-F]-GenX₂ is currently underway in our group.

Chapter 4: Characterization of the Radical SAM Enzyme AprD4 and Its Reductase Partner AprD3

4.1. INTRODUCTION

The aminoglycosides are an important class of antibiotics for the treatment of Gram-negative bacterial infections.^{1-2,185} Their biological activity is a result of their ability to bind the bacterial 30S ribosome and subsequently inhibit protein synthesis. Several years after aminoglycosides were first used to treat human antibacterial chemotherapy, new organisms resistant to these drugs began to appear. The major cause of aminoglycoside resistance is decreased affinity of the drug to the bacterial ribosome. Numerous genetic and biological studies have focused on the mechanisms of bacterial resistance to aminoglycosides.¹⁸⁵⁻¹⁸⁸ Among these mechanisms, enzymatic inactivation of the antibiotic is the most clinically important. Three classes of aminoglycoside-modifying enzymes have been described: the aminoglycoside nucleotidyltransferases (ANTs), the aminoglycoside phosphotransferases (APHs), and the aminoglycoside acetyltransferases (AACs). The regioselective modification site on the targeted aminoglycoside by a specific enzyme is specified in parentheses. For example, AAC(2') is an aminoglycoside acetyltranferase catalyzing the acetylation of hydroxyl group at C2' position.

Aminoglycoside adenyltransferases found in both Gram-negative and Grampositive bacteria can be either plasmid-encoded or chromosomally encoded. The reaction sites are 2" and 3" for enzymes found in Gram-negative bacteria, and 4', 6, and 9 for those derived from Gram-positive organisms.¹⁸⁹ These enzymes catalyze *O*-adenylation of aminoglycosides using Mg-ATP and in the process generate magnesium chelated inorganic pyrophosphate (Figure 4-1). Adenylation at the 2" and 4' are clinically the most significant and have been well-studied.



Figure 4-1. 4'-O-adenylation of kanamycin A by ANT(4')

The aminoglycoside phosphotransferases represent a large group of aminoglycoside-modifying enzymes that play important roles in the clinical resistance of *Staphylococcal* and *Enterococcal* species to aminoglycosides.¹⁸⁵ They can regioselectively transfer the γ -phosphate of ATP to one of the hydroxyl groups of the aminoglycoside (Figure 4-2). Among them, the plasmid-encoded APH(3')-IIIa enzymes from *Enterococcus faecalis* that have been the focus of biochemical investigations.¹⁹⁰⁻¹⁹⁴



Figure 4-2. 3'-*O*-phosphorylation of kanamycin A by APH(3')

Acetyl coenzyme A-dependent aminoglycoside acetylases (AACs) catalyze *N*acetylation of the four amino groups of aminoglycosides: the 1- or 3-amino groups of the central 2-deoxystreptamine ring or the 2' or 6'-amino groups of the 6-amino-6-deoxy glucose ring (Figure 4-3). Two types of AAC have been identified in *E. coli* and *Actinomycete* bacterial strains; however, they are not clinically important, because these enzymes are not virulence factors with respect to infection of humans.¹⁹⁵⁻¹⁹⁶



Figure 4-3. 4'-*N*-acetylation of kanamycin A by AAC(6')

A number of aminoglycosides, such as apramycin (2),⁶ tobramycin (3),¹⁹⁷ nebramycin 5' (4),¹⁹⁸ and lividomycin B (5),¹⁹⁹ possess the 2-deoxystreptamine (2-DOS; 1) core decorated with a variety of carbohydrates necessary for activity; however, they are characterized by a C3'-deoxyhexose linked to the C4 position of 2-DOS via a glycosidic bond (Figure 4-4). The absence of a C3' hydroxyl group in these aminoglycosides renders them more effective as antimicrobial agents, because they are less susceptible to modification by aminoglycoside phosphotransferase (APH(3')), which is a known mechanism of resistance.¹⁸⁵



Figure 4-4. Selected examples of aminoglycosides.

Although the biosynthesis of paromamine has been investigated (see Section 1.1.3.2), little is known about the mechanism of C3'-deoxygenation despite its importance to the development and improvement of clinically useful aminoglycoside antibiotics.²⁰⁰ C3'-deoxygenation of paromamine is similar to the reactions catalyzed by Cbl-dependent dehydratases like diol dehydratases (DD), propanediol hydro-lyase, glycerol dehydratase (GD), and glycerol hydro-lyase, which catalyze the conversion of 1,2-propanediol and other 1,2-diols to the corresponding aldehydes, or deaminases such as ethanolamine ammonia-lyase (EAL) or ethanolamine deaminase which catalyze the conversion of ethanolamine to acetaldehyde in the presence of AdoCbl.²⁰¹⁻²⁰² The mechanism of DDs is shown in Figure 4-5. Two water molecules are bound in the active site of the holoenzyme and are displaced upon substrate binding. The adenosyl radical generated by homolytic cleavage of the carbon-cobalt bond in adenosylcobalamin abstracts a hydrogen atom from the C1 position of the substrate. The resulting substrate radical then undergoes hydroxyl

group migration from C2 to C1, generating a 1,1-diol-2-yl radical. The resulting radical then abstracts a hydrogen atom from the C5' methyl of 5'-deoxyadenosine to form the aldehyde product and adenosyl radical, the latter of which recombines with cob(II)alamin to regenerate AdoCbl. The mechanism of EAL is similar to that of the DDs. One particular thing is that DDs and EAL are not enantioselective: the adenosyl radical can abstract either the pro-*R* or pro-*S* hydrogen from the substrate.²⁰² Another example of a radical mediated dehydration is the reaction catalyzed by Cbl-independent glycerol dehyratase.²⁰³ In the case of both, Cbl-dependent DDs and Cbl-independent glycerol dehydratase, the radical initiator is typically regenerated.



Figure 4-5. Mechanism of the diol dehydratases.

The gene clusters responsible for the biosynthesis of apramycin (2), tobramycin (3), and lividomycin B (5) in *Streptomyces* and *Streptoalloteichius* species have been described,^{8,204-206} and significant progress has been made in elucidating the reactions involved in the tobramycin pathway.^{55,68} Although no gene product in the tobramycin

cluster has been implicated in catalyzing C3'-deoxygenation,^{8,204} gene-knockout studies²⁰⁷ of the aparamycin and lividomycin B pathway have identified the AprD4/AprD3 and LivW/LivY enzyme pairs as likely candidates for their activity. For example, incorporation of the *aprD4* and *aprD3* genes into the kanamycin-producing strain *S. kanamyceticus* led to the isolation of 3'-deoxykanamycins (such as 7),²⁰⁴ which are not normally produced. Likewise previous studies have also shown the conversion of neamine (9) to nebramine (10) in cell-free extracts of *S. venezuelae* expressing protein AprD4 and AprD3.²⁰⁴ However, no conversion of kanamycin B (8) to tobramycin (3) was observed under the same conditions, leading to the suggestion that the substrate for AprD4/AprD3 may be a pseudo-disaccharide (such as 9) rather than a pseudo-trisaccharide (such as 8).²⁰⁴ Since apramycin (2) and tobramycin (3) are produced together in the same bacterial strains, it is possible that AprD4/AprD3 serves as the C3'-deoxygenation machinery for both pathways.

This chapter describes the purification and biochemical characterization of AprD4 and AprD3 from *S. tenebraris* and demonstrates their ability to catalyze 3'-deoxygenation of paromamine (**11**) to lividamine (**13**) in vitro (Scheme 4-1). Using various paromamine derivatives, the mechanistic details regarding the AprD4-catalyzed conversion of paromamine to 4-oxolividamine (**12**) are considered.



Scheme 4-1. AprD4/AprD3 catalysis.

4.2. EXPERIMENTAL PROCEDURES

4.2.1. Materials and Equipment

All chemicals were purchased from Fisher Science (Pittsburgh, PA) or Sigma-Aldrich (St. Louise, MO) and used without further purification unless otherwise specified. DNA gel extraction and spin miniprep were obtained from Qiagen (Valencia, CA). All reagents for sodium dodecyl sulfate polyacrylamide gel electrophoresis (SDS-PAGE) and Amicon and Microcon YM-10 filtration products were purchased from Bio-Rad (Hercules, CA) and Millipore (Billerica, MA), respectively. *Escherichia coli* DH5a cells were bought from Bethesda Research Laboratories (Muskegon, MI). The vector pET24b (+) and enzyme KOD DNA polymerase were purchased from Novagen (Madison, WI). DNA modifying enzymes (for restriction digestion and ligation), PCR primers, and the overexpression host E. coli BL21 star (DE3) were acquired from Invitrogen (Carlsbad, CA) and New England Biolabs (NEB, Beverly, MA). Luria Bertani (LB) media are products of Difco (Detroit, MI) or Fisher Scienctific (Pittsburgh, PA). Pre-stained protein markers were bought from NEB. Kits for Protein concentrations were determined by Bradford Assay¹⁴⁹ using bovine serum albumin as the standard. The relative molecular mass and purity of enzyme samples were determined using SDS-PAGE. The general methods and protocols for recombinant DNA manipulations were as described by Sambrook *et al.*¹⁵⁰ DNA sequencing was performed at the Core Facilities of the Institute of Cellular and Molecular Biology, University of Texas at Austin. All reactions involving AprD4 were performed in a Coy Anaerobic Chamber (glovebox) under an atmosphere of approximately 97.5% N_2 and 2.5% H_2 with less than 1 ppm $O_2.$ All solvents using AprD4 reaction were deaerated by bubbling with nitrogen gas before they were transferred into the glovebox. The solvents in the glove box were stirred open

to the anaerobic atmosphere overnight to allow equilibration before use. Paromamine (11) was obtained by acidic hydrolysis of paromomycin.¹⁶⁶ SAM was prepared as described in Section 2.2.4. [5-²H₂]-SAM were prepared as previous described.¹⁷³ Tetrahydrofuran (THF) was distilled from sodium benzophenone and dichloromethane (CH₂Cl₂) was distilled from calcium hydride under a nitrogen atmosphere. Other anhydrous solvents were purchased from Acros Organics (Pittsburgh, PA). Analytic thin layer chromatography (TLC) was performed on pre-coated TLC glass plates (Silica gel, grade 60, F254, 0.25 mm layer thickness) purchased from EMD chemicals (Madison, WI). Flash column chromatography was carried out on silica gel (230-400 mesh, grade 60) from Sorbent Technologies (Atlanta, GA) by eluting with the specific solvents. ¹H and ¹³C NMR spectra were recorded at 400, 500, 600 MHz and 100, 125, 150 MHz with a Varian Gemini spectrometer. Chemical shifts are reported as parts per million (ppm) from ppm relative to the deuteriochloroform (CDCl₃), 7.26 ppm for ¹H NMR and 77.16 ppm for ¹³C NMR, respectively. Visualization was effected with *p*-phosphoryl molybdic The MS analyses were carried out at the Mass Spectrometry and acid solution. Proteomics Facility of the Department of Chemistry, University of Texas at Austin.

4.2.2. Cloning and Expression of Streptomyces tenebrarius AprD3

The apramycin/tobramycin producer *S. tenebrarius* was obtained from the Agricultural Research Service of the US Department of Agriculture as a lyophilized sample. It was inoculated into 10 mL of tryptic soy broth (TSB) starter culture,¹⁵¹ and the resulting solution was incubated overnight at 30 °C with shaking at 250 rpm. The following day, 0.1 mL of the starter culture was inoculated into 10 mL of fresh TSB, and the resulting culture was incubated at 30 °C for 72 hr. Chromosomal DNA was extracted from 1.5 mL of this culture using a Qiagen DNeasy tissue kit. The *aprD3* gene was PCR

amplified from the isolated S. tenebrarius DNA with the following primers: 5'-CATATGGAGCAACG-GTACGTGCTGGTCACCGGGGCCAGTCGCGGTCTGGG-3' (reverse), and cloned into a pCR:blunt vector (Invitrogen) prior to excision with NdeI and HindIII ligation into the pET28b(+) vector (Invitrogen). The resulting expression plasmid, pET28b(+):aprD3, was used to transform E. coli BL21 cells for expression of the native, recombinant AprD3. An overnight culture of E. coli transformed with the pET28b:aprD3 plasmid was grown at 37 °C in LB medium containing 34 g/mL kanamycin, and used to inoculate 6 x 1 L of the same medium in a 100-fold dilution. Then 1 L cultures were incubated at 37 °C until cell had reached an optical density of 0.6 at 600 nm. At which point AprD3 expression was induced by 1mM IPTG. The cells were harvested by centrifugation at 20000 x g for 30 min, thawed and suspended in 50 mM EPPS (pH 8.0) with 20 mM imidazole and 300 mM NaCl. After sonication (8 x 1 min), the lysate was spun at 26500 x g for 40 min. 1 mM PMSF was added. The supernatant was mixed with 8 mL of Ni-NTA resin at 4 °C. The slurry was transferred to an empty column and washed with 100 mL of 50 mM EPPS buffer (pH 8.0) containing 300 mM NaCl and 20 mM imidazole. The protein was eluted with 50 mL of 50 mM EPPS buffer (pH 8.0) containing 300 mM NaCl and 300 mM imidazole. The purified protein was dialyzed twice against 1 L of 50 mM EPPS buffer (pH 8.0) containing 300 mL NaCl. The dialyzed protein was frozen with liquid nitrogen and stored at -80 °C until use.

4.2.3. Cloning, Expression and Reconstitution of Streptomyces tenebrarius AprD4

The aprD4 gene was PCR amplified from the isolated *S. tenebrarius* DNA with the following primers: 5'-CATATGCGACGAATGCGGCTCGGCACGGTAC-3' (forward), 5'-AAGCTTCATCAGGCG-TCACCGGTCGACCAGGCGTG-3' (reverse), 194

and cloned into a pCR:blunt vector (Invitrogen) prior to excision with NdeI and HindIII and ligation into the pET28 vector (Invitrogen). The resulting expression plasmid, pET28:aprD4, was used to transform E. coli Rosetta DE3 (Novagen) for expression of native, recombinant AprD4. Transformants were grown in 6 x 1 L of LB medium with 34 g/mL kanamycin and 68 g/mL chloramphenicol at 37 °C with shaking at 200 rpm. AprD4 expression was induced with 1 mM IPTG when the cells had reached an optical density of 0.6 at 600 nm. Cells were harvested by centrifugation at 4500 x g for 20 min, thawed and suspended in 50 mM Tris buffer (pH 8.0, HCl) with 20 mM imidazole. After sonication (8 x 1 min), the lysate was spun at 26500 x g for 30 min. Lysate was loaded on a Ni-NTA column (8 mL) pre-equilibrated with 50 mM Tris buffer (pH 8.0, HCl) and 20 mM imidazole. After washing with 10 mL of 50 mM Tris buffer (pH 8.0, HCl) and 20 mM imidazole, the column was developed with 8 mL of 50 mM Tris buffer (pH 8.0, HCl) and 40 mM imidazole, 8 mL of 50 mM Tris buffer (pH 8.0, HCl) and 100 mM imidazole, and 8 mL of 50 mM Tris buffer (pH 8.0, HCl) and 300 mM imidazole. The fractions were collected and centrifuged with a YM-10 filter at $4500 \times g$ for 1 min. The protein was dialyzed twice against 1 L of 50 mM Tris buffer (pH 8.0, HCl). The dialyzed protein was frozen with liquid nitrogen and stored at -80 °C until reconstitution with iron and sulfur.

Reconstitution followed the previously published procedure.¹⁵² Approximately 3– 5 mg of AprD4 was thawed, and transferred into a conical vial. The vial was transferred to the glovebox. The enzyme was diluted with 20 mM NH₄HCO₃ buffer (pH 7.8) containing 5 mM DTT to a volume of ca. 2 mL and gently stirred at ca. 10 °C in an open vial for ca. 2 h in order to equilibrate with the anaerobic atmosphere. The solution was made 1.88 mM in dithiothreitol and allowed to stir for 15 min. To this solution was slowly added 42 μ L of a 50 mM solution of Fe(NH₄)₂(SO₄)₂ in anaerobic water over 10 min. After 10 min, 42 μ L of a 50 mM solution of Na₂S in anaerobic buffer containing 20 mM NH₄HCO₃ (pH 7.8) and 5 mM DTT was added over 10 min. The solution was gently stirred at ca. 10 °C for 2 hr. The reconstituted enzyme was loaded onto a 50 mL Sephadex G-25 column pre-equilibrated with anaerobic buffer containing 20 mM NH₄HCO₃ (pH 7.8) and 5 mM DTT, and the column was eluted with the same buffer. Protein fractions were collected, concentrated to approximately 100 μ M of enzyme and stored in the glovebox at ca. 10 °C until use. Iron content in the reconstituted AprD4 was assessed using the ferrozine assay,¹⁵³ and the sulfide content reported in the text is an average of three measurements.

4.2.4. AprD4 Activity Assays and AprD3 Activity Assays

The reaction mixture contained the following: 50 mM NH₄HCO₃ (pH 7.8), 10 mM DTT, 1 mM paromamine (**11**), 2 mM enzymatically prepared SAM, 2 mM sodium dithionite, and 0.01 mM AprD4. Assays were initiated by the addition of AprD4, incubated for 8 h at room temperature, and quenched by passage through YM-10 Microcon centrifugal filters to remove enzymes. The filtrate of each sample was frozen and stored at -80 °C until analysis of HPLC and mass spectroscopy.

To find proper reducing systems for the AprD4 reaction, all reactions were treated with 1 mM of the chemical reductants being tested. When flavodoxin and flavodoxin reductase from *E. coli* (obtained as previously described¹⁷⁰) were used as the reducing system, the protein concentration was 0.1 mM.

To an assay mixture of AprD4, 0.01 mM AprD3 and 1 mM NAD(P)H were added. Assays were incubated for 1 h at room temperature, quenched by passage through

YM-10 Microcon centrifugal filters to remove enzymes. The filtrate of each sample was frozen and stored at -80 °C until analysis of HPLC and mass spectroscopy.

4.2.5. HPLC Detection of 5'-Deoxyadenosine (5'-dAdo)

A 20 μ L aliquot of the quenched AprD3/AprD4 reaction or AprD4 reaction described above was analyzed using HPLC equipped with a Varian Microsorb-MV 100-5 C18 (4.6 x 250 nm) column pre-equilibrated in H₂O containing 0.1% trifluoroacetic acid (solvent A). After sample loading, the column was eluted with a linear gradient from 0– 20% acetonitrile containing 0.1% trifluoroacetic acid (Solvent B) over 30 min. The UV detector was set at 260 nm. An authentic, commercially obtained 5'-dAdo was injected as a retention time standard for each. The stoichiometric ratio reported in the text is an average of three measurements.

4.2.6. HPLC Detection of AprD4/AprD3 Reaction Product

A 10 L aliquot of the quenched AprD4/AprD3 reaction or the AprD4 reaction was analyzed using HPLC equipped with a Varian Microsorb-MV 100-5 C18 (4.6 x 250 nm) and eluted with H₂O containing 0.4% trifluoroacetic acid for 15 min. The elution was monitored by Corona[®] charged aerosol detector. An authentic paromamine was injected as a retention time standard.

4.2.7. Mass Spectroscopic Analysis of AprD3/AprD4 Activity and AprD4 Activity

Assay mixtures were diluted in water and injected directly into an Agilent 6350 Accurate Mass QTOF-MS at 0.15 mL/min. The analyte was ionized by electrospray ionization and detected in positive ion mode.

4.2.8. Synthesis of Lividamine

The overall synthetic scheme is shown in Scheme 4-2.



Scheme 4-2. Synthetic scheme for the preparation of lividamine.

(*3R*,*5S*,*6R*)-6-(acetoxymethyl)-3-azidotetrahydro-2H-pyran-2,5-diyl diacetate (29): Compound 27 was prepared as previously described.²⁰⁹ Copper sulfate (0.158 g, 1.01 mmol), triethylamine (14.09 mL, 101.09 mmol), and triflic azide (0.4 M solution in dichloromethane, 80 mL) was added to a solution of compound 27 (2.0181 g, 10.11 mmol) in methanol (60 mL) at 0 °C. The reaction mixtures were stirred at room temperature for 24 h, and concentrated *in vacuo*. The residue was dissolved in acetonitrile and washed with water. The remaining aqueous layer was extracted with acetonitrile five times, and the combined organic layers were dried over magnesium sulfate, filtered, and concentrated under reduced pressure. The residue was filtered by short column filled with silica gel by 20% methanol in dichloromethane. Acetic anhydride (50 mL) was added to a solution of the crude compound 28 in pyridine (50 mL) at 0 °C. The reaction mixture was stirred at room temperature for 12 h, and concentrated. The residue was diluted with dichloromethane and washed with saturated aqueous solution of sodium bicarbonate (50
mL) two times. The organic layer was dried over magnesium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 2:1) to give compound **29** (1.75 g, 55% over two steps) as a light yellow liquid. ¹H NMR (500 MHz, CDCl₃) α isomer δ 6.18 (d, J = 3.3 Hz, 1H, H-1), 4.88 (ddd, J = 13.2, 8.3, 4.9 Hz, 1H, H-4), 4.29–4.24 (m, 1H, H-6), 4.12-4.09 (m, 1H, H-6), 4.0 (ddd, J = 10.3, 4.5, 2.3, 1H, H-5), 3.54-3.50 (m, 1H, H-2), 2.48 (dt, J = 11.5, 4.8 Hz, H-3), 2.21 (s, 3H, OAc), 2.09 (s, 3H, OAc), 2.08 (s, 3H, OAc), 2.05–1.98 (m, 1H, H-3). β isomer δ 5.53 (d, J = 8.4 Hz, H-1), 4.80 (ddd, J = 12.9, 8.1, 4.9Hz, H-4), 4.29-4.24 (m, 1H, H-6), 4.16-4.12 (m, 1H, H-6), 3.79 (ddd, J = 10.0, 5.2, 2.6Hz, 1H, H-5), 3.60 (ddd, J = 12.8, 7.9, 5.0 Hz, 1H, H-2), 2.55 (dt, J = 12.6, 4.9 Hz, 1H, H-3), 2.19 (s, 3H, OAc), 2.08 (s, 3H, OAc), 2.07 (s, 3H, OAc), 1.59–1.51 (m, 1H, H-3). ¹³C NMR (125 MHz, CDCl₃) α isomer δ 170.71 (OAc), 169.58 (OAc), 168.77 (OAc), 89.6 (C-1), 69.9 (C-5), 65.4 (C-4), 61.9 (C-6), 55.3 (C-2), 28.5 (C-3), 21.05 (OAc), 20.89 (OAc, 20.74 (OAc). β isomer δ 170.71 (OAc), 169.69 (OAc), 168.93 (OAc), 94.7 (C-1), 75.8 (C-5), 65.2 (C-4), 61.9 (C-6), 57.6 (C-2), 33.2 (C-3), 21.03 (OAc), 20.87 (OAc), 20.76 (OAc). HRMS (ESI, positive ion mode) calculated for $C_{12}H_{17}N_3O_7^+$ $[M + Na]^+$ 338.0959, found 338.0960.

((2*R*,3*S*,5*R*)-3-acetoxy-5-azido-6-(phenylthio)tetrahydro-2H-pyran-2-yl)methyl acetate (30): Boron trifluoride etherate (ca. 48%, 0.99 mL, 2.99 mmol) was added to a solution of compound 29 (0.787 g, 2.50 mmol) and thiophenol (0.51 mL, 4.99 mmol) in dichloromethane (40 mL) at 0 °C under nitrogen atmosphere, and the reaction mixture was stirred at room temperature for 6 h. The reaction was quenched by the addition of saturated aqueous solution of sodium bicarbonate (20 mL) at 0 °C, and the aqueous layer was extracted by dichloromethane (40 mL \times 3). The combined organic layers were dried over magnesium sulfate, filtered, and concentrated under reduced pressure. The residue

was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:1) to provide compound **30** (α/β = 4:1, 0.759 g, 83%) as a clear liquid. ¹H NMR (500 MHz, CDCl₃) α isomer δ 7.53–7.51 (m, 2H, Ph), 7.33–7.28 (m, 3H, Ph), 5.53 (d, J = 4.6 Hz, 1H, H-1), 4.82 (ddd, *J* = 13.1, 8.1, 4.9 Hz, 1H, H-4), 4.49 (ddd, *J* = 10.0, 5.6, 2.3 Hz, 1H, H-5), 4.27 (dd, J = 12.2, 5.6 Hz, 1H, H-6), 4.11 (dd, J = 12.2, 2.3 Hz, 1H, H-6), 4.0 (ddd, J = 12.7, 1H, H-6)5.1, 4.3 Hz, 1H, H-2), 2.45–2.41 (m, 1H, H-3), 2.08 (s, 3H, OAc), 2.02 (s, 3H, OAc), 1.9–1.82 (m, 1H, H-3). β isomer δ 7.59–7.57 (m, 2H, Ph), 7.30–7.27 (m, 3H, Ph), 4.70 (ddd, J = 12.9, 8.2, 4.8 Hz, 1H, H-4), 4.46 (d, J = 8.9 Hz, 1H, H-1), 4.24–4.16 (m, 2H, H-6, H-6), 3.62 (ddd, J = 9.9, 5.4, 2.7 Hz, 1H, H-5), 3.42 (ddd, J = 12.3, 8.4, 4.9 Hz, 1H, H-2), 2.60 (dt, J = 12.4, 4.9 Hz, 1H, H-3), 2.07 (s, 3H, OAc), 2.04 (s, 3H, OAc), 1.55 (q, J =11.8 Hz, 1H, H-3). ¹³C NMR (125 MHz, CDCl₃) α isomer δ 170.60 (OAc), 169.72 (OAc), 132.91 (Ph), 132.32 (Ph), 129.13 (Ph), 127.85 (Ph), 87.94 (C-1), 68.71 (C-5), 66.11 (C-4), 62.42 (C-6), 57.23 (C-2), 30.64 (C-3), 20.91 (OAc), 20.71 (OAc). β isomer δ 170.62 (OAc), 169.64 (OAc), 133.43 (Ph), 131.36 (Ph), 128.92 (Ph), 128.38 (Ph), 88.18 (C-1), 77.87 (C-5), 65.76 (C-4), 62.66 (C-6), 57.88 (C-2), 35.47 (C-3), 20.86 (OAc), 20.76 (OAc). HRMS (ESI, positive) calculated for $C_{27}H_{27}N_3O_5S^+ [M + Na]^+$ 528.1569, found 528.1559.

(2*R*,3*S*,5*R*)-5-azido-2-(hydroxymethyl)-6-(phenylthio)tetrahydro-2H-pyran-3-ol (31): Sodium methoxide (4.35 M in methanol, 95.4 μ L, 0.42 mmol) was added to a solution of compound **30** (0.759 g, 2.08 mmol) in methanol (40 mL). The reaction mixture was stirred at room temperature for 10 h, and neutralized with Amberlite IR-120 (H⁺ form). The mixture was filtered through a Celite pad, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 1:2) to give compound **31** (α/β mixtures, 0.450 g, 85%) as a white solid. As time goes, β isomer converted into α isomer. ¹H NMR (400 MHz, CDCl₃) α isomer δ 7.55–7.50 (m, 2H, Ph), 7.35–7.28 (m, 3H, Ph), 5.49 (d, J = 5.0 Hz, 1H, H-1), 4.09 (m, 1H, H-4), 3.91– 3.77 (m, 3H, H-2, H-6, H-5), 2.33–2.28 (m, 1H, H-3), 1.91–1.82 (m, 1H, H-3). β isomer δ 7.55–7.50 (m, 2H, Ph), 7.35–7.28 (m, 3H, Ph), 4.50 (d, J = 9.9 Hz, 1H, H-1), 3.91–3.77 (m, 2H, H-4, H-6), 3.74–3.68 (m, 1H, H-4), 3.39–3.33 (m, 1H, H-2), 3.32–3.27 (m, 1H, H-5), 2.50 (dt, J = 12.5, 4.7 Hz, 1H, H-3), 1.59 (q, J = 11.6 Hz, 1H, H-3). ¹³C NMR (125 MHz, CDCl₃) α isomer δ 133.21 (Ph), 132.51 (Ph), 129.16 (Ph), 127.84 (Ph), 88.2 (C-1), 72.9 (C-4), 65.84 (C-5), 62.5 (C-6), 57.8 (C-2), 33.7 (C-3). β isomer δ 133.06 (Ph), 132.83 (Ph), 129.08 (Ph), 128.26 (Ph), 88.2 (C-1), 81.9 (C-5), 65.4 (C-4), 62.7 (C-6), 58.6 (C-2), 33.9 (C-3). HRMS (ESI, positive ion mode) calculated for C₁₂H₁₅N₃O₃S⁺ [M + Na]⁺ 304.0726, found 304.0726.

(3R,5S,6R)-3-azido-5-(benzyloxy)-6-((benzyloxy)methyl)-2-(phenylthio)tetrahydro-2-

H-pyran (32): Sodium hydride (60% dispersion in mineral oil, 0.165 g, 3.91 mmol) was added to a solution of compound **31** (0.458 g, 1.63 mmol) in dry dimethylformamide (25 mL) under nitrogen gas at 0 °C. The reaction mixture was stirred at room temperature for 30 min. The reaction solution was cooled to 0 °C and benzyl bromide (0.48 mL) was added. The reaction mixture was stirred at room temperature for 8 h. After adding methanol (5 mL) at 0 °C, the solution was concentrated under reduced pressure. The residue was diluted with dichloromethane (30 mL) and washed with water (20 mL). The organic layer was dried over magnesium sulfate, filtered, and concentrated *in vacuo*. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 5:1) to give compound **32** (α/β mixtures, 0.721 g, 96%) as a clear liquid. ¹H NMR (400 MHz, CDCl₃) α isomer δ 7.56–7.53 (m, 2H, Ph), 7.36–7.29 (m, 7H, Ph),7.29–7.25 (m, 6H), 5.55 (d, *J* = 5.1 Hz, 1H, H-1), 4.62 (d, *J* = 12.0 Hz, 1H, Bn), 4.62 (d, *J* = 11.2 Hz, 1H, Bn), 4.48 (d, *J* = 12 Hz, 1H, Bn), 4.46 (d, *J* = 11.5 Hz, 1H, Bn), 4.35–4.31 (m, 1H, H-5), 3.90–3.81 (m, 2H, H-2, H-6), 3.74–3.68 (m, 2H, H-6, H-4), 2.46–4.40 (m, 1H, H-3), 201

1.89–1.80 (m, 1H, H-3). β isomer δ 7.62–7.60 (m, 2H, Bn), 7.36–7.24 (m, 13H, Bn), 4.62 (d, J = 12.1 Hz, 1H, Bn), 4.58 (d, J = 11.6 Hz, 1H, Bn), 4.56 (d, J = 11.9 Hz, 1H, Bn), 4.50 (d, J = 9.9 Hz, 1H, H-1), 4.45 (d, J = 11.5 Hz, 1H, Bn), 3.82 (dd, J = 10.9, 1.8 Hz, 1H, H-6), 3.73 (dd, J = 10.9, 4.5 Hz, 1H, H-6), 3.59–3.48 (m, 2H, H-4, H-5), 3.35 (ddd, J = 13.3, 8.4, 4.9 Hz, 1H, H-2), 2.61 (dt, J = 12.4, 4.5 Hz, 1H, H-3), 1.61–1.52 (m, 1H, H-3). ¹³C NMR (125 MHz, CD₃Cl₃) α isomer δ 138.03 (Ph), 137.71 (Ph), 133.63 (Ph), 132.24 (Ph), 128.99 (Ph), 128.33 (Ph), 127.90 (Ph), 127.81 (Ph), 127.77 (Ph), 127.65 (Ph), 127.51 (Ph), 88.5 (C-1), 73.4 (Bn), 71.62 (C-5), 71.48 (C-4), 71.34 (Bn), 68.49 (C-6), 57.8 (C-2), 31.0 (C-3). β isomer δ 138.35 (Ph), 127.72 (Ph), 132.89 (Ph), 132.19 (Ph), 128.87 (Ph), 128.44 (Ph), 128.31 (Ph), 127.88 (Ph), 127.72 (Ph), 127.59 (Ph), 127.53 (Ph), 88.26 (C-1), 81.36 (C-5), 73.41 (Bn), 71.46 (Bn), 71.21 (C-4), 69.1 (C-6), 58.48 (C-2), 35.96 (C-3). HRMS (ESI, positive) calculated for C₂₆H₂₇N₃O₃S⁺[M + Na]⁺ 484.1665, found 484.1669.

(15,25,3*R*,45,6*R*)-4,6-diazido-3-(((2*S*,3*R*,5*S*,6*R*)-3-azido-5-(benzyloxy)-6-((benzyloxy)methyl)tetrahydro-2H-pyran-2-yl)oxy)cyclohexane-1,2-diyl diacetate (34): Compound 33 (0.244 g, 0.821 mmol) and compound 32 (0.417 g, 0.903 mmol) were coevaporated from dry toluene three times and further dried under high vacuum overnight. A solution of compound 33 and compound 32 in anhydrous dichloromethane (10 mL) was added to a preactivated 4 Å molecular sieves and *N*-iodosuccinimide (0.295 g, 1.31 mmol). After stirring for 20 min at room temperature, the mixture was cooled to -78 °C. Trifluoromethanesulfonic acid (21 µL, 0.246 mmol) was slowly added, and the reaction was warmed to 0 °C slowly. The reaction was quenched by adding solid sodium sulfite (0.3 g), sodium bicarbonate (0.3 g) and a few drops of water. The color of solution turned to clear from dark yellow. The reaction mixture was diluted with dichloromethane, filtered, and washed with saturated solution of sodium bicarbonate and brine. The residue was dried over magnesium sulfate, filtered, concentrated under reduced pressure, and purified by flash column chromatography on silica gel (Hex:EtOAc =3:1) to give compound **34** as a clear oil. ¹H NMR (600 MHz, CDCl₃) δ 7.34–7.26 (m, 8H, Bn), 7.23– 7.21 (m, 2H, Bn), 5.14 (t, J = 9.8 Hz, 1H, H-5), 5.07 (d, J = 3.5 Hz, 1H, H-1'), 4.92 (t, J =10.0 Hz, 1H, H-6), 4.64 (d, J = 12.2 Hz, 1H, Bn), 4.59 (d, J = 11.4 Hz, 1H, Bn), 4.50 (d, J = 12.2 Hz, 1H, Bn), 4.43 (d, J = 11.3 Hz, 1H, Bn), 4.08 (ddd, J = 9.9, 3.0, 2.1 Hz, 1H, H-5'), 3.83 (dd, J = 10.7, 3.1 Hz, 1H, H-6'), 3.73–3.69 (m, 2H, H-4, H-4'), 3.66 (dd, J =10.7, 2.0 Hz, 1H, H-6'), 3.62 (ddd, J = 12.5, 10.0, 4.5 Hz, 1H, H-1), 3.43 (ddd. J = 12.4, 9.9, 4.6 Hz, 1H, H-3), 3.10 (ddd, J = 13.1, 4.5, 3.6 Hz, 1H, H-2'), 2.38 (dt, J = 13.4, 4.6 Hz, 1H, C-2), 2.32 (dt, J = 11.5, 4.5 Hz, 1H, C-3'), 2.08 (s, 3H, OAc), 2.07 (s, 3H, OAc), 1.98 (dt, J = 13.2, 11.3 Hz, 1H, C-3'), 1.58 (dt, J = 13.4, 12.5 Hz, 1H, C-2). ¹³C NMR (150 MHz, CDCl₃) δ 169.82 (OAc), 169.48 (OAc), 138.04 (Ph), 137.81 (Ph), 128.39 (Ph), 128.34 (Ph), 127.84 (Ph), 127.73 (Ph), 127.65 (Ph), 98.5 (C-1'), 77.7 (C-4), 74.15 (C-6), 73.84 (C-5), 73.53 (Bn), 71.63 (C-5'), 71.30 (Bn), 71.13 (C-4'), 68.05 (C-6'), 58.75 (C-3), 57.72 (C-1), 56.09 (C-2'), 31.82 (C-2), 27.83 (C-3'), 20.67 (OAc), 20.55 (OAc). HRMS (ESI, positive) calculated for $C_{30}H_{35}N_9O_8^+ [M + Na]^+ 672.2501$, found 672.2515. (1S,2R,3R,4S,6R)-4,6-diazido-3-(((2S,3R,5S,6R)-3-azido-5-(benzyloxy)-6-((benzyloxy-)methyl)tetrahydro-2H-pyran-2-yl)oxy)cyclohexane-1,2-diol (35): Sodium methoxide (4.35 M in methanol, 7.8 mL, 0.034 mmol) was added to a solution of compound 34 (0.221 g, 0.341 mmol) in methanol (10 mL). The reaction mixture was stirred at room

temperature for 4 h, and neutralized with Amberlite IR-120 (H⁺ form) resin. The mixture was filtered through a Celite pad, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:2) to give compound **35** (0.146 g, 76%) as a clear oil. ¹H NMR (500 MHz, CDCl₃) δ 7.37–7.28 (m, 8H, Bn), 7.24–7.21 (m, 2H, Bn), 5.20 (d, *J* = 3.5 Hz, 1H, H-1'), 4.64 (d, *J* = 12.0 Hz, 1H,

Bn), 4.60 (d, J = 11.4 Hz, 1H, Bn), 4.50 (d, J = 11.4 Hz, 1H, Bn), 4.43 (d, J = 12.1 Hz, 1H, Bn), 4.07 (ddd, J = 9.9, 3.8, 2.00 Hz, 1H, H-5'), 3.95 (d, J = 3.0 Hz, 1H, OH), 3.80 (dd, J = 10.6, 3.8 Hz, 1H, H-6'), 3.75-3.62 (m, 2H, H-4', H-6'), 3.57-3.43 (m, 2H, C-2')H-5), 3.43-3.35 (m, 3H, H-1,H-4, H-6), 3.31 (ddd, J = 12.3, 9.9, 4.5 Hz, 1H, H-3), 3.13(s, 1H, OH), 2.38 (ddt, J = 11.6, 4.6 Hz, 1H, H-3'), 2.28 (dt, J = 13.3, 4.2 Hz, 1H, H-2), 2.00 (dt. J = 12.9, 11.3 Hz, 1H, H-3'), 1.53–1.36 (m, 1H, H-2), ¹³C NMR (125 MHz, CDCl₃) § 137.87 (Ph), 137.76 (Ph), 128.45 (Ph), 128.39 (Ph), 127.97 (Ph), 127.88 (Ph), 127.76 (Ph), 97.59 (C-1'), 82.0 (C-6), 75.98 (C-5), 75.56 (C-4), 73.56 (Bn), 71.40 (C-5), 71.30 (Bn), 71.18 (C-4'), 68.37 (C-6'), 59.65 (C-1), 58.76 (C-3), 57.58 (C-2'), 32.09 (C-3'), 28.57 (C-2). HRMS (ESI, positive ion mode) calculated for $C_{26}H_{31}N_9O_6^+ [M + Na]^+$ 588.2290, found 588.2294.

Lividamine (13): Trimethylphosphine (1 M solution in THF, 1.12 mL, 1.123 mmol) was added to a solution of compound 35 (0.105 g, 0.187 mmol) in THF (6 mL) and 0.1 M NaOH (2 mL). The reaction mixture was stirred at 50 °C for 2 h, cooled to room temperature, and concentrated in vacuo. The residue was dissolved in a mixture of acetic acid (1 mL), water (3 mL) and methanol (3 mL). The solution was deaerated by evacuating the air inside of the flask and refilling it with nitrogen five times. Palladium hydroxide on carbon (20%, Degussa type, 0.1 g) was added and the solution was charged with hydrogen gas. The reaction mixture was stirred at room temperature under hydrogen gas for 24 hr. The solution was filtered through a Celite pad, washed with water and methanol, and concentrated under reduced pressure. The residue was purified by chromatography on Amberlite CG-50 (NH_4^+ form) resin (5-10% concentrated ammonium hydroxide in water), acidified with hydrochloric acid (pH ca. 4), and lyophilized to provide compound 13 as an oily solid. ¹H NMR (500 MHz, D₂O) δ 5.51 (d, J = 3.5 Hz, 1H, H-1'), 3.95–3.88 (m, 2H, H-5, H-5'), 3.84 (ddd, J = 9.3, 7.1, 2.5 Hz, 1H, H-6'), 3.80–

3.65 (m, 4H, H-6', H-2', H-6, H-4'), 3.65–3.57 (m, 2H, H-4, H-1), 3.38 (ddd, J = 12.5, 10.2, 4.3 Hz, 1H, H-2), 2.54 (dt, J = 12.7, 4.3 Hz, 1H, H-2), 2.31 (dt, J = 12.3, 4.4 Hz, 1H, H-3'), 1.99 (td, J = 12.0, 10.0 Hz, 1H, H-3'), 1.91 (q, J = 12.6 Hz, 1H, H-2). ¹³C NMR (125 MHz, D₂O) δ 95.3, 80.2, 75.13, 74.54, 72.26, 63.16, 60.04, 49.61, 48.86, 48.10, 29.76, 28.17. HRMS (ESI, positive) calculated for C₁₂H₂₅N₃O₆⁺ [M + Na]⁺ 330.1636, found 330.1645.

4.2.9. Synthesis of [4'-²H]-Paromamine (47)

The overall synthetic scheme is shown in Scheme 4-3.



Scheme 4-3. Synthetic scheme for the preparation of [4'-²H]-paromamine.

(4aR,6S,7R,8R,8aS)-7-azido-2-(4-methoxyphenyl)-6-(phenylthio)hexahydropyrano-

[3,2-d][1,3]dioxin-8-ol (37): Compound 36 was prepared as previously described.¹⁵⁹ Compound **36** (12.88 g, 43.32 mmol) and anisaldehyde dimethyl acetal (9.14 mL, 53.72 mmol) were dissolved in dimethylformamide (200 mL). p-Toluenesulfonic acid monohydrate (0.745 g, 4.33 mmol) was added and the flask was attached to a rotary evaporator, rotated, evacuated, and lowered into a water bath at approximately 60 °C to remove the methanol, which was formed during the reaction. After 1 h, triethylamine (10 mL) was added and all solvent was removed by evaporation. The residue was diluted with ethyl acetate (200 mL) and saturated aqueous solution of sodium bicarbonate (100 mL). The aqueous layer was extracted with dichloromethane (3 \times 200 mL). The combined organic extracts were washed with brine, dried over magnesium sulfate, filtered and concentrated under reduced pressure. The residue was purified by flash chromatography on silica gel (Hex/EtOAc = 2:1) to afford compound 37 (13.50 g, 72%) as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 7.61–7.54 (m, 2H, Ph), 7.40–7.26 (m, 5H, Ph), 6.92–6.87 (m, 2H, Ph), 5.49 (s, 1H, PMPCH), 4.55 (d, J = 10.2 Hz, 1H, H-1), 4.36 (dd, J = 10.2, 4.1 Hz, 1H, H-4), 3.80 (s, 3H, OMe), 3.79-3.73 (m, 2H, 2×H-6), 3.50-3.43(m, 2H, H-5, H-3), 3.36 (dd, J = 10.1, 9.0 Hz, 1H, H-2), 2.68 (d, J = 2.7 Hz, 1H, OH). ¹³C NMR (150 MHz, CD₃OD) δ 161.7, 134.9, 132.5, 130.8, 130.6, 130.0, 129.2, 115.2, 103.2, 88.0, 81.6, 75.2, 71.7, 69.7, 66.6, 56.7. HRMS (ESI, positive ion mode) calculated for $C_{20}H_{21}N_3O_5S^+[M+Na]^+$ 438.1100, found 438.1093.

(*4aR*,*6S*,*7R*,*8R*,*8aS*)-7-azido-8-(benzyloxy)-2-(4-methoxyphenyl)-6-(phenylthio)hexahydropyrano[3,2-d][1,3]dioxine (38): Sodium hydride (60% dispersion in mineral oil, 0.945g, 23.62 mmol) was added to a solution of compound 37 (7.55 g, 18.17 mmol) in anhydrous dimethylformamide (120 mL) at 0 °C under nitrogen atmosphere. The mixture was warmed to room temperature and was stirred for 30 min. The mixture was cooled to

0 °C and benzyl bromide (2.81 mL, 23.62 mmol) was added dropwise. The reaction was then allowed to stand for 12 h at room temperature. The mixture was treated with methanol at 0 °C and all the solvent was evaporated under reduced pressure. The residue was diluted with dichloromethane (200 mL) and washed with saturated aqueous solution of sodium bicarbonate (100 mL). The aqueous layer was extracted with dichloromethane $(3 \times 100 \text{ mL})$. The combined organic extracts were washed with brine, dried over magnesium sulfate, filtered and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:1) to afford compound **38** (6.68 g, 73 %) as a white solid. ¹H NMR (500 MHz, CDCl₃) δ 7.58–7.56 (m, 2H, Ph), 7.42–7.40 (m, 2H, Ph), 7.39–7.30 (m, 8H, Ph), 6.96–6.91 (m, 2H, Ph), 5.54 (s, 1H, PMPCH), 4.92 (d, J = 10.9 Hz, 1H, PhCH₂), 4.79 (d, J = 11.0 Hz, 1H, PhCH₂), 4.50 (d, J = 10.2, 1H, H-1), 4.38 (dd, J = 10.5, 5.5 Hz, 1H, H-4), 3.83 (s, 3H, OCH₃), 3.78 (t, J = 10.3 Hz, 1H, H-3), 3.65–3.61 (m, 2H, H-6), 3.47–3.43 (m, 1H, H-5), 3.37 (dd, J = 10.3, 8.8 Hz, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 160.24, 137.70, 134.01. 130.77, 129.67, 129.23, 128.83, 128.55, 128.54, 128.43, 128.11, 127.40, 127.40, 113.78, 101.37, 86.73, 81.36, 81.08, 75.29, 70.63, 68.56, 64.81, 55.42. HRMS (ESI, positive) calculated for $C_{27}H_{27}N_3O_5S^+$ $[M + Na]^+$ 528.1569, found 528.1559.

(2R,3S,4R,5R,6S)-5-azido-4-(benzyloxy)-2-(((4-methoxybenzyl)oxy)methyl)-6-(phenylthio)tetrahydro-2H-pyran-3-ol (39): A mixture of compound 38 (4.544 g, 8.99 mmol), sodium cyanoborohydride (4.518 g, 71.91 mmol), and freshly activated 4 Å molecular sieve (1 g) in dry dimethylformamide (90 mL) was stirred at room temperature under nitrogen for 30 min. Trifluoroacetic acid (6.87 mL, 89.88 mmol) was added dropwise at 0 °C. After stirring at room temperature for 12 h, the reaction mixture was filtered through a Celite pad and poured into ice-cold saturated aqueous solution of sodium bicarbonate (100 mL). The mixture was diluted with ethyl acetate (200 mL), washed with saturated aqueous solution of sodium bicarbonate (50 mL), brine, dried over magnesium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:1) to give compound **39** (4.326 g, 95%) as a clear oil. Compound **39** was unstable, only mass spectra were measured. HRMS (ESI, positive) calculated for $C_{27}H_{29}N_3O_5S^+$ [M + Na]⁺ 530.1726, found 530.1718.

(2R,3R,4R,5R,6S)-5-azido-4-(benzyloxy)-2-(((4-methoxybenzyl)oxy)methyl)-6-(phenylthio)tetrahydro-2H-pyran-3-d-3-ol (40): Dess-Martin periodinane (7.229 g, 17.04 mmol) was added to a mixture of compound 39 (4.326 g, 8.52 mmol) in dry dichloromethane (100 mL). After stirring for 12 h under nitrogen gas at room temperature, mixtures (60 mL) of 10% aqueous solution of sodium thiosulfate and saturated aqueous solution of sodium bicarbonate (1:1) were added to the reaction mixture. The reaction was allowed to stir for an additional 30 min. The organic layer was washed with aqueous solution of sodium thiosulfate/sodium bicarbonate (1:3) (3 \times 60 mL) and then brine, dried over magnesium sulfate, filtered, and concentrated *in vacuo*. The crude ketone compound was dissolved in anhydrous methanol (50 mL) and dry dichloromethane (50 mL). Sodium borodeuteride (0.428 g, 10.23 mmol) was added portionwise to this solution 0 °C. After stirring for 10 min at 0 °C, the reaction mixture was stirred at room temperature for an additional 3 h. The reaction was guenched by the addition of saturated aqueous solution of ammonium chloride (50 mL), and the aqueous layer was extracted with dichloromethane. The combined organic layers were dried over magnesium sulfate, filtered, and concentrated under reduced pressure. Purification by flash column chromatography on silica gel (Hex/EtOAc = 3:1) gave compound 40 (3.294g, 76%) as a white solid.

¹H NMR (600 MHz, CDCl₃) δ 7.60– 7.57 (m, 2H, Ph), 7.39– 7.32 (m, 4H, PH, 7.28–7.23 (m, 6H, PMP, Ph)), 6.89–6.85 (m, 2H, Ph), 4.69 (d, *J* = 11.5 Hz, 1H, Bn), 4.65 (d, *J* = 11.5 Hz, 1H, Bn), 4.52–4.47 (m, 2H, PMP), 4.37 (d, *J* = 10.2 Hz, 1H, H-1), 3.79 (s, 3H, OMe), 3.77–3.70 (m, 2H, 2×H-6), 3.64 (dd, *J* = 10.2, 9.5 Hz, 1H, H-2), 3.52 (t, *J* = 5.7 Hz, 1H, H-5), 3.37 (d, *J* = 9.5 Hz, 1H, H-3). ¹³C NMR (150 MHz, CDCl₃) δ 159.43, 137.08, 133.20, 131.78, 129.97, 129.55, 129.05, 128.71, 128.34, 128.23, 128.14, 113.95, 86.39, 81.12, 73.49, 72.01, 69.08, 61.08, 55.38. HRMS (ESI, positive) calculated for C₂₇H₂₈DN₃O₅S⁺[*M* + *Na*]⁺ 531.1788, found 531.1780.

(2R,3R,4R,5R,6S)-5-azido-4-(benzyloxy)-2-(((4-methoxybenzyl)oxy)methyl)-6-(phenylthio)tetrahydro-2H-pyran-3-yl-3-d methanesulfonate (41): Methanesulfonyl chloride (1.77 mL, 22.93 mmol) and dimethylaminopyridine (0.028 g, 0.23 mmol) was added to a solution of compound 40 (2.332 g, 4.59 mmol) in anhydrous pyridine (40 mL) at 0 °C. The reaction mixture was stirred at room temperature for 10 h. Methanol (10 mL) was added to quench the reaction at 0 °C. The mixture was concentrated under reduced pressure and co-evaporated with toluene (3×30 mL). The residue was diluted with dichloromethane (50 mL) and water (50 mL). The aqueous layer was extracted with dichloromethane (3×50 mL), dried over magnesium sulfate, filtered, and concentrated. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:1) to afford compound 41 (2.081g, 77%) as a light yellowish oil. ¹H NMR (400 MHz, CDCl₃) § 7.60–7.57 (m, 2H, Ph), 7.42–7.30 (m, 10H, Ph), 6.91–6.89 (m, 2H, Ph), 4.81 (d, J = 10.6, 1H, Bn, 4.58 (dd, J = 10.7, 2.0 Hz, 2H, PMP), 4.43 (d, J = 10.8 Hz, 1H, Bn), 4.41 (d, J = 9.56 Hz, 1H, H-1) 3.81 (s, 3H, OCH₃), 3.75–3.69 (m, 3H, 2×H-6, H-3), 3.54– 3.44 (m, 2H, H-2, H-5), 2.92 (s, 3H, SCH₃). ¹³C NMR (100 MHz, CDCl₃) δ 159.51,

136.40, 133.52, 131.26, 130.03, 129.77, 129.06, 128.72, 128.66, 128.54, 128.50, 113.95, 86.39, 79.36, 75.56, 73.60, 72.81, 67.51, 61.14, 55.37, 39.16. HRMS (ESI, positive) calculated for $C_{28}H_{30}DN_3O_7S_2^+$ [M + Na]⁺ 609.1564, found 609.1559.

(2R,3S,4R,5R,6S)-5-azido-4-(benzyloxy)-2-(((4-methoxybenzyl)oxy)methyl)-6-(phenylthio)tetrahydro-2H-pyran-3-yl-3-d benzoate (42): Sodium benzoate (1.214 g, 10.64 mmol) and 15-crown-5 (50 μ L) were added to a solution of compound 41 (2.081 g, 3.55 mmol) in anhydrous dimethylformamide (DMF, 100 mL). The reaction mixture was heated to reflux for 15 h. The solvent was evaporated under reduced pressure, and the residue was diluted with dichloromethane (50 mL) and water (50 mL). The aqueous layer was extracted with dichloromethane $(3 \times 50 \text{ mL})$, dried over magnesium sulfate, filtered, and concentrated. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:1) to give compound 42 (1.316 g, 60 %) as a white solid. ¹H NMR (600 MHz, CDCl₃) & 7.97-7.95 (m, 2H, Bz), 7.61-7.59 (m, 3H, Ph, Bz), 7.45-7.43 (m, 2H, Bz), 7.32–7.29 (m, 3H, Ph), 7.16–7.11 (m, 7H, PMB, Bn), 6.77–6.75 (m, 2H, PMB), 4.72 (d, J = 10.8 Hz, Bn), 4.58 (d, J = 10.8 Hz, Bn), 4.51 (d, J = 10.8 Hz, 1H, H-1), 4.41 (m, 10.1 Hz)2H, PMB), 3.76 (s, 3H, OCH₃), 3.72 (m, 1H, H-6), 3.66 (d, *J* = 9.3 Hz, 1H, H-3), 3.59 (m, 2H, H-5, H-6), 3.47 (dd, J = 10.1, 9.2 Hz, 1H, H-2). ¹³C NMR (150 MHz, CDCl₃) δ 165.31, 159.24, 137.09, 133.63, 133.51, 131.17, 130.00, 129.88, 129.51, 129.45, 129.20, 128.61, 128.59, 128.43, 128.37, 128.05, 113.78, 86.25, 82.48, 78.07, 75.51, 73.32, 69.23, 64.93, 55.35. HRMS (ESI, positive) calculated for $C_{34}H_{32}DN_3O_6S^+$ [M + Na]⁺ 632.2051, found 635.2013.

(2R,3S,4R,5R,6S)-5-azido-4-(benzyloxy)-2-(((4-methoxybenzyl)oxy)methyl)-6-(phenylthio)tetrahydro-2H-pyran-3-d-3-ol (43): Sodium methoxide (4.37 M in methanol, 0.18 mL, 0.78 mmol) was added dropwise to a solution of compound 42 (2.399 g, 3.92 mmol) in anhydrous methanol (10 mL) and anhydrous dichloromethane (40 mL). The

reaction mixture was stirred at 45 °C for 10 h. After cooling down to room temperature, the reaction mixture was neutralized with Amberlite IR-120 (H⁺ form) resin, filtered, and concentrated. The residue was purified by flash column chromatography (Hex/EtOAc = 3:1) on silica gel to give compound **43** (1.215 g, 61%) as a clear oil. Compound **34** was unstable so it was characterized only by mass spectroscopy. HRMS (ESI, positive) calculated for $C_{27}H_{28}DN_3O_5S^+$ [M + Na]⁺ 531.1788, found 531.1781.

(2S,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-(((4-methoxybenzyl)oxy)methyl)-2-

(phenylthio)tetrahydro-2H-pyran-5-d (44): Sodium hydride (60% dispersion in mineral oil, 0.215g, 5.39 mmol) was added portionwise to a solution of compound 43 (2.283 g, 4.49 mmol) in anhydrous DMF (50 mL) at 0 °C under nitrogen atmosphere. The mixture was warmed to room temperature and stirred for 30 min. The solution was cooled to 0 °C and benzyl bromide (0.64 mL, 5.39 mmol) was added dropwise. The reaction was continued with stirring for 9 h at room temperature. The mixture was quenched with methanol at 0 °C and the solvent was evaporated in vacuo. The residue was diluted with dichloromethane (100 mL) and washed with saturated aqueous solution of sodium bicarbonate (50 mL). The aqueous layer was repeatedly extracted with dichloromethane $(3 \times 50 \text{ mL})$. The combined extracts were washed with brine, dried over magnesium sulfate, filtered and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 4:1) to provide compound 44 (2.208 g, 82 %) as a white solid. ¹H NMR (500 MHz, CDCl₃) δ 7.70–7.68 (m, 2H, Ph), 7.41–7.33 (m, 13H, Ph), 7.28–7.27 (m, 2H, Ph), 6.96–6.94 (m, 2H, Ph), 4.95–4.90 (m, 2H, Bn), 4.86 (d, J = 10.9 Hz, 1H, Bn), 4.66–4.63 (m, 2H, PMB), 4.55 (d, J = 11.6 Hz, 1H, Bn), 4.50 (d, J = 10.1 Hz, 1H, H-1), 3.86 (s, 3H, OCH₃), 3.84-3.78 (m, 2H, H-6), 3.58 (d, J = 9.5 Hz, 1H, H-3), 3.53 (dd, J = 4.12, 2.1 Hz, 1H, H-5), 3.44 (t, J = 9.7, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 159.3, 138.0, 137.7, 131.4, 130.3, 129.4, 129.0, 128.6, 128.5, 128,4, 128.2, 128.1, 127.92, 127.88, 113.9, 86.1, 85.1, 79.4, 75.9, 75.0, 73.2, 68.4, 65.2, 55.3. HRMS (ESI, positive) calculated for $C_{34}H_{34}DN_3O_5S^+$ [M + Na]⁺ 621.2258, found 621.2250.

(1S,2S,3R,4S,6R)-4,6-diazido-3-(((2S,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-(((4methoxybenzyl)oxy)methyl)tetrahydro-2H-pyran-2-yl-5-d)oxy)cyclohexane-1,2-diyl diacetate (45): Compound 44 (2.208 g, 3.69 mmol) and compound 33 (0.916 g, 3.07 mmol) were co-evaporated from dry toluene (500 mL) three times and further dried under high vacuum overnight. A solution of compound 44 and 33 in anhydrous diethyl ether (30 mL) and anhydrous dichloromethane (10 mL) was added to a preactivated 4 Å MS (3 g). The mixture was stirred for 30 min at room temperature, and cooled to -40 °C. N-Iodosuccinimide (1.797 g, 7.99 mmol) was added, and the reaction mixture was stirred for 20 min. Trifluoromethanesulfonic acid (136 μ L, 1.53 mmol) was slowly added, and the reaction was warmed to -20 °C and kept stirring for 30 min. The reaction mixture was diluted with dichloromethane (100 mL), filtered through a Celite pad, and washed with 10% aqueous solution of sodium bisulfite (100 mL) and saturated aqueous solution of sodium bicarbonate (100 mL). The aqueous layers were extracted with dichloromethane $(3 \times 50 \text{ mL})$. The combined organic phase was washed with brine, dried over magnesium sulfate, filtered and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:2) to give compound **45** (2.212g, 92%) as a sticky liquid. ¹H NMR (400 MHz, CDCl₃) & 7.33–7.23 (m, 12H, Ph), 7.13–7.11 (m, 2H, Ph), 6.83 (d, J = 8.4 Hz, 2H, Ph), 5.14 (t, J = 9.2 Hz, 1H, H-5), 5.13 (d, J = 4 Hz, 1H, H-1'), 4.92 (t, J = 10.0 Hz, 1H, H-6), 4.84 (dd, J = 25.0, 10.5 Hz, 2H, PMB), 4.75 (d, J = 10.8 Hz, 1H, Bn), 4.60 (d, J = 11.6 Hz, 1H, Bn), 4.47 (d, J = 11.2 Hz, 1H, Bn), 4.40 (d, J = 11.6 Hz, 1H, Bn), 4.13 (s, 1H, H-5'), 3.93 (d, J = 10.4Hz, 1H, H-3'), 3.80 (dd, J = 10.6, 2.0 Hz, 1H, H-6'), 3.75 (s, 3H, -OCH₃), 3.66–3.58 (m, 212

3H, H-1, H-6', H-4), 3.43–3.37 (m, 1H, H-3), 3.35 (dd, J = 10.2, 3.6 Hz, 1H, H-2'), 2.40 (dt, J = 13.6, 4.4 Hz, 1H, H-2), 1.585 (q, J = 12.8 Hz, 1H, H-2). ¹³C NMR (100 MHz, CDCl₃) δ 169.95, 169.61, 159.43, 138.05, 137.92, 129.92, 129.77, 128.56, 128.49, 128.08, 127.97, 127.86, 127.77, 113.92, 99.50, 79.60, 78.40, 75.42, 74.97, 74.23, 73.67, 73.32, 71.72, 67.52, 63.25, 58.81, 57.83, 55.32, 31.94, 20.75, 20.68. HRMS (ESI, positive) calculated for C₃₈H₄₂DN₉O₁₀⁺ [M + Na]⁺ 809.3093, found 809.3079.

(1S,2R,3R,4S,6R)-4,6-diazido-3-(((2S,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-(((4methoxybenzyl)oxy)methyl)tetrahydro-2H-pyran-2-yl-5-d)oxy)cyclohexane-1,2-diol (46): Sodium methoxide (4.37 M in methanol, 64 µL, 0.28 mmol) was added dropwise to a solution of compound 45 (2.196 g, 2.79 mmol) in methanol (30 mL) and dichloromethane (10 mL). The reaction mixture was stirred at room temperature for 10 h. The reaction mixture was neutralized with Amberlite IR-120 (H⁺ form) resin, filtered, and concentrated. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:2) to give compound 46 (1.215 g, 92%) as a clear oil. ¹H NMR (500 MHz, CDCl₃) & 7.35–7.23 (m, 10H, Ph), 7.14–7.10 (m, 2H, Ph), 6.85–6.81 (m, 2H, Ph), 5.13 (d, J = 3.2 Hz, 1H, H-1), 4.88 (dd, J = 31.2, 10.8 Hz, 2H, PMB), 4.75 (d, J = 10.4Hz, 1H, Bn), 4.60 (d, J = 12.0 Hz, 1H, Bn), 4.48 (d, J = 10.8 Hz, 1H, Bn), 4.40 (d, J =11.6 Hz, 1H, Bn), 4.19 (s, 1H, OH), 4.07 (t, J = 2.0, 1H, H-5'), 3.96 (d, J = 10.4 Hz, 1H, H-3'), 3.78 (dd, J = 10.6, 2.8 Hz, 1H, H-6'), 3.75 (s, 3H, OCH₃), 3.66–3.62 (m, 2H, H-2', H-6'), 3.49-3.38 (m, 3H, H-4, H-5, H-1), 3.25 (dd, J = 5.6, 2.8 Hz, 2H, H-3, H-6), 2.33-2.28 (m, 1H, H-2), 1.49–1.45 (m, 1H, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 159.45, 137.90, 137.67, 129.87, 129.83, 128.62, 128.56, 128.18, 128.11, 127.98, 127.85, 113.92, 99.71, 84.12, 81.03, 75.98, 75.76, 75.46, 75.04, 73.35, 71.73, 67.61, 64.38, 59.76, 58.88, 55.34, 32.23. HRMS (ESI, positive) calculated for $C_{34}H_{38}DN_9O_8^+ [M + Na]^+$ 725.2882, found 725.2876.

[4'-²H]-Paromamine (47): Trimethylphosphine (1 M solution in THF, 2.21 mL) was added to a solution of compound 46 (0.259 g, 0.37 mmol) in THF (9 mL) and 0.1 M NaOH (3 mL). The reaction mixture was stirred at 50 °C for 2 h. After cooling down to room temperature, the solution was evaporated under reduced pressure. The material was dissolved in a mixture of acetic acid (2 mL), water (6 mL) and methanol (6 mL). The solution was deaerated by evacuating the air in the flask and refilling it with nitrogen five times. Palladium hydroxide on carbon (20%, Degussa type, 0.095 g) was added and the solution was charged with hydrogen. The reaction mixture was stirred at room temperature under hydrogen gas for 24 h. The solution was filtered through a Celite pad and concentrated. The residue was purified by chromatography on Amberlite CG-50 resin (NH₄⁺ form) (2.5% concentrated ammonium hydroxide in water), concentrated, acidified with hydrochloric acid (pH ca. 4), and lyophilized to provide compound 47 (0.086 g, 73%) as an oily solid. ¹H NMR (400 MHz, D₂O) δ 5.65 (d, J = 4.0 Hz, 1H, H-1'). 3.96– 3.79 (m, 4H, H-3', H-5, H-6', H-5'), 3.75-3.72 (m, 1H, H-6'), 3.67 (dd, J = 12.0, 6.5 Hz,1H, H-6), 3.61-3.55 (m, 2H, H-1, H-4), 3.45 (dd, J = 12.2, 6.6, 1H, H-2'), 3.37-3.31 (m, 1H, H-3), 2.50 (dt, J = 12.6, 4.3 Hz, 1H, H-2), 1.86 (q, J = 12.6 Hz, 1H, H-2). ¹³C NMR (150 MHz, D₂O) § 97.1 (C-1'), 80.4 (C-5), 74.7 (C-6), 73.6 (C-5'), 72.3 (C-4), 69.0 (C-4'), 60.3 (C-6'), 54.0 (C-2'), 49.6 (C-3), 48.8 (C-1), 28.3 (C-2). HRMS (ESI, positive) calculated for $C_{12}H_{24}DN_{3}O_{7}^{+}[M+H]^{+}$ 325.1829, found 325.1834.

4.2.10. Synthesis of [3'-²H]-Paromamine (65)

The overall synthetic scheme is shown in Scheme 4-4. Yu-Cheng Yeh synthesized this compound.



Scheme 4-4. Synthetic scheme for the preparation of $[3'-{}^{2}H]$ -paromamine.

(*3aR*,*5S*,*6S*)-5-((*S*)-2,2-dimethyl-1,3-dioxolan-4-yl)-2,2-dimethyltetrahydrofuro[2,3-d][1,3]dioxol-6-d-6-ol (52): Compound 51 was prepared as previously described.¹⁷⁰ Compound 51 (1.059 g, 2.90 mmol) was dissolved in 5% solution of sodium hydroxide in methanol (0.1 M, 29 mL), and the reaction mixture was stirred for 2 h. The solvent was removed under reduced pressure, and the residue was dissolved in dichloromethane (40

mL). The organic layer was washed with water (40 mL × 2), brined, dried over sodium sulfate, filtered, and concentrated. The crude product was purified by flash column chromatography on silica gel (Hex/EtOAc = 1:1) to give compound **52** (0.652 g, 86%) as a white solid. ¹H NMR (500 MHz, CDCl₃) δ 5.95 (d, *J* = 3.6 Hz, 1H, H-1), 4.53 (d, *J* = 3.7 Hz, 1H, H-2), 4.34 (ddd, *J* = 7.6, 6.2, 5.4 Hz, 1H, H-5), 4.17 (dd, *J* = 8.6, 6.2 Hz, 1H, H-6), 4.07 (d, *J* = 7.6 Hz, 1H, H-4), 3.98 (dd, *J* = 8.6, 5.4 Hz, 1H, H-6), 2.48 (s, 1H, OH), 1.50 (d, *J* = 0.7 Hz, 3H, Me), 1.44 (d, *J* = 0.7 Hz, 3H, Me), 1.36 (d, *J* = 0.7 Hz, 3H, Me), 1.32 (d, *J* = 0.7 Hz, 3H, Me). ¹³C NMR (125 MHz, CDCl₃) δ 111.99(acetal), 109.82 (acetal), 105.47 (C-1), 85.18 (C-2), 81.21 (C-4), 73.71 (C-5), 67.83 (C-6), 27.01 (Me), 26.92 (Me), 26.34 (Me), 25.29 (Me). HRMS (ESI, positive) calculated for C₁₂H₁₉DO₆⁺ [*M* + *Na*]⁺ 284.1215, found 284.1215.

(*3aR*,*5R*,*6S*)-6-(benzyloxy)-5-((*S*)-2,2-dimethyl-1,3-dioxolan-4-yl)-2,2-dimethyltetrahydrofuro[2,3-d][1,3]dioxole-6-d (53): Sodium hydride (60% dispersion in mineral oil, 1.15 g, 28.70 mmol) was added to a solution of compound 52 (6.25 g, 23.92 mmol) in dimethylformamide (205 mL) at 0 °C. The reaction mixture was stirred at room temperature for 30 min. Benzyl bromide (3.5 mL, 28.70 mmol) was added at 0 °C, and the reaction mixture was stirred for 15 h. Methanol (10 mL) was added slowly at 0 °C. The reaction solution was concentrated under reduced pressure. The residue was diluted with dichloromethane (200 mL), and washed with water (150 mL × 2). The combined aqueous layers were extracted with dichloromethane (200 mL × 2). The combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The crude compound 53 (8.047 g, 96%) as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 7.38–7.28 (m, 5H, Ph), 5.90 (d, *J* = 3.7 Hz, 1H), 4.72–4.61 (m, 2H, 2×Bn), 4.58 (d, *J* = 3.7 Hz, 1H, H-2), 4.37 (dt, *J* = 7.7, 6.0 216 Hz, 1H, H-5), 4.17–4.08 (m, 2H, H-6, H-4), 4.05–3.94 (m, 1H, H-6), 1.49 (d, J = 0.6 Hz, 3H, OAc), 1.43 (d, J = 0.7 Hz, 3H, OAc), 1.37 (d, J = 0.7 Hz, 3H, OAc), 1.31 (s, 3H, OAc). HRMS (ESI, positive) calculated for $C_{19}H_{25}DO_6^+$ [M + Na]⁺ 374.1684, found 374.1690.

(3R,4S,5R,6R)-6-(acetoxymethyl)-4-(benzyloxy)tetrahydro-2H-pyran-2,3,5-triyl-4-d triacetate (54): Compound 53 (8.047 g, 22.90 mmol) was dissolved in acetic acid (69 mL) and H₂O (46 mL), and the reaction mixture was stirred at 90 °C for 12 h. The reaction solution was evaporated under reduced pressure and co-evaporated with toluene three times. The residue was dissolved in pyridine (0.4 M, 57 mL) and acetic anhydride (0.4 M, 57 mL) was added at 0 °C. The reaction mixture was stirred at room temperature for 6 h. The reaction solution was evaporated under reduced pressure and co-evaporated with toluene three times. The residue was dissolved in ethyl acetate (100 mL) and washed with saturated aqueous solution of sodium bicarbonate (60 mL \times 2) and brine. The organic layer was dried over magnesium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 1:1) to provide compound 54 (9.47 g, 94%) as a clear oil. ¹H NMR (500 MHz, CDCl₃) α isomer δ 7.36–7.21 (m, 5H, Ph), 6.32 (d, J = 3.6 Hz, 1H), 5.19 – 5.12 (m, 1H, H-4), 5.06 (d, J = 3.7 Hz, 1H, H-2), 4.71 (d, J = 11.8 Hz, 1H, Bn), 4.65–4.56 (m, 1H), Bn, 4.21-4.18 (m, 1H, H-6), 4.10-4.05 (m, 1H, H-6), 4.02 (ddd, J = 10.3, 4.5, 2.4Hz, 1H, H-5), 2.17 (s, 3H, OAc), 2.08 (s, 3H, OAc), 1.99 (s, 3H, OAc), 1.97 (s, 3H). β isomer δ 7.36–7.21 (m, 5H, Ph), 5.66 (d, J = 8.2 Hz, 1H, H-1), 5.19–5.12 (m, 2H, H-2, H-4), 4.65–4.56 (m, 2H, $2 \times Bn$), 4.21 (ddd, J = 12.5, 9.9, 4.7 Hz, 1H, H-6), 4.08 (td, J =12.4, 2.4 Hz, 1H, H-6), 3.74 (ddt, J = 10.1, 4.9, 2.5 Hz, 1H, H-5), 2.10 (s, 3H, OAc), 2.07 (s, 3H, OAc), 1.98 (s, 6H, 2×OAc). ¹³C (125 MHz, CDCl₃) α isomer 170.75 (O(C=O)), 169.61 (O(C=O)), 169.29 (O(C=O)), 168.79 (O(C=O)), 137.99 (Ph), 128.49 (Ph), 127.87

(Ph), 127.55 (Ph), 89.54 (C-1), 74.77 (Bn), 71.50 (C-2), 70.32 (C-5), 69.07 (C-4), 61.87 (C-6), 20.93 (OAc), 20.76 (OAc), 20.74 (OAc), 20.62 (OAc). β isomer 170.72 (O(C=O)), 169.32 (O(C=O)), 169.24 (O(C=O)), 169.11 (O(C=O)), 137.63 (Ph), 128.53 (Ph), 127.99 (Ph), 127.83 (Ph), 92.04(C-1), 74.18 (Bn), 73.04 (C-5), 71.55 (C-2), 69.07 (C-4), 61.87 (C-6), 20.89 (OAc), 20.77 (OAc), 20.76 (OAc), 20.73 (OAc). HRMS (ESI, positive) calculated for C₂₁H₂₅DO₁₀⁺ [*M* + *Na*]⁺ 462.1481, found 462.1487.

(2R,3R,4S,5R,6S)-2-(acetoxymethyl)-4-(benzyloxy)-6-(phenylthio)tetrahydro-2H-

pyran-3,5-diyl-4-d diacetate (55): Thiophenol (2.43 mL, 23.7 mmol) and boron trifluoride diethyl etherate (6.3 mL, 23.7 mmol) were added to a solution of compound 54 (9.47 g, 21.55 mmol) in dichloromethane (215 mL) at 0 °C. The reaction mixture was warmed to room temperature and stirred for 5 h. The reaction mixture was guenched by adding saturated aqueous solution of sodium bicarbonate (150 mL) at 0 °C. The aqueous layer was extracted with dichloromethane (100 mL \times 3). The combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The crude product was purified by flash column chromatography on silica gel (Hex/EtOAc = 2:1) to provide compound 55 (6.35 g, 60%) as a clear solid. ¹H NMR (500 MHz, CDCl₃) δ 7.52–7.45 (m, 2H, Ph), 7.34–7.25 (m, 6H, Ph), 7.23–7.19 (m, 2H, Ph), 5.08-5.04 (m, 2H, H-4, H-2), 4.64 (d, J = 10.0 Hz, 1H, H-1), 4.63-4.57 (m, 2H, $2 \times Bn$, 4.19–4.15 (m, 2H, $2 \times H$ -6), 3.63 (ddd, J = 10.1, 5.1, 3.2 Hz, 1H, H-5), 2.07 (s, 3H, OAc), 2.04 (s, 3H, OAc), 1.96 (s, 3H, OAc). ¹³C NMR (125 MHz, CDCl₃) δ 170.76 (O(C=O)), 169.44 (O(C=O)), 169.31 (O(C=O)), 137.80 (Ph), 132.71 (Ph), 132.63 (Ph), 128.96 (Ph), 128.57 (Ph), 128.16 (Ph), 127.99 (Ph), 127.88 (Ph), 86.34 (C-1), 76.21 (C-5), 74.27 (Bn), 71.41 (C-2), 69.69 (C-4), 62.67 (C-6), 21.06 (OAc), 20.87 (OAc), 20.84 (OAc). HRMS (ESI, positive) calculated for $C_{25}H_{27}DO_8S^+$ $[M + Na]^+$ 512.1460, found 512.1465.

(2R,3R,4S,5R,6S)-4-(benzyloxy)-2-(hydroxymethyl)-6-(phenylthio)tetrahydro-2H-

pyran-4-d-3,5-diol (56): Sodium methoxide (5.4 M solution in methanol, 0.24 mL, 1.29 mmol) was added to a solution of compound **55** (6.35 g, 12.97 mmol) in methanol (130 mL). The reaction mixture was stirred for 10 h, neutralized with Amberlite IR-120 (H⁺ form) resin, filtered, and concentrated under reduced pressure. The crude product was purified by flash column chromatography on silica gel (EtOAc/DCM = 1:1) to provide compound **56** (4.37 g, 93%) as a white solid. No characterization of this compound was conducted.

(4aR,6S,7R,8R,8aR)-8-(benzyloxy)-2-phenyl-6-(phenylthio)hexahydropyrano[3,2-

d][1,3]dioxin-8-d-7-ol (57): p-Toluenesulfonic acid monohydrate (0.247 g, 1.30 mmol) and compound 56 (4.72 g, 12.99 mmol) was co-evaporated from dry toluene (30 mL) three times and further dried under high vacuum overnight. Compound 56 and ptoluenesulfonic acid were dissolved in dimethylformamide, and benzaldehyde dimethyl acetal (3.90 mL, 26 mmol) was added. The reaction mixture stirred at 70 °C for 10 h under house vacuum. The solvent was removed under reduced pressure. The residue was diluted with dichloromethane (70 mL), and washed with saturated aqueous solution of sodium bicarbonate (60 mL \times 2). The combined aqueous layers were extracted with dichloromethane (70 mL \times 2). The combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated. The crude product was purified by flash column chromatography on silica gel (Hex:EtOAc = 1:1) to provide compound 57 (4.44 g, 76%) as a white solid. ¹H NMR (600 MHz, CDCl₃) δ 7.56–7.53 (m, 2H, Ph), 7.51-7.47 (m, 2H, Ph), 7.40-7.35 (m, 6H, Ph), 7.35-7.31 (m, 5H, Ph), 5.58 (s, 1H, PhCHO₂), 4.96 (d, J = 11.5 Hz, 1H, Bn), 4.80 (d, J = 11.6 Hz, 1H, Bn), 4.64 (d, J = 9.8Hz, 1H, H-1), 4.39 (dd, J = 10.5, 5.0 Hz, 1H, H-6), 3.80 (t, J = 10.3 Hz, 1H, H-6), 3.66 219

(d, J = 9.4 Hz, 1H, H-4), 3.58-3.49 (m, 2H, H-2, H-5).¹³C NMR (150 MHz, CDCl₃) δ 138.30 (Ph), 137.31 (Ph), 133.32 (Ph), 131.44 (Ph), 129.17 (Ph), 129.15 (Ph), 128.60 (Ph), 128.50 (Ph), 128.39 (Ph), 128.25 (Ph), 128.02 (Ph), 126.13 (Ph), 101.40 (PhCHO₂), 88.59 (C-1), 81.17 (C-4), 74.91 (Bn), 72.31 (C-2), 70.87 (C-5), 68.76 (C-6). HRMS (ESI, positive) calculated for $C_{26}H_{25}DO_5S^+ [M + Na]^+ 474.1456$, found 474.1459.

(4aR,6S,7S,8R,8aR)-8-(benzyloxy)-2-phenyl-6-(phenylthio)hexahydropyrano[3,2-

d][1,3]dioxin-8-d-7-ol (58): Dess-Martin periodinane (8.341 g, 19.66 mmol) was added to a solution of compound 57 (4.44 g, 9.83 mmol) in dichloromethane (98 mL). The reaction mixture was stirred for 12 h. Saturated aqueous solutions of sodium sulfite (50 mL) and sodium bicarbonate (50 mL) were added, and the reaction mixture was stirred for additional 30 min. The organic layer was washed with saturated aqueous solutions of sodium sulfite (30 mL) and sodium bicarbonate (90 mL) three timed. The combined organic layers were washed with brine, dried over magnesium sulfate, filtered, and concentrated. The resulting ketone was dissolved in methanol (98 mL), and cooled to 0 °C. Sodium borohydride (0.470 g, 11.80 mmol) was added portionwise during 10 min. The reaction mixture was stirred at 0 °C for 10 min and at room temperature for 1 h. The reaction was quenched by adding saturated aqueous solution of ammonium chloride (50 mL). The solvent was evaporated under reduced pressure. The residue was dissolved in dichloromethane (100 mL) and washed with water (100 mL). The aqueous layer was extracted with dichloromethane (120 mL \times 3). The combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated. The crude product was purified by flash column chromatography on silica gel (EtOAc/DCM = 1:9) to give compound **58** (2.51 g, 57%) as a white solid. Compound **58** was unstable, only ¹H NMR and mass spectra were measured. ¹H NMR (400 MHz, CDCl₃) δ 7.55–7.45 (m, 4H, Ph), 7.43–7.26 (m, 11H, Ph), 5.62 (s, 1H, PhCHO₂), 4.93–4.81 (m, 2H, H-1, Bn), 4.76 (d, J =

12.1 Hz, 1H, Bn), 4.38–4.28 (m, 2H, H-6, H-2), 4.17 (d, J = 9.4 Hz, 1H, H-4), 3.93 (t, J = 10.3 Hz, 1H, H-6), 3.49–3.36 (m, 1H, H-5), 2.82 (d, J = 1.8 Hz, 1H, OH). HRMS (ESI, positive) calculated for C₂₆H₂₅DO₅S⁺ [M + Na]⁺ 474.1456, found 474.1458.

(4aR,6S,7S,8S,8aR)-8-(benzyloxy)-2-phenyl-6-(phenylthio)hexahydropyrano[3,2d][1,3]dioxin-7-yl-8-d trifluoromethanesulfonate (59): Trifluoromethanesulfonic anhydride (4.69 mL, 27.9 mmol) was added to a solution of compound 58 (2.52 g, 5.58 mmol) in dry pyridine (28 mL) and dry dichloromethane (28 mL) at -15 °C. The reaction mixture was warmed to room temperature, and diluted with ethyl acetate (50 mL). The reaction was quenched by adding few drops of water. The reaction solution was washed with saturated aqueous solutions of ammonium chloride (40 mL) and sodium bicarbonate (40 mL). The organic layer was washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The crude product was used in next step without further purification.

(4aR,6S,7R,8R,8aS)-7-azido-8-(benzyloxy)-2-phenyl-6-(phenylthio)hexahydro-

pyrano[3,2-d][1,3]dioxine-8-d (60): Sodium azide (1.814 g, 27.9 mmol) was added to a solution of the crude compound 59 in dimethylformamide (56 mL) at 0 °C. The reaction mixture was stirred at room temperature for 10 h, and diluted with ethyl acetate (50 mL). The organic layer was washed with water (40 mL × 3). The combined aqueous layers were extracted with ethyl acetate (60 mL × 2). The combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The crude product was purified by flash column chromatography on silica gel (Hex/EtOAc = 4:1) to provide compound 60 (2.32 g, 87%) as a clear oil. Compound 60 was unstable, only ¹H NMR and mass spectra were measured. ¹H NMR (400 MHz, CDCl₃) δ 7.60–7.54 (m, 2H, Ph), 7.50–7.44 (m, 2H, Ph), 7.42–7.28 (m, 11H, Ph), 5.57 (s, 1H, PhCHO₂), 4.91 (d, *J* = 10.9 Hz, 1H, Bn), 4.78 (d, *J* = 10.9 Hz, 1H, Bn), 4.49 (d, *J* = 221

10.2 Hz, 1H, H-1), 4.39 (dd, J = 10.5, 5.0 Hz, 1H, H-6), 3.84–3.75 (m, 1H, H-6), 3.63 (d, *J* = 9.4 Hz, 1H, H-4), 3.46 (ddd, *J* = 10.0, 9.4, 5.0 Hz, 1H, H-5), 3.36 (d, *J* = 10.3 Hz, 1H, H-2). HRMS (ESI, positive) calculated for $C_{26}H_{24}DN_3O_5S^+$ $[M + Na]^+$ 499.1521, found 499.1518.

(2R,3S,4R,5R,6S)-5-azido-4-(benzyloxy)-2-(hydroxymethyl)-6-(phenylthio)tetra-

hydro-2H-pyran-4-d-3-ol (61): p-toluenesulfonic acid monohydrate (0.046 g, 0.242 mmol) was added to a solution of compound 60 (1.154 g, 2.42 mmol) in methanol (24 mL). The reaction mixture was stirred for 14 h. The solvent was removed under reduced pressure, and the residue was diluted with ethyl acetate (25 mL). The organic layer was washed with saturated aqueous solution of sodium bicarbonate (20 mL \times 2). The combined aqueous layers were extracted with ethyl acetate (30 mL \times 2). The combined organic layers were washed with brine, dried over magnesium sulfate, filtered, and concentrated under reduced pressure. The crude product was purified by flash column chromatography on silica gel (Hex/EtOAc = 1:1) to give compound **61** (0.895 g, 95%) as a white solid. ¹H NMR (500 MHz, CDCl₃) & 7.57-7.51 (m, 2H, Ph), 7.39-7.35 (m, 4H, Ph), 7.34–7.32 (m, 4H, Ph), 4.95 (d, J = 11.2 Hz, 1H, Bn), 4.76 (d, J = 11.3 Hz, 1H, Bn), 4.48 (d, J = 10.1 Hz, 1H, H-1), 3.87 (dd, J = 12.0, 3.3 Hz, 1H, H-6), 3.76 (dd, J = 12.0, 4.9 Hz, 1H, H-6), 3.54 (d, J = 9.7 Hz, 1H, H-4), 3.32 (ddd, J = 10.1, 5.0, 3.6 Hz, 2H, H-5, H-2). ¹³C NMR (125 MHz, CDCl₃) δ 137.87 (Ph), 133.44 (Ph), 131.34 (Ph), 129.27 (Ph), 128.89 (Ph), 128.65 (Ph), 128.41 (Ph), 128.29 (Ph), 86.56 (C-1), 79.48 (C-5), 75.58 (Bn), 70.31 (C-4), 65.00 (C-2), 62.57 (C-6). HRMS (ESI, positive) calculated for $C_{19}H_{20}DN_{3}O_{4}S^{+}[M+Na]^{+}411.1208$, found 411.1212.

(2S,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-((benzyloxy)methyl)-2-(phenylthio)tetrahydro-2H-pyran-4-d (62): Sodium hydride (60% dispersion in mineral oil, 0.216 g, 5.41 mmol) was added to a solution of compound 61 (0.841 g, 2.16 mmol) in

dimethylformamide (22 mL) at 0 °C. The reaction mixture was stirred at room temperature for 30 min. Benzyl bromide (0.65 mL, 5.41 mmol) was added at 0 °C, and the reaction mixture was stirred for 11 h. Methanol (5 mL) was added slowly at 0 °C. The reaction solution was concentrated under reduced pressure. The residue was diluted with dichloromethane (30 mL), and washed with water (20 mL \times 3). The combined aqueous layers were extracted with dichloromethane (30 mL \times 3). The combined organic layers were washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The crude compound was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:1) to provide compound 62 (1.189 g, 97%) as a white solid. ¹H NMR (600 MHz, CDCl₃) δ 7.62–7.58 (m, 2H, Ph), 7.36–7.23 (m, 16H, Ph), 7.22–7.18 (m, 2H, Ph), 4.86 (d, J = 10.5 Hz, 1H, Bn), 4.82 (d, J = 10.6 Hz, 1H, Bn), 4.78 (d, J =10.9 Hz, 1H, Bn), 4.61 (d, J = 12.0 Hz, 1H, Bn), 4.58 (d, J = 10.9 Hz, 1H, Bn), 4.54 (d, J= 12.0 Hz, 1H, Bn), 4.41 (d, J = 10.2 Hz, 1H, H-1), 3.77 (dd, J = 10.9, 2.0 Hz, 1H, H-6), 3.74 (dd, J = 11.0, 4.2 Hz, 1H, H-6), 3.60 (d, J = 9.8 Hz, 1H H-4), 3.47 (ddd, J = 9.8, 4.2)2.0 Hz, 1H, H-5), 3.34 (d, J = 10.2 Hz, 1H, H-2). ¹³C NMR (150 MHz, CDCl₃) δ 138.33 (Ph), 137.98 (Ph), 137.74 (Ph), 133.76 (Ph), 131.29 (Ph), 129.11 (Ph), 128.65 (Ph), 128.60 (Ph), 128.50 (Ph), 128.48 (Ph), 128.33 (Ph), 128.14 (Ph), 128.02 (Ph), 127.97 (Ph), 127.74 (Ph), 127.70 (Ph), 86.08 (C-1), 79.46 (C-5), 77.60 (C-4), 75.97 (Bn), 75.18 (Bn), 73.57 (Bn), 68.87 (C-6) 65.11 (C-2). HRMS (ESI, positive) calculated for $C_{33}H_{32}DN_{3}O_{4}S^{+}[M+Na]^{+}$ 591.2147, found 591.2149.

(1S,2S,3R,4S,6R)-4,6-diazido-3-(((2S,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-((benzyloxy)methyl)tetrahydro-2H-pyran-2-yl-4-d)oxy)cyclohexane-1,2-diyl diacetate (63): Compound 33 (0.087 g, 0.293 mmol) and compound 62 (0.200 g, 0.352 mmol) were co-evaporated from dry toluene three times and further dried under high vacuum overnight. A solution of compound 33 and 62 in anhydrous diethyl ether (3 mL) and 223

anhydrous dichloromethane (1 mL) was then added to preactivated 4 Å molecular sieves. After stirring for 30 min at room temperature, the mixture was cooled to -40 °C. N-Iodosuccinimide (0.171 g, 0.762 mmol) was added, and the reaction mixture was stirred for 20 min. Trifluoromethanesulfonic acid (0.013 mL, 0.147 mmol) was slowly added, and the reaction was warmed to -20 °C and kept stirring for 30 min. Sodium bisulfite (0.1 g), sodium bicarbonate (0.1 g), and water (1 mL) were added at 0 °C, and the mixture was stirred for 10 min at room temperature. The reaction mixture was diluted with dichloromethane (10 mL), filtered through a Celite pad, and washed with a saturated solution of aqueous sodium bicarbonate (15 mL). The aqueous layers were extracted with dichloromethane (15 mL \times 3), and the combined organic phase was washed with brine, dried over sodium sulfate, filtered, and concentrated under reduced pressure. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 2:1) to give compound **63** (0.135 g, 61%) as a clear oil. ¹H NMR (600 MHz, CDCl₃) δ 7.36–7.26 (m, 13H, Ph), 7.18–7.14 (m, 2H, Ph), 5.17–5.11 (m, 2H, H-1', H-5), 4.92 (t, J = 10.0 Hz, 1H, H-6), 4.88 (d, J = 10.5 Hz, 1H, Bn), 4.82 (d, J = 10.7 Hz, 1H, Bn), 4.79 (d, J = 10.9 Hz, 1H, Bn), 4.64 (d, J = 12.1 Hz, 1H, Bn), 4.54 (d, J = 10.9 Hz, 1H, Bn), 4.49 (d, J = 12.1Hz, 1H, Bn), 4.18–4.12 (m, 1H, H-5'), 3.83 (dd, J = 10.8, 2.9 Hz, 1H, H-6'), 3.78 (d, J = 10.1 Hz, 1H, H-4'), 3.68 (dd, J = 10.8, 2.1 Hz, 1H, H-6'), 3.66–3.58 (m, 2H, H-4, H-1), 3.41 (ddd, J = 12.5, 10.0, 4.6 Hz, 1H, H-3), 3.36 (d, J = 3.8 Hz, 1H, H-2'), 2.39 (dt, J =13.4, 4.5 Hz, 1H, H-2), 2.09 (s, 3H, OAc), 2.08 (s, 3H, OAc), 1.65–1.54 (m, 1H, H-2). ¹³C NMR (150 MHz, CDCl₃) δ 169.99 (O(C=O)), 169.65 (O(C=O)), 138.04 (Ph), 137.94 (Ph), 137.91 (Ph), 128.60 (Ph), 128.56 (Ph), 128.54 (Ph), 128.12 (Ph), 128.04 (Ph), 128.02 (Ph), 127.93 (Ph), 127.90 (Ph), 127.85 (Ph), 99.51 (C-1'), 78.44 (C-4), 78.19 (C-4'), 75.43 (Bn), 75.14 (Bn), 74.24 (C-6), 73.74 (Bn), 73.68 (C-5), 71.83 (C-5'), 68.08 (C-6'), 63.23 (C-2'), 58.82 (C-3), 57.85 (C-1), 31.97 (C-2), 20.79 (OAc), 20.72 (OAc).

HRMS (ESI, positive) calculated for $C_{37}H_{40}DN_9O_9^+$ $[M + Na]^+$ 779.2982, found 779.3050.

(1S,2R,3R,4S,6R)-4,6-diazido-3-(((2S,3R,4R,5S,6R)-3-azido-4,5-bis(benzyloxy)-6-

((benzyloxy)methyl)tetrahydro-2H-pyran-2-yl-4-d)oxy)cyclohexane-1,2-diol (64): Sodium methoxide (5.4 M in methanol, 0.006 mL, 0.032 mmol) was added dropwise to a solution of compound 63 (0.245 g, 0.324 mmol) in methanol (3.2 mL). The reaction mixture was stirred at room temperature for 8 h and then neutralized with Amberlite IR-120 (H⁺ form) resin, filtered and concentrated. The residue was purified by flash column chromatography on silica gel (Hex/EtOAc = 3:2) to give compound **64** (0.187 g, 86%) as a pale yellow oil. ¹H NMR (400 MHz, CDCl₃) δ 7.37–7.27 (m, 13H, Ph), 7.18–7.12 (m, 2H, Ph), 5.15 (d, J = 3.6 Hz, 1H, H-1'), 4.93 (d, J = 10.6 Hz, 1H, Bn), 4.85 (d, J = 10.6Hz, 1H, Bn), 4.79 (d, J = 10.8 Hz, 1H, Bn), 4.65 (d, J = 12.0 Hz, 1H, Bn), 4.54 (d, J =10.8 Hz, 1H, Bn), 4.49 (d, J = 12.0 Hz, 1H, Bn), 4.21 (s, 1H, OH), 4.10 (ddd, J = 10.1, 3.4, 2.1 Hz, 1H, H-5'), 3.85-3.75 (m, 2H, H-6', H-4'), 3.69 (dd, J = 10.8, 2.0 Hz, 1H, H-6'), 3.65 (d, J = 3.6 Hz, 1H, H-5), 3.56-3.38 (m, 3H, H-6, H-1, H-4), 3.30-3.22 (m, 2H, H-3, H-2'), 2.37–2.24 (m, 1H, H-2), 1.56–1.41 (m, 1H, H-2). HRMS (ESI, positive) calculated for $C_{33}H_{36}DN_9O_7^+ [M + Na]^+ 695.2771$, found 695.2770.

[3'-²H]-Paromamine (65): Trimethylphosphine (1 M solution in THF, 1.418 mL) was added to a solution of compound 64 (0.159 g, 0.236 mmol) in THF (5.9 mL) and 0.1 M NaOH (2 mL). The reaction mixture was stirred at 50 °C for 2 h. After cooling down to room temperature, the solution was evaporated under reduced pressure. The material was dissolved in a mixture of acetic acid (1.3 mL), water (3.9 mL) and methanol (3.9 mL). The solution was deaerated by evacuating the air in the flask and refilling it with nitrogen five times. Palladium hydroxide on carbon (20%, Degussa type, 0.1 g) was added and the solution was charged with hydrogen. The reaction mixture was stirred at room temperature under hydrogen gas for 24 h. The solution was filtered through a Celite pad and concentrated. The residue was purified by chromatography on Amberlite CG-50 resin (NH₄⁺ form) (2.5% concentrated ammonium hydroxide in water), concentrated, acidified with hydrochloric acid (pH ca. 4), and lyophilized to provide compound **65** (0.067 g, 81%) as an oily solid. ¹H NMR (500 MHz, D₂O) δ 5.72 (d, *J* = 3.9 Hz, 1H, H-1'), 3.98 (dd, *J* = 12.0, 2.2 Hz, 1H, H-6'), 3.95 (dd, *J* = 10.2, 9.1 Hz, 1H, H-5), 3.89 (ddd, *J* = 9.9, 6.6, 2.1 Hz, 1H, H-5'), 3.80 (dd, *J* = 12.1, 6.6 Hz, 1H, H-6'), 3.73 (t, *J* = 9.2 Hz, 1H, H-6), 3.69–3.61 (m, 2H, H-1, H-4), 3.55 (d, *J* = 9.9 Hz, 1H, H-4'), 3.51 (d, *J* = 4.0 Hz, 1H, H-2'), 3.40 (ddd, *J* = 12.4, 10.3, 4.2 Hz, 1H, H-3), 2.57 (dt, *J* = 12.7, 4.3 Hz, 1H, H-2), 1.93 (q, *J* = 12.6 Hz, 1H, H-2). ¹³C NMR (125 MHz, D₂O) δ 97.07 (C-1'), 80.33 (C-5), 74.69 (C-6), 73.65 (C-5'), 72.33 (C-4), 69.30 (C-4'), 60.36 (C-6'), 53.91 (C-2'), 49.62 (C-3), 48.85 (C-1), 28.27 (C-2). HRMS (ESI, positive) calculated for C₁₂H₂₄DN₃O₇⁺ [*M* + *H*]⁺ 325.1829, found 325.1824.

4.3. RESULTS AND DISCUSSION

4.3.1. Gene Analysis

Sequence analysis led to the assignment of AprD3 as an NAD-dependent dehydrogenase and AprD4 as a member of the radical *S*-adenosyl-L-methionine (SAM) family of enzymes. The catalytic [4Fe-4S] cluster characteristic of radical SAM enzyme is typically coordinated by a highly conserved CxxxCxxC motif, however, in the case of AprD4, a noncanonical CxxxCxxxC motif is found instead. Recently a number of examples of radical SAM enzymes with different [4Fe-4S]-binding motifs have been reported. PhnJ, which catalyzes cleavage of the C-P bond of phosphonates, has a $Cx_2Cx_{21}Cx_5C$ motif.²⁰⁹⁻²¹⁰ HmdB, which is involved in the biosynthesis of hydrogenase

cofactor, contains a Cx_5Cx_2C cluster-binding motif.²¹¹ QueE participates in the synthesis of 7-carboxy-7-deazaguanine and contains a $Cx_{14}Cx_2C$ cluster-binding motif.²¹² ThiC, the thiamine pyrimidine synthase found in plants and bacteria, has a Cx_2Cx_4C motif.²¹³ Also GenD1 has a Cx_4Cx_2C motif.⁸¹ Although these radical SAM enzymes do indeed exhibit atypical [4Fe-4S] binding motifs, the noncanonical motif of AprD4 still raises questions, as to whether AprD4 operates as a radical SAM enzyme and the mechanism by which C3'-deoxygenation is effected during the biosynthesis of aminoglycosides.

4.3.2. Purification of AprD3 and AprD4

Both the *aprD3* and *aprD4* genes were amplified by polymerase chain reaction from *S. tenebrarius* genomic DNA, cloned in to a pET28 vector, and heterologously overexpressed in *Escherichia coli* (Figure 4-6 for AprD3 and Figure 4-7 for AprD4). The His₆-tagged AprD4 was isolated from cell lysates using a nickel-nitrilotriacetate (Ni-NTA) affinity column and was then subjected to anaerobic incubation with [Fe(NH)₄(SO₄)₂] and Na₂S in a Coy anaerobic chamber with an atmosphere of more than 97% N₂ and 3% H₂ to reconstituted the [4Fe-4S] cluster. Iron¹⁵³ and Sulfur analysis¹⁵⁴ of the reconstituted AprD4 revealed the presence of 9.2 ± 0.6 equivalents of iron and 8.2 ± 0.7 equivalents of sulfide per AprD4 monomer. These results indicated the presence of two [4Fe-4S] clusters bound to each AprD4 monomer. Likewise, the UV/Vis absorption spectrum of the reconstituted AprD4 exhibits a shoulder around $\lambda = 420$ nm, which is typical of [4Fe-4S]²⁺-containing proteins.²¹⁴⁻²¹⁵ Furthermore, addition of sodium dithionite (2mM) to the reconstituted enzyme led to a decrease in the absorbance at $\lambda =$ 420 nm, implying reduction of the cluster.



Figure 4-6. SDS PAGE of AprD3. The calculated molecular weight of AprD3 is 29.4 kDa.



Figure 4-7. UV-visible absorption spectra and SDS PAGE of AprD4. The blue, red, and green spectra correspond to no reconstituted AprD4 (8 mM), reconstituted AprD4 (15 mM), and reconstituted AprD4 (15 mM) treated with 1 mM sodium dithionite. The bleaching of the absorbance shoulder at 420 nm is characteristic of a bound iron-sulfur cluster. The calculated molecular weight of AprD4 is 52.5 kDa.

4.3.3. Synthesis of Lividamine

As shown in Scheme 4-2 in Section 4.2.8, D-lividosamine (27) was obtained after four steps from *N*-acetyl-D-glucosamine.²⁰⁸ *N*-acetyl-D-glucosamine was converted into a furanosyl oxazoline intermediate using dry acetone at reflux. The hydroxyl group at C3

position was removed using Barton-McCombie deoxygenation.²¹⁶ Deprotection of oxazoline ring was conducted in 2 N HCl at reflux to give D-lividosamine. After introduction of an azide group at the C2 position and acetylation of the remaining alcohols, a thiophenyl group was incorporated at C1. After changing the protecting group on the alcohols, a coupling reaction with 2-streptamine derivative **33** was used to give pseudo-disaccharide **34**. Basic hydrolysis of the acetyl group with sodium methoxide, Staudinger reaction, and hydrogenation using palladium hydroxide on carbon gave lividamine (**13**) in 11.4% over 9 steps from compound **27**.

4.3.4. In vitro AprD3 and AprD4 Assay

Co-incubation of reconstituted AprD4 (0.01 mM) with 1 mM paromamine and 2 mM SAM in the presence of 10 mM dithiothreitol (DTT) and 2 mM dithionite (50 mM NH₄HCO₃ buffer, pH 7.8) at room temperature led to the formation of 5'-dAdo, as detected by monitoring the change in the absorption at λ = 260 nm (Figure 4-8 , trace b). Although exclusion of dithionite from the reaction did not lead to detectable levels of 5'-dAdo formation, the replacement of dithionite with either methyl viologen and NADPH or the *E. coli* flavodoxin/flavodoxin reductase/NADPH system were effective at doing so (Figure 4-9). In contrast, benzyl viologen with NADPH was not suitable for observing enzyme activity. Likewise, when the incubation was performed without prior reconstitution of the enzyme, no 5'-dAdo was discernible in the absence of paromamine, the extent of SAM turnover to 5'-dAdo was significantly lower than that of the full reaction (traces d and b in Figure 4-8). These results strongly imply that AprD4 is able to catalyze radical homolysis of SAM consistent with its assignment as a radical SAM enzyme.^{101,117}



Figure 4-8. HPLC traces of AprD4 reaction mixtures for production of 5'-dAdo observed by UV detection at 260 nm. Trace (a) corresponds to 5'-dAdo. The remaining traces (b-h) are from reaction mixtures containing AprD4 (10 μ M), paromamine (1 mM), SAM (2 mM), DTT (10 mM), dithionite (2 mM) in NH₄HCO₃ (50 mM; pH 7.8) buffer with the following modifications: b) no changes, c) no AprD4, d) no paromamine, e) nonreconstituted AprD4, f) no SAM, g) with AprD3 (10 μ M) and NADPH (1 mM), and h) with AprD3 (10 μ M) and no NADPH.



Figure 4-9. Competence of various reducing systems for supporting AprD4 activity. HPLC traces with UV detection at 260 nm showed 5'-dAdo formation.

The AprD4 reaction was monitored by HPLC using a Corona charged aerosol detector (CAD)²¹⁷ as well as FDNB derivatization method with UV absorbance detect at 340 nm, because the pseudo-disaccharide substrate paromamine and its products lack a UV-absorbing chromophore (Figure 4-10). Despite the detected conversion of SAM to 5'dAdo, neither significant consumption of paromamine nor new product formation was noted, unless AprD3 (10 μ M) and NADPH (1mM) were also include in the reaction mixture (trace b and g in Figure 4-10). In trace g in Figure 4-10A, a new peak with a retention time of 7.2 min was detected, in addition to the consumption of paromamine. Electrospray ionization mass spectra (ESIMS) of the new product revealed a signal at m/z330.1638, consistent with the sodium adduct of lividamine (13; calc. 330.1636 [M + Na^{+} ; Figure 4-11, Figure 4-12E). In trace g in Figure 4-10B, a new peak with the same retention time with lividamine standard at 12.9 min was detected. This result demonstrated that the position where the deoxygenation reaction occurs is the C3' of paromamine. These findings led to the hypothesis that AprD4 operates as a radicalmediated dehydratase with AprD3 serving as its reductase partner to effect the overall reduction of paromamine to lividamine $(11 \rightarrow 12 \rightarrow 13)$.



Figure 4-10. HPLC traces of AprD4 reaction mixtures showing the consumption of paromamine, monitored using A) a Corona CAD, and B) UV detection at 340 nm after the assay mixture was subjected to the reaction with 1-fluoro-2,4-dinitrobenzene (FDNB). Traces b–h are as described in the legend of **Figure 4-8**. The small peaks at ~12 min in trace c, g, and h are unrelated to product and substrate.



Figure 4-11. (A) Low resolution, and (B) high resolution mass spectra of the collected peak at 7.2 min shown in Figure 4-9A, trace g. (A) LRMS (ESI, positive) calculated for $C_{12}H_{25}N_3O_6^+[M+H]^+$ 308.18 and $[M+Na]^+$ 330.16, observed 308.18 and 330.16, 232

respectively. (B) HRMS (ESI, positive) calcd $[M + Na]^+$ 330.1636, observed 330.1638. These results were consistent with the chemical structure of lividamine (13).

As a first test of this hypothesis, reaction mixtures excluding AprD3 were analyzed by ESIMS to check for low concentrations of the putative dehydration product of paromamine that may have gone undetected by Corona CAD and FDNB derivatization. As shown in Figure 4-12A, a signal was detected at m/z 328.15, consistent with the sodium adduct of such an intermediary species (12). To further support assignment of this MS signal to the putative ketone intermediate, the AprD4 reaction in the absence of AprD3 was run in the presence of phenylhydrazine (10 mM) for 4 h prior to analysis by ESIMS. In this case, a new signal at m/z 418.20 was detected whereas the signal at m/z 328.15 was significantly decreased, suggesting formation of a phenylhydrazone derivative (14; Figure 4-12B; Scheme 4-6). Moreover, treatment of the AprD4-only reaction with NaBH₄ or NaBD₄ resulted in a shift of the signal at m/z 328.15 to 330.17 and 331.17, respectively (Figure 4-12C, D; Scheme 4-6). These results support the hypothesis that AprD4 catalyzes the first of two half-reactions, resulting in the dehydration of paromamine to form a keto intermediate such as 12.



Figure 4-12. Electrospray ionization (ESI) mass spectra of the AprD4 reaction with paromamine (11) (A), together with the addition of (B) phenylhydrazine, (C) NaBH₄ and (D) NaBD₄ as described in the text. (E) ESI mass spectrum of the AprD4/AprD3 co-incubation with paromamine and NADPH.



Scheme 4-6. The reduction of paromamine (11) to lividamine (13) catalyzed by AprD4 and AprD3. Derivatization reactions carried out in the absence of AprD3 are also shown.
4.3.5. Proposed Mechanism of AprD4 Catalysis

Based on the results in the Section 4.3.4, at least two mechanisms can be envisioned for the AprD4-catalyzed dehydration of paromamine (Scheme 4-7). Consistent with radical SAM chemistry, both mechanisms begin with initial hydrogenatom abstraction from the C4' position of paromamine $(11\rightarrow 20)$ by the 5'-dAdo radical (18) generated by the reductive homolysis of SAM. The resulting substrate radical (20) may then be deprotonated to form a ketyl radical 21/22 that subsequently undergoes β elimination of the 3'-OH group to yield the enol radical 23 (Scheme 4-7, route A). This mechanism is similar to the mechanisms proposed for (*R*)-2-hydroxyacyl-CoA dehydratase¹⁷⁸ and ribonucleotide reductase.²¹⁸ One-electron reduction of the resulting enol radical intermediate would then complete the dehydration reaction (23/24 \rightarrow 12). Alternatively, the initially formed substrate radical 20 may instead undergo a radicalinduced 1,2-hydroxy shift to form the *gem*-diol radical 25 (Scheme 4-7, route B), in direct analogy to the mechanisms of the Cbl-dependent diol-dehydratases and ethanolamine ammonia lyase.²⁰¹⁻²⁰² Following reduction , the *gem*-diol intermediate 26 would then be converted into 12 upon elimination of water.



Scheme 4-7. Proposed mechanisms for the AprD4-catalyzed dehydration of paromamine.

4.3.6. Synthesis of [4'-²H]-Paromamine

As shown in Scheme 4-3 in section 4.2.9, $[4'^{2}H]$ -paromamine (47) was prepared from 36 as summarized below. Intermediate 39 was obtained from 36^{159} via 4,6-diol protection with anisaldehyde dimethyl acetal ($36 \rightarrow 37$), followed by benzyl protection of the 3-OH group ($37 \rightarrow 38$) and selective reduction using sodium cyanoborohydride and trifluoroacetic acid.²¹⁹ Equatorial introduction of deuterium at C4 was achieved via Dess-Martin oxidation and sodium borodeuteride reduction ($39 \rightarrow 40$).¹⁷⁴ The stereochemistry at C4 of 40 was inverted in a two-step procedure with sodium benzoate serving as a nucleophile under reflux conditions ($40 \rightarrow 41 \rightarrow 42$).²²⁰ Alkaline hydrolysis of the benzoate protecting group in **42** followed by benzyl protection of the exposed hydroxyl at C4 provided **44**. Intermediate **44** was then coupled with an azide derivative of 2-deoxystreptamine $(33)^{157}$ in the presence of *N*-iodosuccinimide and trifluoromethanesulfonic acid to provide the pseudodisaccharide **45**. Next, the acetyl protecting groups were removed by alkaline hydrolysis to give **46**, and the azide groups were converted to amines using Staudinger reduction. Final deprotection of benzyl group by hydrogenolysis provided **47** in 7.4% yield over 13 steps.

4.3.7. Synthesis of [3'-²H]-Paromamine

As shown in Scheme 4-4 in Section 4.2.10, the synthesis of $[3'-{}^{2}H]$ -paromamine (65) began from commercially available 1,2,5,6-di-O-isopropylidene- α -D-glucofuranose (48). Deuterium was incorporated at the C3 position with double inversion of stereochemistry to provide compound 52. After benzyl protection of the C3 alcohol of compound 52, furanose was converted into pyranose by acid hydrolysis and the remaining alcohols were acetylated to form the triacetylate 54. A thiophenyl group was introduced at the C1 position using the Lewis acid boron trifluoride diethyletherate to provide compound 55. The acetyl group was hydrolyzed by basic hydrolysis using catalytic amounts of sodium methoxide. Selective protection of the C4 and C6 alcohols with benzylidine acetal was conducted to provide the intermediate 57. An azide group was introduced at C2 again with double inversion of stereochemistry using oxidation, reduction, and $S_N 2$ type reaction to give compound 60. After acid hydrolysis to remove benzylidine acetal, the free alcohols of compound 61 were protected as benzyl group. Coupling reaction of compound 62 with 2-deoxystreptamine derivative 33, basic hydrolysis, Staudinger reaction, and hydrogenation gave $[3'-{}^{2}H]$ -paromamine (65) in 3.2 % over 21 steps from compound 48.

4.3.8. Mechanistic Studies of AprD4 Catalysis

When AprD4 (10 μ M) was incubated with [4'-²H]-paromamine (1 mM), SAM (2 mM), and dithionite (2 mM) in NH₄HCO₃ (50 mM; pH 7.8) for 8 h at room temperature, formation of 5'-dAdo was again detected and this time with single-deuterium incorporation beyond natural abundance as determined by ESIMS (**Figure 4-13**). A complimentary assay with unlabeled paromamine (11) and AprD4 using the dideuterated [5'-²H₂]-SAM isotopologue¹⁷³ in the incubation revealed hydrogen transfer from **11** to [5'-²H₂]- 5'-dAdo• to yield [5'-²H₂]- 5'-dAdo as the product (**Figure 4-14**). These results further substantiated the proposed early steps of AprD4 catalysis shown Scheme 4-7, where AprD4 acts as a radical SAM enzyme that initiates a radical-mediated dehydration of paromamine through hydrogen-atom abstraction from the C4' position.



Figure 4-13. Mass spectra of (A) $[5'-{}^{2}H]$ -5'-dAdo isolated from the incubation of SAM and AprD4/D3 with $[4'-{}^{2}H]$ -paromamine (47). (B) assay mixture showing formation of product (13) in the incubation of $[4'-{}^{2}H]$ -paromamine (47) with AprD4/AprD3.



Figure 4-14. Mass spectra of $[5'-{}^{2}H_{2}]$ -5'-dAdo produced during the AprD4/AprD3 reaction with $[5'-{}^{2}H_{2}]$ -SAM and unlabeled paromamine (11).

Proton transfer manner at C3' position from paromamine to lividamine by AprD4 catalysis was determined by ¹H NMR spectroscopic analysis with the labeled substrate, [3'-²H]-paromamine. Assays were conducted with 0.06 mM AprD4/D3, 20 mM dithionite, 20 mM SAM, 10 mM NADPH, 100 mM DTT in NH₄HCO₃ buffer (500 mM; pH 7.8) with 10 mM of paromamine, [3'-²H]-paromamine, lividamine, or without substrate. After a 16 h incubation, assay mixtures were analyzed by ¹H NMR spectroscopy. Additionally, to confirm AprD4 reaction with [3'-²H]-paromamine, ESIMS of [3'-²H]-lividamine was verified at m/z 331.1701 (calc. [M + Na]⁺ 331.1698). Paromamine and lividamine NMR standards in pH 7.8 buffer were prepared for comparison with the ¹H NMR spectra. A new peak in the assay with paromamine was observed at approximately 1.7 ppm (Figure 4-15a). In the assay with [3'-²H]-paromamine, this peak was not seen (Figure 4-15c). This peak is also observed in the assays with lividamine and the lividamine standard spectra (Figure 4-15b and 4-15e), but

not without any substrate and paromamine standard (Figure 4-15d and 4-15f). The peak around 1.7 ppm corresponds to the axial hydrogen at C3' of lividamine. These results demonstrated that the axial hydrogen at C3' of paromamine is retained in lividamine during AprD4/D3 catalysis, whereas the equatorial hydroxyl group at C3' of paromamine is replaced with hydrogen.



Figure 4-15. ¹H NMR spectra of AprD4/D3 reaction with a) paromamine, b) lividamine, c) [3'-²H]-paromamine, d) without any substrate. e) and f) are respectively ¹H NMR spectra from the lividamine and paromamine standards at the same pH.

4.4. CONCLUSION

C3'-deoxygenation of aminoglycosides results in their decreased susceptibility to phosphorylation thereby increasing their efficacy as antibiotics. However, the biosynthetic mechanism of C3'-deoxygenation is unknown. To address this issue, the aprD4 and aprD3 genes from the apramycin gene cluster in Streptomyces tenebrarius were expressed in *E. coli* and the resulting gene products were purified and characterized in vitro. AprD4 was shown to be a radical S-adenosyl-L-methionine (SAM) enzyme, catalyzing homolysis of SAM to 5'-dAdo in the presence of paromamine. C3'deoxygenation of paromamine catalyzed by AprD4/D3 was confirmed by comparing the lividamine standard with reaction mixtures treated with FDNB and analyzed by HPLC. [4'-²H]-Paromamine was prepared and used to show that its C4'-H is transferred to 5'dAdo by AprD4, after which the substrate is dehydrated to a product consistent with 4'oxolividamine. In contrast, paromamine is reduced to a deoxy product when incubated with AprD4/AprD3/NADPH. The AprD4/D3 assay with [3'-²H]-paromamine showed that axial C3'-H is retained during the reaction, and the equatorial hydroxyl group is replaced with new hydrogen. These results provide evidence for the assignment of functions to the enzymes AprD4 and AprD3 in the biosynthesis of C3'-deoxyaminoglycosides. AprD4 appears to be the first example of a radical SAM diol-dehydratase and catalyzes the radical-mediated dehydration of paromamine. AprD3 is shown to be a dehydrogenase that acts as the reductase counterpart to AprD4 in order to facilitate the net C3'deoxygenation of the pseudo-disaccharide substrates in the biosynthesis of apramycin and tobramycin. Enzymes LivW and LivY are expected to play analogous roles in the lividomycin biosynthetic pathway.

Appendix



A.1. SPECTRAL DATA FOR CHAPTER 2






























































 $\overline{}$





























































-






















































59:87 05:67 26:52 92.59 92.59 02.99 22.29 25.69 25.69 25.69 25.69 25.69 25.62 11.08

88.96— -101.03




















































































いいわたい ちんちいかい しまいちのし シックスト ちまちんかん ちん ころう ちょうちょう いっちょう ちょうちょう ちょうちょう ちょうちょう ちょうちょう ちょうちょう しょうちょう しょうちょう しょうちょう しょうちょう しょうしょう しょうしょう しょうしょう しょうしょう しょうしょう




















































A.3. SPECTRAL DATA FOR CHAPTER 4



































 $\overline{}$






































































Bibliography

1. Moazed, D.; Noller, H. F., Interaction of antibiotics with functional sites in 16S ribosomal RNA. *Nature* **1987**, *327*, 389–394.

2. Becker, B.; Cooper, M. A., Aminoglycoside Antibiotics in the 21st Century. ACS Chem. Biol. 2013, 8, 105–115.

3. Waksman, S. A., Streptomycin: Background, Isolation, Properties, and Utilization. *Science* **1953**, *118*, 259–266.

4. Weistein, M. J.; Keudemann, G. M.; Oden, E. M.; Wagman, G. H.; Rosselet, J. P.; Marquez, J. A.; Coniglio, S. T.; Charney, W.; Herzog, H. L.; Black, J., Gentamicin, a New Antibiotic Complex from Micromonospora. *J Med. Chem.* **1963**, *6*, 463–464.

5. Higgens, C. E.; Kastner, R. E., Nebramycin a new broad-spectrum antibiotic complex. II. Description of Streptomyces tenebrarius. *Antimicrob. Ag. Chemother.* **1967**, *7*, 324–331.

6. O'Connor, S.; Lam, L. K. T.; Jones, N. D.; Chaney, M. O., Apramycin, a unique aminocyclitol antibiotic. *J. Org. Chem.* **1976**, *41*, 2087–2092.

7. Yamaguchi, M.; Mishashi, S., Antibacterial activity of lividomycin against several bacteria isolated from clinical specimens. *Jpn. J. Antibiot.* **1972**, *25*, 336–339.

8. Piepersberg, W.; Aboshanab, K. M.; Schmidt-Beissner, H. U.; Wehmer, F., In *Aminoglycoside Antibiotics* (Eds; D. P. Arya), Wiley-VCH, Weinhein, **2007**, pp 15–118.

9. Miyake, T.; Tsuchiya, T.; Umezawa, S.; Umezawa, H., A synthesis of 3',4'-dideoxykanamycin B. *Carbohydr Res.* **1976**, *49*, 141–151.

10. Kawaguchi, H.; Naito, T.; Nakagawa, S.; Fujisawa, K. BB-K8, a new semisynthetic aminoglycoside antibiotics. *J. Antibiot.* **1972**, *25*, 695–708.

11. Kondo, S.; Ilnuma, K.; Yamamoto, H.; Maeda, K.; Umezawa, H., Syntheses of 1-*N*-{(*S*)-4-amino-w-hydroxybutyryl}-kanamycin B and -3',4'-dideoxykanamycin B active against kanamycin-resistant bacteria. *J. Antibiot.* **1973**, *26*, 412–415.

12. Goering, R. V.; Sanders, C. C.; Sanders, W. E. Jr., In vivo analysis of structureactivity relationships among four aminoglycosides: gentamicin netimicin, 1-NHAPA gentamicin B and amikamicin. *Curr. Ther. Res.* **1979**, *26*, 329–341.

13. Miller, G. H.; Arcieri, G.; Weinstein, m. J.; Waitz, J. A., Biological activity of netilmicin, a broad-spectrum semisynthetic aminoglycoside antibiotic. *Antimicrob. Agents Chemother.* **1976**, *10*, 827–836

14 9 (10). Poelsgaard, J.; Douthwaite, S., The bacterial ribosome as a target for antibiotics. *Nat. Rev. Microbiol.* **2005**, *3*, 870–881.

15. Carter, A. P.; Clemons, W. M.; Brodersen, D. E.; Morgan-Warren, R. J.; Wimberly, B. T.; Ramakrishnan, V., Functional insights from the structure of the 30S ribosomal subunit and its interactions with antibiotics. *Nature* **2000**, *407*, 340–348.

16. Pfister, P.; Hobbie, S.; Vicens, Q.; Böttger, E. C.; Westhof, E., The Molecular Basis for A-Site Mutations Conferring Aminoglycoside Resistance: Relationship between Ribosomal Susceptibility and X-ray Crystal Structures. *ChemBioChem* **2003**, *4*, 1078–1088.

17. Davis, B. D.; Chen, L.; Tai, P. C., Misread protein creates membrane channels: An essential step in the bactericidal action of aminoglycosides. *Proc. Natl. Acad. Sci. USA* **1986**, *83*, 6164–6168.

18. Bryan, L. E.; Kwan, S., Roles of Ribosomal Binding, Membrane Potential, and Electron Transport in Bacterial Uptake of Streptomycin and Gentamicin. *Antimicrob. Agents Chemother.* **1983**, *23*, 835–845.

19. Davis, B. D., Mechanism of Bactericidal Action of Aminoglycosides. *Microbiol. Rev.* **1987**, *51*, 341–350.

20. Kohanski, M. A.; Dwyer, D. J.; Hayete, B. Lawrence, C. A.; Collins, J. J., A Common Mechanism of Cellular Death induced by Bactericidal Antibiotics. *Cell* **2007**, *130*, 797–810.

21. Flatt, P. M.; Mahmud, T., Biosynthesis of aminocyclitol-aminoglycoside antibiotics and related compounds. *Nat. Prod. Rep.* **2007**, *24*, 358–392.

22. Catt, K. J.; Hunyady, L.; Balla, T., Second messengers derived from inositol lipids. *J. Bioenerg. Biomembr.* **1991**, *23*, 7–27.

23. Exton, J. H., Mechanisms of action of calcium-mobilizing agonists: some variations on a young theme. *FASEB J.* **1988**, *2*, 2670–2676.

24. Biswas, B. B.; Ghosh, B.; Majumder, A. L., *myo*-Inositol Polyphosphates and Their Roles in Cellular Metabolism. *Subcell. Biochem.* **1984**, *10*, 237–280.

25. Majumder, A. L.; Johnson, M. D.; Henry, S. A., 1L-myo-inotisol-1-phosphate synthase. *Biochim. Biophys. Acta*. **1997**, *1348*, 245–256.

26. Mahmud, T.; Tornus, I.; Egelkrout, E.; Wolf, E.; Uy, C.; Floss, H. G.; Lee. S., Biosynthetic Studies on the α-Glycosidase Inhibitor Acrosse in *Actinoplanes* sp.: 2-*epi*-5-*epi*-Valiolone Is the Direct Precurosr of the Valienamine Moiety. *J. Am. Chem. Soc.* **1999**, *121*, 6973–6983.

27. Dong, H.; Mahmud, T.; Tornus, I.; Lee, S.; Floss, H. G., Biosynthesis of the Validamycins: Identification of Intermediates in the Biosynthesis of Validamycin A by *Streptomyces hygroscopicus* var. *limoneus. J. Am. Chem. Soc.* **2001**, *123*, 2733–2742.

28. Mahmud, T.; Lee. S.; Floss, H. G., The biosynthesis of acarbose and validamycin. *Chem. Rec.* **2001**, *1*, 300–310.

29. Bhuyan, B. K., Pactamycin Production by *Streptomyces pactum. Appl. Microbiol.* **1962**, *10*, 302–304.

30. Sakuda, S.; Isogai, A.; Matsumoto, S.; Suzuki, A., Search for microbial insect growth regulators II. Allosamidin , a novel insect chitinase inhibitor. *J. Antibiot.* **1987**, *40*, 296–300.

31. Rinehart, Jr., K. L.; Weller, D. D.; Pearce, C. J., Recent Biosynthetic Studies on Antibiotics. J. Nat. Prod. 1980, 43, 1–20.

32. Wehmeier, U. F.; Pieepersberg, W., Enzymology of Aminoglycoside Biosynthesis – Deduction from Gene Clusters. *Methods Enzymol.* **2009**, *459*, 459–491.

33. Kudo, F.; Eguchi, T., Biosynthetic genes for aminoglycoside antibiotics. *J. Antibiot.* **2009**, *62*, 471–481.

34. Rinehart, Jr. K. L.; Stroshane, R. M., Biosynthesis of aminocyclitol antibiotics. J. Antibiot. 1976, 29, 319–353.

35. Rinehart, Jr. K. L., Biosynthesis and Mutasynthesis of Aminocyclitol Antibiotics. J. Antibiot. 1979, 32, S32–S46.

36. Furumai, T.; Takeda, K.; Kinumaki, A.; Ito, Y.; Okuda, T., Biosynthesis of butirosins. II Biosynthetic pathway of butirosins elucidated from cosynthesis and feeding experiments. *J. Antibiot.* **1979**, *32*, 891–899.

37. Distler, J.; Braun, C.; Ebert, A.; Piepersberg, W., Gene cluster for streptomycin biosynthesis in *Streptomyces griseus*: Analysis of a central region including the major resistance gene. *Mol. Gen. Genet.* **1987**, *208*, 204–210.

38. Distler, J.; Mansouri, K.; Mayer, G.; Stockmann, M.; Piepersberg, W., Streptomycin biosynthesis and its regulation in Streptomycetes. *Gene* **1992**, *115*, 105–111.

39. Kudo, F.; Hosomi, Y.; Tamega, H.; Kakinuma, K., Purification and characterization of 2-deoxy-*scyllo*-inose synthase derived from *Bacillus circulans*. A crucial carbocyclization enzyme in the biosynthesis of 2-deoxystreptamine-containing aminoglycoside antibiotics. *J. Antibiot.* **1999**, *52*, 81–88.

40. Fourmy, D.; Recht, M. I.; Blanchard, S. C.; Puglisi, J. D., Structure of the A Site of *Escherichia coli* 16S Ribosomal RNA Complexed with an Aminoglycoside Antibiotic. *Science* **1996**, *274*, 1367–1371.

41. Fourmy, D.; Recht, M. I.; Puglisi, J. D., Binding of Neomycin-class Aminoglycoside Antibiotics to the A-site of 16 S rRNA. *J. Mol. Biol.* **1998**, *277*, 347–362.

42. Llewellyn, N. M.; Spencer, J. B., Biosynthesis of 2-deoxystreptamine-containing aminoglycoside antibiotics. *Nat. Prod. Rep.* **2006**, *23*, 864–874.

43. Kudo, F.; Eguchi, T., Biosynthetic Enzymes for the Aminoglycosides Butirosin and Neomycin. *Methods Enzymol.* **2009**, *459*, 493–519.

44. Kudo, F.; Eguchi, T., Aminoglycoside Antibiotcs: New Insights into the Biosynthetic Machinery of Old Drugs. *Chem. Rec.* **2016**, *16*, 4–18.

45. Rinehart, K. L.; Malik, J. M.; Nystrom, R. S.; Stroshane, R.; Truitt, S. T.; Taniguchi, M; Rolls, J. P.; Haak, W. J.; Ruff, B. A., Carbon-13 as a biosynthetic tool. IV. Biosynthetic incorporation of glucosamine-1-13C and glucose-6-13C into neomycin. *J. Am Chem. Soc.* **1974**, *96*, 2263–2265.

46. Kakinuma, K.; Ogawa, Y.; Sasaki, T.; Seto, H.; Otake, N., Streochemistry of ribostamycin biosynthesis. Application of hydrogen-2 NMR spectroscopy. *J. Am Chem Soc.* **1981**, *103*, 5614–5616.

47. Kakinuma, K.; Ogawa, Y.; Sasaki, T.; Seto, H.; Otake, N., Mechanism and stereochemistry of the biosynthesis of 2-deoxystreptamine and neosamine C. *J. Antibiot.* **1989**, *42*, 926–933.

48. Goda, S. K.; Akhtar, M., The Involvement of C-4 of D-glucose in the Biosynthesis of the 2-Deoxystreptamine Ring of neomycin. *J. Chem. Soc. Chem. Commun.* **1987**, 12–14.

49. Goda, S. K.; Akhtar, M., Neomycin biosynthesis: the incorporation of D-6-deoxyglucose derivatives and variously labeled glucose into the 2-deoxystreptamine ring. Postulated involvement of 2-deoxyinosose synthase in the biosynthesis. *J. Antibiot.* **1992**, *45*, 984–994.

50. Yamauchi, N.; Kakinuma, K., Confirmation of in vitro synthesis of 2-deosy-*scyllo*inose, the earliest intermediate in the biosynthesis of 2-deoxystreptamine, using cell free preparations of *Streptomyces Fradiae*. J. Antibiot. **1992**, 45, 774–780.

51. Yamauchi, N.; Kakinuma, K., Enzymatic Carbocycle Formation in Microbial Secondary Metabolism. The Mechanism of the 2-Deoxy-*scyllo*-inose Synthase Reaction as a Crucial Step in the 2-Deoxystreptamine Biosynthesis in *Streptomyces fradiae*. J. Org. Chem. **1995**, 60, 5614–5619.

52. Ota, Y.; Tamegai, H.; Kudo, F.; Kuriki, H., Koike-Takeshita, A.; Eguchi, T.; Kakinuma, K., Butirosin-biosynthetic Gene Cluster from *Bacillus circulans*. J. Antibiot. **2000**, *53*, 1158–1167.

53. Subba, B.; Kharel, M. K.; Lee, H. C.; Liou, K.; Kim, B.-G.; Sohng, J. K., The Ribostamycin Biosynthetic Gene Cluster in *Streptomyces ribosidificus*: Comparison with Butirosin Biosynthesis. *Mol. Cells* **2005**, *20*, 90–96.

54. Kharel, M. K.; Basnet, D. B.; Lee, H. C.; Liou, K.; Woo, J. S.; Kim, B.-G.; Sohng, J. K., Isolation and characterization of the tobramycin biosynthetic gene cluster from *Streptomyces tenebrarius. FEMS Microbiol. Lett.* **2004**, *230*, 185–190.

55. Park, S. R.; Park, J. W.; Ban, Y. H.; Sohng, J. K.; Yoon, Y. J., 2- Deoxystreptaminecontaining aminoglycoside antibiotics: Recent advances in the characterization and manipulation of their biosynthetic pathways. *Nat. Prod. Rep.* **2013**, *30*, 11–20. 56. Nango, E.; Kumasaka, T.; Hirayama, T.; Tanaka, N.; Eguchi, T., Structure of 2-deoxy-*scyllo*-inose synthase, a key enzyme in the biosynthesis of 2-deoxystreptamine-containing aminoglycoside antibiotics, in complex with a mechanism-based inhibitor and NAD⁺. *Proteins: Struct., Funct., Bioinf.* **2008**, *70*, 517–527.

57. Carpenter, E. P.; Hawkins, A. R.; Frost, J. W.; Brown, K. A., Structure of dehydroquinate synthase reveals an active site capable of multistep catalysis. *Nature* **1998**, *394*, 299–302.

58. Hirayama, T.; Kudo, F.; Huang, Z.; Eguchi, T., Role of glutamate 243 in the active site of 2-deoxy-*scyllo*-inose synthase from *Bacillus circulans*. *Bioorg. Med. Chem.* **2007**, *15*, 418–423.

59. Yokoyama, K.; Yamamoto, Y.; Kudo, F.; Eguchi, T., Involvement of Two Distinct *N*-Acetylglucosaminyltransferases and a Dual-Function Deacetylase in Neomycin Biosynthesis. *ChemBioChem* **2008**, *9*, 865–869.

60. Truman, A. W.; Huang, F.; Llewellyn, N. M.; Spencer, J. B., Characterization of the enzyme BtrB from *Bacillus circulans* and revision of its functional assignment in the biosynthesis of butirosin. *Angew. Chem. Int. Ed.* **2007**, *46*, 1462–1464.

61. Huang, F.; Spiteller, D.; Koorbanally, N. A.; Li, Y.; Llewellyn, N. M.; Spencer, J. B., Elaboration of Neosamine Rings in the Biosynthesis of Neomyicn and Butirosin. *ChemBiochem* **2007**, *8*, 283–288.

62. Kudo, F.; Fujii, T.; Kinoshita, S.; Eguchi, T., Unique *O*-ribosylation in the biosynthesis of butirosin. *Bioorg. Med. Chem.* **2007**, *15*, 4360–4368.

63. Kurumbang, N. P.; Liou, K.; Sohng, J. K., Biosynthesis of Ribostamycin Derivatives by Reconstitution and Heterologous Expression of Required Gene Sets. *Appl. Biochem. Biotechnol.* **2011**, *163*, 373–382.

64. Fan, Q.; Huang, F.; Leadlay, P. F.; Spencer, J. B., The neomycin biosynthetic gene cluster of *Streptomyces fradiae* NCIMB 8233: genetic and biochemical evidence for the

roles of two glycosyltransferases and a deacetylase. Org. Biomol. Chem. 2008, 9, 3306–3314.

65. Huang, F.; Haydock, S. F.; Mironenko, T.; Spiteller, D.; Li, Y.; Spencer, J. B., The neomycin biosynthetic gene cluster of *Streptomyces fradiae* NCIMB 8233: characterization of an aminotransferase nvolved in the formation of 2-deoxystreptamine. *Org. Biomol. Chem.* **2005**, *3*, 1410–1418.

66. Kudo, F.; Kawashima, T.; Yokoyama, K.; Eguchi, T., Enzymatic preparation of neomycin C from ribostamycin. *J. Antibiot.* **2009**, *62*, 643–646.

67. Clausnitzer, D.; Piepersberg, W.; Wehmeier, U. F., The oxidoreductase LivQ and NeoQ are responsible for the different 6'-modifications in the aminoglycosides lividomycin and neomycin. *J. Appl. Microbiol.* **2011**, *111*, 642–651.

68. Park, J. W.; Park, S. R.; Nepal, K. K.; Han, A. R.; Ban, Y. H.; Yoo, Y. J.; Kim, E. J.; Kim, E. M.; Kim, D.; Sohng, J. K.; Yoon, J. Y., Discovery of parallel pathways of kanamycin biosynthesis allows antibiotic manipulation. *Nat. Chem. Biol.* **2011**, *7*, 843–852.

69. Kudo, F.; Sucipto, H.; Eguchi, E., Enzymatic activity of a glycosyltransferase KanM2 encoded in the kanamycin biosynthetic gene cluster. *J. Antibiot.* **2009**, *62*, 707–710.

70. Sucipto, H.; Kudo, F.; Eguchi, T., The Last Step of Kanamycin Biosynthesis: Unique Deamination Reaction Catalyzed by the α -Ketoglutarate-Dependent Nonheme Iron Dioxygenase KanJ and the NADPH-Dependent Reductase KanK. *Angew. Chem. Int. Ed.* **2012**, *51*, 3428–3431.

71. Nagabhushan, T. L.; Daniels, P. J. L.; Jaret, R. S.; Morton, J. B., Gentamicin antibiotics. 8. Structure of gentamicin A2. J. Org. Chem. **1975**, 40, 2835–2836.

72. Nagabhushan, T. L.; Turner, W. n.; Daniels, P. J. L.; Morton, J. B., Gentamicin antibiotics. 7. Structures of the gentamicin antibiotics A1, A3, and A4. *J. Org. Chem.* **1975**, *40*, 2830–2304.

73. Daniels, P. J. L.; Luce, C.; Nagabhushan, T. L., The gentamicin antibiotics. 6. Gentamicin C_{2b} , an aminoglycoside antibiotic produced by *Mircomonospora purpurea* mutant JI-33. *J. Antibiot.* **1975**, *28*, 35–41.

74. Ilavsky, J.; Bayan, A. P.; Charney, W.; Reimann, H., Antibiotic from *Micromonospora purpurea*, 1975; US3903072, September 2, 1975.

75. Testa, R. T.; Tilley, B. C. Biotransformation, a new approach to aminoglycoside biosynthesis: II. Gentamicin. *J. Antibiot.* **1976**, *29*, 140–146.

76. Kharel, M. K.; Basnet, D. B.; Lee, H. C.; Liou, K.; Moon, Y. H.; Kim, J.-J.; Woo, J. S.; Sohng, J. K., Molecular Cloning and Characterization a 2-Deoxystreptamine Biosynthetic Gene Cluster in Gentamicin-producing *Micromonospora echinospora* ATCC 15835. *Mol. Cells.* **2004**, *18*, 71–78.

77. Unwin, J.; Standage, S.; Alexander, D.; Hosted Jr., T.; Horan A. C.; Wellington, E. M., Gene Cluster in *Micromonospora echinospora* ATCC15835 for the Biosynthesis of Gentamicin C Complex. *J. Antibiot.* **2004**, *57*, 436–445.

78. Hong, W.-R.; Ge, M.; Zeng, Z.-H.; Zhu, L.; Luo, M.-U.; Shao, L.; Chen, D.-J., Molecular cloning and sequence analysis of the sisomicin biosynthetic gene cluster from *Micromonospora inyoensis. Biotechnol Lett.* **2009**, *31*, 449–455.

79. Park, J. W.; Hong, J. S. J.; Parajuli, N.; Jung, W. S.; Park, S. Y.; Lim, S.-K.; Sohng, J. K.; Yoon, Y. J., Genetic dissection of the biosynthetic route to gentamicin A2 by heterologous expression of its minimal gene set. *Proc. Nat. Acad. Sci. USA* **2008**, *105*, 8399–8404.

80. Junhong, G.; Huang, F.; Huang, C.; Duan, X.; Jian, X.; Leeper, F.; Deng, Z.; Leadlay, P. F.; Sun, Y., Specificity and Promiscuity at the Branch point in Gentamicin Biosynthesis. *Chem. Biol.* **2014**, *21*, 608–618.

81. Huang, C.; Huang, F.; Moison, E.; Guo, J.; Jian, X.; Duan, X.; Deng, Z.; Leadlay, P. F.; Sun, Y., Delineating the Biosynthesis of Gentamicin X2, the Common Precursor of the Gentamicin C Antibiotic Complex. *Chem. Biol.* **2015**, *22*, 251–261.

82. Ni, X.; Zong, T.; Zhang, H.; Gu, Y.; Huang, M.; Tian, W.; Xia, H., Biosynthesis of 3"-demethyl-gentamicin C components by *gen*N disruption strain of *Micromonospora echinospora* and test their antimicrobial activities *in vitro*. *Microbiol*. *Res.* **2016**, *185*, 36–44.

83. Gu, Y.; Ni, X.; Gao, H.; Wang, D.; Xia, H., Biosynthesis of Epimers C2 and C2a in the Gentamicin C Complex. *ChemBioChem* **2015**, *16*, 1933–1942.

84. Shao, L.; Chen, J.; Wang, C.; Li, J.-a.; Tang, Y.; Chen, D.; Liu, W., Characterization of a key aminoglycoside phosphotransferase in gentamicin biosynthesis. *Bioorg. Med. Chem. Lett.* **2013**, *23*, 1438–1441.

85. Chirpich, T. P.; Zappia, V.; Costilow, R. N.; Barker, H. A., Lysing 2,3-aminomtase. Purification and properties of a pyridoxal phosphate and *S*-adenosylmethionine-activated enzyme. *J. Biol. Chem.* **1970**, *245*, 1778–1789.

86. Zappia, V.; Baker, H. A., Studies on lysine-2,3-aminomutase. Subunit structure and sulfhydryl groups. *Biochim. Biophys. Acta*. **1970**, *207*, 505–513.

87. Moss, M.; Frey, P. A., The role of *S*-adenosylmethionine in the lysine 2,3-aminomutase reaction. *J. Biol. Chem.* **1987**, *262*, 14859–14862.

88. Petrovich, R. M.; Ruzicka, f. J.; Reed, G. H.; Frey, P. A., Metal cofactors of lysine-2,3-aminomutase. *J. Biol. Chem.* **1991**, *266*, 7656–7660.

89. Petrovich. R. M.; Ruzicka, F. J.; Reed, G. H.; Frey, P. A., Characterization of ironsulfur clusters in lysine 2,3-aminomutase by electron paramagnetic resonance spectroscopy. *Biochemistry* **1992**, *31*, 10774–10781. 90. Knappe, J.; Neugebauer, F. A.; Blaschkowski, H. P.; Gänzler, H., Post-translational activation introduces a free radical into pyrivate formate-lyase. *Proc. Natl. Acad. Sci. USA* **1984**, *81*, 1332–1335.

91. Knappe, J.; Schacht, J.; Möckel, W.; Höpner, Th.; Vetter Jr., H.; Edenharder, R., Pyruvate Formate-Lyase Reaction in *Escherichia coli*. The Enzymatic System Converting an Inactive Form of the Lyase into the Catalytically Active Enzyme. *Eur. J. Biochem.* **1969**, *11*, 316–327.

92. Wagner, A. F. V.; Frey, M.; Neugebauer, F. A.; Schäfer, W.; Knappe, J., The free radical in pyruvate formate-lyase is located on glycin-734. *Proc. Natl. Acad. Sci. USA* **1992**, *89*, 996–1000.

93. Broderick, J. B.; Duderstadt, R. E.; Fernandez, D. C.; Wojtuszewski, K.; Henshaw. T. F.; Johnson, M. L., Pyruvate Formate-Lyase Activating Enzyme Is an Iron-Sulfur Protein. *J. Am. Chem. Soc.* **1997**, *119*, 7396–7397.

94. Knappe, J.; Schmitt, T., A novel reaction of S-adenosyl-L-methionine correlated with the activation of pyruvate formate-lyase. *Biochem. Biophys, Res. Commun.* **1976**, *71*, 1110–1117.

95. Mulliez, E.; Fontecave, M.; Gaillard, J.; Reichard, P., An Iron-Sulfur Center and a Free Radical in the Active Anaerobic Ribonucleotide Reductase of *Escherichia coli*. *J. Biol. Chem.* **1993**, *268*, 2296–2299.

96. Sun, X.; Harder, J.; Krook, M.; Jörnvall, H.; Sjöberg, B.-M.; Reichard, P., A possible glycine radical in anaerobic ribonucleotide reductase from *Escherichia coli*: Nucleotide sequence of the cloned *nrdD* gene. *Natl. Acad. Sci. USA* **1993**, *90*, 577–581.

97. Babbit, P.; Gerlt, J., The Regents of the University of California; 2013; Vol. 2013, http://sfld.rbvi.ucsf.edu/django/.

98. Sofia, H. J.; Chen, G.; Hetzler. B G.; Reyes-Spindola, J. F.; Miller, N. E., Radical SAM, a novel protein superfamily linking unsolved steps in familiar biosynthetic

pathways with radical mechanisms: Functional characterization using new analysis and information visualization methods. *Nucleic Acids Res.* **2001**, *29*, 1097–1106.

99. Walsby, C. J.; Ortillo, D.; Broderick, W. E.; Broderick, J. B.; Hoffman, B. M., An Anchoring Role for FeS Clusters: Chelation of the Amino Acid Moiety of *S*-Adenosylmethionine to the Unique Iron Site of the [4Fe-4S] Cluster of Pyruvate Forate-Lyase Activating Enzyme. *J. Am. Chem. Soc.* **2002**, *124*, 11270–11271.

100. Chen, D.; Walsby, C.; Hoffman, B. M.; Frey, P. A., Coordination and Mechanism of Reversible Cleavage of *S*-Adenosylmethionine by the [4Fe-4S] Center in Lysine 2,3-Aminomutase. *J. Am. Chem. Soc.* **2003**, *125*, 11788–11789.

101. Broderick, J. B.; Duffus, B. R.; Duschene, K. S.; Shepard, E. M., Radical *S*-Adenosylmethionine Enzymes. *Chem. Rev.* **2014**, *114*, 4229–4317.

102. Langraf, B. L.; Arcinas, A. J.; Lee, K.-H.; Booker, S. J., Identification of an Intermediate Methyl Carrier in the Radical *S*-Adenosylmethionine Methylthiotransferases RimO and MiaB. *J. Am. Chem. Soc.* **2013**, *135*, 15404–15416.

103. Driesener, R. C.; Duffus, B. R.; Shepard, E. M.; Bruzas, I. R.; Duschene, K. S.; Coleman, N. J.R. Marrison, A. P. G.; Salvadori, E.; Kay, C. W. M.; Peters, J. W.; Broderick, J. B.; Roach, P. L., Biochemical and Kinetic Characterization of Radical *S*-Adenosyl-*L*-methionine Enzyme HydG. *Biochemistry* **2013**, *52*, 8696–8707.

104. Kuchenreuther, J. M.; Myers, W. K.; Stich, T. A.; George, S. J.; NejatyJahromy, Y.; Swartz, J. R.; Britt, R. D., A Radical Intermediate in Tyrosine Scission to the CO and CN⁻ Ligands of FeFe Hydrogenase. *Science* **2013**, *342*, 472–475.

105. Jameson, G. N.; Cosper, M. M.; Hernandez, H. L.; Johnson, M. K.; Huynh, B. H. Role of the [2Fe-2S] cluster in recombinant Escherichia coli biotin synthase. *Biochemistry* **2004**, *43*, 2022-2031.
106. Yokoyama, K.; Numakura, M.; Kudo, F.; Ohmori, D.; Eguchi, E., Characterization and Mechanistic Study of Radical SAM Dehydrogenase in the Biosynthesis of Butirosin. *J. Am. Chem. Soc.* **2007**, *129*, 15147–15155.

107. Yokoyama, K.; Ohmori, D.; Kudo, F.; Eguchi, T., Mechanistic Study on the Reaction of a Radical SAM Dehydrogenase BtrN by Electron Paramagnetic Resonance Spectroscopy, *Biochemistry* **2008**, *47*, 8950–8960.

108. Grove, T. L.; Ahlum, J. H.; Sharma, P.; Krebs, C.; Booker, S. J., A Consensus Mechanism for Radical SAM-Dependent Dehydrogenation? BtrN Contains Two [4Fe-4S] Clusters, *Biochemistry* **2010**, *49*, 3783–3785.

109. Goldman, P. J.; Grove, T. L.; Booker, S. J.; Drennan, C. L., X-ray analysis of butirosin biosynthetic enzyme BtrN redefines structural motifs for AdoMet radical Chemistry, *Proc. Natl. Acad. Sci. USA* **2013**, *110*, 15949–15954.

110. Kudo, F.; Hoshi, S.; Kawashima, T.; Kamachi, T.; Eguchi, T., Characterization of a Radical *S*-Adenosyl-L-methionine Epimerase, NeoN, in the Last Step of Neomycin B Biosynthesis, *J. Am. Chem. Soc.* **2014**, *136*, 13909–13915.

111. Goldman, P. J.; Grove, T. L.; Sites, L. A.; McLaughlin, M. L.; Booker, S. J.; Drennan, C. L., X-ray structure of an AdoMet radial Activase reveals an anaerobic solution for formyglycine posttranslational modification. *Proc. Natl. Acad. Sci. USA* **2013**, *110*, 8519–8524.

112. Tehlivets, O.; Malanovic, N.; Visram, M.; Pavkov-Keller, T.; Keller, W., *S*-adenosyl-*L*-homocysteine hydrolase and methylation disorders: yeast as a model system. *Biochim. Biophys. Acta* **2013**, *1832*, 204–215.

113. Prokhortchouk, E.; Defossez, P.-A., The cell biology of DNA methylation in mammals. *Biochim. Biophys. Acta* **2008**, *1783*, 2167–2173.

114. Grillo, M. A.; Colombatto, S., S-adenosylmethionine and its products. *Amino Acids* **2008**, 34, 187–193.

115. Zhang, Q.; van der Donk, W. A.; Liu, W., Radical-mediated enzymatic methylation: a tale of two SAMS. *Acc. Chem. Res.* **2012**, *45*, 555–564.

116. Bauerle, M. R.; Schwalm, E. L.; Booker, S. J., Mechanistic Diversity of Radical *S*-Adenosylmethionine (SAM)-dependent Methylation. *J. Biol. Chem.* **2015**, *290*, 3995–4002.

117. Frey, P. A.; Hegeman, A. D.; Ruzicka, F. J., The Radical SAM Superfamily. *Crit. Rev. Biochem. Mol. Biol.* 2008, 43, 63–88.

118. Grove, T. L.; Benner, J. S.; Radle, M. I.; Ahlum, J. H.; Landgraf, B. J.; Krebs, C.; Booker, S. J., A Radically Different Mechanism for *S*-Adenosylmethionine-Dependent Methyltansferases. *Science* **2011**, *332*, 604–607.

119. McCusker, K. P.; Medzihradszky, K. F.; Shiver, A. L.; Nichols, R. J.; Yan, F.; Maltby, D. A.; Gross, C. A.; Fujimori, D. G., Covalent Intermediate in the Catalytic Mechanism of the Radical S-Adenosyl-L-methionine Methyl Synthase RlmN Trapped by Mutagenesis. J. Am. Chem. Soc. **2012**, 134, 18074–18081.

120. Silakov, A.; Grove, T. L.; Radle, M. I.; Bauerle, M. R.; Green, M. T.; Rosenzweig, A. C.; Boal, A. K.; Booker, S. J., Characterization of a Cross-Linked Protein-Nucleic acid Substrate Radical in the Reaction Catalyzed by RlmN. *J. Am. Chem. Soc.* **2014**, *136*, 8221–8228.

121. Yan, F.; LaMarre, J. M.; Röhrich, R.; Wiesner, J.; Jomaa, H.; Mankin, A. S.; Fujimori, D. G., RlmN and Cfr are Radical SAM Enzymes Involved in Methylation of Ribosomal RNA. *J. Am. Chem. Soc.* **2010**, *132*, 3953–3964.

122. Yan, F.; Fujimori, D. G., RNA methylation by Radical SAM enzyme RlmN and Cfr proceeds via methylene transfer and hydride shift. *Proc. Natl. Acad. Sci. USA* **2011**, *108*, 3930–3934.

123. Boal, A. K.; Grove, T. L.; McLaughlin, M. I.; Yennawar, N. H.; Booker, S. J.; Rosenzweig, A. C., Structural Basis for Methyl Transfer by a Radical SAM Enzyme. *Science* **2011**, *332*, 1089–1092.

124. Grove, T. L.; Radle, M. I.; Krebs, C.; Booker, S. J., Cfr and RlmN Contain a Single [4Fe-4S] Cluster, which Directs Two Distinct Reactivities for S-Adenosylmethionine: Methyl Transfer by $S_N 2$ Displacement and Radical Generation. J. Am. Chem. Soc. 2011, 133, 19586–19589.

125. Kaminska, K. H.; Purta, E.; Hansen, L. H.; Bujnicki, J. M.; Vester, B.; Long, K. S., Insights into the structure, function and evolution of the radical-SAM 23S rRNA methyltransferase Cfr that confers antibiotic resistance in bacteria. *Nucleic Acids. Res.* **2010**, *38*, 1652–1663.

126. Grove, T. L.; Livada, J.; Schwalm, E. L.; Green, M. T.; Booker, S. J.; Silakov, A., A substrate radical intermediate in catalysis by the antibiotic resistance protein Cfr. *Nat. Chem. Biol.* **2013**, *9*, 422–427.

127. Pierre, S.; Guillot, A.; Benjdia, A.; Sandström, C.; Langella, P.; Berteau, O., Thiostrepton tryptophan methyltransferase expands the chemistry of radical SAM enzymes. *Nat. Chem. Biol.* **2012**, *8*, 957–959

128. Benjdia, A.; Pierre, S.; Gherasim, C.; Guillot, A.; Carmonia, M.; Amara, P.; Banerjee, R.; Berteau, O., The thiostrepton A tryptophan methyltransferase TsrM catalyses a cob(II)alamin-dependent methyl transfer reaction. *Nature Comm.* **2015**, *6*, 8377.

129. Blaszczyk, A. J.; Silakov, A.; Zhang, B.; Maiocco, S. J.; Lanz, N. D.; Kelly, W. L.; Ellliott, S. J.; Krebs, C.; Booker, S. J., Spectroscopic and Electrochemical Characterization of the Iron-Sulfur and Cobalamin Cofactors of TsrM, an Unusual Radical *S*-Adenosylmethionine Methylase. *J. Am. Chem. Soc.* **2016**, *138*, 3416–3426.

130. Allen, K. D.; Wang, S. C., Initial characterization of Fom3 from *Streptomyces wedmorensis*: The methyltransferase in fosfomycin biosynthesis. *Arch. Biochem. Biophys.* **2014**, *543*, 67–73.

131. Woodyer, R. D.; Li, G.; Zhao, H.; van der Donk, W. A., New insight into the mechanism of methyl transfer during the biosynthesis of fosfomycin. *Chem. Commun.* **2007**, 359–361.

132. Marous, D. R.; Lloyd, E. P.; Buller, A. R.; Moshos, K. A.; Grove, T. L.; Blaszczyk, A. J.; Booker, S. J.; Townsend, C. A., Consecutive radical *S*-adenosylmethionine methylations form the ethyl side chain in thienamycin biosynthesis. *Proc. Natl. Acd. Sci. USA* **2015**, *112*, 10354–10358.

133. Werner, W. J.; Allen, K. D.; Hu, K.; Helms, G. L.; Chen, B. S.; Wang, S. C., In vitro phosphinate methylation by PhpK from *Kitasatospora phosalacinea*. *Biochemistry* **2011**, *50*, 8986–8988.

134. Allen, K.; Wang, S. C., Spectroscopic characterization and mechanistic investigation of P-methyl transfer by a radical SAM enzyme from marine bacterium *Shewanella denitrificans* OS217. *Biochim. Biophy. Acta* **2014**, *1844*, 2135–2144.

135. Hu, K.; Werner, W. J.; Allen, K. D.; Wang, S. C., Investigation of enzymatic C-P bond formation using multiple quantum HCP nuclear magnetic resonance spectroscopy, *Magn. Reson. Chem.* **2015**, *53*, 267–272.

136. Yu, Y.; Duan, L.; Zhang, Q.; Liao, R.; Ding, Y.; Pan, H.; Wendt-Pienkowski, E.; Tang, G.; Shen, B.; Liu, W., Nosiheptide Biosynthesis Featuring a Unique Indole Side Ring Formation on the Characteristic Thiopeptide Framework. *ACS Chem. Biol.* **2009**, *4*, 855–864.

137. Ding, Y.; Yu, Y.; Pan, H.; Guo, H.; Li, Y.; Liu, W., Moving posttranslational modifications forward to biosynthesize the glycosylated thiopeptide nocathiacin I in *Nocardia* sp. ATCC202099. *Mol. Biosyst.* **2010**, *6*, 1180–1185.

138. Huang, W.; Xu, H.; Li, Y.; Zhang, F.; Chen, X.-Y.; He, Q.-L.; Igarashi, Y.; Tang, G.-L., Characterization of Yatakemycin Gene Cluster Revealing a Radical *S*-Adenosylmethionine Dependent Methyltransferase and Highlighting Spirocyclopropane Biosynthesis. *J. Am. Chem. Soc.* **2012**, *134*, 8831–8840.

139. Hiratsuka, T.; Suzuki, H.; Kariya, R.; Seo, T.; Minami, A.; Oikawa, H., Biosynthesis of the Structurally Unique Polycyclopropanated Polyketide-Nucleoside Hybrid Jawsamycin (FR-900848). *Angew. Chem. Int. Ed.* **2014**, *53*, 5423–5426.

140. Watanabe, H.; Tokiwano, T.; Oikawa, H., Biosynthetic Study of FR-900848: Origin of the Aminodexoynucleoside Part. J. Antibiot. **2006**, *59*, 607–610.

141. Watanabe, H.; Tokiwano, T.; Oikawa, H., Biosynthetic study of FR-900848: unusual observation on polyketide biosynthesis that did not accept acetate as origin of acetyl-CoA. *Tetrahedron Lett.* **2006**, *47*, 1399–1402.

142. Allen, K. D.; Xu, H.; White, R. H., Identification of a Unique Radical *S*-Adenosylmethionine Methylase Likely Involved in Methanopterin Biosynthesis in *Methanocaldococcus jannaschii. J. Bacteriol.* **2014**, *196*, 3315–3323.

143. Zhou, P., O'Hagan, D., Mocek, U., Zeng, Z., Yuen, L.-D., Frenzel, T., Unkefer, C. J., Beale, J. M., and Floss, H. G., Biosynthesis of the antibiotic thiostrepton. Methylation of tryptophan in the formation of the quinaldic acid moiety by transfer of the methionine methyl group with net retention of configuration. *J. Am. Chem. Soc.* 1989, *111*, 7274–7276.

144. Seto, H.; Hidaka, T.; Kuzuyama, T.; Shibahara, S.; Usui, T.; Sakanaka, O.; Imai, S., Studies on the biosynthesis of fosfomycin 2. Conversion of 2-hydroxypropyl-phosphonic acid to fosfomycin by blocked mutants of *Streptomyces wedmorensis*. J. Antibiot. **1991**, 44, 1286–1288.

145. Kuzuyama, T.; Hidaka, T.; Kamigiri, K.; Imai, S.; Seto, H. J., Studies on the biosynthesis of fosfomycin 4. The biosynthetic origin of the methyl group of fosfomycin. *J. Antibiot.* **1992**, *45*, 1812–1814.

146. Hammerschmidt, F., Biosynthessis of Natural products with a P–C Bond, IX. Synthesis and Incorporataion of (*S*)- and (*R*)-Hydroxy- $[2-^{2}H_{1}]$ ethylphosphonic Acid into Fosfomycin by *Streptomyces fradiae*. *Liebigs Ann. Chem.* **1992**, *1992*, 553–557.

147. Wagman, G. H.; Weinstein, M. J., Antibiotics from Micromonospora. Annu. Rev. Microbiol. 1980, 34, 537-557.

148. Hong, W.; Yan, L., Identification of *gntK*, a gene required for the methylation of purpurosamine C-6' in gentamicin biosynthesis, *J. Gen. Appl. Microbiol.* **2012**, *58*, 349–356.

149. Bradford, M. M., A Rapid and Sensitive Method for the Quantitation of Microgram Quantities of Protein Utilizing the Principle of protein-Dye Bindind. *Anal. Biochem.* **1976**, *72*, 248–254.

150. Sambrook, J.; Fritsch, E. F.; Maniatis, T., *Molecular Cloning: A Laboratory Manual*; Cold Spring Harbor Laboratory Press: Plainview, NY, 1989.

151. Kieser, T.; Bibb, J. K.; Buttner, M. J.; Chater, K. F.; Hopwood, D. A., Practical Streptomyces Genetics; John Innes Foundation: Norwich, 2000.

152. Ruszczycky, M. W.; Choi, S.-h.; Liu, H.-w., Stoichiometry of the Redox Neutral Deamination and Oxidative Dehydrogenation Reactions Catalyzed by the Radical SAM Enzyme DesII. *J. Am. Chem. Soc.* **2010**, *132*, 2359–2369.

153. Fish, W. W., Rapid colorimetric Micromethod for the Quantitation of Complexed Iron in Biological Samples. *Methods Enzymol.* **1988**, *158*, 357–364.

154. Beinert, H., Semi-micro Methods for Analysis of Labile Sulfide and of Labile Sulfide plus Sulfate Sulfur in Unusually Stable Iron-Sulfur proteins. *Anal. Biochem.* **1983**, *131*, 373–378.

155. McCarty, R. M.; Krebs, C.; Bandarian, V., Spectroscopic, Steady-State Kinetic, and Mechanistic Characterization of the Radical SAM Enzyme QueE, Which Catalyzes a Complex Cyclization Reaction in the Biosynthesis of 7-Deazapurines, *Biochemistry*. **2013**, *52*, 188–198.

156. Nicoli, S.; Santi, P., Assay of amikacin in the skin by high-perfomance lliquid chromatography. *J. Pharm. Biomed. Anal.* **2006**, *41*, 994–997.

157. Greenberg, W. A.; Priestley, E. S.; Sears, P. S.; Alper, P. B.; Rosenbohm, C.; Hendrix, M.; Hung, S.-C.; Wong, C.-H., Design and Synthesis of New Aminoglycoside Antibiotics Containing Neamine as an Optimal Core Structure: Correlation of Antibiotic Activity with in Vitro inhibition of Translation. *J. Am Chem. Soc.* **1999**, *121*, 6527–6541.

158. Page, P. C. B.; Chan, Y.; Liddle, J.; Elsegood, M. R. J., Carbohydrate-derived iminium salt organocatalysts for the asymmetric epoxidation of alkenes. *Tetrahedron* **2014**, *70*, 7283–7305.

159. Buskas, T.; Garegg, P. J.; Konradsson, P.; Maloisel, J.-L., Facile Preparation of Glycosyl Donors for Oligosaccharide Synthesis: 2-Azido-2-deoxyhexopyranosyl Building Blocks. *Tetrahedron: Asymmetry* **1994**, *5*, 2187–2194.

160. Hogendorf, W. F. J.; Gisch, N.; Schwudke, D.; Heine, H.; Bols, M.; Pedersen, C. M., Total Synthesis of Five Lipoteichoic acids of *Clostridium difficile. Chem. Eur. J.* **2014**, *20*, 13511–13516.

161. Daragics, K.; Fugedi, P. Regio- and chemoselective reductive cleavage of 4,6-*O*-benzylidene-type acetals of hexopyranosides using BH3·THF-TMSOTf, *Tetrahedron Letters* **2009**, *50*, 2914–2916.

162. Cheng, M. S.; Wang, W. L.; Tian, Q.; Song, H. Y.; Liu, Y. X.; Li, Q.; Xu, X.; Miao, H. D.; Yao, X. S.; Yang, Z., Total Synthesis of Methyl Protodioscin: A Potent Agent with Antitumor Activity. *J. Org. Chem.* **2003**, *68*, 3658–3662.

163. Chen, L.; Hainrichson, M.; Bourdetsk, D.; Mor, A.; Yaron, A.; Baasov, T. Structure-toxicity relationship of aminoglycosides: Correlation of 2'-amine basicity with acute toxicity in pseudo-disaccharide scaffolds. *Bioorg. Med. Chem.* **2008**, *16*, 8940–8951.

164. Marzilli, L. G. The Two B12 Cofactors: Influence of the *trans* Nitrogen Ligand on Homolytic and Heterolytic process In *Bioinorganic Catalysis* (Eds; Reedijk, J. ; Bouewman, E.), Marcel Dekker, Inc.: New York, **2005**, pp 423-468.

165. Nyffeler, P. T.; Liang, C.-H.; Koeller. K. M.; Wong, C.-H., The Chemistry of Amine-Azide Interconversion: Catalytic Diazotransfer and Regioselective Azide Reduction. *J. Am. Chem. Soc.* **2002**, *124*, 10773–10778.

166. Haskell, T. H.; French, J. C.; Bartz, Q. R., Paromomycin. I. Paromamine, a glycoside of D-glucosamine. *J. Am. Chem. Soc.* **1959**, *81*, 3480–3481.

167. Leach, B. E.; Teeters, C. M., Neamine, an Antibacterial Degradation Product of Neomycin. J. Am Chem. Soc. 1950, 73, 2794–2797.

168. Liscombe, D. K.; Louie, G. V.; Noel, J. P., Architectures, mechanisms and molecular evolution of natural product methyltransferases. *Nat. Prod. Rep.* **2012**, *29*, 1238–1250.

169. Williamson, J. M.; Inamine, E.; Wilson, K. E.; Douglas, A. W.; Liesch, J. M.; Albers-Schönberg, G., Biosynthesis of the β -Lactam Antiviotic, Thienamycin, by *Streptomyces cattleya*. J. Biol. Chem. **1985**, 260, 4637–4647.

170. Szu, P. H.; Ruszczycky, M. W.; Choi, S.-h.; Yan, F.; Liu, H.-w., Characterization and Mechanistic Studies of DesII: A Radical S-Adenosyl-L-methionine Enzyme Involved in the Biosynthesis of TDP-D-Desosamine. *J. Am. Chem. Soc.* **2009**, *131*, 14030–14042.

171. Ruszczycky, M. W.; Choi. S.-h.; Mansoorabadi, S. O.; Liu, H.-w., Mechanistic Studies of the Radical S-Adenosyl-L-methionine Enzyme DesII: EPR Characterization of a Radical Intermediate Generated During Its Catalyzed Dehydrogenation of TDP-D-Quinovose. J. Am Chem. Soc. 2011, 133, 7292–7295.

172. Ruszczycky, M. W.; Ogasawara, Y.; Liu, H.-w., Radical SAM enzymes in the biosynthesis of sugar-containing natural products. *Biochim. Biophys. Acta Proteins Proteomics* **2012**, *1824*, 1231–1244.

173. Ruszczycky, M. W.,; Choi, S.-h.; Liu, H.-w., EPR-kinetic isotope effect study of the mechanism of radical-mediated dehydrogenation of an alcohol by the radical SAM enzyme DesII. *Proc. Natl. Acad. Sci. USA* **2013**, *110*, 2088–2093.

174. Ko, Y.; Ruszczycky, M. W.; Choi, S.-h.; Liu, H.-w., Mechanistic Studies of the Radical *S*-Adenosylmethionine Enzyme DesII with TDP-D-Fucose. *Angew. Chem. Int. Ed.* **2015**, *54*, 860–863.

175. Lin, G.-M.; Choi, S.-h.; Ruszczycky, M. W.; Liu, H.-w., Mechanistic Investigation of the Radical S-Adenosyl-L-methionine Enzyme DesII Using Fluorinated Analogues. J. Am. Chem. Soc. **2015**, *137*, 4964–4967.

176. Ruszczycky, M. W.; Liu, H.-w., Mechanistic Enzymology of the Radical SAM Enzyme DesII. *Isr. J. Chem.* **2015**, *55*, 315–324.

177. Kim, J.; Darley, D. J.; Buckel, W.; Pierik, A. J., An allyl ketyl radical intermediate in clostridial amino-acid fermentation. *Nature* **2008**, *452*, 239–242.

178. Buckel, W., Radical and Electron Recycling in Catalysis. *Angew. Chem. Int. Ed.* **2009**, *48*, 6779–6787.

179. Matthews, R. G., Cobalamin-Dependent Methyltransferases. Acc. Chem. Res. 2001, 34, 681–689.

180. Matthews, R. G.; Koutmos, M.; Datta, S., Cobalamin-dependent and cobamidedependent methyltransferases. *Curr. Opin. Struct. Biol.* **2008**, *18*, 658–666.

181. Epp, J. B.; Widlanski, T. S., Facile Preparation of Nucleoside-5'-carboxylic Acids. J. Org. Chem. 1999, 64, 293–295.

182. Xu, L.; Price, N. P., Stereoselective synthesis of chirally deuterated (S)-D-(6-²H₁)glucose. *Carbohydr. Res.* **2004**, *339*, 1173–1178.

183. Salnikov, D. S.; Dilaghi-Dumitrescu, R.; Makarov, S. V.; van Eldik, R.; Boss, G. R., Cobalamin reduction by dithionite. Evidence for the formation of a six-coordinate cobalamin(II) complex. *Dalton Trans.* **2011**, *40*, 9831–9834.

184. Fonseca, M. V.; Escalante-Semerena, J. C., Reduction of Cob(III)alamin to Cob(II)alamin in *Samonella enterica* Serovar Typhimurium LT2. *J. Bacteriol.* **2000**, *182*, 4304–4309.

185. Magnet, S.; Blanchard, J., Molecular Insights into Aminoglycoside Action and Resistance. *Chem. Rev.* 2005, 105, 477–479.

186. Kotra, L. P.; Haddad, J.; Mobashery, S., Aminoglycosides: Perspectives on Mechanisms of Action and Resistance and Strategies to Counter Resistance. *Antimicrob. Agents Chemother.* **2000**, *44*, 3249–3256.

187. Vicens, Q.; Westhof, E., Molecular Recognition of Aminoglycoside Antibiotics by Ribosomal RNa and resistance Enzymes: An Analysis of X-Ray Crystal Structures. *Biopolymers* **2003**, *70*, 42–57.

188. Vakulenko, S. B.; Mobashery, S., Versatility of Aminglycosides and Prospects for Their Future. *Clini. Microbiol. Rev.* **2003**, *16*, 430–450.

189. Shaw, K. J.; Rather, P. N.; Hare, R. S.; Miller, G. H., Molecular Genetics of Aminoglycoside Resistance Genes and Familial Relationships of the Aminoglycoside-Modifying Enzymes. *Microbiol. Rev.* **1993**, *57*, 138–163.

190. McKay, G. A.; Thompson, P. R.; Wright, G. D., Broad Spectrum Aminoglycoside Phosphotransferase Type III from Enterococcus: Overexpression, Purification, and Substrate Specificity. *Biochemistry* **1994**, *33*, 6936–6944.

191. Trieu-Cuot, P.; Courvalin, P., Nucleotide sequence of the Streptococcus faecalis plasmid gene encoding the 3'5"-aminoglycoside phosphotransferase type III. *Gene* **1983**, *23*, 331–341.

192. Cox, J. R.; McKay, G. A.; Wright, G. C.; Serpersu, E. H., Arrangement of Substrates at the Active Site of an Aminoglycoside Antibiotic 3'-Phosphotransferase As Determined by NMR. *J. Am. Chem. Soc.* **1996**, *118*, 1295–1301.

193. Hon, W.-C.; McKay, G. A.; Thompson, P. R.; Sweet, R. M.; Tang, D. S.; Wright, G. D.; Berghuis, A. M., Structure of an Enzyme Required for Aminoglycoside Antibiotic Resistance Reveals Homology to Eukaryotic Protein Kinases. *Cell* **1997**, *89*, 887–895.

194. Fong, D. H.; Berghusis, A. M., Substrate promiscuity of an aminoglycoside antibiotic resistance enzyme via target mimicry. *EMBO J.* **2002**, *21*, 2323–2331.

195. Lovering, A. M.; White, L.O.; Reeves, D. S., AAC(1): a new aminoglycosideacetylating enzyme modifying the Cl aminogroup of apramycin. *J. Antimicrob. Chemother.* **1987**, *20*, 803–813.

196. Sunada, A.; Nakajima, M.; Ikeda, Y.; Kondo, S.; Hotta, K., Enzymatic 1-*N*-acetylation of paromomycin by an actinomycete strain #8 with multiple aminoglycoside resistance and paromomycin sensitivity. *J. Antibiot.* **1999**, *52*, 809–814.

197. Stark, W. M.; Hoehn, M. M.; Knox, N. G., Nebramycin, a new broad-spectrum antibiotic complex. I. Detection and biosynthesis. *Antimicrob. Agents Chemother*. *(Bethesda)*, **1967**, 7, 314–323.

198. Koch, K. F.; Davis, F. A.; Rhoades, J. A., Nebramycin: Separation of the complex and identification of factors 4, 5, and 5'. *J. Antibiot.* **1973**, *12*, 745–751.

199. Mori, T.; Kyotani, Y.; Watanabe, I.; Oda, T., Chemocal conversion of lividomycin A into lividomycin B. J. Antibiot. 1972, 25, 149–150.

200. Kondo, S.; Hotta, K. J., Semisynthetic aminoglycoside antibiotics: Development and enzymatic modifications. *Infect. Chemother.* **1999**, *5*, 1–9.

201. Toraya, T., Radical Catalysis in Coenzyme B₁₂-Dependent Isomerization (Eliminating) Reactions. *Chem. Rev.* **2003**, *103*, 2095–2027.

202. Toraya, T. Cobalamin-dependent dehydratases and a deaminase: Radical catalysis and reactivating chaperones. *Arch. Biochem. Biophys.* **2014**, *544*, 40–57.

203. Demick, J. M.; Lanzilotta, W. N., Radical SAM Activation of the B₁₂-Independent Glycerol Dehydratase Results in Formation of 5'-Deoxy-5'-(methylthio)adenosine and Not 5'-Deoxyadenosine. *Biochemistry* **2011**, *50*, 440–442.

204. Hirayama, T.; Tamegai, H.; Kudo, F.; Kojima, K.; Kakinuma, K.; Eguchi, T., Biosynthesis of 2-Deoxystreptamine-containing Antibiotics in *Streptoalloteichus hindustanus* JCM 3268: Characterization of 2-Deoxy-*scyllo*-inose Synthase. *J. Antibiot.* **2006**, *59*, 358–361.

205. Oda, T.; Mori, T.; Ito, H.; Kunieda, T.; Munakata, K., Studies on new antibiotic lividomycins. I. Taxonomic studies on the lividomycin-producing strain *Streptomyces lividus* nov. sp. *J. antibiot.* **1971**, *24*, 333–338.

206. Mori, T.; Ichiyanagi, T.; Kondo, H.; Tokunaga, K.; Oda, T.; Munakata, K., Studies on new antibiotic lividomycins. II. Isolation and characterization of lividomycins A, B and other aminoglycosidic antibiotics produced by Streptomyces lividus. *J. Antibiot.* **1971**, *24*, 339–346.

207. Ni, X.; Li, D.; Yang, L.; Huang, T.; Li, H.; Xia, H., Construction of kanamycin B overproducing strain by genetic engineering of *Streptomyces tenebrarius*. *Appl. Microbiol. Biotechnol.* **2011**, *89*, 723–731.

208. Zhan, Z.-L.; Ren, F.-X.; Zhao, Y.-M., Racile synthesis of D-lividosamine. *Carbohydr. Res.* 2010, 345, 315–317.

209. Kamat, S. S.; Williams, H. J.; Raushel, F. M., Intermediates in the transformation of phosphonates to phosphate by bacteria. *Nature* **2011**, *480*, 570–573.

210. Kamat, S. S.; Williams, H. J.; Dangott, L. J.; Chakrabarti, M.; Raushel, F. M., The catalytic mechanism for aerobic formation of methane by bacteria. *Nature* **2013**, *497*, 132–136.

211. McGlynn, S. E.; Boyd, E. S.; Shepard, E. M.; Lange, R. K.; Gerlach, R.; Broderick, J. B.; Peters, J. W., Identification and Characterization of a Novel Member of the Radical AdoMet Enzyme Superfamily and Implications for the Biosynthesis of the Hmd Hydrogenase Active Site Cofactor. *J. Bacteriol.* **2010**, *192*, 595–598.

212. Dowling D. P.; Bruender, N. A.; Young, A. P.; McCarty, R. M.; Bandarian, V.; Drennan, C. L., Radical SAM enzyme QueE defines a new minical core fold and metal-dependent mechanism. *Nat. Chem. Biol.* **2014**, *10*, 106–112.

213. Chatterjee, A.; Li, S.; Zhang, Y.; Grove, T.; Lee, M.; Krebs, C.; Booker, S. J.; Begley, T. P.; Ealick, S. E., Reconstitution of ThiC in thiamine pyrimidine biosynthesis expands the radical SAM superfamily. *Nat. Chem. Biol.* **2008**, *4*, 758–765.

214. Ugulava, N. B.; Gibney, B. R.; Jarrett, J. T., Iron-Sulfur Cluster Interconversions in Biotin Synthase: Dissociation and Reassociation of Iron during Conversion of [2Fe-2S] to [4Fe-4S] Clusters. *Biochemistry* **2000**, *39*, 5206–5214.

215. Ollagnier, S.; Meier, C.; Mulliez, E.; Gaillard, J.; Schuenemann, V.; Trautwein, A.; Mattioli, T.; Lutz, M.; Fontecave, M., Assembly of 2Fe-2S and 4Fe-4S Clusters in the Anaerobic Ribonucleotide Reductase from *Escherichia coli. J. Am. Chem. Soc.* **1999**, *121*, 6344–6350.

216. Barton, D. H. R.; McCombie, S. W., A new method for the deoxygenation of secondary alcohols. J. Chem. Soc., Perkin, Trans. 1. 1975, 16, 1574–1585.

217. Vehovec, T.; Obreza, A., Review of operating principle and applications of the charged aerosol detector. *J. Chromatogr. A* **2010**, *1217*, 1549–1556.

218. Stubbe, J.; van der Donk, W. A., Ribonucleotide reductases: radical enzymes with suicidal tendencies. *Chem. Biol.* **1995**, *2*, 793–801.

219. Sakakibara, K.; Nakatsubo, F.; French, A. D.; Roseanu, T., Chiroptical properties of an alternatingly functionalized cellotriose bearing two porphyrin groups. *Chem. Commun.* **2012**, *48*, 7672–7674.

220. Sharma, M.; Bernacki, R.; Paul. B.; Korytnyk, W., Fluorinated carbohydrates as potential plasma membrane modifiers. Synthesis of 4- and 6-fluoro derivatives of 2-acetamido-2-deoxy-D-hexopyranoses. *Carbohydr. Res.* **1990**, *198*, 205–221.